

2025

Franchise Disclosure Document



FRANCHISE DISCLOSURE DOCUMENT

Taco John's International, Inc.
(a Wyoming corporation)
1650 West End Blvd., Suite 200
St. Louis Park, MN 55416
(307) 635-0101
www.tacojohns.com
www.tacojohnsfranchise.com
FranDev@tacojohns.com

The franchise offered is to operate a Taco John's Restaurant that serves mainly Mexican food for carry out or consumption on the premises.

The total investment necessary to begin operation of a Taco John's franchise varies depending upon the size and seating capacity of the Taco John's Restaurant. The total investment (except for real estate) for a freestanding traditional restaurant range from \$1,365,000 to \$2,120,000, for an endcap traditional restaurant ranges from \$805,000 to \$1,342,000, and for a DT Digital Olé traditional restaurant ranges from \$985,000 to \$1,745,000. For a non-traditional restaurant the total investment ranges from \$390,000 to \$632,000. This includes \$15,000 to \$40,000 that must be paid to the franchisor for the Initial Franchise Fee.

This disclosure document summarizes certain provisions of your franchise agreement and other information in plain English. Read this disclosure document and all accompanying agreements carefully. You must receive this disclosure document at least 14 calendar days before you sign a binding agreement with, or make any payment to, the franchisor or an affiliate in connection with the proposed franchise sale. **Note, however, that no government agency has verified the information contained in this document.**

The terms of your contract will govern your franchise relationship. Do not rely on the disclosure document alone to understand your contract. Read all of your contract carefully. Show your contract and this disclosure document to an advisor, like a lawyer or accountant.

Buying a franchise is a complex investment. The information in this disclosure document can help you make up your mind. More information on franchising, such as "A Consumer's Guide to Buying a Franchise," which can help you understand how to use this disclosure document, is available from the Federal Trade Commission. You can contact the FTC at 1-877-FTC-HELP or by writing to the FTC at 600 Pennsylvania Avenue NW, Washington, DC 20580. You can also visit the FTC's home page at www.ftc.gov for additional information. Call your state agency or visit your public library for other sources of information on franchising.

There may also be laws on franchising in your state. Ask your state agencies about them.

Issuance Date: March 31, 2025, as amended July 17, 2025.

FOR USE IN: AL, AK, AZ, AR, CA, CO, CT, DE, DC, GA, FL, ID, IL, IN, IA, KS, KY, LA, ME, MA, MI, MN, MS, MO, MT, NE, NV, NH, NJ, NM, NY, NC, ND, OH, OK, OR, PA, RI, SC, SD, TN, TX, UT, VT, VA, WA, WV, WI, WY, and U.S. TERRITORIES (see State Effective Dates page for effective dates in certain states.)

NOT FOR USE IN: HI or MD.

How to Use This Franchise Disclosure Document

Here are some questions you may be asking about buying a franchise and tips on how to find more information:

QUESTION	WHERE TO FIND INFORMATION
How much can I earn?	Item 19 may give you information about outlet sales, costs, profits or losses. You should also try to obtain this information from others, like current and former franchisees. You can find their names and contact information in Item 20 or Attachments D and E.
How much will I need to invest?	Items 5 and 6 list fees you will be paying to the franchisor or at the franchisor's direction. Item 7 lists the initial investment to open. Item 8 describes the suppliers you must use.
Does the franchisor have the financial ability to provide support to my business?	Item 21 or Attachment F includes financial statements. Review these statements carefully.
Is the franchise system stable, growing, or shrinking?	Item 20 summarizes the recent history of the number of company-owned and franchised outlets.
Will my business be the only Taco John's business in my area?	Item 12 and the "territory" provisions in the franchise agreement describe whether the franchisor and other franchisees can compete with you.
Does the franchisor have a troubled legal history?	Items 3 and 4 tell you whether the franchisor or its management have been involved in material litigation or bankruptcy proceedings.
What's it like to be a Taco John's franchisee?	Item 20 or Attachments D and E lists current and former franchisees. You can contact them to ask about their experiences.
What else should I know?	These questions are only a few things you should look for. Review all 23 Items and all Exhibits in this disclosure document to better understand this franchise opportunity. See the table of contents.

What You Need To Know About Franchising Generally

<u>Continuing responsibility to pay fees</u>. You may have to pay royalties and other fees even if you are losing money.

Business model can change. The franchise agreement may allow the franchisor to change its manuals and business model without your consent. These changes may require you to make additional investments in your franchise business or may harm your franchise business.

<u>Supplier restrictions</u>. You may have to buy or lease items from the franchisor or a limited group of suppliers the franchisor designates. These items may be more expensive than similar items you could buy on your own.

<u>Operating restrictions</u>. The franchise agreement may prohibit you from operating a similar business during the term of the franchise. There are usually other restrictions. Some examples may include controlling your location, your access to customers, what you sell, how you market, and your hours of operation.

<u>Competition from franchisor</u>. Even if the franchise agreement grants you a territory, the franchisor may have the right to compete with you in your territory.

Renewal. Your franchise agreement may not permit you to renew. Even if it does, you may have to sign a new agreement with different terms and conditions in order to continue to operate your franchise business.

<u>When your franchise ends</u>. The franchise agreement may prohibit you from operating a similar business after your franchise ends even if you still have obligations to your landlord or other creditors.

Some States Require Registration

Your state may have a franchise law, or other law, that requires franchisors to register before offering or selling franchises in the state. Registration does not mean that the state recommends the franchise or has verified the information in this document. To find out if your state has a registration requirement, or to contact your state, use the agency information in Attachment G.

Your state also may have laws that require special disclosures or amendments be made to your franchise agreement. If so, you should check the State Specific Addenda. See the Table of Contents for the location of the State Specific Addenda.

Special Risks to Consider About This Franchise

Certain states require that the following risk(s) be highlighted:

1. Out-of-State Dispute Resolution. The franchise agreement requires you to resolve disputes with the franchisor by mediation, arbitration and/or litigation only in Minnesota. Out-of-state mediation, arbitration, or litigation may force you to accept a less favorable settlement for disputes. It may also cost more to mediate, arbitrate, or litigate with the franchisor in Minnesota than in your own state.

Certain states may require other risks to be highlighted. Check the "State Specific Addenda" (if any) to see whether your state requires other risks to be highlighted.

INFORMATION FOR PROSPECTIVE FRANCHISEES IN MICHIGAN

The State of Michigan prohibits certain unfair provisions that are sometimes in franchise documents. If any of the following provisions are in these franchise documents, the provisions are void and cannot be enforced against you.

Each of the following provisions is void and unenforceable if contained in any documents relating to a franchise:

- (A) A prohibition on the right of a franchisee to join an association of franchisees.
- (B) A requirement that a franchisee assent to a release, assignment, novation, waiver, or estoppel which deprives a franchisee of rights and protections provided in this act (the Michigan Franchise Investment Law]. This shall not preclude a franchisee, after entering into a franchise agreement, from settling any and all claims.
- (C) A provision that permits a franchisor to terminate a franchise prior to the expiration of its term except for good cause. Good cause shall include the failure of the franchisee to comply with any lawful provisions of the franchise agreement and to cure such failure after being given written notice thereof and a reasonable opportunity, which in no event need be more than 30 days, to cure such failure.
- (D) A provision that permits a franchisor to refuse to renew a franchise without fairly compensating the franchisee by repurchase or other means for the fair market value at the time of expiration, of the franchisee's inventory, supplies, equipment, fixtures, and furnishings. Personalized materials which have no value to the franchisor and inventory, supplies, equipment, fixtures, and furnishings not reasonably required in the conduct of the franchise business are not subject to compensation. This subsection applies only if: (i) the term of the franchise is less than five years; and (ii) the franchisee is prohibited by the franchise or other agreement from continuing to conduct substantially the same business under another trademark, service mark, trade name, logotype, advertising, or other commercial symbol in the same area subsequent to the expiration of the franchise or the franchisee does not receive at least six months advance notice of franchisor's intent not to renew the franchise.
- (E) A provision that permits the franchisor to refuse to renew a franchise on terms generally available to other franchisees of the same class or type under similar circumstances. This section does not require a renewal provision.
- (F) A provision requiring that arbitration or litigation be conducted outside this state. This shall not preclude the franchisee from entering into an agreement, at the time of arbitration, to conduct arbitration at a location outside this state. (The above language has been included in this Disclosure Document as a condition for registration. We and you do not agree that the parties are restricted from choosing to conduct arbitration outside of Michigan and believe that each of the provisions of the Franchise Agreement, including each of the arbitration provisions, is fully enforceable. We and you intend to rely on the federal pre-emption under the Federal Arbitration Act.)
- (G) A provision which permits a franchisor to refuse to permit a transfer of ownership of a franchise, except for good cause. This subdivision does not prevent a franchisor from exercising a right of first refusal to purchase the franchise. Good cause shall include, but is not limited to:
- (i) The failure of the proposed franchisee to meet the franchisor's then-current reasonable qualifications or standards.

- (ii) The fact that the proposed transferee is a competitor of the franchisor or subfranchisor.
- (iii) The unwillingness of the proposed transferee to agree in writing to comply with all lawful obligations.
- (iv) The failure of the franchisee or proposed transferee to pay any sums owing to the franchisor or to cure any default in the franchise agreement existing at the time of the proposed transfer.
- (H) A provision that requires the franchisee to resell to the franchisor items that are not uniquely identified with the franchisor. This subdivision does not prohibit a provision that grants to a franchisor a right of first refusal to purchase the assets of a franchise on the same terms and conditions as a bona fide third party willing and able to purchase those assets, nor does this subdivision prohibit a provision that grants the franchisor the right to acquire the assets of a franchise for the market or appraised value of such assets if the franchisee has breached the lawful provisions of the franchise agreement and has failed to cure the breach in the manner provided in subdivision (C).
- (I) A provision which permits the franchisor to directly or indirectly convey, assign, or otherwise transfer its obligations to fulfill contractual obligations to the franchisee unless provision has been made for providing the required contractual services.

The fact that there is a notice of this offering on file with the Attorney General does not constitute approval, recommendation, or endorsement by the Attorney General.

If the franchisor's most recent financial statements are unaudited and show a net worth of less than \$100,000, the franchisor shall, at the request of a franchisee, arrange for the escrow of initial investment and other funds paid by the franchisee or subfranchisor until the obligations to provide real estate, improvements, equipment, inventory, training, or other items included in the franchise offering are fulfilled. At the option of the franchisor, a surety bond may be provided in place of escrow.

Any questions regarding the notice should be directed to:

State of Michigan
Department of Attorney General
Franchise Section - Consumer Protection Division
G. Mennen Williams Building, 1st Floor
525 W. Ottawa Street
Lansing, Michigan 48933
Telephone Number: (517) 373-7117

TABLE OF CONTENTS

<u>ITE</u>	<u>M</u>		PAGE
1.	The Franchison	r and Any Parents, Predecessors and Affiliates	1
2.		ience	
3.	-		
4.	O		
5.			
6.			
7.		d Initial Investment	
7. 8.		Sources of Products and Services	
9.		bligations	
10.	U		
11.		ssistance, Advertising, Computer Systems and Training	
12.	•		
13.			_
14.		ights and Proprietary Information	
15.	_	Participate in the Actual Operation of the Franchise Business	
16.	Restrictions on	What the Franchisee May Sell	53
17.	Renewal, Term	nination, Transfer and Dispute Resolution	53
18.	Public Figures		57
19.	Financial Perfo	ormance Representations	57
20.	Outlets and Fra	anchisee Information	61
21.	Financial State	ments	66
22.	Contracts		66
23.	Receipt		67
A T 7	TACHMENTS		
_	ACHMENT A	Franchise Agreement	
		Exhibit I – Addendum to Franchise Agreement	
		Exhibit II – Authorization Agreement for Direct Payments	
		Exhibit III – Statement of Ownership	
		Exhibit IV – Guaranty	
		Exhibit V – Non-Traditional Facility Rider	
		Exhibit VI – State Riders	
ATT	ACHMENT B	Area Development Agreement	
A TT	A CHIMENIT C	Exhibit I – Addendum to Area Development Agreement	
	ACHMENT C	Nondisclosure and Noncompetition Agreement	
	ACHMENT D ACHMENT E	List of Franchisees and Area Developers List of Former Franchisees	
	ACHMENT F	Financial Statements	
	ACHMENT G	State Administrators/Agents for Service of Process	
	ACHMENT H	Operations Manual Table of Contents	
	ACHMENT I	Disclosure Acknowledgment	
	ACHMENT J	Gift Card Onboarding and Change Form	
	ACHMENT K	Lease Rider	
	ACHMENT L	Operational Assessment Program Confidentiality/Release Agreement	
	ACHMENT M	State Addenda	
ATT	ACHMENT N	Receipts	

ITEM 1. THE FRANCHISOR AND ANY PARENTS, PREDECESSORS AND AFFILIATES

The Franchisor is Taco John's International, Inc. To simplify the language in this Franchise Disclosure Document ("Disclosure Document"), Taco John's International, Inc. will be referred to as "we," "us" or "Taco John's." We will refer to the person or entity that buys or is considering buying the franchise as "you" throughout the Disclosure Document. If the Franchisee will operate through a corporation or partnership, "you" also includes the Franchisee's owners or partners.

We are a Wyoming corporation incorporated on May 7, 1985. Our principal business address is 1650 West End Blvd., Suite 200, St. Louis Park, Minnesota 55416 and our telephone number is 307-635-0101. We do business under our corporate name and as "Taco John's." Our agents for service of process are listed on Attachment G. There are no parents or predecessors that are required to be disclosed in the Disclosure Document.

In March 2020, we relocated our franchise support services departments, consisting of operations, marketing, supply chain, research and development, training, and restaurant technology, to 1650 West End Boulevard, Suite 200, St. Louis Park, Minnesota 55416, telephone number 800-854-0819. Between February and July 2025, we relocated our administrative services departments to our St. Louis Park office.

We are in the business of operating and granting single-unit franchises and franchise development rights to operate restaurants known as "Taco John's Restaurants" which offer for sale various food products prepared on site, mainly Mexican food, including a wide variety of traditional and cross-ethnic menu items. We will refer to a Taco John's Restaurant in this Disclosure Document as either a "Restaurant" or a "Taco John's Restaurant." Customers may purchase food for dine-in, carryout, or drive-thru. Most Restaurants provide seating and/or a drive-thru window. Taco John's Restaurants operate under certain trademarks, service marks and other commercial symbols, including the mark TACO JOHN'S®, all of which we may change or add to (the "Marks"), and use a system for the establishment and operation of the Restaurants that includes distinctive recipes, interior and exterior designs, color schemes, and fixtures and equipment, all of which we may modify, develop and supplement periodically (the "Restaurant System").

The Taco John's concept was first developed in 1969 by Woodson-Holmes Enterprises, Inc., a Wyoming corporation formed on February 10, 1969 ("Woodson-Holmes"). Prior to our formation in 1985, Woodson-Holmes operated Taco John's Restaurants and franchised Taco John's Restaurants to third parties. In 1985, Woodson-Holmes was merged into what is presently known as Taco John's International, Inc., at which time we began operating our own Taco John's Restaurants and franchising Taco John's Restaurants. As of December 31, 2024, there were a total of 340 Taco John's Restaurant units in 22 states. This includes 333 franchised Taco John's Restaurants and seven company-owned and operated Taco John's Restaurants. We are not engaged in any other line of business. Woodson-Holmes may be deemed a predecessor of ours.

In 2015, we began offering Area Development Agreements for the sale of multiple Taco John's Restaurants. As of December 31, 2024, there were seven Area Development Agreements to develop Restaurants at various locations.

Spicy Seasonings, LLC, a Wyoming limited liability company formed on September 24, 2020 and located at 67 Goose Lane, Sheridan, Wyoming 82801 ("SS, LLC"), is an affiliate of ours that owns all of the Marks and certain of the secret recipes which make up part of the Restaurant System. (See Items 8 and 13 of this Disclosure Document.) In September 2020, SS, LLC acquired all of its assets from Taco John's Seasonings Limited Partnership, a Wyoming limited partnership ("TJS"), through a reorganization of

commonly owned entities. TJS was subsequently dissolved on December 31, 2020. Neither we, our predecessor, nor our affiliates have granted or sold franchises in any other lines of business.

We have, in some limited situations, allowed a Franchisee to operate its restaurant facility as a dual branded unit, offering products authorized under the Restaurant System together with products licensed from an independent third party. As of December 31, 2024, there was one Franchisee-owned Taco John's/Steak Escape restaurant located in South Dakota, and two Franchisee-owned Taco John's/Good Times Burgers and Frozen Custard restaurants located in Wyoming.

This Disclosure Document describes the Taco John's Restaurant franchise and the development rights we currently offer in the states for which this Disclosure Document is used. The terms of the agreements described in this Disclosure Document may differ substantially from those we have used in the past, and we reserve the right to change the forms and terms of the agreements used in the future.

Taco John's Franchise Agreement

A franchise offered under this Disclosure Document is for a qualified person (a "Franchisee" or collectively, the "Franchisees") to operate a Taco John's Restaurant under a franchise agreement with us. Attached to this Disclosure Document, as <u>Attachment A</u>, is our current standard form of franchise agreement for the operation of a Taco John's Restaurant (the "Franchise Agreement"). Under the Franchise Agreement, you will be given the right to own and operate a Taco John's Restaurant at a designated location accepted by us (the "Premises") using the Marks and the Restaurant System.

We offer two franchise concepts. Our feature concept is our traditional restaurant concept, which has three prototype building models: a freestanding building, an endcap location in a shopping mall, strip mall or alternative facility with our approval, and a DT (drive-thru) Digital Olé building. The freestanding building and endcap location models typically have a Premises size of 1,800 to 2,250 square feet with the capacity to seat 20 to 40 patrons and with drive-thru window service. The DT Digital Olé models are designed for smaller markets and typically have a Premises size of 946 to 1,500 square feet and focus on drive-thru, pickup, and walk-up service with limited or no seating. These DT Digital Olé models are currently being offered on a limited test basis only by invitation from us based on several factors including, but not limited to, the franchisee's geographic location, overall business experience, operating experience within the Taco John's system, and available capitalization and funding resources. Our DT Digital Olé models include the "DT Digital Olé" (without seating) and the "DT Digital Olé+" (with seating). Unless otherwise noted, references in this Disclosure Document to our traditional restaurant refers to the freestanding, endcap, and DT Digital Olé models. See Item 7 for additional information on each of these restaurant designs.

The second franchise concept is for non-traditional locations. A non-traditional location is a venue or facility where the primary purpose of the consumer's visit is typically for reasons other than the purchase of goods from a Taco John's Restaurant. Examples include airports and other transportation hubs, hospitals, college campuses and other educational facilities, convention centers, grocery stores, department stores, "big box" retail centers, sports arenas and stadiums, hotels and office buildings, military installations (Army and Air Force Exchange Services), and food courts. Typically, these are full-service restaurants with a Premises size of 675 to 2,000 square feet. Non-traditional locations typically do not have a drive-thru window.

We franchise both concepts under our standard form of Franchise Agreement. In some cases, we may modify the Franchise Agreement for non-traditional locations to address different conditions and circumstances. See the Non-Traditional Facility Rider which is attached to the Franchise Agreement as Exhibit V.

Taco John's Area Development Agreement

Certain qualified persons or entities may also be offered the right to develop and operate multiple Taco John's Restaurants in a designated geographical area (the "Development Area") under the terms of an Area Development Agreement ("ADA"). You must meet our multi-unit development criteria to enter an ADA with us. Our current standard form of ADA for the development of Taco John's Restaurants is attached as Attachment B. In the sections of this Disclosure Document where we describe the ADA, we will refer to a person who has been granted rights under such agreement as "you" or the "Developer." A Developer must sign our then-current form of Franchise Agreement for each Taco John's Restaurant opened in the Development Area, which Franchise Agreement may be materially different from the form of Franchise Agreement attached to this Disclosure Document. The ADA states the number of Taco John's Restaurants that you must open in your Development Area, at specific locations accepted by us. The deadlines for opening and commencing operations of those Restaurants will be described in a Development Schedule which establishes the number of Restaurants you must have open and continuously operating by certain established dates.

Competition

The products of Taco John's Restaurants are sold to the general public. You will have to compete with other national and regional chains and local businesses offering similar products and, to a lesser extent will have to compete for customers with grocery stores, which offer frozen or prepared Mexican food. You should recognize that the quick service restaurant industry is highly competitive and that such competition affects both us and our Franchisees. An investment in a Taco John's Restaurant, like any other business, involves business risks and the success of the venture depends primarily on your business abilities and efforts. We do not warrant or represent that a specific location will be successful or that you will be able to achieve any specific sales or profit levels.

Laws and Regulations

The food industry is regulated. You must investigate and comply with all local, state and federal health and sanitation laws, and safety standards relating to food handling and preparation, and the sale of food. You will need to understand and comply with those laws in the operation of your Taco John's Restaurant. In addition, franchisees must comply with other laws which affect the operation of a business generally, including employment, workers' compensation, business entities, insurance, corporate, taxing, licensing, zoning, and similar laws and regulations. You should consult with an attorney regarding state and local laws, rules, and regulations that may affect the operation of your Taco John's Restaurant at your particular location.

ITEM 2. BUSINESS EXPERIENCE

Director: John V. Blankenship

Mr. Blankenship became a member of our Board of Directors in January 2022. From March 2019 to the present, he has served as a member of the board of Juniper Grill Enterprise, LLC.

Director: Kelley Digby

Ms. Digby became a member of our Board of Directors in December 2014.

Director: Greg Haggis

Mr. Haggis became a member of our Board of Directors in January 2020. From April 2016 to August 2023, he served as Chief Financial Officer of The Wenzak Companies, a Wendy's and Wingstop franchisee based out of Naperville, Illinois.

Director: Erik Hess

Mr. Hess became a member of our Board of Directors in January 2022. From August 2020 to May 2021, he served as President of Focus Brands LLC - Restaurant Category (Moe's Southwest Grill, McAlister's Deli, and Schlotzsky's Deli) in Atlanta, Georgia. From August 2019 to August 2020, he served as President of Moe's Southwest Grill in Atlanta, Georgia. From June 2021 to the present, he served as President of Hess Consulting Group, LLC in St. Charles, Illinois.

Director: Les Karel

Mr. Karel was elected to our Board of Directors in January 2020. Since December 2020, he has also served as Executive Vice President of National DCP, LLC, in Duluth, Georgia. National DCP distributes products to Dunkin' franchise restaurants.

Director: Clint Langer

Mr. Langer is an attorney and since October 2001 has practiced with the firm Davis & Cannon, LLP, in Sheridan, Wyoming. He became a member of our Board of Directors in December 2014.

Director: Gerard Lewis

Mr. Lewis has been a member of our board since June 2018 and since January 2021 serves as the Chairman of the Board. He served as our Interim President from January 2024 to February 2024. In January 2014, he founded G&S Food Group located in Castle Pines, Colorado, which he still runs.

Director: Brandy O'Connor

Mr. O'Connor became a member of our Board of Directors in January 2017. He has been Chief Revenue Officer of TRC Healthcare, a Colibri Group Company, in Roseville, California since December 2016.

Director: Bart Taylor

Mr. Taylor has been a member of our Board of Directors since June 2005. He became a manager of SS, LLC upon its formation in September 2020. Mr. Taylor also actively participates in the management of various family trusts.

Director: Scott Weisberg

Mr. Weisberg became a member of our Board of Directors in January 2020. From April 2018 to the present, Mr. Weisberg has been providing executive coaching, talent and organizational services through Scott Weisberg Coaching and Advisory, LLC (formerly Incisive Consulting, LLC), a company he founded that is located in Boerne, Texas.

President, Chief Executive Officer, and Director: Heather Leed Neary

Ms. Neary joined Taco John's as President and Chief Executive Officer in February 2024. She became a member of our Board of Directors in March 2024. From March 2021 to January 2024, Ms. Neary served as President of KBP Bells, a Taco Bell franchisee located in Leawood, Kansas. Prior to that, from November 2015 to January 2021, she served as President of Auntie Anne's LLC in Lancaster, Pennsylvania. Ms. Neary is located in our St. Louis Park, Minnesota office.

Chief Financial Officer and Corporate Treasurer: Richard Bundy

Mr. Bundy joined us in February 2021 as Chief Financial Officer. He became our Corporate Treasurer in January 2024. From June 2018 to January 2021, he served as Senior Vice President and Chief Financial Officer for Christopher & Banks Corporation, located in Plymouth, Minnesota. Mr. Bundy is located in our St. Louis Park, Minnesota office.

Chief Operating Officer: Jackie Secor

Ms. Secor joined Taco John's in June 2025 as our Chief Operating Officer. Prior to joining us, she served in various roles with GoTo Foods (formerly known as Focus Brands) located in Atlanta, Georgia, including as Senior Vice President-GTF Operations & Training for seven brands from December 2024 to May 2025, and as Senior Vice President-GTF Operations & Training, Specialty Category for four brands from October 2020 to December 2024. Ms. Secor is based out of our St. Louis Park, Minnesota office.

Chief Marketing Officer: Kevin Flaherty

Mr. Flaherty joined Taco John's in October 2024 as our Chief Marketing Officer. From May 2024 to October 2024, he was self-employed as a restaurant marketing consultant located in Seattle, Washington. From April 2023 to February 2024, he served as Interim Chief Marketing Officer for MOD Pizza located in Bellevue, Washington. Prior to that, he served in various roles for MOD Pizza located in Bellevue, Washington, including as Senior Vice President, Digital Marketing and Off-Premises Ordering from January 2022 to March 2023, and as Vice President, Digital Marketing from December 2015 to December 2021. Mr. Flaherty is located in our St. Louis Park, Minnesota office.

Chief of Staff: Nicole Hostetter

Ms. Hostetter joined Taco John's in May 2024 as our Chief of Staff. From April 2022 to May 2024, she served as the Executive Assistant at KBP Brands located in Leawood, Kansas. She previously served as the Executive Assistant at Benchmark Construction located in Lancaster, Pennsylvania, from August 2021 to April 2022, and also served as the Executive Assistant at Penn Medicine Lancaster General Health located in Lancaster, Pennsylvania from January 2021 to August 2021. Prior to that, she was the Manager, Administration with Auntie Anne's, Inc., located in Lancaster, Pennsylvania, from January 2018 to January 2021. Ms. Hostetter is based out of our St. Louis Park, Minnesota office.

General Counsel and Corporate Secretary: Kristin M. Nuss

Ms. Nuss became our General Counsel and Corporate Secretary in January 2025. From May 2021 to December 2024, she served as our Associate General Counsel and Director of Franchise Compliance & Contracts. From April 2015 to May 2021, she served as a Supervising Attorney General in the Wyoming Attorney General's Office. Ms. Nuss is based out of our St. Louis Park, Minnesota office.

Vice President/Technology: Thomas A. Perella, Jr.

Mr. Perella joined Taco John's as Vice President/Technology in March 2025. From April 2023 to January 2025, he served as Vice President of Technology, and from May 2022 to April 2023, he served as Director of Technology, for Duck Donuts located in Mechanicsburg, Pennsylvania. Prior to that, from October 2020 to May 2022, he served as Director – Retail Information Implementation for FOCUS Brands, LLC located in Atlanta, Georgia. Mr. Perella is located in our St. Louis Park, Minnesota office.

Vice President/Development: Shannon Iverson

Ms. Iverson joined Taco John's as Vice President/Development in February 2024. From January 2022 to February 2024, she served as Vice President of Franchise Growth for Marco's Franchising, LLC located in Toledo, Ohio. She also served in various franchise sales roles with Marco's Franchising from November 2018 to January 2022. Ms. Iverson is based out of our St. Louis Park, Minnesota office.

Vice President/Supply Chain: Nick Davis

Mr. Davis joined Taco John's as Vice President/Supply Chain in July 2024. From March 2023 to June 2024, he served as Senior Vice President, Procurement & Supply Chain for Promise Confections (Pearson's Candy Company, Edward Marc Brands, Seattle Gourmet Foods and Annabelle's Candy Company) headquartered in Pittsburgh, Pennsylvania. Prior to that, from October 2016 to February 2023, he served as Vice President, Procurement, Supply Chain and Co-Manufacturing Business Development for Pearson's Candy Company located in St. Paul, Minnesota. Mr. Davis is located in our St. Louis Park, Minnesota office.

Vice President/Human Resources: Damian Hanft

Mr. Hanft joined Taco John's as Vice President/Human Resources in April 2025. Prior to joining us, he served in various roles with Inspire Brands located in Atlanta, Georgia, including as VP/Talent Experience from April 2023 to April 2025, as VP/Brand Ambassador from April 2022 to April 2023, and as VP/Learning & Development from October 2018 to April 2022. Mr. Hanft is located in our St. Louis Park, Minnesota office.

ITEM 3. LITIGATION

Arbitration Actions Brought by Us Against Franchisees and Guarantors to Collect Royalty Payments

<u>Taco John's International, Inc. v. Bierschenk, Inc., Bierschenk-Worthington, Inc., and Todd A. Bierschenk, No. 2024-1765A</u> (Judicial Arbiter Group, Denver, Colorado office, December 16, 2024).

No other litigation is required to be disclosed in this Disclosure Document.

ITEM 4. BANKRUPTCY

Our CFO, Richard Bundy, was Senior Vice President and Chief Financial Officer for Christopher & Banks Corporation from June 2018 to January 2021. On January 13, 2021, Christopher & Banks filed a petition to reorganize under Chapter 11 of the U.S. Bankruptcy Code in the District of New Jersey (Case No. 21-10269-ABA). Christopher & Banks is located at 2400 Xenium Lane North, Plymouth, Minnesota. It is not affiliated with us.

Except for the one bankruptcy matter noted above, no bankruptcies are required to be disclosed in this Disclosure Document.

ITEM 5. INITIAL FEES

The initial fees and other initial payments required under the ADA and the Franchise Agreement are described below. All initial fees are due to us upon signing of the ADA or Franchise Agreement and are non-refundable once paid.

Area Development Agreement

In addition to the Initial Franchise Fee, stated below, for your initial Taco John's restaurant, we currently charge a non-refundable development fee (referred to in the sections of this Disclosure Document that describe the ADA as the "Development Fee") of \$5,000 multiplied by the number of Taco John's Restaurants to be developed and operated by you under the ADA, payable in a lump sum when you sign the ADA. The formula to determine the Development Fee is applied uniformly for all Developers that sign this form of agreement. The Development Fee paid for each Restaurant will be applied toward the Initial Franchise Fee payable under each Franchise Agreement for your Taco John's Restaurants executed pursuant to the ADA.

Franchise Agreement

For a single traditional Taco John's Restaurant, we currently charge an Initial Franchise Fee (referred to in those sections of this Disclosure Document in which we describe the Franchise Agreement as the "Initial Franchise Fee") of \$40,000, payable in full when you sign the Franchise Agreement. Except for the differences described below, the Initial Franchise Fee is imposed uniformly for all new franchisees who sign our current Franchise Agreement. Some of our existing franchisees have the right to develop additional Restaurants under existing agreements and on different terms.

The Initial Franchise Fee may be discounted if you qualify for one of our promotional incentives described below. We reserve the right to negotiate the Initial Franchise Fee with prospective franchisees in the future and may, in our sole discretion, consider a number of factors in those negotiations including, but not limited to, previous experience, prior relationship with us, the number of franchise agreements to be entered, and local market conditions.

For a non-traditional Taco John's Restaurant, we currently charge an Initial Franchise Fee of \$15,000, payable in full when you sign the Franchise Agreement.

If you sign an ADA with us, the Initial Franchise Fee for the subsequent outlets is payable in full when you sign each Franchise Agreement, and that Initial Franchise Fee will be the then-current fee that

we charge. We will apply \$5,000 of your Deposit or Development Fee to the Initial Franchise Fee due when you sign the Franchise Agreement for each Restaurant.

Promotional Incentives

Our current promotional incentives are listed below. Promotional incentives cannot be combined or transferred, and cannot be combined with other pre-existing incentive agreements.

Military Service. We currently charge a discounted Initial Franchise Fee as follows:

- 1. \$24,000 for qualified U.S. veterans for the first Restaurant.
- 2. \$0 for qualified U.S. veterans with a 50% or more military service-connected disability rating for the first Restaurant.
- 3. \$28,000 for qualified U.S. veterans for the second or subsequent Restaurant.

Military Service incentives are available to veterans who have received an honorable discharge and to active-duty personnel. If the franchisee is an entity, the veteran participant must have and maintain 51% or more ownership interest in the entity to qualify for the incentive. To apply for the incentive, you must provide us a copy of Form DD-214 or evidence of active-duty status before the Franchise Agreement is signed.

<u>Multi-Unit Franchisees (New)</u>. We currently charge a discounted Initial Franchise Fee for qualified first-time franchisees who are acquiring franchises for multiple Restaurants, based on the total number of Restaurants initially purchased, as follows:

- 1. \$32,000 per Franchise Agreement for two Restaurants.
- 2. \$28,000 per Franchise Agreement for three or four Restaurants.
- 3. \$24,000 per Franchise Agreement for five or more Restaurants.

To qualify for this incentive, we generally require that the Franchise Agreements be signed at the same time.

We currently reduce the Royalty to 4% of Net Sales for each Restaurant for the first 12 months after the fourth Restaurant opening. All four Restaurants must be opened and continuously operating by the 18th month after the Franchise Agreements are signed.

To be eligible for the promotional incentives, you must meet our financial, creditworthiness, and operational criteria and be approved by us to open each Restaurant. If the franchisee is an entity, the qualifying owner must have and maintain at least 51% or more ownership interest in the entity and in each Restaurant to qualify for the incentive and must comply with all program requirements.

The development and opening timeframe for a multi-unit purchase should not exceed 18 months for 2 to 4 units and should not exceed 24 months for five or more units.

<u>Multi-Unit Franchisees (Existing)</u>. We currently charge a discounted Initial Franchise Fee for qualified existing franchisees who are acquiring franchises for multiple Restaurants, based on the total number of new-built Restaurants (excluding acquired Restaurants) which are opened and continuously operating, as follows:

1. \$32,000 per Franchise Agreement for 2 to 4 total new-built Restaurants.

- 2. \$28,000 per Franchise Agreement for 5 to 10 total new-built Restaurants.
- 3. \$24,000 per Franchise Agreement for 11 or more total new-built Restaurants.

To qualify for this incentive, we generally require that the Franchise Agreements be signed at the same time.

We currently reduce the Royalty to 4% of Net Sales for each Restaurant for the first 12 months after the fourth Restaurant opening. All four Restaurants must be opened and continuously operating by the 18th month after the Franchise Agreements are signed.

To be eligible for the promotional incentives, you must meet our financial, creditworthiness, and operational criteria and be approved by us to open each Restaurant. If the franchisee is an entity, the qualifying owner must have and maintain at least 51% or more ownership interest in the entity and in each Restaurant to qualify for the incentive and must comply with all program requirements.

Other Information

You are not obligated under the Franchise Agreement to purchase any other pre-opening goods or services from us or our affiliates.

Except as described above, the Initial Franchise Fee is uniform for all Franchise Agreements executed pursuant to this Disclosure Document.

ITEM 6. OTHER FEES

Area Development Agreement

There are no "continuing fees" under the ADA, except for the following:

Column 1	Column 2	Column 3	Column 4
Name of Fee	Amount	Due Date	Remarks
Extension Fee	4.5% of the monthly average	Due on the 20th of each	If you fail to meet your Development
	system-wide sales of both	month	Schedule under an ADA, at your discretion
	franchised and company-		and in lieu of termination of the ADA or loss
	owned outlets in the prior		of exclusivity, you may request an extension
	month		to the Development Schedule and pay an
			Extension Fee for up to a maximum of 13
			months. If we agree to grant such extension
			in our discretion, then this fee is paid to us
			and is non-refundable.

The ADA contemplates that you will execute one or more Franchise Agreements. The additional fees that you will be expected to pay under the Franchise Agreement are described below.

Franchise Agreement

You will be obligated to make certain payments to us under the Franchise Agreement in addition to the Initial Franchise Fees described in Item 5. Unless otherwise noted, these payments are non-refundable and are imposed by and payable to us. All of these fees are generally imposed uniformly on our

franchisees who sign our current Franchise Agreement, but we may in unique situations modify certain fees. Certain fees in the current Franchise Agreement have changed from the amounts charged in the past and may change in the future. Therefore, existing and future franchisees may have fees imposed on them that are different from those represented in this table.

Column 1 Name of Fee ¹	Column 2 Amount	Column 3 Due Date	Column 4 Remarks
Royalty	5% of Net Sales ^{2,5}	Currently due on the 20 th day of each month	We reserve the right to change the frequency of the payment and/or the due date(s) of the payment on 60 days' prior advance written notice to you.
Advertising and Marketing Fees	4% of Net Sales ² for traditional Restaurants; 2% of Net Sales ² for non- traditional Restaurants	Payable at the same time as the Royalty	For traditional Restaurants, we will not change the amount of this fee before January 1, 2026, but thereafter we may modify this fee upon 60 days' notice up to a maximum of 4.25% of the Net Sales of the Restaurant during the Initial Term of the Franchise Agreement. We will not increase this fee for non-traditional Restaurants during the Initial Term of the Franchise without your consent.
Financial Reporting System Fee	Currently none; fee will be determined once implemented, based on price charged by the vendor, which shall not exceed \$15 per month when initially implemented	As incurred	We reserve the right to implement this fee upon 180 days' notice to you. Once implemented, this fee will be payable to us or our third-party vendor as we direct for a system that allows you to upload and deliver to us electronic reports containing the Net Sales of the Restaurant, Profit and Loss statements, and any other data, information, and supporting records concerning the financial condition of the Restaurant that we reasonably require. Once implemented, this fee may be changed upon 30 days' notice to you.
Grand Opening Marketing	\$25,000, plus up to an additional \$10,000 in those markets we define as underpenetrated	As incurred	You must conduct a grand opening advertising program that we approve. We may require a higher expenditure for underpenetrated markets.
Required Purchases	Will vary under circumstances	Upon receipt of merchandise or installation of equipment	You must purchase only products to be sold from the Restaurant that we have approved, and you must purchase only from suppliers and distributors that we approve, which may include us or our affiliates. For more information about required purchases, see Items 8, 11 and 16.
The Olé! Way Online Learning Management System License Fee	\$240 per year; and subject to change annually in an amount equal to any increase in the price charged by the vendor	Annually	Payable to us or our third-party online vendor as we direct for access to our online learning management system called The Olé! Way, which provides you with certain ongoing training materials and access to the Operations Manual, defined in Item 8.

Column 1 Name of Fee ¹	Column 2 Amount	Column 3 Due Date	Column 4 Remarks
Gift Card Fee	Currently none; fee will be determined once implemented, subject to change annually to an amount designated by the gift card processor, estimated not to exceed \$100	Annually	You must participate in the Gift Card System and pay us a fee which we will submit to the gift card processor on your behalf. You will be required to enter into the Gift Card Onboarding and Change Form, in the form attached to this Disclosure Document as <u>Attachment J.</u>
Back of House Information System	\$140 per month, and subject to change annually in an amount equal to any increase in the price charged by the vendor	Monthly	Payable to us or our third-party vendor as we direct for the back of house information system.
Geofencing Service Fee	\$35 per month, and subject to change annually in an amount equal to any increase in the price charged by the vendor	Monthly	Payable to us or our third-party vendor as we direct for the geofencing service integrated with the mobile ordering platform.
Digital Menu Board License Fee (only applicable to certain existing Restaurants)	\$18.80 per screen per month, subject to change annually	Monthly	Our current standards require that franchisees acquire the digital menu boards from our approved third-party vendor and pay any fees to that vendor rather than to us (the third party fees may vary from the fee stated here). However, we may charge this fee in limited cases for existing Restaurants operating under an existing digital menu board system, including existing Restaurants that are transferred.
Loyalty/Rewards Program Fee	Currently none; fee will be determined once implemented, subject to change annually in an amount equal to any increase in the price charged by the vendor	As incurred	You must, at your expense, participate in our customer loyalty/rewards program. You must honor loyalty/rewards presented by members in your Restaurant. We reserve the right to modify or terminate the program.
Guest Experience Program Fees	Mystery Shop: As designated by the suppliers. Currently \$95 monthly plus cost of the food, subject to change annually in an amount equal to any increase in the price charged by the vendor. Guest Surveys: Currently none; subject to change annually in an amount equal to any increase in the price charged by the vendor.	As determined	We have established a guest experience program to evaluate guest satisfaction of your Restaurant, which consists of mystery shops and guest surveys. You must participate in it and pay the fees or charges associated with it. We collect this fee from you and submit it to the vendors on your behalf. We have the right to review the results of the service evaluations and reserve the right to charge you for the cost of resolving complaints.

Column 1	Column 2	Column 3	Column 4
Name of Fee ¹ Technology Fee ³	Amount Currently none; fee will be determined once implemented, subject to change annually in an amount equal to any increase in the underlying prices charged by the vendors	Due Date Once implemented, payable at the same time as the Royalty	Remarks We reserve the right to implement this fee upon 60 days' notice to you. Once implemented, this fee will replace some or all of the following fees as we specify: the Digital Menu Board License Fee, The Olé! Way Online Learning Management System License Fee, financial reporting system fee, back of house information system fee, geofencing service fee, gift card fee, loyalty/rewards program fee, and guest experience program fees noted above, and other technology-related implementation, subscription, or license costs. The fee may be in the amount of the total fees charged to us by our vendors of the underlying services, plus up to 10% of that total cost which amount we will use toward additional development and support of technology.
Additional Training	Then-current rate	As incurred	We do not charge a fee for initial training, and we do not charge any additional fee for training on The Olé! Way system, but we may charge a fee for any other training programs, workshops or seminars that we may periodically offer and for additional inunit training requested by a Franchisee from time to time. For more information about training requirements, see Item 11.
Certified Manager Training	Will vary as designated by the method of training	As incurred	Your management team must participate in a training program that we provide for no charge (our Management Certification Program). They must also obtain a food safety certification through programs offered from third-party suppliers that we approve. You are responsible for expenses of attendance. See Item 11 for more information on the Management Certification Program requirements applicable to your Active Certified Managers.
Non-Attendance Fee	\$1,000	As incurred immediately following the convention	If you do not attend the Association of Taco John's Franchisees' convention or if you do not have at least one manager per Restaurant attend the manager convention, we have the right to charge you a non-attendance fee.
Non-Compliance Fees	Will vary under circumstances	As incurred	If you are determined by us to be non- compliant with remodel, renovation, refresh, reimage, refurbishment, modernization, equipment, financial reporting system, or other technology requirements, or fail to sign up for third- party delivery or catering as required in the Franchise Agreement, we have the

Column 1	Column 2	Column 3	Column 4
Name of Fee ¹	Amount	Due Date	Remarks
			right to charge you a fee of 3% of Net Sales ² to offset the administrative and management costs incurred as a result of your non-compliance. This fee will be charged from the date that is 15 days after the date you are notified of the non-compliance issue and must be paid weekly through the date when the non-compliance
			issue is cured, up to a maximum of six months for each act of non-compliance.
			If you are determined by us to be non-compliant with third-party delivery, catering, or hours of operations standards in the Operations Manual, we have the right to charge you a fee of \$1,000 per day for any partial day of non-compliance or \$3,000 per day for any full day of non-compliance, to offset the administrative and management costs incurred as a result of your non-compliance. This fee will be
			payable immediately upon notification by us.
Delayed Opening Fee	\$250 for each support member, up to a maximum of \$2,000	As incurred	If your scheduled opening date is delayed for any reason, we have the right to charge you a delayed opening fee. This fee is to help offset the cost we incur in preparing to attend your opening, including rescheduling our team member resources and modifying travel arrangements.
Media Relations Support Fee	\$500	As incurred, for each instance of media support provided by us	If your restaurant or organization has a crisis that could generate negative media attention for the brand, and you do not report it to us within 24 hours, we have the right to charge you a fee. This fee is to help offset the cost we incur in assisting you with media relations or engaging public relations professionals.
Insurance Cost Reimbursement	Will vary under circumstances		You are required to obtain the insurance we require (see Item 8). If you do not secure or maintain the specified coverage or give us evidence of that coverage, we may obtain the insurance and charge you for it and any expenses we incur.
Premium for Loss of Business Income and Incident Response for Foodborne Illness Insurance Coverage	\$340 per year; and subject to change annually in an amount equal to any increase in the price charged by the insurance carrier	As incurred	We will provide this insurance coverage. It covers our locations and each individual franchised location. You must participate in this required coverage and pay the premium associated with it. We collect this amount from you and submit it to the insurance carrier on your behalf.
Transfer Fee	\$10,000	Upon our approval of the transferee	Payable by you.

Column 1	Column 2	Column 3	Column 4
Name of Fee ¹	Amount	Due Date	Remarks
Renewal Fee	25% of our then current Initial Franchise Fee	•	You must pay us a renewal fee for each renewal of your Franchise Agreement.
Conceptual Floor Plan and Conceptual Site Plan	Will vary under circumstances	As incurred	We will provide one set of conceptual plans at no charge. Additional layouts or revisions will be invoiced at our then-current rate. Our current rate is \$200.00 per hour to prepare additional layouts or revisions to your conceptual plans.
Site Evaluation and Site Acquisition Support Services	Will vary under circumstances	As incurred	We may require you to reimburse us for actual expenses if we assist you with site evaluation and site acquisition support if you request our assistance.
Temporary Management Fee	Will vary under circumstances	As incurred	If you cannot operate your Restaurant due to death or permanent disability, we may appoint a manager and charge you a fee for the management services of 15% of the Net Sales for the days we managed the restaurant.
Advertising Cooperative Bookkeeping Fee ⁴	Optional, but if obtained, currently \$250 to \$350 per month, depending on the amount of services provided	As incurred	If a local or regional Advertising Cooperative requests we provide bookkeeping services to it, it pays us this fee.
Relocation Fee	\$10,000 (or, for a non-traditional location, 25% of our then current Initial Franchise Fee)	At time we accept the new site	If you lose the right to possess the Premises, you request relocation of the Restaurant, or we determine that your current location is not suitable under our standards, you may relocate to a site which meets our then-current criteria and is accepted by us. This fee is due for each Restaurant that is approved for relocation when we accept the new site. We may charge you this fee for our time and efforts in approving your new location regardless if a relocation actually occurs.
Interest on Late Payments	Not more than 1.5% on amounts due per month	As incurred	After the date that they are due, all payments you owe to us or our affiliates will bear interest of up to 1.5% per month.
Nonsufficient Funds Fee	Currently \$35, subject to change annually; not to exceed \$100	As incurred	We may charge a fee if there are insufficient funds in your account to process payment of fees due to us.

Column 1	Column 2	Column 3	Column 4
Name of Fee ¹	Amount	Due Date	Remarks
Remodeling Expenses	Varies	As incurred	At our request, you must remodel, renovate,
			refresh, reimage, refurbish, and modernize
			the Premises and the Restaurant to reflect
			changes in the operations of Taco John's
			Restaurants that are prescribed and required
			of new Franchisees. We currently cannot
			estimate a range of these expenses. Material
			alterations to remodel or renovate will not
			be required more often than once every 10
			years, but may be required within 12
			months after a transfer. Non-material
			alterations to refresh, reimage, refurbish, or
			modernize will not be required more often
			than once every 5 years, but may be
			required within 12 months after a transfer.
			Maintenance and repair of the Restaurant
			are required on an ongoing basis.
Indemnification	Will vary under circumstances	As incurred	You must reimburse us if we defend
			against or are held liable for claims that
			arise from your Restaurant operations
			unless your operations were in strict
			compliance with our Operations Manual,
			standards or procedures or the claims result
			from negligence or willful or reckless
			misconduct on our part or on the part of our
			agents.

NOTES:

- 1. We currently require that all fees due to us be paid directly and automatically from your bank account. Before you commence operations of your Taco John's Restaurant and as we may from time-to-time request, you must sign and deliver to us and your bank the authorization agreement for direct payments attached as Exhibit II to the Franchise Agreement and any other documents needed to permit us to debit your bank account for the fees and other payments due to us under the Franchise Agreement or otherwise. We reserve the right to require payment of fees by other similar technology now or hereafter developed.
- 2. "Net Sales" means the total receipts from all sales of the Restaurant, including all receipts from sales made at or away from the Premises of the Restaurant and revenue from the redemption of Taco John's gift certificates, customer loyalty rewards, gift cards and other prepaid cards, and sales through mobile ordering, delivery and catering, and all other income of every kind and nature related to, derived from, or originating from the Restaurant, including proceeds of any business interruption insurance policies, but does not include sales tax or equivalent taxes. We reserve the right to modify our policies and practices regarding inclusion or exclusion of certain revenue from "Net Sales" as circumstances, business practices, and technology change.
- 3. Once implemented, when we collect the Technology Fee, we will submit a portion of it to the third-party vendors who provide the services covered by this fee on your behalf. The up to 10% portion retained by us shall be used by us for technology development and support services. Such development and support may be conducted in-house and/or with third parties, and we may compensate ourselves from the Technology Fee for any personnel time or expenditures that we incur in providing these services. The portion retained by us will be used for further development and support of technology by or for us, including changes or enhancements to ordering and AI

- platforms, the mobile app, data reporting platforms, custom development, or other technology innovations. We may increase the Technology Fee at any time, provided that the increase will never be more than the increase in actual prices charged by the vendors to provide the products and services plus 10% of the total cost to provide the products and services.
- 4. We are a member of the advertising cooperatives for the designated marketing areas in Cheyenne, Wyoming and Minneapolis, Minnesota. We have controlling voting power in the Cheyenne cooperative.
- 5. The Royalty may be reduced to 4% of Net Sales under our Multi-Unit Franchisee incentives described in Item 5, with such reduction applying for all of your Restaurants opened as part of the multi-unit group for a period of 12 months following the opening of your fourth Restaurant.

ITEM 7. YOUR ESTIMATED INITIAL INVESTMENT

Area Development Agreement

As described in Item 5 of this Disclosure Document, when you sign an ADA, you must pay a Development Fee equal to \$5,000 multiplied by the number of Taco John's Restaurants you agree to develop under the ADA. We will apply \$5,000 of the Development Fee to the Initial Franchise Fee payable under each Franchise Agreement you sign under the ADA.

The Development Fee is non-refundable under all circumstances. Except for the payment of the Development Fee and certain working capital funds that you may need to cover miscellaneous expenses to establish your business operations (which will vary greatly depending upon the circumstances of each Developer and which we are not able to estimate), no initial investment is required of you when you execute the ADA. An initial investment will be required, however, for each Taco John's Restaurant developed under the ADA, and those amounts are set forth in the following charts. You will be required to execute the Franchise Agreement for your initial Taco John's Restaurant upon signing your ADA. You should be aware that while there are no additional requirements for working capital or additional funds that apply to you as a Developer, you will need funds sufficient for the Development Fee as described above before you execute the ADA.

Franchise Agreement

We allow you to build one of the following restaurant designs when you sign a Franchise Agreement. There is a difference in the initial investment you will incur between the designs. You may also remodel an existing building to meet our image, or place a Restaurant in a non-traditional location.

Traditional Restaurants - The freestanding traditional restaurant location has a premises size typically between 1,900 to 2,250 square feet with seating for 30 to 40. It requires a lot between 26,000 and 39,000 square feet, drive-thru capability, and space to park 20 to 30 cars.

The following chart describes your estimated initial investment to establish a freestanding traditional Taco John's Restaurant.

Column 1	Column 2	Column 3	Column 4	Column 5
Expenditures	Estimated Amount or Estimated Low-High Range	Method of Payment	When Payable	To Whom Paid
Initial Franchise Fee (1)	\$40,000	Cash	In full on signing the Franchise Agreement	Us
Real Estate Purchase or Lease (2)	(See Note 2)	As arranged	As arranged	Seller or landlord
Site Preparation and Completion Costs (3)	\$250,000 to \$500,000	As arranged	As scheduled	Contractors
Construction Costs (4)	\$650,000 to \$900,000	As arranged	As scheduled	Contractors
Operating Equipment (5)	\$185,000 to \$250,000	As arranged	As arranged	Suppliers
Furniture, Fixtures and Decor (6)	\$35,000 to \$55,000	As arranged	As arranged	Suppliers
Signage and Installation (7)	\$30,000 to \$70,000	As arranged	As arranged	Manufacturers and installers
Point of Sale System (8)	\$20,000 to \$35,000	As arranged	As arranged	Suppliers
Digital Menu Boards (9)	\$35,000 to \$65,000	As arranged	As arranged	Suppliers
Initial Inventory (10)	\$15,000 to \$22,000	As arranged	As arranged	Suppliers
Grand Opening Marketing (11)	\$25,000 to \$35,000	As incurred	As arranged	Suppliers
Pre-Opening Training (Travel and Living Expenses) (12)	\$35,000 to \$48,000	As incurred	Before opening	Suppliers
Miscellaneous Opening Costs (4 and 13)	\$20,000 to \$60,000	As incurred	As arranged	Governmental authorities, providers, and other third parties
Additional Funds (14)	\$25,000 to \$40,000	As incurred	As arranged	Various suppliers and providers
TOTAL ESTIMATED INITIAL INVESTMENT (without real estate costs) (15)	\$1,365,000 to \$2,120,000			

The endcap traditional restaurant location is typically located in a shopping mall, strip mall or alternative facility with our approval. It has a premises size typically between 1,800 to 2,200 square feet with seating for 20 to 40.

The following chart describes your estimated initial investment to establish an endcap traditional Taco John's Restaurant.

Column 1 Expenditures	Column 2 Estimated Amount	Column 3 Method of	Column 4 When Payable	Column 5 To Whom Paid
Expenditures	or Estimated Low-High Range		When I ayable	10 Whom I alu
Initial Franchise Fee (1)	\$40,000	Cash	In full on signing the Franchise Agreement	Us
Real Estate Purchase or Lease (2)	(See Note 2)	As arranged	As arranged	Seller or landlord
Leasehold Improvement Costs (4)	\$350,000 to \$650,000	As arranged	As scheduled	Contractors
Operating Equipment (5)	\$185,000 to \$254,000	As arranged	As arranged	Suppliers

Column 1	Column 2	Column 3	Column 4	Column 5
Expenditures	Estimated Amount or Estimated Low-High Range	Method of Payment	When Payable	To Whom Paid
Furniture, Fixtures and Decor (6)	\$35,000 to \$55,000	As arranged	As arranged	Suppliers
Signage and Installation (7)	\$20,000 to \$38,000	As arranged	As arranged	Manufacturers and installers
Point of Sale System (8)	\$20,000 to \$35,000	As arranged	As arranged	Suppliers
Digital Menu Boards (9)	\$35,000 to \$65,000	As arranged	As arranged	Suppliers
Initial Inventory (10)	\$15,000 to \$22,000	As arranged	As arranged	Suppliers
Grand Opening Marketing (11)	\$25,000 to \$35,000	As incurred	As arranged	Suppliers
Pre-Opening Training (Travel and Living Expenses) (12)	\$35,000 to \$48,000	As incurred	Before opening	Suppliers
Miscellaneous Opening Costs (4 and 13)	\$20,000 to \$60,000	As incurred	As arranged	Governmental authorities, providers, and other third parties
Additional Funds (14)	\$25,000 to \$40,000	As incurred	As arranged	Various suppliers and providers
TOTAL ESTIMATED INITIAL INVESTMENT (without real estate costs) (15)	\$805,000 to \$1,342,000			

Some franchisees may be granted the opportunity to participate in test programs for new unit concepts being evaluated by us. Our test programs currently include two modular building designs: a DT Digital Olé (without seating) and a DT Digital Olé+ (with seating), which are further discussed in Item 1. The DT Digital Olé traditional restaurant location is a drive-thru focused building targeted for small market development. It has a premises size typically between 946 to 1,500 square feet with limited or no seating inside. It requires a lot size between 15,000 to 25,000 square feet, drive-thru capability, and space to park 5 to 10 cars.

The following chart describes your estimated initial investment to establish a DT Digital Olé traditional Taco John's Restaurant.

Column 1	Column 2	Column 3	Column 4	Column 5
Expenditures	Estimated Amount or Estimated Low-High Range	Method of Payment	When Payable	To Whom Paid
Initial Franchise Fee (1)	\$40,000	Cash	In full upon signing the Franchise Agreement	Us
Real Estate Purchase or Lease (2)	(See Note 2)	As arranged	As arranged	Seller or Landlord
Site Work (3)	\$125,000 to \$450,000	As arranged	As scheduled	Contractors
DT Digital Olé Manufacturing (4)	\$450,000 to \$650,000	As arranged	As arranged	Manufacturer
Operating Equipment (5)	\$175,000 to \$240,000	As arranged	As arranged	Suppliers
Furniture, Fixtures and Décor (6)	\$0	As arranged	As arranged	Suppliers
Signage & Installation (7)	\$20,000 to \$35,000	As arranged	As arranged	Manufacturers and Installers

Column 1	Column 2	Column 3	Column 4	Column 5
	Estimated Amount			
Expenditures	or Estimated	Method of	When Payable	To Whom Paid
	Low-High Range	Payment		
Point of Sale System (8)	\$20,000 to \$60,000	As arranged	As arranged	Suppliers
Digital Exterior Menu	\$35,000 to \$65,000	As arranged	As arranged	Manufacturers and
Boards (9)				Installers
Initial Inventory (10)	\$15,000 to \$22,000	As arranged	As arranged	Suppliers
Grand Opening Marketing	\$25,000 to \$35,000	As arranged	Set aside	Suppliers
(11)				
Pre-Opening Training	\$35,000 to \$48,000	As incurred	Before opening	Suppliers
(Travel & Living				
Expenses (12)				
Miscellaneous Opening	\$20,000 to \$60,000	As incurred	As arranged	Governmental
Costs (13)				authorities, providers,
				and other third parties
Additional Funds (14)	\$25,000 to \$40,000	As incurred	As arranged	Various suppliers and
				providers
Total Estimated Initial	\$985,000 to \$1,745,000			
Investment (without real				
estate costs) (15)				

NOTES: The following notes apply to the freestanding, endcap, and DT Digital Olé traditional Restaurant models.

- 1. <u>Initial Franchise Fee</u>. The Initial Franchise Fee is \$40,000 unless you qualify for any discounts described in our current promotional incentives. The Initial Franchise Fee is non-refundable in all circumstances once paid.
- 2. Real Estate Purchase or Lease. You may either purchase or lease the land for your Restaurant. Land purchase costs generally range between \$15 and \$45 per square foot, but such costs may vary substantially based upon the location of the site, and are generally higher in urban/suburban areas than in rural areas. Ground lease costs vary depending on the market, generally range between \$2 and \$5 per square foot per year for the site and, as with land purchase costs, the square footage costs for ground leases may vary substantially based upon the location of the site. You will consult with a real estate agent or broker approved by us for a better estimate of your real estate costs.
- 3. <u>Site Preparation and Completion Costs</u>. The cost of site improvements (such as site demolition and preparation, grading, curbing, gutters, sidewalks, paving, utility extensions, site lighting, landscaping, exterior trash enclosure) will vary widely based on the condition of the site, its size, local code requirements, as well as local construction costs. The range of costs shown is for a 26,000 to 39,000 square foot site for a freestanding traditional Taco John's Restaurant and 15,000 to 25,000 square foot site for a DT Digital Olé Taco John's Restaurant. There may be unusual costs associated with a site.
- 4. Construction Costs; Leasehold Improvement Costs. If you acquire the right to open and operate a freestanding traditional restaurant model, you will need to construct a freestanding steel or wood-frame building of approximately 1,900 to 2,250 square feet, having a synthetic exterior finish according to brand standards and commonly accepted quick service restaurant interior finishes, which may include wall murals, painting and tile. The costs for the building will vary based upon local construction costs, local codes and standards, as well as factors such as the time of year the building is constructed, the availability of a skilled work force, construction financing and other

variables. You are required to use design and construction professionals approved by us. In addition, you will be required to obtain various permits and licenses associated with the development and construction of a new restaurant or for improvements to an existing restaurant. These amounts are reflected in this chart as Miscellaneous Opening Costs.

If you acquire the right to open and operate an endcap restaurant model, your costs will vary greatly. The square footage of the location, the as-is condition, tenant improvement allowances from the landlord, and the type and availability of utilities and services are typically the largest monetary items affecting your leasehold improvement buildout costs. A location that previously served as a restaurant may significantly reduce these costs depending on final layout. The costs shown in the chart are estimated costs to build out space in an endcap leased location of approximately 1,800 to 2,200 square feet. You may be able to lower or even eliminate these costs if you have obtained a built-to-suit lease space, where the landlord will construct the tenant improvement and incorporate the costs of it into your monthly rent spread out over the life of the lease, or the location is "move-in-ready" space.

In the event an existing freestanding or endcap facility is being converted to a Taco John's Restaurant, construction costs are estimated to range between \$350,000 and \$750,000, but costs may vary substantially depending upon the prior use, condition, and size of the facility.

If you acquire the right to open and operate a DT Digital Olé restaurant model, you will need to work with our approved vendor who will build the steel building in a warehouse and ship to your restaurant location. The costs for the building will vary based upon local construction costs, local codes and standards, as well as factors such as the time of year the building is constructed, the availability of a skilled work force, construction financing and other variables. In addition, you will be required to obtain various permits and licenses associated with the development and construction of a new restaurant or for improvements to an existing restaurant. These amounts are reflected in this chart as Miscellaneous Opening Costs.

- 5. **Operating Equipment**. The estimate of your investment for operating equipment is based upon equipping a newly constructed facility with specified food preparation and storage, and equipment. These costs will increase if you purchase additional optional equipment for the Restaurant.
- 6. <u>Furniture, Fixtures and Décor</u>. This estimate is based on a freestanding or endcap Restaurant with a range of 20 to 40 seats. Your costs may differ depending on the type of facility and the location and seating capacity of your Restaurant. A DT Digital Olé Restaurant will typically have limited or no seating.
- 7. <u>Signage and Installation</u>. Your investment in signage may be higher or lower depending on the size and height requirements mandated by local laws and regulations.
- 8. <u>Point of Sale System</u>. You must purchase a Point of Sale System from our approved supplier. The Point of Sale System includes electronic cash registers, freestanding kiosks, kitchen video monitors, receipt printers, a drive-thru speed of service timing system, and headset system.
- 9. <u>Digital Menu Boards</u>. You must purchase exterior digital menu boards, and you have the option to purchase interior digital menu boards, from our approved supplier. You must pay the license fee to us or our third-party supplier as we direct.
- 10. <u>Initial Inventory</u>. You must purchase all food products and supplies from distributors or suppliers that we approve, and which may include us or our affiliates.

11. **Grand Opening Marketing.** You are required to implement a grand opening local advertising and initial promotion of the Restaurant approved in advance by us. Although not required under the Franchise Agreement, we anticipate that many Franchisees will spend more than the minimum amount required to promote the opening of their Restaurants.

If you open a Restaurant in a market that is defined by us as underpenetrated, you must conduct a grand opening advertising program that we will develop with you that will be designed to introduce Taco John's to the market. In markets defined by us as underpenetrated, you may be required to spend up to an additional \$10,000, reflected in the high estimate in this table. An underpenetrated market is one where there is limited awareness of the Taco John's concept and minimal Taco John's advertising penetration.

- 12. **Pre-Opening Training (Travel and Living Expenses)**. This amount includes airfare, hotel, food and transportation expenses for four people for 4 to 6 weeks of training in a Taco John's Restaurant we designate.
- 13. <u>Miscellaneous Opening Costs</u>. This estimate includes the costs of permits and fees required by local governmental rules and regulations that will be associated with your project development, site preparation and building construction. These fees include standard permitting fees, but do not include tap fees or impact fees. Many jurisdictions impose these fees. In some cases, they may be paid by your developer. It also reflects the costs for professional services you may incur for architects, engineers, attorneys and/or accountants.
- Additional Funds. The disclosure laws require us to include this estimate of costs and expenses to operate your Taco John's Restaurant during the "initial period" of the business, which is defined as three months or a longer period if "reasonable for the industry." We are not aware of any established longer "reasonable period," so our disclosures cover a three-month period. This estimate reflects the funds you should have available to cover operating costs during the initial period as a cash reserve for incidental costs not covered by cash flow during the start-up period and that are not included in any of the above categories. These figures are estimates, and we cannot guarantee that you will not incur additional expenses. Such operating costs may vary substantially depending upon many factors, including your management skill, experience, and business acumen, the size and location of the Restaurant, the number of persons you choose to employ and their experience and efficiency, local economic conditions, local market conditions, prevailing wage rates in your community, competition, and the sales level reached in the initial period. This item does not include the cost of any financing, interest, or the amount of debt service obligation that you may undertake.
- 15. <u>Total Estimated Initial Investment</u>. This amount reflects our current estimate of your initial investment for a traditional Taco John's Restaurant. The amounts shown are our best estimates of the amounts that Franchisees may spend for the purposes indicated. However, your actual costs may be higher or lower depending upon your particular circumstances including the size and/or location of the Premises of the Restaurant and whether you lease or purchase the land and/or building for the Premises, construction and labor costs in the area, your discretionary expenditures, and other factors, such as tenant allowances negotiated with the landlord, if any.

We relied on more than 56 years of experience operating Taco John's Restaurants to compile these estimates and on the information we have obtained from our current Franchisees. The estimates are averages and reflect expenses for a single Restaurant under typical circumstances. Your situation may not be typical, and unforeseen circumstances may arise. You should review these figures

carefully with various business advisors before making a decision to purchase a Taco John's Restaurant franchise. You should be aware that a variety of factors such as market demand, demographics, the extent to which you follow our methods and procedures, local economic conditions and your own business experience may affect many of the described expenses. We do not offer financing directly or indirectly for any part of the initial investment.

Non-traditional Restaurant - The non-traditional restaurant is generally attached to or within another structure. It requires 675 to 2,000 square feet with seating for 0 to 40.

The following chart describes your estimated initial investment to establish a non-traditional Taco John's Restaurant.

Column 1	Column 2	Column 3	Column 4	Column 5
Expenditures	Estimated Amount or Estimated Low-High Range	Method of Payment	When Payable	To Whom Paid
Initial Franchise Fee (1)	\$15,000	Cash	In full upon signing the Franchise Agreement	Us
Real Estate Lease (2)	(See Note 2)	As arranged	As arranged	Seller or landlord
Leasehold Improvements (3)	\$125,000 to \$175,000	As arranged	As scheduled	Contractors
Operating Equipment (4)	\$100,000 to \$175,000	As arranged	As arranged	Suppliers
Furniture, Fixtures and Decor (5)	\$5,000 to \$35,000	As arranged	As arranged	Suppliers
Signage and Installation (6)	\$5,000 to \$25,000	As arranged	As arranged	Manufacturers and installers
Point of Sale System (7)	\$20,000 to \$30,000	As arranged	As arranged	Suppliers
Digital Menu Boards (8)	\$10,000 to \$12,000	As arranged	As arranged	Suppliers
Initial Inventory (9)	\$15,000 to \$22,000	As arranged	As arranged	Suppliers
Grand Opening Marketing (10)	\$25,000	As incurred	Set aside	Suppliers
Pre-Opening Training (Travel and Living Expenses) (11)	\$35,000 to \$48,000	As incurred	Before opening	Suppliers
Miscellaneous Opening Costs (3 and 12)	\$10,000 to \$30,000	As incurred	As arranged	Governmental authorities, providers, and other third parties
Additional Funds - (13)	\$25,000 to \$40,000	As incurred	As arranged	Various suppliers and providers
TOTAL ESTIMATED INITIAL INVESTMENT (without real estate costs) (14)	\$390,000 to \$632,000			

NOTES:

- 1. <u>Initial Franchise Fee</u>. The Initial Franchise Fee for a non-traditional Taco John's Restaurant is currently \$15,000. The Initial Franchise Fee is non-refundable in all circumstances once paid.
- 2. **Real Estate Lease**. You will typically enter into a lease for your non-traditional Restaurant. Lease costs generally range between \$15 and \$45 per square foot per year but the square footage costs may vary substantially based upon the location of the site. You will consult with a real estate agent or broker approved by us for a better estimate of your real estate costs.

- 3. <u>Leasehold Improvements.</u> Costs vary widely, but generally range between \$125,000 and \$175,000. This estimate assumes that you will construct tenant finishes having a synthetic stucco exterior finish and commonly accepted quick service restaurant interior finishes, which may include wallpaper, painting, tile and /or wood paneling. The costs for the project will vary based upon local construction costs, local codes and standards, as well as factors such as the availability of a skilled work force, construction financing and other variables. In addition, you will be required to obtain various permits and licenses associated with the development of a new restaurant or for improvements to an existing restaurant. These amounts are reflected in this chart as Miscellaneous Opening Costs.
- 4. **Operating Equipment**. The estimate of your investment for operating equipment is based upon equipping a newly constructed facility with specified food preparation and storage, and equipment. These costs will increase if you purchase additional optional equipment for the Restaurant.
- 5. <u>Furniture, Fixtures and Décor</u>. This estimate is based on a Restaurant with a range of 0 to 40 seats. Your costs may differ depending on the type of facility and the location and seating capacity of your Restaurant.
- 6. <u>Signage and Installation</u>. Your investment in signage may be higher or lower depending on the size and height requirements mandated by local laws and regulations.
- 7. <u>Point of Sale System</u>. You must purchase a Point of Sale System from our approved supplier. The Point of Sale System includes electronic cash registers, freestanding kiosks, kitchen video monitors, receipt printers, and a drive-thru speed of service timing system and headset system if the non-traditional location has a drive-thru.
- 8. <u>Digital Menu Boards</u>. You must purchase exterior digital menu boards, and you have the option to purchase interior digital menu boards, from our approved supplier. You must pay the license fee to us or our third-party supplier as we direct.
- 9. <u>Initial Inventory</u>. You must purchase all food products and supplies from distributors or suppliers that we approve, and which may include us or our affiliates.
- 10. <u>Grand Opening Marketing</u>. You are required to implement a grand opening local advertising and initial promotion of the Restaurant approved in advance by us. However, because non-traditional restaurants are generally attached to or located within other structures with a captive audience, the grand opening marketing and initial promotion of the Restaurant will be determined based on the specific circumstances of the location and venue.
- 11. <u>Pre-Opening Training (Travel and Living Expenses)</u>. This amount includes airfare, hotel, food and transportation expenses for four people for 4 to 6 weeks of training in a Taco John's Restaurant we designate.
- 12. <u>Miscellaneous Opening Costs</u>. This estimate includes the costs of permits and fees required by local governmental rules and regulations that will be associated with your project development. It also reflects the costs for professional services you may incur for architects, engineers, attorneys and/or accountants.
- 13. <u>Additional Funds</u>. The disclosure laws require us to include this estimate of costs and expenses to operate your Taco John's Restaurant during the "initial period" of the business, which is defined

as three months or a longer period if "reasonable for the industry." We are not aware of any established longer "reasonable period," so our disclosures cover a three-month period. This estimate reflects the funds you should have available to cover operating costs during the initial period as a cash reserve for incidental costs not covered by cash flow during the start-up period and that are not included in any of the above categories. These figures are estimates, and we cannot guarantee that you will not incur additional expenses. Such operating costs may vary substantially depending upon many factors, including your management skill, experience, and business acumen, the size and location of the Restaurant, the number of persons you choose to employ and their experience and efficiency, local economic conditions, local market conditions, prevailing wage rates in your community, competition, and the sales level reached in the initial period. This item does not include the cost of any financing, interest, or the amount of debt service obligation that you may undertake.

14. <u>Total Estimated Initial Investment</u>. This amount reflects our current estimate of your initial investment for a Taco John's non-traditional Restaurant. The amounts shown are our best estimates of the amounts that Franchisees may spend for the purposes indicated. However, your actual costs may be higher or lower depending upon your particular circumstances including the size and/or location of the Premises of the Restaurant and whether you lease or purchase the land and/or building for the Premises, construction and labor costs in the area, your discretionary expenditures, and other factors, such as tenant allowances negotiated with the landlord, if any.

We relied on more than 56 years of experience operating Taco John's Restaurants to compile these estimates and on the information we have obtained from our current Franchisees. The estimates are averages and reflect expenses for a single Restaurant under typical circumstances. Your situation may not be typical, and unforeseen circumstances may arise. You should review these figures carefully with various business advisors before making a decision to purchase a Taco John's Restaurant franchise. You should be aware that a variety of factors such as market demand, demographics, the extent to which you follow our methods and procedures, local economic conditions and your own business experience may affect many of the described expenses. We do not offer financing directly or indirectly for any part of the initial investment.

The amounts set forth in the above tables are non-refundable.

ITEM 8. RESTRICTIONS ON SOURCES OF PRODUCTS AND SERVICES

You must follow our standards and specifications in operating your Taco John's Restaurant. We formulate our standards and specifications according to our expectations of price and quality on an itemby-item basis, and we generally do not issue these specifications and standards to Franchisees.

Operations Manual

We provide access to you, during the term of the franchise, one or more manuals that contain mandatory and suggested specifications, standards and operating procedures and rules, as well as information that relates to your other obligations under the Franchise Agreement and the operation of your Restaurant (the "Operations Manual"). The entire contents of the Operations Manual are confidential and will remain our property. We may add to and otherwise modify the Operations Manual if we deem it appropriate in our discretion to improve the standards of service, or product quality or the efficient operation of Taco John's Restaurants, to protect or maintain the goodwill associated with the Marks, to meet competition, or to otherwise improve or benefit the Restaurant System. We provide you the Operations Manual in electronic format via The Olé! Way online learning management system described in Items 6

and 11. The Operations Manual, as modified periodically, and mandatory specifications, standards and operating procedures and rules we prescribe periodically and communicate to you in writing, or by other electronically stored data, including through a restricted website on the Internet, will constitute provisions of the Operations Manual. All references in this Disclosure Document to the Operations Manual include the provisions of the Operations Manual and all mandatory specifications, standards and operating procedures and rules.

Our Operations Manual is provided online and consists of 250 pages plus linked documents. The Table of Contents is attached to this Disclosure Document as Attachment H.

The Operations Manual and other standards and specifications we provide are designed to protect our reputation and the goodwill of the Marks. They are not designed, nor are they intended, to control the day-to-day operations of your Taco John's Restaurant.

Approved Real Estate, Design, and Construction Professionals

We require you to use only qualified real estate brokers approved by us to locate a site for your Restaurant. We also require you to use only qualified building design professionals, including architects and engineers, that are approved by us, for the build-out and improvement of your Restaurant. We also require you to complete the construction and build-out of your Restaurant using only a qualified general contractor approved by us. Upon reasonable request, we will consider the approval of other professionals, but we are under no obligation to approve any particular professional or any minimum number of them. All professionals are required to be licensed if required by the state in which the Restaurant is located. You must deal directly with any such real estate brokers, architects, engineers, or general contractors. We will not receive any payments from approved real estate brokers, architects, engineers, or general contractors, and we will not have any liability to you for their acts or omissions.

Approved Modular Building Manufacturers

If you build a DT Digital Olé model Restaurant, we require you to use only the modular building manufacturers approved by us. We will not receive any payments from approved modular building manufacturers and we will not have any liability to you for their acts or omissions.

Equipment, Fixtures, Furnishings and Signs

When you develop and operate your Restaurant, you must use only the brands, types, and/or models of equipment and furniture, computer hardware and software, fixtures, furnishings, digital menu boards, and signs that meet our standards and specifications. You may only purchase these from our approved suppliers. You will be required to purchase certain types of restaurant equipment and supplies for the kitchen of the Restaurant. We approve certain manufacturers to supply you with these supplies and equipment items. We may require that you pay a fee to us or our designated suppliers for purposes of obtaining access to our learning management system or financial reporting system, and we may implement a Technology Fee to cover various technology services in the future.

Except as described above, neither we nor our affiliates will derive revenue as a result of your purchase of such items in accordance with our standards and specifications other than as described above.

Condition and Appearance of Restaurant

You must maintain the condition and appearance of the Restaurant, its equipment, furniture, fixtures, furnishings, signs, and the Premises according to our standards and consistent with the image of a

Taco John's Restaurant as an efficiently operated business that offers high quality products, beverages and services and observes the highest standards of cleanliness, sanitation, efficient, courteous service and pleasant ambiance. You must meet these standards during the term of the Franchise Agreement by (i) thoroughly cleaning, repainting and redecorating the interior and exterior of the Premises at reasonable intervals, (ii) performing interior and exterior repair of the Premises (iii) repairing or replacing damaged, worn out, outdated or obsolete equipment, furniture, furnishings and signs; and (iv) engaging a pest control service on an ongoing basis that meets our specifications.

You may not make any material alterations to the Premises, or to the appearance of the Restaurant, as originally developed, without our prior approval. If you do so, we have the right, at your expense, to rectify alterations we have not previously approved. You must replace or add new equipment when we require to meet changing standards, or to accommodate new menu items, or new preparation methods, and service methods, including the manner or method used to deliver food to the guest. At our request, you must remodel, renovate, refresh, reimage, refurbish, and modernize the Premises and the Restaurant, including the building design, parking lot, landscaping, equipment, signs, interior and exterior décor items, fixtures, furnishings, trade dress, and color scheme, to meet our then-current standards, specifications, and design criteria required of new Franchisees, and including, without limitation, such structural changes, remodeling, and redecoration, and such modifications to existing improvements as may be necessary to do so, all within the time frame we require.

Material alterations to remodel or renovate will not be required more often than once every 10 years. Non-material alterations to refresh, reimage, refurbish, or modernize will not be required more often than once every five years. The frequency limitations do not apply to your obligations to regularly maintain, repair, and replace the Restaurant and related property, and they also do not apply to your obligation to replace or add equipment to meet our standards. Material or non-material alterations may be required within 12 months after a Transfer. If you do not comply with equipment or remodel requirements, you may be required to pay a non-compliance fee.

You may place or display at the Premises only the signs, emblems, designs, artwork, lettering, logos, and display and advertising materials that we approve.

Compliance with Laws and Good Business Practices

You agree to secure and maintain in force in your name all required licenses, permits, and certificates relating to the construction and operation of the Restaurant. You also agree to construct and operate the Restaurant in full compliance with all applicable laws, ordinances, and regulations.

Additional Quality Control Requirements

You further agree to notify us of any crisis within your restaurant or organization that could result in negative media attention in accordance with our crisis management procedures contained in the Operations Manual.

You must participate, at your sole expense, in a guest experience program which includes a mystery shop service and guest satisfaction survey service or such other similar service that we may designate from time to time to evaluate the conduct and quality of your Restaurant, including but not limited to cleanliness, food quality, merchandising and customer service. You must use our approved providers of such services and utilize the service at a frequency and in the manner prescribed by us. We have the right to review the results of the service evaluations and reserve the right to charge you for the cost of resolving guest complaints.

Insurance

Before you begin constructing your Restaurant, you must obtain and have in effect the following insurance coverage under policies issued by carriers of your choice whom we approve, unless we designate a required carrier for systemwide brand protection policies: (1) comprehensive public, product and automobile liability insurance against claims for personal injury, death, and property damage caused by or occurring in conjunction with the operation of the Restaurant or your conducting business under the Franchise Agreement under one or more policies of insurance containing minimum liability coverage we prescribe from time to time; (2) general casualty insurance including fire and extended coverage, vandalism, and malicious mischief insurance for the replacement value of the Restaurant and its contents; (3) cyber liability insurance; (4) loss of business income and incident response for foodborne illness insurance; (5) employment practices liability insurance, (6) umbrella insurance, (7) workers' compensation insurance required by your state, and (8) business interruption insurance. We may require that you carry additional insurance policies in the future.

You must maintain the policies described above in force during the entire term of the Franchise Agreement. You are required to comply with applicable state laws in obtaining sufficient coverage. We may periodically establish the amounts of coverage required under these insurance policies and require different or additional kinds of insurance at any time, including umbrella liability insurance, to reflect inflation, additional risks, changes in law or standards of liability, higher damage awards or other relevant changes in circumstances.

Currently, we require for each Restaurant location comprehensive general liability insurance coverage of at least \$1 million per occurrence and \$2 million in the aggregate; cyber liability insurance coverage of at least \$50,000; loss of business income and incident response for foodborne illness insurance coverage of at least \$1 million (which we purchase on behalf of the system as discussed below); and employment practices liability insurance coverage of at least \$25,000. In addition to those required coverages, you must also maintain a separate \$1 million umbrella policy for each Restaurant location. If you operate two to five Restaurants, your umbrella policy requirement is \$2 million total umbrella coverage. If you operate six to nine Restaurants, your umbrella policy requirement is \$5 million total umbrella coverage. If you operate ten or more Restaurants, your umbrella policy requirement is \$10 million total umbrella coverage. We currently require you to carry business interruption insurance that provides coverage for actual loss sustained over a 12-month time period.

We currently maintain the insurance coverage for loss of business income and incident response for foodborne illness, which covers our locations and each individual franchised location in the required amounts. We may designate a specific required carrier and collect your share of the premium for this insurance from you and pay it to the insurance carrier on your behalf. We may require you to obtain this insurance directly from the insurance carrier in the future.

Each insurance policy must be endorsed to name us as Additional Insured – Grantor of Franchise, and must provide for 30 days' advance written notice to us of any material modification, cancellation, or expiration of the policy. Before the expiration of the term of each insurance policy, you must furnish us with a certificate of insurance evidencing coverage in the required amounts and provide an endorsement naming us as Additional Insured – Grantor of Franchise. We reserve the right to require you to send us a complete copy of each insurance policy on request, evidence of the payment of the premium for each, and copies of the Additional Insured Endorsements. If you do not maintain the required insurance coverage, or do not furnish us with satisfactory evidence of your insurance coverage and the payment of the premiums for them, we may obtain the required insurance coverage on your behalf. If we do that, you must fully cooperate with us in our effort to obtain the insurance policies, promptly execute all forms or instruments

required to obtain or maintain the insurance, allow any inspections of the Restaurant which are required to obtain or maintain the insurance, and pay us for any costs and premiums we incur.

Neither we nor our affiliates will derive revenue as a result of your purchase of required insurance coverage according to our standards and specifications.

Gift Cards and Loyalty Program

You must participate in our gift card program by offering customers the ability to buy and make purchases with gift cards, using the system we specify (the "Gift Card System"). The Gift Card System is designed to work with your Point of Sale System. The Gift Card System is provided and managed by the unaffiliated third-party supplier that we designate. Your restaurant must use the designated equipment that is purchased from approved suppliers. In addition, you must sign a participation agreement with our approved provider for transaction processing. Currently, the Advertising Fund pays for the production costs of the gift cards. We charge an annual gift card fee as described in Item 6 above, which fee is used to pay the third-party gift card supplier for the transaction costs. You will be required to complete our Gift Card Onboarding and Change Form. A copy of our Gift Card Onboarding and Change Form is attached to this Disclosure Document as Attachment J.

We require participation in our guest loyalty/rewards program, and you must participate in it in the manner we prescribe using the system we specify.

Approved Products

We have developed standards and specifications for food products, ingredients, beverages, materials, and supplies incorporated in or used in marketing, preparing, serving, packaging, and delivering prepared food and beverage products authorized for sale at Taco John's Restaurants. We have approved and will continue to periodically approve suppliers and distributors of products that meet our standards and requirements. We will provide these standards to prospective suppliers as described below. You must purchase all food products, beverages, menus, paper and plastic products, packaging or other materials, and utensils only from distributors and suppliers we have approved.

Collectively, the purchases and leases you will make according to our requirements, standards and specifications will represent virtually 100% of your total purchases and leases to establish, and virtually 100% of your total purchases and leases to operate your Restaurant.

Approved and Designated Suppliers and Distributors

Except as described in this Item 8, you are not required to purchase or lease from us or our designees any goods, services, supplies, fixtures, equipment, inventory, or real estate relating to the establishment or operation of a Taco John's Restaurant under the terms of the agreements described in this Disclosure Document. There are no approved suppliers in which any of our officers owns an interest.

We may approve a single supplier for many of the products and supplies you will use in your Taco John's Restaurant.

If your Restaurant is located outside of our primary distribution area, you may be required to pay an additional delivery fee ("cross dock fee") to the distributor.

We currently have purchasing arrangements with certain national beverage suppliers for fountain beverage ingredients and products. You are required to offer those fountain beverages in your Taco John's Restaurant and will be generally prohibited from using other beverage products in your Restaurant.

You will be required to obtain from approved suppliers, and use in your Taco John's Restaurant, specific equipment related to the fountain beverage services.

Some advertising materials are available without cost from the Advertising Fund (as defined in Item 11). Certain advertising materials produced by the Advertising Fund are only available for purchase from the Fund. Additional promotional materials are available for purchase directly from us or from our approved supplier.

We may negotiate other purchase agreements for the benefit of franchisees and the Restaurant System. Currently, we have a purchasing arrangement with a national distributor as our products and supplies distributor. You are required to purchase those products and supplies distributed by our approved distributor and its affiliates. You will enter into any and all agreements and documents required of franchisees by the approved distributor.

We may approve other suppliers for any product or service and may approve a single supplier only as to certain products or services. If we require, you must purchase from the suppliers or distributors we specify. We will identify all designated and approved suppliers in our Operations Manual or through other written or electronic communications. We may concentrate purchases with one or more suppliers to obtain lower prices or the best advertising support or services for any group of Taco John's Restaurants franchised or operated by us, but we will use reasonable efforts to negotiate agreements with suppliers which, in our good faith belief, are in the best interest of all Taco John's Franchisees. Approval of a supplier may be conditioned on requirements relating to the frequency of delivery, concentration of purchases, standards of service or other criteria, and approval may be temporary, pending our continued evaluation of the supplier.

Under the Franchise Agreement, you authorize us to act as your sole agent to enter into contracts with parties that offer promotions, discounts or other programs under which you would receive rebates or marketing allowances ("Rebates") from handling items offered for sale by such parties. Except as discussed below, all Rebates will be paid to us and will be our sole property to be used at our discretion, unless we designate otherwise. In certain situations, we may agree with certain suppliers for portions of the Rebates to be paid to you, but we have no obligation to arrange for any particular Rebate amounts to be paid to you.

As of the date of this Disclosure Document, our beverage suppliers pay us Rebates between \$0.80 to \$1.00 per gallon purchased by our franchisees. Other suppliers pay us Rebates of between \$0.08 and \$2.26 per case of franchisee purchases. Certain suppliers also pay us fixed amounts of Rebates, which in our last fiscal year totaled \$127,000.

Our agreement with our products and supplies distributor provides that it shall pay us an administrative allowance of \$0.10 per case of franchisee purchases. Any administrative allowance amounts paid to us are currently required to be used at our discretion to address supply chain issues or other supply chain initiatives.

Except as discussed above, by signing the Franchise Agreement you assign all of your right, title and interest in all Rebates to us, and authorize us to furnish any proof of purchase evidence as may be required in accordance with such contracts.

If you would like to purchase any items or services from an unapproved supplier, you must submit to us a written request for approval of the proposed supplier. Upon receipt of your request to approve an

alternate supplier, we will furnish the supplier with our relevant standards and specifications (except standards and specifications which are or include trade secrets) for suppliers of items or services of the type to be furnished by the proposed supplier as long as the proposed supplier agrees to execute a confidentiality agreement in a form acceptable to us. We will have the right to inspect the proposed supplier's facilities, and to require product samples from the proposed supplier to be delivered at our option either directly to us or to any independent, certified laboratory which we designate for testing in order to determine if the proposed supplier meets our requirements for approved suppliers. We do not generally make available to our Franchisees the criteria used in approving suppliers. We will inform you of our decision to approve or disapprove a proposed supplier within a reasonable time considering the nature of the products and specifications we must review. We reserve the right to charge the proposed supplier a fee (not to exceed the reasonable cost of the inspection and the actual cost of the testing) to make the evaluation. We reserve the right to revoke our approval upon written notice to the supplier and any Franchisees purchasing from that supplier if the supplier does not continue to meet any of our criteria. The time for our approval or disapproval of a proposed supplier ranges from 90 to 120 days from the date we receive all required information about the supplier.

You may not contract with alternative suppliers who meet our criteria unless and until they are approved by us as described above.

Additional Information.

We may derive revenue from purchases or leases of inventory, supplies and equipment from any approved supplier of ours on account of their dealings with you and other franchisees. Currently, any amounts we receive as Rebates as a result of purchases of products or services by Franchisees from approved vendors are retained by us. In the past fiscal year ending December 31, 2024, we had \$36,370,278 in total revenues. Of these amounts, \$1,184,397 (3.3% of our total revenues) was derived based on Rebates paid by suppliers as described above. However, in the last fiscal year, all such Rebates were contributed by us to the Advertising Fund on a full pass-through basis, and funds from those Rebates were spent for the benefit of all franchisees. Any amounts we receive as administrative allowances as a result of purchases of products or services by Franchisees from approved suppliers are retained by us. However, in the last fiscal year, all such administrative allowance amounts were used by us to address supply chain issues on behalf of franchisees such as inventory disposition and minimizing outages, and we therefore did not retain any revenues from those administrative allowances.

We do not provide you or other Franchisees material benefits based on purchases from approved suppliers. We do not derive revenues from real estate leases. Neither we nor any affiliated or parent company of ours currently derives any revenues from the sale of any products or services to our franchisees, but we reserve the right to do so in the future. Neither we nor any affiliated or parent company of ours currently derive any revenue from the lease of any products or services to our franchisees, but we reserve the right to do so in the future.

Except as is described in this Item 8, you do not receive a material benefit from us based on your use of any particular designated or approved sources or your purchase of particular products or services.

ITEM 9. FRANCHISEE'S OBLIGATIONS

This table lists your principal obligations under the Franchise Agreement and other agreements. It will help you find more detailed information about your obligations in these agreements and in other items of this Disclosure Document.

Obligation	Section in Agreement	Item in Disclosure
Obligation	Section in Agreement	Document
(a) Site selection and acquisition/ lease	Section 3.1-3.3 of Franchise Agreement.	Items 7, 8, 11 and 12
(b) Pre-opening purchases/leases	Section 3.1 - 3.5 of Franchise Agreement.	Items 5, 7, 8 and 11
(c) Site development and other pre-opening requirements	Article 3 of Franchise Agreement.	Items 5, 7, 8 and 11
(d) Initial and ongoing training	Article 4 of Franchise Agreement.	Items 5, 6, 7 and 11
(e) Opening	Section 3.6 of Franchise Agreement.	Item 11
(f) Fees	Article 3 and 4 of ADA, Article 9 of Franchise Agreement, Section 6 of Non- Traditional Facility Rider.	Items 5 and 6
(g) Compliance with standards and policies/Operations Manual	Sections 5.2 and Article 10 of Franchise Agreement, Sections 7 through 10 of Non- Traditional Facility Rider.	Items 11, 14 and 16
(h) Trademarks and proprietary information	Section 2.E of ADA, Article 6 of Franchise Agreement.	Items 13 and 14
(i) Restrictions on products/services offered	Sections 10.4 and 10.5 of Franchise Agreement.	Items 8 and 16
(j) Warranty and customer service requirements	Not Applicable.	Item 16
(k) Territorial development and sales quotas	Article 4 of ADA, Article 3 of Franchise Agreement.	Item 12
(l) Ongoing product/service purchases	Section 10.5 of Franchise Agreement.	Item 8
(m) Maintenance, appearance and remodeling requirements	Section 10.2 of Franchise Agreement.	Item 11
(n) Insurance	Section 10.8 of Franchise Agreement.	Items 7 and 8
(o) Advertising	Sections 9.3 and 9.5 and Article 11 of Franchise Agreement, Section 6 of Non-Traditional Facility Rider.	Items 6 and 11
(p) Indemnification	Section 11.B of ADA, Sections 6.5 and 7.3 of Franchise Agreement, Section 5 of Non-Traditional Facility Rider.	Item 6
(q) Owner's participation/ management and staffing	Sections 2.2 and 10.7 and Article 4 of Franchise Agreement.	Item 15
(r) Records/reports	Article 12 of Franchise Agreement.	Item 6
(s) Inspections/audits	Article 13 of Franchise Agreement.	Item 6
(t) Transfer	Article 8 of ADA, Article 14 of Franchise Agreement.	Items 6 and 17
(u) Renewal	Section 2.4 of Franchise Agreement.	Item 17
(v) Post-termination obligations	Section 7.C, D, and F of ADA, Article 16 of Franchise Agreement.	Item 17
(w) Non-competition covenants	Article 9 of ADA, Sections 8.3 and 16.5 of Franchise Agreement.	Item 17
(x) Dispute resolution	Article 13 of ADA, Section 17.10 of Franchise Agreement.	Item 17

Obligation	Section in Agreement	Item in Disclosure	
		Document	
(y) Gift Cards	Sections 9.2, 9.4, 10.4, and 11.3 of Franchise Agreement.	Items 6 and 8	

ITEM 10. FINANCING

Neither we nor any agent or affiliate of ours offer direct or indirect financing to our franchisees. We do not guarantee any notes, leases or other obligations of our franchisees.

ITEM 11. FRANCHISOR'S ASSISTANCE, ADVERTISING, COMPUTER SYSTEMS AND TRAINING

Except as listed below, we are not required to provide you with any assistance.

OBLIGATIONS TO BE PERFORMED BEFORE COMMENCING OPERATIONS

Pre-Approval Operational Assessment

Before you sign an ADA or Franchise Agreement, we have the right to require all financially qualified and otherwise pre-qualified franchisee candidates to attend an Operational Assessment ("OA"). There is no cost paid by you to us to participate in the OA, but you are responsible for travel expenses. The objective of the OA is to provide both you and us with sufficient information about your in-store operational disposition and temperament to help predict your performance in the initial training program, including your ability to successfully handle customer contact, learn the fundamentals of Mexican quick-service restaurant operations, understand the challenges and rigors of the food service industry, and comprehend basic store administrative people leadership and management functions. The total time commitment for the OA is a minimum of three hours at a Taco John's Restaurant during peak hours. You will be provided with a summary report checklist and recommendation by the responsible trainer. Before participation in the OA, you must sign a standard non-disclosure agreement and release of liability in the form attached as Attachment L. Satisfactory completion of the OA does not guarantee that we will enter into a Franchise Agreement with you, or that you will have success in the initial training program or its tests, or in actual operation of a Restaurant, and we make no warranties or representations that completion of the OA will result in the grant of a Taco John's Restaurant franchise or ensure the success of your business.

Area Development Agreement

If you sign an ADA, we will grant you the right to locate and develop sites to own and operate multiple Taco John's Restaurants within the Development Area so long as you are in compliance with the ADA and the Franchise Agreements previously executed pursuant to the ADA. You must execute a separate Franchise Agreement for each Restaurant to be established pursuant to the ADA. The Franchise Agreement for each Restaurant developed pursuant to the ADA will be our then-current form of Franchise Agreement, as applicable, which may be substantially different from the Franchise Agreements described in this Disclosure Document. (ADA Sections 3 and 4.)

Franchise Agreement

Before you open the Restaurant, we will:

- 1. Restaurant Development. Evaluate your site and accept it if it meets our criteria; provide you with conceptual floor plans and specifications for construction, equipment, computer hardware and software, furniture, fixtures, furnishings, and signs; review your construction documents and hold regularly-scheduled Stage Gate New Restaurant Opening meetings. We will conduct at least one inspection of the Restaurant during construction (Franchise Agreement-Article 3).
- 2. <u>Training</u>. Provide initial training for one owner and the number of managers required for your Restaurant specified in the Operations Manual, as more fully described below (Franchise Agreement-Article 4).

OBLIGATIONS TO BE PERFORMED DURING OPERATIONS

Franchise Agreement

During the operation of your Restaurant, we will:

- 1. <u>Guidance, Assistance and Additional Training</u>. Provide guidance, assistance and additional training on how to operate and manage a Taco John's Restaurant as more fully described below (Franchise Agreement Section 4.3).
- 2. <u>Operations Manual</u>. Make available to you our Operations Manual (Franchise Agreement Section 5.2).
- 3. <u>Marketing</u>. Provide you with marketing and promotional materials as more fully described below (Franchise Agreement Article 11).
- 4. Pricing. At our discretion, provide guidance and assistance to you in recommending the prices to be charged by you for the services and products provided by you. At our discretion and to the extent permitted by law, we may periodically provide specifications to you for any maximum or minimum prices we establish to be charged by you for products and services in your Restaurant for limited time offerings or offers and items we designate for inclusion in our value programs. You must comply with the maximum and minimum pricing guidelines we establish. If we do not establish pricing limits for certain products and services, we may provide you with suggested prices for those items at our discretion. In that case, any prices that we recommend to you are merely recommendations (Franchise Agreement Section 10.4(j)).
- 5. <u>License of Marks</u>. Permit you to use our Marks in accordance with our specifications (Franchise Agreement Sections 2.1 and 6.2).
- 6. <u>Use of Restaurant System</u>. Permit you to use our Restaurant System, as it may be modified (Franchise Agreement Section 2.1).

Advertising Requirements

You must use your best efforts to promote and advertise your Restaurant and you must participate in all national or special advertising and promotional programs we establish in the manner we direct.

We will maintain, and provide to you upon your written request, a list of approved advertising and promotional materials which may be used by Taco John's Franchisees. You must submit to us for our approval samples of all local advertising and promotional materials which are not on the list of approved materials before using them. We will provide approval or disapproval within five business days after the date we receive the materials. You agree not to use any advertising or promotional materials that we have disapproved.

Grand Opening Marketing

Under the Franchise Agreement, you must set aside a sum of money that you will spend for the grand opening advertising of the Restaurant within the first 6 to 12 weeks. Your grand opening advertising program must be planned with the assistance of our marketing department and be approved by us. We require you to spend a minimum of \$25,000, but some franchisees may spend more depending upon local market circumstances. In markets defined by us as underpenetrated, you may be required to spend an additional \$10,000. We may require you to provide us with an accounting of your grand opening expenditures within 180 days after the date the Restaurant is opened.

Advertising and Marketing Fee

By signing the Franchise Agreement, you agree to pay to the Advertising Fund, discussed below, an Advertising and Marketing Fee (the "Advertising and Marketing Fee") equal to 4% of the Net Sales of the traditional Restaurant. The Advertising and Marketing Fee for Non-traditional Restaurants is 2% of Net Sales. We will not increase this fee for traditional Restaurants before January 1, 2026, but thereafter, we may modify the Advertising and Marketing Fee for traditional Restaurants on 60 days' notice to an amount equal to the amount we charge to new Franchisees, but in no event shall the Advertising and Marketing Fee exceed 4.25% of the Net Sales. We will not increase this fee for a Non-traditional Restaurant during the initial term of your Franchise Agreement. The Advertising and Marketing Fee is payable at the same time and in the same manner as your Royalty payment. Taco John's Restaurants which we own pay Advertising and Marketing Fees on the same basis as franchised Taco John's Restaurants.

Although we try to use the Advertising and Marketing Fees to place advertising in a manner that will benefit all Taco John's Restaurants, we cannot ensure that expenditures of Advertising and Marketing Fees in or affecting any geographic area are proportionate or equivalent to Advertising and Marketing Fees paid by Taco John's Franchisees operating in that geographic area or that any Taco John's Franchisee will benefit from the placement of advertising directly or in proportion to the amounts it has paid as Advertising and Marketing Fees. Advertising and Marketing Fees paid to an Advertising Cooperative will be used for marketing and advertising programs conducted in the area where the Taco John's Restaurants owned by the members of the Advertising Cooperative are located.

Certain Franchisees whose franchise agreements are dated prior to the issuance date of this Disclosure Document may pay a different amount of advertising fees than you will pay under your Franchise Agreement, and those fees may be allocated in a different manner than how fees are allocated above.

Advertising Fund

We have established an advertising fund (the "Advertising Fund" or "Fund") in which we deposit the Advertising and Marketing Fee we receive from Franchisees.

The Advertising Fund is administered by us. Unless we have established a National Marketing Program as provided below, we will confer with an advertising production committee (the "Advertising Production Committee"), which is made up of seven Taco John's Franchisees and one of our management level employees. Five of the Franchisee members of the Advertising Production Committee are nominated by Franchisees and are elected by ballot vote to serve three years on the Committee. Two of the Franchisee members are appointed by the Board of Directors of the Association of Taco John's Franchisees, Inc. to serve three years on the Committee. The Advertising Production Committee is organized and governed pursuant to its bylaws that we approve. We may disband the Advertising Production Committee in our sole discretion. The Advertising Production Committee serves in an advisory role only. The artistic and creative aspects of advertising/marketing materials will be determined by us after conferring with the Advertising Production Committee.

The Advertising Fund is used to create and produce advertising materials that are made available to you, to other Taco John's Franchisees, and to advertising cooperatives formed by Taco John's Franchisees ("Advertising Cooperatives"), as they may exist from time to time. We or our advertising agency may charge a fee for any advertising materials acquired by a Franchisee or Advertising Cooperative.

The funds in the Advertising Fund are spent to foster recognition and promotion of the Marks, the brand, and retail sales of Taco John's Restaurants, generally, and to pay for the development, support, and dissemination of marketing materials and programs. Materials produced using the Advertising Fund may include television and radio spots, newspaper and billboard displays, freestanding inserts ("FSI") and flyers, signs, menus, menu boards, e-mail or other social media campaigns, digital displays, and visual displays. We may use a variety of sources for advertising materials including in-house and regional or national advertising agencies.

In the year ended December 31, 2024, the Advertising Fund was allocated to advertising production costs (99%) and administrative expenses (1%). No part of the Advertising Fund was used to solicit the sale of Taco John's franchises; however, a toll-free number for franchise sales information occasionally appears on marketing materials provided for retail promotions. Neither we nor our affiliates receive payment for any goods or services we or they provide to the Advertising Fund. If fees that were collected in a particular year are not entirely spent in that year, the balance is carried over to the Advertising Fund's budget for the following year. An annual audited statement of monies collected and expended by the Advertising Fund will be made available to you on written request.

We may in the future establish a program for national, regional, or local media placement, digital advertising, and national advertising (the "National Marketing Program"). If we create a National Marketing Program, we may also use some or all of the Advertising Fund to support it. At that time, the Advertising Production Committee will be disbanded, and the functions of that committee shall become the responsibility of a National Marketing Program advisory committee (the "National Marketing Program Advisory Committee"). We currently do not have any standards for a National Marketing Program or for a National Marketing Program Advisory Committee. We will provide you at least six months' advance notice of the establishment of a National Marketing Program.

Advertising Cooperatives

We reserve the right to establish and require your participation in either a local or regional Advertising Cooperative at our sole discretion (each an Advertising Cooperative). If we require you to join a local Advertising Cooperative, it will consist of Franchisees in your Designated Market Area ("DMA") or such other contiguous areas determined by us, in our sole discretion, to be appropriate in your market area. Currently, a DMA is determined by the A.C. Nielsen Company's Nielsen Station Index and is defined as "generally a group of counties in which the commercial TV stations in the Metro/Central area achieve the largest audience share." We may change how the DMA is established based on changes in the advertising industry.

If we require you to join a regional Advertising Cooperative, we will designate the market areas included within the regional Advertising Cooperative. We may change your assignment to an Advertising Cooperative in our sole discretion. We will provide you at least six months' advance notice of any change in your assignment to an Advertising Cooperative.

We require that every local or regional Advertising Cooperative be legally incorporated, and as a corporation, each Advertising Cooperative must operate under bylaws that we approve. We will provide model bylaws to all Advertising Cooperatives. We may dissolve an Advertising Cooperative or change the structure and/or function of an Advertising Cooperative. If we establish a National Marketing Program, we will confer with the National Marketing Program Advisory Committee comprised of franchisees and our representatives, and the committee will operate under guidelines established by us.

Franchisees assigned to a local or regional Advertising Cooperative shall elect officers of their Advertising Cooperatives, and these officers are responsible for administration of the Advertising Cooperative and the expenditure of funds contributed to the Advertising Cooperative. We require that at a minimum, a President and Treasurer for each Advertising Cooperative be elected. We may require the Advertising Cooperative to delegate the daily administration of the Advertising Cooperative to an advertising agency, media buyer, or other professional marketing services provider.

If you are required to join a local or regional Advertising Cooperative, then the Fund will contribute to your Advertising Cooperative monthly an amount equal to the lesser of the amount you have paid toward your Advertising and Marketing Fee or 2.85% of your total Net Sales (or 1% for non-traditional facility franchisees) to be used for media placement in your DMA. An Advertising Cooperative may determine, by a vote of its members, to collect additional advertising fees for use by the Advertising Cooperative on approved advertising and marketing programs, although non-traditional locations are not required to contribute to an Advertising Cooperative. Advertising Cooperatives are also required to prepare annual financial statements and to present these to its members (or prospective members) and to us upon request.

If we establish a National Marketing Program, it may replace any local and regional Advertising Cooperatives.

Local Advertising

You are permitted to use advertising and promotional materials you develop and to conduct independent local promotions so long as you obtain our prior written consent to do so. You must submit to us for our approval any advertising and promotional materials you develop. We will provide approval or disapproval within five business days after the date we receive the materials.

If you are operating a traditional Taco John's Restaurant and are not in an Advertising Cooperative, you will be required, at your expense, to conduct local advertising within your DMA using one of our

approved media buyers. In that event, the Advertising Fund will pay 2.85% of your Net Sales to your media buyer to be applied toward the media buyer's fees and the costs of your local advertising program developed by the media buyer. You must be current on paying us your Advertising and Marketing Fee or we may withhold amounts paid to your media buyer, in which case you will be responsible for all amounts due to the media buyer. We will not pay any amounts from the Advertising Fund to your media buyer if you are in an Advertising Cooperative or if you operate a non-traditional Restaurant.

We do not reimburse food cost for coupons redeemed at your restaurant.

Websites and Social Media

We consider presence on the Internet, as it may evolve over time, including websites and social platforms, to be advertising and subject to our review and written approval before use in connection with the Restaurant. As used in the Franchise Agreement, the term Internet includes, but is not limited to, what are today known as World Wide Web home pages and social platforms. However, these types of global electronic communications evolve and change over time and such changes are intended to be included within the reasonable intent of this clause. Collectively, and for ease of reference, the term "Internet presence" will encompass the current Internet, along with websites and social platforms, and future similar concepts. If we permit you to have an independent Internet presence, then the following requirements will apply: (1) You may not establish or use the Internet presence without our prior written approval; (2) before establishing any Internet presence, you must submit to us for our written approval, a proposal in the form, manner, and containing the content that we may reasonably require, including: (a) the domain name or platform name, (b) format, and (c) visible and non-visible content (including meta-tags), and (d) if you propose any material revision to the Internet presence or any of the information contained in the Internet presence, you must submit the revisions to us for our prior written approval; (3) you must establish hyperlinks to our website and others as we may require from time to time; and (4) you must properly use our Marks on the Internet presence.

We may revoke our approval of the Internet presence at any time and require you to discontinue using it. We may, in the future, restrict any Internet presence to web pages on our Taco John's website or other similar Internet presence as the same may evolve. In such event, you will be required, at your own expense, to convert your Internet presence to the format we reasonably specify.

Electronic Cash Register/Point of Sale System

You must purchase, from our approved vendor, and install the specific electronic cash register, freestanding kiosks, kitchen video monitors, receipt printers, drive-thru speed of service timing system, and headset system (collectively, the "Point of Sale System") we specify from time to time. You must also purchase and install the required computer hardware, software, internet connections and service, and other computer-related accessories, peripherals and equipment necessary to support the Point of Sale System, mobile ordering and delivery platforms that utilize geofencing services, firewall and security service, and the required back of house information system we specify from time to time (collectively, with the Point of Sale System, the "Computer System"). The cost of the Point of Sale System is estimated to be between \$20,000 and \$35,000, with a license fee currently in the amount of \$475 to \$650 per month depending on the number of terminals selected. The cost of the back of house information system is currently \$140 per month. The current cost of geofencing services is \$35 per month, which we may require you to pay to us or the approved supplier.

Your Computer System must be set up in accordance with our specifications, have the capability to process credit and debit card payments, comply with PCI DSS (as defined below), be able to receive sales from mobile ordering, third-party delivery, and catering, be able to record all sales information in

connection with the Restaurant, and be able to comply with your reporting requirements to us. We may require you to upgrade, at your expense, any part or all of the Computer System when deemed necessary by us.

You understand and agree that we have full rights and access to the information and data stored in your Point of Sale System.

We may revise our specifications for the Point of Sale System from time to time as we determine reasonably necessary. We may also change the required Point of Sale System and all other elements of the Computer System from time to time in our discretion, and you will be required to acquire any updated system at your sole expense. Consequently, you may be required to upgrade or update your equipment. There are no contractual limitations on the frequency or cost of this obligation. We cannot estimate the cost of maintaining, updating or upgrading your Computer System or its components because it will depend on your repair history, local costs of computer maintenance services in your area and technological advances we cannot predict at this time. If you do not comply with technology or equipment requirements, you may be required to pay a non-compliance fee.

We require use of a financial reporting system that we designate and which you must acquire from us or our designated vendor, allowing you to upload and deliver electronic reports of your Net Sales, Profit and Loss statements, and other data to us. There is currently no fee for this system, but we may require you to start paying a fee to us or our approved third-party vendor for use of that service upon 180 days' notice. The amount of the fee will be determined if and when the fee is implemented, and once implemented is subject to change on 30 days' notice.

You must accept credit and debit cards from customers of your Taco John's Restaurant. You shall not charge customers any additional fees or service charges if they elect to use a credit card or debit card for payment of any services or products provided in your Taco John's Restaurant. The Payment Card Industry ("PCI") requires all companies that process, store, or transmit credit or debit card information to protect the cardholders' information by complying with the PCI Data Security Standard ("PCI DSS"). Therefore, you must be PCI compliant by following and adhering to the then-current PCI DSS, currently found at www.pcisecuritystandards.org, or any similar or subsequent standard for the protection of cardholder data throughout the term of your Franchise Agreement. PCI mandates the PCI DSS compliance.

We currently have an online learning management system, which we call The Olé! Way, which you may access to obtain information on various aspects of the operation of a Taco John's Restaurant, including the Operations Manual. You will enter into any and all agreements and documents required of our vendor for this online system and pay to us or the vendor of this system, as we direct, the Online Learning Management System Fee, currently in the amount of \$240 per year. We have no obligation to maintain this system and intranet site indefinitely. We can, after 60 days' advance written notice to you, remove this site and system entirely, and require you to subscribe, at your sole cost and expense, to a service or provider we designate to obtain access this or similar information, including the Operations Manual.

You must purchase exterior digital menu boards, and you have the option to purchase interior digital menu boards, from our approved third-party supplier. The cost of the Digital Menu Board License fee from the approved third-party supplier is currently \$75 per month for exterior digital menu boards and \$65 per month for interior digital menu boards, and you will pay that directly to the supplier. For certain existing Restaurants operating with an existing digital menu board system, we may collect the license fee for that system, currently in the amount of \$18.80 per screen license per month.

We may in the future establish a Technology Fee as described in Item 6 for further development of technology provided by or for us. If we create a Technology Fee, it will replace some or all of the following:

the Digital Menu Board License Fee, The Olé! Way online learning management system license fee, financial reporting system fee, back of house information system fee, geofencing service fee, gift card fee, loyalty/rewards program fee, guest experience program fee, and other technology-related subscription costs. The Technology Fee will also be used for further development and support of technology by or for us, including changes or enhancements to ordering and AI platforms, the mobile app, data reporting platforms, custom development, or other technology innovations. We reserve the right to implement this fee on 60 days' notice to you.

Site Selection

Under the Franchise Agreement, including each Franchise Agreement signed under an ADA, you must locate, within 90 days after the date of the Franchise Agreement, a site that is suitable for the operation of a Taco John's Restaurant (a "Proposed Site"). You will work with a real estate broker approved by us to locate a site which you believe in good faith to be suitable for development as a Taco John's Restaurant, and you will submit to us full and complete site documentation in our required form, a copy of the proposed purchase agreement or lease, and any other information about the site and its development that we may reasonably request. We will use reasonable efforts to evaluate and accept or reject a Proposed Site and will not unreasonably withhold acceptance of sites that meet our minimum standards for market type, trade area, general location, physical and demographic characteristics of the market or sub-market, traffic pattern, parking, restaurant competition, proximity of other businesses (including other Taco John's Restaurants), the nature of other businesses in proximity to the site, layout and other physical characteristics, development costs, rental, lease duration and other lease terms and conditions for the Taco John's Restaurant.

You may not execute a purchase agreement for a Proposed Site until you have provided us a copy of the purchase agreement and we have accepted the site and purchase agreement in writing. We have 30 days after receipt of your full and complete site documentation and purchase agreement to review and accept it.

You may not execute a lease for a Proposed Site until you have provided us a copy of the lease and have demonstrated to our reasonable satisfaction that your lease contains the required provisions described in our Lease Rider (currently in the form attached as Attachment K but subject to modification) and we have accepted the site and lease in writing. We have 30 days after receipt of your full and complete site documentation and lease to review and accept it.

We may provide you with suggestions and recommendations concerning the terms and conditions of the purchase agreement or lease. We have the right to require that certain provisions are included in the lease relating to duration, use, default, notices, assignment, remodeling, personal property rights, the right for our employees to enter the Premises, our right to protect our Marks, and the right of first refusal. Some provisions may be contained in our Lease Rider form. We make no representations or warranties as to the legal validity of these provisions. It ultimately is your responsibility, however, to negotiate and accept the final terms of these documents. Our involvement in the review and approval of the purchase agreement or lease for the Proposed Site does not constitute a recommendation, endorsement, or guarantee by us of the suitability or profitability of the location or the purchase agreement or lease, and you will need to take any additional steps necessary to ascertain whether such location and documentation are acceptable to you, which may include engaging your own legal counsel.

An "Accepted Site" is a Proposed Site for which we have reviewed your full and complete site documentation, a copy of the proposed purchase agreement or lease, and any other required information, and which we have accepted in writing. Our acceptance of a site is not a representation or warranty, express or implied, that the Proposed Site is suitable for a Taco John's Restaurant or for any other purpose. Upon execution of the Franchise Agreement, you may begin proposing locations immediately. We cannot

estimate the time it will take you to secure an Accepted Site, but the time it takes you may depend on many factors including the local market and suitability of existing buildings. If you are unable to locate an Accepted Site within the specified time, we may terminate the Franchise Agreement. Upon our written acceptance of a Proposed Site for your Restaurant, you agree to attempt to obtain lawful possession of the Accepted Site through purchase or lease.

You are solely responsible for developing, constructing, remodeling and decorating the site for your Restaurant, for all expenses associated with it, for conforming the premises to local ordinances and building codes and obtaining any required permits, and for compliance with the requirements of any applicable federal, state or local laws. You must engage building design professionals, including architects and engineers, and qualified general contractors, that are approved by us, for the buildout and improvements of your Restaurant.

We do not generally own the sites for Taco John's Restaurants or lease them to franchisees.

Time for Commencement of Operations

We estimate the typical length of time between the signing of the Franchise Agreement and the opening of your Taco John's Restaurant will be approximately 9 to 12 months. Some factors which may affect this timing are pre-existing obligations, completion of your training, the time to acquire and build out your Restaurant location, and your ability to secure financing. You must open your Restaurant for business within the earlier of: (i) two hundred ten (210) days after receiving your building permits for the Accepted Site, or (ii) 10 days after obtaining the Certificate of Occupancy for the Restaurant, and in any event within 12 months after the signing of the Franchise Agreement. If you are opening multiple outlets under the multi-unit franchisee incentives described in Item 5, we will agree to modify the opening date requirements as described in Item 5.

You must continue to actively operate your Taco John's Restaurant in accordance with the Operations Manual and Restaurant System after you have commenced operations.

Training

Area Development Agreement

We do not provide initial or additional training programs under the ADA. However, you will have training requirements as a Franchisee as described below.

Franchise Agreement

<u>Owner Training Program</u>. We may, at our discretion, require a franchise owner to complete up to one week in a franchise ownership, business management, and operations training program (the "Owner Training Program" or "OTP"). If required, the OTP will occur prior to the Management Certification Program, described below. You must pay all expenses incurred by any owners of the restaurant in connection with attending the OTP. However, we do not charge a fee for the OTP. This program will be conducted in a Restaurant location or in our Restaurant Support Center located in the Minneapolis, Minnesota metropolitan area at the time we designate.

<u>Initial Training</u>. We will provide an initial training program in the operation of a Taco John's Restaurant for a specific number of Restaurant managers (Franchise Agreement - Section 4). Before the Restaurant opens for business, one owner, or if the owner is not an individual, the person designated by the owner to assume primary operational responsibility for the Restaurant ("Designated Principal"), and a

minimum of three additional managers must attend the "Management Certification Program" at the time and location we designate. If the owner or Designated Principal will not have primary operating responsibilities, then four managers must attend this training and become certified as Active Certified Managers (defined below) in addition to the owner or Designated Principal. If you operate multiple Taco John's Restaurants, we may require you to have members of your management team who oversee multiple Restaurant locations (each an "Above Store Leader"), attend additional training. If you open a new restaurant in an underpenetrated market, we may require you and your managers to attend additional training. You must pay all expenses incurred by any owners, Above Store Leaders, and/or manager(s) of the Restaurant in connection with attendance at the Management Certification Program. All persons who complete the Management Certification Program will be designated an Active Certified Manager, as defined below. We do not charge a fee for the Management Certification Program for up to four individuals, but you will be required to pay a fee for the food safety program and certification provided by a third party licensed to teach food safety-that we approve. If you elect to train additional managers, there may be an additional fee. We pay no compensation to you or your employees while in training.

The Management Certification Program for Taco John's includes instructions on guest service, operating and maintaining equipment, preparing Taco John's food, food safety techniques, food purchasing and food cost analysis, business management, personnel management, and other details and procedures for operating a Taco John's Restaurant. The four to six week Management Certification Program is a hands-on training course that is conducted in a Restaurant we designate or in our Restaurant Support Center located in the Minneapolis, Minnesota metropolitan area at the time we designate, with other training provided online. Your Field Training Manager or Franchise Business Consultant oversees and coordinates the training course. The initial Management Certification Program must be completed to our satisfaction prior to opening for business, typically at least four to six weeks prior to the opening. We reserve the right to suspend or terminate the training program for any training candidate who fails to adhere to the established standards and guidelines. This includes, but is not limited to, non-compliance with operational procedures, failure to meet performance benchmarks, or any conduct deemed detrimental to the franchise system's reputation and integrity. The decision to suspend or terminate training will be at our sole discretion and is intended to ensure the highest quality and consistency across all franchised locations.

The training program described in the chart below is conducted under the supervision of Jackie Secor, our Chief Operating Officer. Ms. Secor has over 26 years of experience in restaurant operations and training. She became employed by us in June 2025. The instructional materials consist of our Operations Manual, the Management Certification Program Course, and our online Learning Management System, available on The Olé! Way (monthly subscription required). The Olé! Way contains orientation training, job station training, and four levels of management development, including certified trainer, shift supervisor, certified manager, and advanced management training. Training content is delivered through test-based modules, videos, games, and simulation models.

MANAGEMENT CERTIFICATION PROGRAM

Subject	Hours of Classroom Training ¹	Hours of On-the-Job Training ²	Location
Food Safety Certification	8 hours		Online
Orientation, Guest Service and Station Training Program	10 hours	10 days (90 hours); see schedule below	Online, and at a Taco John's Restaurant location we designate or our Restaurant Support Center located in the Minneapolis, Minnesota metropolitan area
Shift Supervision Training	5 hours	5 days (45 hours)	Online, and at a Taco John's Restaurant location we designate or our Restaurant Support Center
Management Training	5 hours	5 days (45 hours)	Online, and at a Taco John's Restaurant location we designate or our Restaurant Support Center
Systems Training	8-10 hours	4-5 days (32-40 hours)	Online, and at a Taco John's Restaurant location we designate or our Restaurant Support Center
Totals	36-38 hours	24-25 days (220 hours)	Total Training 4-6 weeks

The training identified here as classroom training is conducted online.

Week 1:

Review of the Taco John's Operations Manual.

Team member orientation and job station training. Job stations include fry station, recipe preparation, menu assembly, and cashier.

Satisfactory completion of job station quizzes is required.

Management Training:

Manager walks, guest service, crew development, portion control, dating procedures and First In First Out ("FIFO"), Food Quality and Temperature ("FQT") Logs, daily prep guide, drive-thru timer, rotation log and waste control.

At the end of Week 1, a weekly appraisal is provided to the Manager in Training and an Action Plan is created for Week 2.

Week 2:

Learn all aspects related to team member training in guest service. Work on the cook line to become more familiar with food preparation. Introduction to team member training in guest ordering. Introduction to team member opening and closing duties.

Management Training:

Store opening and closing management duties, shift change procedures, pre-shift meetings, and steps to creating great guest service.

The on-the-job Management Certification Program shall consist of the following topics:

Tests given: Relevant tests and quizzes relating to restaurant operations and food preparation.

Continue review of Taco John's Operations Manual.

At the end of Week 2, a weekly appraisal is provided to the Manager in Training and an Action Plan is created for Week 3.

Week 3:

Finish review of team member guest service duties. Finish review of all team member job stations.

Management Training:

Shadow and assist with managing all shifts, complete all opening and closing management functions (paperwork, reports and operations), learn processes relating to marketing calendar and Point of Purchase materials ("POP"), receiving and storage, inventory, food cost and quality control, team member scheduling, training and management development.

Test on full menu item assembly.

Finish reading Taco John's Operations Manual.

At the end of Week 3, a written appraisal is provided to the Manager in Training and an Action Plan is created for Week 4.

Week 4:

Management Training:

Manager on Duty responsibilities for all shifts. (Training Manager on site to support.)

Manager in Training Performance Appraisal.

Personnel management.

Cash management.

In-store marketing execution.

Profit & Loss Statement.

A final Performance Assessment is given to the Manager in Training at the end of Week 4.

Week 5:

Systems Training:

Point of Sale System

Back of House System

Digital Sales platforms

Distribution or supply chain

Local store marketing

Week 6:

Additional Required Management training:

If required.

The owner, or if applicable, the Designated Principal, Above Store Leader, and manager(s) of the Restaurant must complete the entire training course and meet our standard certification requirements. (A person who has met our standard certification requirements or has completed the Management Certification Program and works regularly in the Restaurant is referred to as an "Active Certified Manager.") If the owner or the manager(s) fails to complete the training program or is not certified, we will allow another owner or manager, as applicable, to attend training, provided he or she completes our training program and becomes

an Active Certified Manager. You will not be allowed to open your new Restaurant until you have the required number of Active Certified Managers on staff, one of which may be you or your Designated Principal.

For your first Restaurant opening, we will provide, at our expense, qualified Taco John's Operations or Training Team Member(s) who will, for a minimum of three days, assist you and/or your managers with the pre-opening of the Restaurant and, for up to 10 days (which may or may not be consecutive days), assist you and/or your managers with the post-opening of the Restaurant, provided you have notified us in writing of your request no less than 30 days prior to the scheduled opening of the Restaurant. For the second and subsequent Restaurants you open we will provide, at our expense, a minimum of one Taco John's Operations or Training Team Member to assist with opening and initial operation for up to eight days (which may or may not be consecutive days). If your scheduled opening date is delayed for any reason, you may be required to pay a delayed opening fee.

Your Active Certified Managers will provide on-the-job training to your team members immediately before your Restaurant is opened. This on-site team member training is four days in length, taught by your newly certified managers. The on-site training will only be conducted after the Restaurant's construction is completed and a certificate of occupancy has been issued.

<u>Maintaining Active Certified Managers</u>. After opening your Restaurant, you must maintain the number of Active Certified Managers we specify in the Operations Manual. The process to replace an Active Certified Manager is described below.

The requirements to become a replacement Active Certified Manager are as follows: a passing score on the food safety certification course/exam, a score of 80% or higher on each of the guest service classes/exams, the development class/exam, the operations class/exam, and the Taco John's standards test. If a replacement or additional manager needs to be certified, we offer the classes through our online management certification program.

You may obtain food safety certification from a third party licensed to teach food safety that we approve. We do not provide this training, and you will be required to pay the fees associated with that training to the third-party provider. You will need to complete the guest service class, the development class, the operations class, and the Taco John's standards test using a web-based certification program. We continuously evolve our training programs to provide the most thorough training to our franchisees and reserve the right to modify this outline and the training program requirements from time to time.

Additional Training Support.

As described in Item 6, in exchange for the fee of \$240 per year currently paid to us or our third-party online vendor, we will provide you with access to our online learning management system, currently called The Olé! Way, which includes ongoing training programs in webinar and other formats, and training materials for you and your employees. This system also provides you access to our Operations Manual.

We may require your management team members to attend additional training programs, including both training that we may require on a system-wide basis and training only for you or a limited number of franchisees in situations where we determine that you or any of your managers are not meeting our standards. We may also conduct ongoing on-site training for some or all franchisees at their Restaurant locations on a rotating basis or as we otherwise deem appropriate. In addition, from time to time we may also offer optional training programs and seminars at various locations. Some of these programs may be offered via The Olé! Way online learning management system, while others may require in-person attendance or participation. We may require your manager(s) to attend a manager convention. You must

pay all expenses incurred by managers in connection with attendance at such training programs. We may charge a non-attendance fee, at our sole discretion, if you do not have at least one manager per Restaurant attend the manager convention. We may charge a fee for the optional training programs. Additional in-store training is available upon request by a Franchisee at our current rate.

We require you (or at least one of your owners if the Franchisee is an entity) to attend any convention or system-wide meeting of franchisees (collectively referred to as "Franchisee Convention"). You will be responsible to pay any registration fee and all costs of travel, accommodations, wages (if your employees attend), and any other expenses associated with attendance. If you employ an Above Store Leader, that individual may attend the Franchisee Convention in your place. We reserve the right to charge you a non-attendance fee, at our sole discretion.

Not less than 60 days prior to the opening date of a Franchisee Convention or manager convention, you may submit a written request to us for a waiver of this requirement based on your specific circumstances. We will use our best business judgment to determine whether to approve the request.

All training programs are provided to protect our brand and the Marks and not to control the day-to-day operation of your Taco John's Restaurant. You are also solely responsible for hiring and training your employees.

ITEM 12. TERRITORY

Area Development Agreement

Under the ADA, you will have the right to locate multiple sites to develop and operate a specified number of Taco John's Restaurants within a specified geographic territory (referred to in those sections of this Disclosure Document which describe the terms of the ADA as the "Development Area"). A description of the Development Area will be inserted into the ADA before it is executed. Development Areas will typically consist of a trade area located in a demographic market area and may vary in size depending upon the population concentration in the demographic market area and the number of Taco John's Restaurants to be developed.

So long as you have complied with all of the terms and conditions of the ADA, we will not, except with your written permission, during the term of the ADA, operate or grant a franchise or development rights for anyone else to develop and/or operate any Taco John's Restaurant located within the Development Area, except for those franchises granted to you under the ADA. Nothing in the ADA prevents, prohibits, or otherwise restricts at any time the operation of any Taco John's Restaurants in the Development Area which are operated under any existing franchise agreements, or the renewal of franchise rights related to such existing franchise agreements. Further, a delivery of goods, whether by a third-party delivery service or otherwise, to a customer located within your Development Area from any Taco John's Restaurant, is not a violation of your Development Area rights. Your Development Area rights also do not prevent a Taco John's Restaurant located outside of your Development Area from providing catering services within your Development Area.

The Development Area may not be changed during the term of the ADA except by mutual written consent. Prior to development of each Restaurant, you must obtain our acceptance of the proposed site within the Development Area for each of the Restaurants to be opened under the ADA. You will be required to comply with our then-current criteria and standards in selecting each proposed site. The Protected Territory, as defined below, for each Taco John's Restaurant opened pursuant to an ADA will be specified

in each Franchise Agreement for that Taco John's Restaurant based on the accepted site and our thencurrent standards for the establishment of Protected Territories.

Under the ADA, you are granted a right of first refusal for two years following the execution of the final Franchise Agreement signed pursuant to the ADA to establish and operate additional Taco John's Restaurants in your Development Area. We will notify you if we identify a location in the Development Area for a third party to open a Taco John's Restaurant, and you have 30 days to elect to enter our thencurrent Franchise Agreement for the location. If you fail to exercise this right of first refusal for any location, it will be terminated and of no further effect for all future locations.

To maintain your rights to the Development Area, you must have open and operating the cumulative number of Taco John's Restaurants by the dates set out in the Development Schedule, which is attached as an exhibit to the ADA. We will not typically grant development rights unless you agree to develop three or more Taco John's Restaurants within a Development Area within a 24 to 36 month period. If you do not meet your Development Schedule under an ADA, at our option we may treat such failure as a default of the ADA or we may terminate your exclusive rights to the Development Area and corresponding right of first refusal. At your discretion, and in lieu of termination of the ADA or loss of exclusivity, you may request an extension to the Development Schedule. If we agree to grant that extension, you will be charged a non-refundable Extension Fee as set forth in the ADA for each Restaurant not in compliance with the Development Schedule until such time as you comply with the Development Schedule, up to a maximum of 13 months. If you are not in compliance with the Development Schedule after 13 months, we may exercise our right to terminate the ADA or terminate your exclusivity rights and right of first refusal. The Development Area may not be changed during the term of the ADA except by mutual written consent.

Franchise Agreement

Premises.

The Franchise Agreement will grant you the right to operate a Taco John's Restaurant at a specific location, defined in this Disclosure Document as the "Premises." Provided that you are in substantial compliance with the provisions of the Franchise Agreement, if you lose the right to possession of the Premises through no fault on your part or if the Premises are destroyed, condemned or otherwise rendered unusable, or if you request the relocation of your Restaurant and, in our reasonable judgment, the existing site for the Restaurant does not meet our then-current standards and specifications for sites for Taco John's Restaurants, we will grant permission for the relocation of the Restaurant to an alternate location and site which meets our then-current site criteria for Taco John's Restaurants and is accepted by us. If we agree on the relocation of the Restaurant, you agree to expeditiously develop the accepted site in compliance with our then-current specifications and requirements for Taco John's Restaurants, all at your sole cost and expense, and to open the relocated Restaurant within 150 days after receiving our permission to relocate, unless otherwise agreed to by us. You agree to de-identify, to our standards, the vacated location within 30 days of relocation. You will pay us the relocation fee at the time the relocation actually occurs and is nonrefundable once paid.

Protected Territory.

The Franchise Agreement designates a territory in which you will have certain rights (the "Protected Territory"). The size and dimensions of the Protected Territory will depend on the specific market variables of the Premises, including demographics, density, market and development trends, traffic flow, and natural manmade boundaries. A non-traditional Restaurant does not have a Protected Territory,

and non-traditional Restaurants may be located in the Protected Territory of another Taco John's Restaurant, as discussed below.

Except as described in this Item 12, we will not establish, operate, or grant a franchise for the operation of a Taco John's Restaurant within the Protected Territory as long as you are in substantial compliance with all the provisions of the Franchise Agreement. However, a delivery of goods, whether by a third-party delivery service or otherwise, to a customer located within your Protected Territory from a Taco John's Restaurant located outside of your Protected Territory rights. Similarly, your Protected Territory rights do not prevent a Taco John's Restaurant located outside of your Protected Territory from providing catering services within your Protected Territory.

We also reserve the right for us and our affiliates to market, offer, and sell, and to authorize third parties to market, offer, and sell, any and all products and services through alternative channels of distribution (other than Taco John's Restaurants), at any location (including within your Protected Territory), and on any terms and conditions as we determine including, but not limited to, sales through channels such as wholesale or retail distribution, Internet marketing and distribution, social media marketing, mail order, catalog, retail store display, or made-to-order or pre-packaged product sales in grocery stores, at any location. The products and services available through these alternative channels may include those that are the same as or similar to those which you will offer and sell, such as any menu items offered at your Restaurant, or entirely different services and products. The prices advertised and charged by us, our affiliates, and our approved suppliers for the sale of the products and services to operators of alternative channels, and the prices advertised and charged by the operators of the alternative channels to third parties and the public for the products and services, may be higher or lower than the prices at which the same or similar products and services are made available by us, our affiliates, and our approved suppliers to you and the prices charged by you to customers of your Restaurant.

We retain all other rights in your Protected Territory including the right to distribute goods under similar or different marks, and through non-traditional Restaurants at any location.

You must offer delivery services from your Restaurant to customers located in your Protected Territory, through all third-party delivery services approved by us which are available in your Protected Territory. You may also offer delivery services from your Restaurant to customers located outside of your Protected Territory, even if the location of the customer is in the protected territory of another franchisee. We also require you to participate in catering programs through online catering platforms that we designate or approve. You may conduct the catering services at any location, even if the location is in the protected territory of another franchisee. Similarly, another franchisee may offer delivery services from its Restaurant to customers located in your Protected Territory or provide catering services at a location in your Protected Territory. You must comply with all terms imposed from time to time by the third-party delivery services and catering platforms.

In the event (1) we or a Prospective Franchisee desires to locate, open and operate a Taco John's Restaurant, other than a Restaurant in a non-traditional location (discussed below), within two miles outside the boundary of your Protected Territory ("Right of First Refusal Area"); or (2) we or a Prospective Franchisee desires to locate, open and operate an Affiliated Business (as defined in Section 2.3 of the Franchise Agreement), other than in a non-traditional location, within your Protected Territory, then we will first offer you the right to locate, acquire and operate a Taco John's Restaurant in the Right of First Refusal Area under the terms and conditions of our then-current Franchise Agreement for a Taco John's Restaurant or in your Protected Territory for an Affiliated Business. You will have 30 days after notice is given to you in which to accept the offer by signing our then-current Franchise Agreement and paying the then-current Initial Franchise Fee in full. If you do sign our then-current Franchise Agreement and pay the then-current franchise fee in full, you will have 120 days to bring an acceptable site to us in accordance

with our then-current requirements. If you fail to meet this requirement, we will be entitled to operate or grant a franchise for the operation of a Taco John's Restaurant or an Affiliated Business, as applicable, in the applicable area, and you will have no right to prevent or prohibit us from operating or granting a franchise for a Taco John's Restaurant or an Affiliated Business in the applicable area in the future. Our obligation to make the offers described above is subject to the condition that you are in substantial compliance with all of the provisions of the Franchise Agreement.

In the event that you and another Taco John's franchisee have overlapping Right of First Refusal Areas, then you and the other franchisee will each be given consecutive periods of 30 days in which to exercise your respective rights of first option. The franchisee whose Taco John's Restaurant began operating first shall have the first 30 day period. If that franchisee does not exercise its rights by signing a Franchise Agreement by the end of that 30 day period, then the second franchisee shall have the next 30 day period in which to exercise its right of first refusal. If neither you nor the other franchisee signs a Franchise Agreement during the applicable time period, we will be entitled to proceed to operate or grant a franchise in the applicable area as described above.

We also may operate or grant licenses to others to operate Taco John's Restaurants in non-traditional locations that may be within your Protected Territory or Right of First Refusal Area without providing any rights or compensation to you. Non-traditional locations are defined as venues or facilities where the primary purpose of the consumer's visit is typically for reasons other than the purchase of goods from a Taco John's Restaurant. Examples include airports and other transportation hubs, hospitals, college campuses and other educational facilities, convention centers, grocery stores, department stores, "big box" retail centers, sports arenas and stadiums, hotels and office buildings, military installations (Army and Air Force Exchange Services), and food courts, without regard to location. As a result, your Protected Territory is not considered an exclusive territory. You will not receive an exclusive territory. You may face competition from other franchisees, from outlets that we own, or from other channels of distribution or competitive brands that we control.

We do not pay compensation for soliciting or accepting orders inside your Protected Territory. Subject to you complying with the advertising requirements set forth in the Franchise Agreement and elsewhere in this Disclosure Document, you are not restricted from soliciting or accepting orders from customers outside of your Protected Territory. As of the issuance date of this Disclosure Document we do not currently sell anything directly to customers of your Taco John's Restaurant.

Except as described above, we retain all rights with respect to Taco John's Restaurants, the Marks, and the Restaurant System. Your rights in the Protected Territory will not be dependent upon the achievement of minimum sales quotas, market penetration or other such contingencies. Your Protected Territory may not be changed except by mutual written consent. Except as described above, you do not receive any options, rights of first refusal or other similar rights to acquire additional franchises in any territory or contiguous territories. Neither we nor any affiliate of ours currently operates, franchises, or plans to operate or franchise a business under a different trademark and that sells or will sell goods or services similar to those you will offer, although there are no restrictions from us or any affiliate of ours from doing so in the future.

ITEM 13. TRADEMARKS

The ADA does not grant any rights to use the Marks (defined below). Rights to use the Marks are granted solely under the Franchise Agreement which grants you the right to operate your Restaurant at a specific location under the mark "TACO JOHN'S®" and under any other trademarks, trade names, service marks, logotypes and commercial symbols which we may designate (the "Marks").

Our affiliate, SS, LLC, owns all right, title and interest in and to the Marks. Under a license agreement we entered into with TJS on December 7, 1995 that was assumed by SS, LLC (the "License Agreement"), we have the worldwide non-exclusive license to use the Marks and certain unique proprietary confidential information, consisting of formulas, recipes, specifications and procedures for various Mexican food products sold from Taco John's Restaurants and any other franchise or license we may offer that sells products under the Marks (the "Trade Secrets") and the worldwide exclusive license to sublicense the Marks and the Trade Secrets in connection with the sale of our franchises. Under the terms of the License Agreement, we must pay SS, LLC a royalty as set forth in the License Agreement. The License Agreement will expire on May 15, 2095, but may be terminated by either party upon the default of the other party or by written agreement signed by both parties.

Under the Franchise Agreement, we will grant you the nonexclusive right to use the Marks. SS, LLC owns the registration of the following principal Marks that were registered by TJS on the Principal Register of the United States Patent and Trademark Office and subsequently assigned to SS, LLC, as follows:

Mark	Registration Date	Registration Number		
TACO JOHN'S	April 4, 1978	1,088,950		
TACO JOHN'S	October 9, 1990	1,617,184		
TACO JOHN'S	December 11, 1990	1,627,389		
Potato Olés	October 3, 1995	1,923,888		

The registrations of the Taco John's® marks have been renewed. We intend to further renew the registrations of some or all of these Marks at the appropriate time. TJS has filed all affidavits of use required to be filed when due, and SS, LLC will submit timely renewal applications when due. We and SS, LLC may apply for the registration of additional Marks in the future.

In addition to those Marks listed above, we claim common law service or trademark rights to a number of other words, phrases, or designs that you may use in your Taco John's Restaurant. The following statements apply solely to any unregistered trademarks and service marks: We do not have a federal registration for these trademarks. Therefore, these trademarks do not have as many legal benefits and rights as federally registered trademarks. If our right to use these trademarks is challenged, you may have to change to an alternative trademark, which may increase your expenses.

To our knowledge there are no infringing uses which could materially affect your use of the Marks. There are no currently effective determinations of the PTO, the trademark administrator of any state or any court, or any pending interference, opposition or cancellation proceedings, or any pending material litigation involving the Marks.

Other than the License Agreement described above, there are no agreements currently in effect which would limit our rights to use or license the use of the Marks in any manner material to you.

Your Obligations Regarding the Marks

Your right to use the Marks is derived solely from the Franchise Agreement and is limited to your operation of the Taco John's Restaurant in compliance with the Franchise Agreement and all standards, specifications, and operating procedures we prescribe during the term of the franchise. Any unauthorized use of the Marks will constitute a breach of the Franchise Agreement and an infringement of our and SS, LLC's rights in and to the Marks. All your usage of the Marks and any goodwill established by your use

will benefit us and SS, LLC exclusively. The Franchise Agreement does not confer any goodwill or other interests in the Marks on you, other than the right to operate a Taco John's Restaurant in compliance with the Franchise Agreement. All provisions of the Franchise Agreement applicable to the Marks will apply to any additional trademarks, service marks, commercial symbols, designs, artwork and logos we may authorize you to use during the term of the Franchise Agreement.

You must use the Marks as the sole trade identification of the Restaurant, except you must display at the Restaurant a notice, in the form we prescribe, which states that you are the independent owner of the franchise under a Franchise Agreement with us. You agree not to use any Mark as part of any corporate or trade name or with any prefix, suffix, or other modifying words, terms, designs, or symbols (other than logos and additional trade and service marks licensed to you under the Franchise Agreement), or in any modified form, and you may not use any Mark or any name or symbol similar to the Marks, in connection with the performance or sale of any unauthorized services or products or in any other manner we have not expressly authorized in writing. You agree to display the Marks in the manner we prescribe at the Restaurant, on menus and in connection with advertising and marketing materials, and to use, along with the Marks, the notices of trademark and service mark registrations that we specify.

You further agree to obtain any fictitious or assumed name registrations as may be required under applicable law.

If it becomes advisable at any time in our sole judgment for you to modify or discontinue the use of any Mark or for you to use one or more additional or substitute trade or service marks, you agree at your expense to comply with our directions to modify or otherwise discontinue the use of the Mark, or use one or more additional or substitute trade or service marks, within a reasonable time after our notice to you.

Notification of Infringements and Claims

Under the Franchise Agreement, you agree to immediately notify us in writing of any apparent infringement of or challenge to your use of any Mark, or claim by any person of any rights in any Mark or similar trade name, trademark or service mark of which you become aware. You agree not to communicate with anyone except us, our affiliate and our and their counsel in connection with any such infringement, challenge, or claim. We and SS, LLC have the right to exclusively control any litigation or other proceeding arising out of any infringement, challenge or claim relating to any Mark. You agree to sign any documents, render any assistance, and do any acts that our or SS, LLC's attorneys say is necessary or advisable in order to protect and maintain the Marks for the benefit of Taco John's Restaurants and Franchisees.

Indemnification

Under the Franchise Agreement, we agree to indemnify you against and to reimburse you for all damages for which you are held liable in any proceeding arising out of your use of any Mark, and for all costs you reasonably incur in the defense of any such claim in which you are named as a party, provided (i) you have operated fully and completely pursuant to and in compliance with the Franchise Agreement, and (ii) you have timely notified us of the claim and afforded us the opportunity to participate in the action, including, in our sole discretion, representing you in such action.

ITEM 14. PATENTS, COPYRIGHTS AND PROPRIETARY INFORMATION

Patent Rights and Copyrights.

We do not own any patents or copyright registrations or any pending patent or copyright applications which are material to the franchise.

We claim a copyright and treat the information in the Operations Manual as confidential trade secrets. We also claim a copyright in our other written materials, but you are permitted to use the materials as part of your Taco John's Restaurant.

Trade Secrets and Other Confidential Information.

We and our affiliates possess certain unique confidential and proprietary information and Trade Secrets relating to Taco John's Restaurants, which consist of the following information developed by us, our affiliates and our Franchisees:

- 1. Formulas, recipes, menu analyses, and methods of preparing, packaging, and serving products sold at Taco John's Restaurants;
- 2. Knowledge of sales and profit performance, standard performance evaluations of individual Taco John's Restaurants;
- 3. Knowledge of test programs, concepts or results relating to new menu items or categories; brands; sources of products; advertising and promotional programs; restaurant image and decor; and methods for the selection and training of personnel; and
- 4. Methods, techniques, formats, specifications, procedures, systems and knowledge of and experience in the development, operation, and franchising of Taco John's Restaurants.

As described above in Item 13, we have the right to use certain of SS, LLC's Trade Secrets under the License Agreement. We will disclose much of the above-described information to you in training or in the Operations Manual, and in providing guidance and assistance to you under the Franchise Agreement. In addition, in the course of the operation of the Restaurant, you or your employees may develop ideas, concepts, methods, techniques or improvements, relating to the Restaurant which you agree to disclose to us and which we may then authorize you and others in writing to use in the operation of Taco John's Restaurants. (Any such information disclosed to or developed by you will be referred to as "Confidential Information.")

Your relationship with us does not give you any interest in the Confidential Information other than the right to use it to develop and operate a Taco John's Restaurant under a Franchise Agreement with us. Use or duplication of the Confidential Information in any other business would constitute an unfair method of competition. Under the ADA and the Taco John's Franchise Agreement, you acknowledge and agree that the Confidential Information belongs to us and our affiliates, may contain Trade Secrets belonging to us and our affiliates and is disclosed to you or authorized for your use only on the condition that you agree that you:

1. Will not use the Confidential Information in any other business or capacity;

- 2. Will maintain the absolute confidentiality of the Confidential Information during and after the term of the Franchise Agreement;
 - 3. Will not make unauthorized copies of any portion of the Confidential Information; and
- 4. Will use all reasonable procedures we may periodically require to prevent unauthorized use or disclosure of the Confidential Information which may include requiring each owner, officer, director, or multi-unit manager to sign a Nondisclosure and Noncompetition Agreement, in the form contained in Attachment C to this Disclosure Document.

ITEM 15. OBLIGATION TO PARTICIPATE IN THE ACTUAL OPERATION OF THE FRANCHISE BUSINESS

Area Development Agreement

You must at all times during the term of the ADA faithfully, honestly and diligently perform your obligations and continuously exert your best efforts to locate, promote and enhance the development of multiple Taco John's Restaurants within the Development Area. An individual (the "Development Executive") must be designated in writing who will be personally responsible for the Developer's development activities during the term of the ADA and who will devote his or her best efforts to the development of Taco John's Restaurants in the Development Area. The Development Executive must be an owner, shareholder or partner of your business. There is no requirement under the ADA that you or the Development Executive complete the Management Certification Program.

Franchise Agreement

You must continuously operate the Restaurant, perform your obligations under the Franchise Agreement and exert your best efforts to promote and enhance the business of the Restaurant for the full term of the Franchise Agreement. The Restaurant must, during substantially all operating hours, be under the direct, on premises supervision of the Franchisee, an owner or a manager who has completed our required training programs, has been certified by us and who devotes his or her full time and efforts to the management of the Restaurant. Your Active Certified Managers and other managers are not required to have an equity interest in the franchise or to sign non-competition agreements.

You are responsible for recruiting, appointing, hiring, firing, and supervising your Above Store Leaders, Active Certified Managers, employees, independent agents, and other representatives. You will have sole authority and control over the day-to-day operations of your Taco John's Restaurant and its employees and representatives. You are solely responsible for implementing training and other programs for employees related to the legal, safe, and proper performance of their work, regardless of the fact that we may provide advice, suggestions, and certain training programs regarding the overall operation of a Restaurant as described in this Disclosure Document. The advice, suggestions, and training we provide are to protect our brand and the Marks and not to control the day-to-day operation of your Taco John's Restaurant.

Upon our request, you will cause each owner, officer, and director to sign a separate non-disclosure agreement.

If the Franchisee is a business entity, all officers, directors, shareholders, partners, or members, as applicable, will be required to sign a Guaranty when you sign the Franchise Agreement. A copy of the Guaranty is attached to the Franchise Agreement.

ITEM 16. RESTRICTIONS ON WHAT THE FRANCHISEE MAY SELL

Area Development Agreement

The ADA does not restrict the goods or services which you may offer or sell. However, you will be bound by provisions in the Franchise Agreements you sign under the ADA that restrict the goods and services that you may offer and sell from a Taco John's Restaurant.

Franchise Agreement

You must open the Restaurant for business during all required hours of operations, sell all products and services included in the "Core Menu," and offer and sell approved products and services only in the manner we prescribe. You may not offer for sale or sell at the Restaurant, the Premises or any other location any products or services we have not approved and must not use the Restaurant or the Premises for any purpose other than the operation of the Restaurant. You must discontinue selling and offering for sale any products or services that we decide at any time to withdraw from Taco John's Restaurants, and we have the unrestricted right to change any or all of the goods and services you are authorized to sell. We retain the unlimited right to change the types of authorized goods and services offered in your Restaurant. You are not restricted as to the customers to whom you may sell approved products and services. You must meet and maintain the highest health standards and ratings applicable to operating the Restaurant. You must furnish to us, within five days after receipt, a copy of any violation or citation which indicates your failure to maintain local health or safety standards in operating the Restaurant.

ITEM 17. RENEWAL, TERMINATION, TRANSFER AND DISPUTE RESOLUTION

THE FRANCHISE RELATIONSHIP

<u>Area Development Agreement</u> - The following table lists certain important provisions of the Area Development Agreement. You should read these provisions in the Area Development Agreement attached to this Disclosure Document as Attachment B:

Provision	Section in ADA	Summary
(a) Length of the ADA term	Article 6	From execution of the ADA until the date for
		development of the last Restaurant as specified in
		the Development Schedule, or the deadline in the
		Development Schedule for signing the Franchise
		Agreement for the last Restaurant, which is
		typically three to five years.
(b) Renewal or extension of the term	None	You have no rights to renew or extend, but we may,
		in our sole discretion, grant an extension to the
		Development Schedule. You may be charged a fee
		for an extension.
(c) Requirements for you to renew or extend	None	See above.
(d) Termination by you	None	No specific provision.
(e) Termination by us without cause	None	No specific provision.
(f) Termination by us with cause	Section 7.B	We have the right to terminate if you commit any of
		several violations (see (g) and (h) below).

Dwa	.inia	Section in ADA	Summour
	vision "Cause" defined - defaults which can be	Section in ADA Section 7.B	Summary 30 days' notice for any breach, including failure to
(8)	cured		meet the Development Schedule. Default due to
			failure to meet the Development Schedule will not
			cause a default of any Franchise Agreements then
			in place, but all other defaults may cause a default
			of all Franchise Agreements.
(h)	"Cause" defined – non- curable defaults	Section 7.B	A termination of an underlying Franchise
			Agreement or other related agreements.
	Your obligations on termination/non-	Sections 7.C, D and F	You will have no further rights to develop
	renewal		Restaurants and must satisfy the obligations under
			the Franchise Agreements that survive expiration or
			termination of the ADA.
(j)	Assignment of contract by us	Section 8.A	There are no restrictions on our right to
			assign.
(k)	"Transfer" by you - defined	Section 8.B	Includes assignment, sale, gift, exchange or other
			change in ownership, whether voluntary or
			involuntary.
	Our approval of transfer by you	Section 8.B	We must approve all transfers.
(m)	Conditions for our approval of transfer	Sections 8.B, C, D, and E	
	by you		agreement approved; concurrent transfer of
()		7.7	underlying Franchise Agreements.
(n)	Our right of first refusal to acquire your	None	Not Applicable.
(-)	business	NI	NI - 4 A1: 1.1 -
	Our option to purchase your business	None None	Not Applicable.
7 1	Your death or total disability Non-competition covenants during		No specific provisions. You are subject to all restrictive covenants as set
(q)	the term of the franchise	Article 9	
	the term of the franchise		forth in any Franchise Agreement executed
			pursuant to the ADA and in any Nondisclosure and Noncompetition Agreements executed in
			conjunction with a Franchise Agreement.
(r)	Non-competition covenants after the	Article 9	You are subject to all restrictive covenants as set
(1)	franchise is terminated or expires	Afficie	forth in any Franchise Agreement executed
	tranemise is terminated or expires		pursuant to the ADA and in any Nondisclosure
			and Noncompetition Agreements executed in
			conjunction with a Franchise Agreement.
(s)	Modification of the agreement	Section 12.D	No modification unless by mutual written
(5)	Triodiffication of the agreement	Section 12.B	agreement.
(t)	Integration/merger clause	Section 12.D	Only the terms of the ADA are binding.
-	Dispute resolution by arbitration or	Section 13.B	The ADA incorporates the arbitration requirement
	mediation		of the last Franchise Agreement you executed
			pursuant to the ADA.
(v)	Choice of forum	Section 13.B	The ADA incorporates the arbitration requirement
` ′			of the last Franchise Agreement you executed
			pursuant to the ADA. If no Franchise Agreements
			have been signed, arbitration before the American
			Arbitration Association in Minneapolis, Minnesota
			(subject to applicable state law).
(w)	Choice of law	Section 13.A	Wyoming law applies (subject to applicable state
			law).

THE FRANCHISE RELATIONSHIP

<u>Franchise Agreement</u> - The following table lists certain important provisions of the Franchise Agreement. You should read these provisions in the Franchise Agreement attached to this Disclosure Document as <u>Attachment A</u>:

		Section in the Franchise	
Prov	vision	Agreement	Summary
(a)	Length of the franchise term	Section 2.1	20 years for a traditional Taco John's Restaurant; 10 years for a non-traditional Restaurant.
(b)	Renewal or extension of the term	Section 2.4(a)	Opportunity to renew for up to three additional terms of 10 years each.
(c)	Requirements for you to renew or extend	Section 2.4, Section 4 of Non- Traditional Facility Rider	Compliance with the Franchise Agreement; remodel, renovate, refresh, reimage, refurbish, and modernize the Restaurant if we require; notice to us; payment of renewal fee; execution of our then-current Franchise Agreement and ancillary agreements and meet training requirements; others. You may be asked to sign an agreement with materially different terms and conditions than your original agreement.
(d)	Termination by you	Section 15.1	You may terminate if we materially breach the Franchise Agreement and we fail to cure the breach after an opportunity to cure of 90 days.
(e)	Termination by us without cause	Section 15.2(q)	We can terminate if there is a force majeure event that prevents the parties from performing under the Franchise Agreement and continues for six consecutive months or longer.
(f)	Termination by us with cause	Section 15.2	We have the right to terminate if you commit any of several violations (see (g) and (h)).
(g)	"Cause" defined - defaults which can be cured	Section 15.2(a-i)	30 days to cure failure to open the Restaurant as required or to obtain or maintain required certifications; loss of the Premises without your fault and failure to locate another suitable Premises; termination of your lease due to your default; unauthorized transfer; failure to comply with provisions, specifications, standards and procedures; relocation of the restaurant without obtaining our permission or failure to de-identify the location you vacated; failure to offer the Core Menu items; 10 days for failure to pay amounts due or misuse of Marks; 7 days for filing legal action in violation of dispute resolution terms in Franchise Agreement; 72 hours for violation of health and safety laws; others.
(h)	"Cause" defined - non- curable defaults	Section 15.2(j-p, r) and Section 15.3	Your bankruptcy or insolvency; abandonment; false or material misrepresentation; seizure of the Premises; conviction for conduct that reflects adversely on the Restaurant; operation in an unsafe manner; sexual harassment or discrimination; three defaults in three years; defaults of other agreements with us or our affiliates by you or affiliates of yours, others.

Prov	rision	Section in the Franchise Agreement	Summary
(i)	Your obligations on termination/non-renewal	Article 16	Pay us amounts due; pay us liquidated damages, if terminated due to your default, based on your monthly average of Net Sales multiplied by the number of months remaining on the term of this Agreement but not to exceed 24; cease using any items with the Marks and confidential information; de-identify; return the Operations Manual and other materials; comply with non-competition covenants (see (r) below); allow us the right to operate the Restaurant, and comply with our option to purchase (see (o) below).
(j)	Assignment of contract by us	Section 14.2	There is no restriction on our assignment or transfer of our interest.
(k)	"Transfer" by you - defined	Section 14.3	Includes assignment, sale, gift, exchange or other change in ownership, whether voluntary or involuntary.
(1)	Our approval of transfer by you	Section 14.3	By our prior written consent.
(m)	Conditions for our approval of transfer by you	Section 14.4	Notice to us; you pay amounts owed; the transfer fee is paid; the new owner is approved, trained and certified; you assign or sublease the Premises to the new owner, if necessary; the Restaurant meets our image according to current specifications; the purchase agreement is approved; the new owner executes our then-current franchise agreement and ancillary agreements; and you sign a general release.
(n)	Our right of first refusal to acquire your business	Section 14.7	We have the right to match any offer.
(o)	Our option to purchase your business	Sections 16.6 and 16.7	We may purchase your assets at fair market value, and acquire rights to your Premises by lease or purchase, upon prior written notice to you when the franchise terminates or expires.
(p)	Your death or total disability	Section 14.5	Transfer of ownership upon your or your owner's death or total disability does not require our consent or trigger our right of first refusal so long as the transfer is to a family member or a partner that meets our qualifications.
(q)	Non-competition covenants during the term of the franchise	Section 8.3	No involvement in a competing business anywhere.
(r)	Non-competition covenants after the franchise is terminated or expires	Section 16.5, Section 12 of Non-Traditional Facility Rider	No involvement in a competing business for 2 years within five miles of the Protected Territory (or within five miles of the Premises of a non-traditional location) or within 5 miles of the Protected Territory of any other Taco John's Restaurant.
(s)	Modification of the agreement	Section 17.14	No modification unless by mutual written agreement, or, at our option, upon approval of 75% of our franchisees affected by the modification.
(t)	Integration/merger clause	Section 17.14	Only the terms of the Franchise Agreement and the exhibits are binding. However, nothing in the Franchise Agreement or any related agreement is intended to disclaim any representations made by us in this Disclosure Document.

Provision		Section in the Franchise Agreement	Summary
(u)	Dispute resolution by arbitration or mediation	Section 17.10	Arbitration of most claims in Minneapolis, Minnesota before the American Arbitration Association.
(v)	Choice of forum	Section 17.10	Arbitration before the American Arbitration Association in Minneapolis, Minnesota (subject to applicable state law).
(w)	Choice of law	Section 17.16	Wyoming law applies (subject to applicable state law).

ITEM 18. PUBLIC FIGURES

We do not use any public figure to promote the franchise.

ITEM 19. FINANCIAL PERFORMANCE REPRESENTATIONS

The FTC's Franchise Rule permits a franchisor to provide information about the actual or potential financial performance of its franchised and/or franchisor-owned outlets, if there is a reasonable basis for the information, and if the information is included in the Disclosure Document. Financial performance information that differs from that included in Item 19 may be given only if: (1) a franchisor provides the actual records of an existing outlet you are considering buying; or (2) a franchisor supplements the information provided in this Item 19, for example, by providing information about possible performance at a particular location or under particular circumstances.

CHARTS OF PER FRANCHISED RESTAURANTS AVERAGE SALES AND ACTUAL SALES

Included in the charts below are all of the 312 franchised Taco John's Restaurants with drive-thru windows that operated for all of calendar year 2024 and reported their sales to us through our approved Point of Sale System (or an alternative reporting method that we believe to be accurate and have approved) for all of calendar year 2024. The Notes that follow the charts explain the data included in the chart and you should review them carefully.

[CHARTS APPEAR ON FOLLOWING PAGE]

CHART 1

AVERAGE SALES OF FREESTANDING TRADITIONAL TACO JOHN'S RESTAURANTS WITH DRIVE-THRU WINDOWS DURING CALENDAR YEAR 2024								
Taco John's	Number of	Average	Nui	nber of	High	Median	Low	
Restaurants	Restaurants in group	Sales	Restaurants and percentage in the group that met or exceeded the average sales of the group		performer	performer	performer	
Top Quartile	65	\$1,881,809	27	41.5%	\$3,087,487	\$1,788,802	\$1,470,885	
Second Quartile	65	\$1,312,352	33	50.8%	\$1,466,536	\$1,321,375	\$1,185,932	
Third Quartile	65	\$1,068,530	33	50.8%	\$1,181,008	\$1,086,169	\$961,098	
Lower Quartile	64	\$771,452	36	56.3%	\$959,682	\$802,923	\$244,397	
Combined	259	\$1,260,416	104	40.2%	\$3,087,487	\$1,185,932	\$244,397	

CHART 2

	AVERAGE SALES OF ENDCAP TRADITIONAL TACO JOHN'S RESTAURANTS WITH DRIVE-THRU WINDOWS DURING CALENDAR YEAR 2024								
Taco John's Restaurants	Number of Restaurants in group	Average Sales	Number of Restaurants and percentage in the group that met or exceeded the average sales of the group		High performer	Median performer	Low performer		
Top Quartile	7	\$1,472,820	3	42.9%	\$1,667,333	\$1,459,707	\$1,268,805		
Second Quartile	7	\$1,156,611	4	57.1%	\$1,264,434	\$1,157,483	\$1,055,930		
Third Quartile	7	\$878,173	2	28.6%	\$1,022,734	\$864,118	\$804,158		
Lower Quartile	7	\$714,016	5	71.4%	\$766,879	\$741,989	\$550,108		
Combined	28	\$1,055,405	14	50.0%	\$1,667,333	\$1,039,332	\$550,108		

CHART 3

			СНА	KT 3				
AVERAGE SALES OF TACO JOHN'S RESTAURANTS IN CONVENIENCE STORES OR TRAVEL PLAZAS WITH DRIVE-THRU WINDOWS DURING CALENDAR YEAR 2024								
Taco John's Number of Restaurants in group Taco John's Restaurants Restaurants Sales Number of Restaurants and performer Performer Restaurants and performer Performer								
Top Quartile	6	\$1,506,199	2	33.3%	\$2,014,275	\$1,444,984	\$1,236,834	
Second Quartile	6	\$1,003,213	2	33.3%	\$1,196,902	\$959,926	\$890,040	
Third Quartile	6	\$833,587	3	50.0%	\$883,168	\$826,153	\$792,939	
Lower Quartile	7	\$643,610	4	57.1%	\$712,852	\$500,301		
Combined	25	\$982,531	8	32.0%	\$2,014,275	\$883,168	\$500,301	

NOTES TO CHARTS

- 1. Chart 1 is a statement of the average "Sales" of the 259 Taco John's traditional freestanding Restaurants with drive-thru windows and that reported Sales to us through our approved Point of Sale System or an alternative reporting method that we believe to be accurate and have approved (an "Approved Alternative System") for all of calendar year 2024. Chart 2 is a statement of the average "Sales" of 28 Taco John's traditional Restaurants located in the endcap space in strip malls with drive-thru windows that reported Sales to us through our POS System or an Approved Alternative System for all of calendar vear 2024. Chart 3 is a statement of the average "Sales" of 25 Taco John's Restaurants located in Convenience Stores or Travel Plazas with Drive-thru windows that reported Sales to us through our POS System or an Approved Alternative System for all of calendar year 2024. These charts collectively report the average "Sales" of 312 franchised Taco John's Restaurants. The term "Average Sales" as used in these charts means the average of the total Sales of each grouping for 2024 as reported to us through our POS System or an Approved Alternative System. The charts also show the Sales of the highest performer, lowest performer and the median sales. The term "median" means the data point that is the center of all data points used, or in the context of these charts, the results of the franchisee falling in the middle of each grouping. For any groupings with an even number of data points, the two middle numbers are added and then divided by two.
- 2. We had a total of 333 franchised Taco John's Restaurants in our system as of December 31, 2024. Of those, one Restaurant does not report its Sales through our approved Point of Sale System (or an Approved Alternative System) and 10 Restaurants do not have drive-thru windows as part of their Restaurants. Because the restaurant models we are offering under this Disclosure Document require the use of our approved Point of Sale System and are expected to have a drive-thru window, we did not include those Restaurants in these charts. Further, six of our franchised Taco John's Restaurants operating as of December 31, 2024 opened during the course of the year and did not operate a full year. Four additional locations are considered prototypes of the DT Digital Olé or earlier concepts and are not included in these charts. Additionally, 30 Taco John's Restaurants closed during the year 2024,

and are not included in these charts. Three of the outlets that closed permanently in calendar year 2024 had been open for less than 12 months. We also had a total of seven company-owned Taco John's Restaurants that operated for the full 2024 calendar year that are not included in these charts.

- 3. We only included in these charts the Sales figures of those Restaurants that report their Sales through our approved Point of Sale System or an Approved Alternative System because we believe the Sales figures we receive from those Restaurants are more reliable. Further, the franchised Taco John's Restaurants offered pursuant to this Disclosure Document are required to use only our approved Point of Sale System to report their Sales to us, although in certain very limited situations we have authorized use of an Approved Alternative System. The franchised Taco John's Restaurants offered pursuant to this Disclosure Document will most likely have a drive-thru window as part of the Restaurant and will be required to use our approved Point of Sale System to report their Sales to us.
- 4. "Sales" means the total receipts of the Restaurant, but does not include sales tax or equivalent taxes.
- 5. The Taco John's Restaurants included in these charts sell substantially the same products and services as the franchised Taco John's Restaurants offered pursuant to this Disclosure Document.
- 6. As of the start of calendar year 2024, the Taco John's Restaurants included in these charts had been in operation for periods ranging from one year to 56 years.
- 7. There is a large variation in the range of Sales generated by our Franchisees during calendar year 2024. During calendar year 2024, Sales by the Taco John's Restaurants reported in these charts varied from a high of \$3,087,487 to a low of \$244,397.
- 8. The sales information for the Restaurants included in these charts is information as reported by the Franchisees and has not been audited or otherwise verified by us. These charts only show Sales of Taco John's Restaurants and are not a statement of profits or earnings. We show no expenses in these charts. We cannot accurately determine expenses of our franchised Restaurants because franchisees do not report expenses to us.
- 9. These charts are historic financial performance representations based on data from 2024. They are not a forecast or projection of the franchisee's potential future financial performance.
- 10. The primary characteristic of your Restaurant that could differ from the outlets shown in these charts is the model of Restaurant. Each chart shows the results of different models of Restaurant. Since your Restaurant will open under one of these models, your Restaurant will vary from those of the other models represented here. Otherwise, except for the fact the represented outlets in this Item 19 have operated for a longer period of time, there are no financial or operational characteristics of our represented outlets that are reasonably anticipated to differ materially from those of a new franchisee.

Caution: Some outlets have sold this amount. Your individual results may differ. There is no assurance that you will sell as much.

If you rely upon our figures, you must accept the risk of not doing as well. Your actual financial results are likely to differ from the figures presented. If possible, show these figures to someone who can advise you, like a lawyer or accountant.

ADDITIONAL NOTES APPLICABLE TO ALL CHARTS:

In presenting this data, we do not estimate the length of time it will take for any particular franchisee to achieve any revenues levels, sales levels, or other results.

Your ability to achieve any certain level of sales or profits will depend upon factors not within our control, including the occurrence of certain start up and operating expenses and the amount of those expenses, and your level of expertise. It will also depend on your ability to promote your Tacos John's Restaurant to customers, which is a major factor in determining your profitability and is dependent upon your marketing ability and efforts, competition within your market, economic conditions, the amount of time you devote to your business, your management and leadership skills, the time and money spent promoting your business, your profit motivation and other market factors. Expenses will vary from franchisee to franchisee.

Written substantiation of this financial performance representation will be made available to you upon reasonable request.

Other than the preceding financial performance representation, we do not make any financial performance representations. We also do not authorize our employees or representatives to make any such representations either orally or in writing. If you are purchasing an existing Taco John's Restaurant, however, we may provide you with the actual records of that Restaurant. If you receive any other financial performance information or projections of your future income, you should report it to the franchisor's management by contacting Kristin M. Nuss at 1650 West End Blvd., Suite 200, St. Louis Park, Minnesota 55416, and (307) 635-0101, the Federal Trade Commission, and the appropriate state regulatory agencies.

ITEM 20. OUTLETS AND FRANCHISEE INFORMATION

TABLE NO. 1 SYSTEMWIDE OUTLET SUMMARY FOR YEARS 2022 TO 2024

Column 1	Colum 2	Column 3 Outlets at the	Column 4 Outlets at the	Column 5
Outlet Type	Year	Start of the Year	End of the Year	Net Change
	2022	367	361	-6
Franchised	2023	361	357	-4
	2024	357	333	-24
	2022	6	7	+1
Company-Owned	2023	7	7	0
	2024	7	7	0
TOTAL TACO JOHN'S	2022	373	368	-5
RESTAURANT	2023	368	364	-4
OUTLETS	2024	364	340	-24

TABLE NO. 2 TRANSFER OF OUTLETS FROM FRANCHISEES TO NEW OWNERS (OTHER THAN THE FRANCHISOR) FOR YEARS 2022 TO 2024

Column 1	Column 2	Column 3
State	Year	Number of Transfers
	2022	4
Colorado	2023	0
	2024	0
	2022	3
Illinois	2023	2
	2024	1
	2022	0
Indiana	2023	2
	2024	0
	2022	6
Iowa	2023	5
	2024	0
	2022	0
Kansas	2023	0
	2024	2
	2022	0
Kentucky	2023	2
-	2024	0
	2022	0
Massachusetts	2023	0
	2024	1
	2022	0
Minnesota	2023	4
	2024	1
	2022	4
Missouri	2023	1
	2024	0
	2022	3
Montana	2023	0
	2024	3
	2022	3
Nebraska	2023	3
	2024	0
	2022	2
North Dakota	2023	0
2	2024	0
	2022	3
South Dakota	2023	2
Zoun Danou	2024	1
	2022	0
Tennessee	2023	3
Termessee	2024	0
	2027	U

Column 1	Column 2	Column 3
State	Year	Number of Transfers
	2022	1
Wisconsin	2023	2
	2024	0
	2022	0
Wyoming	2023	5
	2024	0
	2022	29
TOTAL	2023	31
	2024	9

TABLE NO. 3 STATUS OF FRANCHISED OUTLETS FOR YEARS 2022 TO 2024 (1)

Column 1 State	Column 2 Year	Outlets at Start of Year	Column 4 Outlets Opened	Column 5 Terminations	Column 6 Non- Renewals	Column 7 Re-Acquired By Franchisor	Column 8 Ceased Operations – Other Reason	Column 9 Outlets at End of Year
Arkansas	2022	1	0	0	0	0	0	1
	2023	1	0	0	0	0	0	1
	2024	1	0	0	0	0	0	1
Colorado	2022	16	0	0	0	0	1	15
	2023	15	0	0	0	0	0	15
	2024	15	0	0	1	0	5	9
Idaho	2022	2	0	0	0	0	1	1
	2023	1	0	0	0	0	0	1
	2024	1	0	0	0	0	0	1
Illinois	2022	12	2	0	0	0	0	14
	2023	14	0	0	0	0	1	13
	2024	13	0	0	0	0	2	11
Indiana	2022	5	0	0	0	0	1	4
	2023	4	1	0	0	0	0	5
	2024	5	0	0	0	0	0	5
Iowa	2022	57	0	0	0	0	1	56
	2023	56	0	0	0	0	3	53
	2024	53	0	0	0	0	1	52
Kansas	2022	17	1	0	0	0	1	17
	2023	17	1	0	0	0	2	16
	2024	16	0	0	0	0	0	16
Kentucky	2022	10	0	0	0	0	1	9
	2023	9	1	0	0	0	0	10
	2024	10	0	0	0	0	0	10
Massachusetts	2022	0	0	0	0	0	0	0
	2023	0	2	0	0	0	0	2
	2024	2	0	0	0	0	0	2

Column 1	Column 2	Column 3	Column 4	Column 5	Column 6	Column 7	Column 8	Column 9
State	Year	Outlets at Start of Year	Outlets Opened	Terminations	Non- Renewals	Re-Acquired By Franchisor	Ceased Operations – Other Reason	Outlets at End of Year
Michigan	2022	0	1	0	0	0	0	1
	2023	1	4	0	0	0	0	5
	2024	5	0	0	0	0	5	0
Minnesota	2022	58	0	0	0	0	2	56
	2023	56	1	0	0	0	3	54
	2024	54	1	2	0	0	1	52
Missouri	2022	11	0	0	0	0	0	11
	2023	11	0	0	0	0	0	11
	2024	11	0	0	0	0	0	11
Montana	2022	17	1	0	0	0	0	18
	2023	18	0	0	0	0	0	18
	2024	18	0	0	0	0	0	18
Nebraska	2022	40	0	0	0	0	2	38
	2023	38	0	0	0	0	2	36
	2024	36	0	1	0	0	0	35
Nevada	2022	1	0	0	0	0	0	1
	2023	1	0	0	0	0	0	1
	2024	1	0	0	0	0	0	1
New	2022	1	0	0	0	0	1	0
York	2023	0	0	0	0	0	0	0
	2024	0	0	0	0	0	0	0
North Dakota	2022	21	0	0	0	0	1	20
	2023	20	0	0	0	0	5	15
-1.	2024	15	0	0	0	0	1	14
Ohio	2022	3	0	0	0	0	0	3
	2023	3	1	0	0	0	0	4
~	2024	4	1	0	0	0	2	3
South Dakota	2022	37	2	0	0	0	1	38
	2023	38	1	0	0	0	1	38
	2024	38	0	1	0	0	1	36
Tennessee	2022	4	2	0	0	0	0	6
	2023	6	1	0	0	0	1	6
TT. 1	2024	6	0	0	0	0	3	3
Utah	2022	0	1	0	0	0	0	1
	2023	1	0	0	0	0	0	1
W1:	2024	1	0	0	0	0	0	1
Washington	2022	2	1	0	0	0	0	3
	2023	3	0	0	0	0	1	2
177'	2024	2	0	0	0	0	0	2
Wisconsin	2022	31	0	0	0	0	2	29
	2023	29	3	0	0	0	0	32
	2024	32	4	0	0	0	1	35

Column 1	Column 2	Column 3	Column 4	Column 5	Column 6	Column 7	Column 8	Column 9
							Ceased	
State	Year	Outlets	Outlets	Terminations	Non-	Re-Acquired		Outlets at
		at Start	Opened		Renewals	By	– Other	End of
		of Year				Franchisor	Reason	Year
***	2022	2.1		0	0	0		1.0
Wyoming	2022	21	0	0	0	0	2	19
	2023	19	0	0	0	0	1	18
	2024	18	0	0	1	0	2	15
TOTALS	2022	367	11	0	0	0	17	361
	2023	361	16	0	0	0	20	357
	2024	357	6	4	2	0	24	333

⁽¹⁾ If multiple events occurred affecting an outlet, this table shows the event that occurred last in time.

TABLE NO. 4 STATUS OF COMPANY-OWNED OUTLETS FOR YEARS 2022 TO 2024

Column 1	Column 2	Column 3	Column 4	Column 5	Column 6	Column 7	Column 8
State	Year	Outlets at Start of Year	Outlets Opened	Re-Acquired From Franchisees	Outlets Closed	Outlets Sold to Franchisees	Outlets at End of Year
Illinois	2022	1	0	0	1	0	0
	2023	0	0	0	0	0	0
	2024	0	0	0	0	0	0
Minnesota	2022	0	2	0	0	0	2
	2023	2	0	0	0	0	2
	2024	2	0	0	0	0	2
Wyoming	2022	5	0	0	0	0	5
	2023	5	0	0	0	0	5
	2024	5	0	0	0	0	5
TOTAL	2022	6	2	0	1	0	7
	2023	7	0	0	0	0	7
	2024	7	0	0	0	0	7

TABLE NO. 5 PROJECTED OPENINGS AS OF DECEMBER 31, 2024

Column 1	Column 2	Column 3	Column 4	
State	Franchise Agreements Signed but Restaurants Not Open	Projected New Franchised Restaurants in Next Fiscal Year	Projected Company- Owned Openings in Next Fiscal Year	
Connecticut	0	1	0	
Iowa	0	2	0	
Indiana	0	1	0	
Kentucky	0	1	0	

Column 1	Column 2	Column 3	Column 4
State	Franchise Agreements Signed but Restaurants Not Open	Projected New Franchised Restaurants in Next Fiscal Year	Projected Company- Owned Openings in Next Fiscal Year
Massachusetts	1	1	0
North Dakota	1	2	0
Wisconsin	0	0	0
TOTAL	2	8	0

Attached to this Disclosure Document as <u>Attachment D</u> is a list of the names, addresses, and telephone numbers of Franchisees and Developers. Attached to this Disclosure Document as <u>Attachment E</u> is a list of the name and current business telephone number (or, if unknown, the last known home telephone number) of every Franchisee who had an outlet terminated, canceled, not renewed, or otherwise voluntarily or involuntarily ceased to do business under the franchise agreement during our recently completed fiscal year, or who had not communicated with us within 10 weeks of the issuance date of this Disclosure Document. If you buy this franchise, your contact information may be disclosed to other buyers when you leave the franchise system. During the last three fiscal years, no current or former Franchisees have signed confidentiality clauses that restrict them from discussing their experiences as a Franchisee in our franchise system.

In the Franchise Agreement, we agree to regularly consult with any Franchisee association which represents 50% or more of our current total of both franchised Restaurants and Franchisees. (Franchise Agreement - Section 17.11) We currently consult with the elected Board of Directors of the Association of Taco John's Franchisees, Inc. (P.O. Box 155, Minneapolis, Minnesota 55387-0155, 952-250-3121, jschindler@atjf.net, www.atjf.net) on various matters.

ITEM 21. FINANCIAL STATEMENTS

Attached as <u>Attachment F</u> are our audited financial statements for the fiscal years ended December 31, 2024, 2023, and 2022.

ITEM 22. CONTRACTS

The following agreements are attached as Attachments to this Disclosure Document:

Attachment A - Franchise Agreement and Exhibits

Attachment B - Area Development Agreement

Attachment C - Nondisclosure and Noncompetition Agreement

Attachment I - Disclosure Acknowledgment

Attachment J - Gift Card Onboarding and Change Form

Attachment K – Lease Rider

Attachment L – Operational Assessment Program Confidentiality/Application Agreement

ITEM 23. RECEIPT

Attachment N of this Disclosure Document contains the receipt pages. Please sign and date the second receipt page as of the date you received this Disclosure Document and promptly return it to us as specified on that page.

ATTACHMENT A FRANCHISE AGREEMENT

TACO JOHN'S INTERNATIONAL, INC. FRANCHISE AGREEMENT



Franchisee:			
Restaurant L	ocation:		

THIS CONTRACT CONTAINS A BINDING ARBITRATION PROVISION WHICH MAY BE ENFORCED BY THE PARTIES.

033125, as amended 071725

TABLE OF CONTENTS

<u>ART</u>	TICLE/S	<u>SECTION</u>	PAGE
1.	INT	RODUCTION	1
2.	GRA	ANT OF FRANCHISE	2
4.	2.1	Grant and Term	
	2.2	Full Term Performance	
	2.3	Protected Territory; Right of First Option; Reservation of Rights	
	2.4	Renewal of Franchise	
3.	DEV	VELOPMENT AND OPENING OF RESTAURANT	6
	3.1	Site Location	
	3.2	Site Acceptance	
	3.3	Site Acquisition	
	3.4	Construction Documents and Specifications	
	3.5	Equipment, Fixtures, Furnishings and Signs	
	3.6	Restaurant Opening.	
4.	TRA	AINING	8
••	4.1	Initial Training and Certification	
	4.2	Active Certified Managers	
	4.3	Ongoing and Additional Training Programs	
	4.4	Purpose of Training	
5.	CIII	DANCE; OPERATIONS MANUAL	10
J.	5.1	Guidance and Assistance	
	5.2	Operations Manual	
(N/T A 1	nve	11
6.		RKS	
	6.1	Ownership and Goodwill of Marks	
	6.2	Limitations on Franchisee's Use of Marks	
	6.3	Notification of Infringements and Claims	
	6.4	Discontinuance of Use of Marks	
	6.5	Indemnification of Franchisee	12
7.		LATIONSHIP OF PARTIES; INDEMNIFICATION	
	7.1	Independent Contractor; No Fiduciary Relationship	
	7.2	No Liabililty, No Warranties	
	7.3	Indemnification	13
8.	CON	NFIDENTIAL INFORMATION; NON-COMPETITION	13
	8.1	Types of Confidential Information	
	8.2	Non-disclosure Agreement	14
	8.3	Non-competition Agreement	14
9.	FEE	ZS	14
	9.1	Initial Franchise Fee	15
	9.2	Royalty	

	9.3	Advertising and Marketing Fee	15
	9.4	Technology Fee	
	9.5	Grand Opening Marketing	
	9.6	Interest on Late Payments; Nonsufficient Funds Fees	
	9.7	No Withholding	
	9.8	Transfer of Ownership Fee	
	9.9	Renewal Fee.	
	9.10	Relocation Fee	
	9.11	Non-compliance Fee	
	9.12	Delayed Opening Fee.	
	9.13	Media Relations Support Fee	
	9.14	Allocation of Payments	
	9.15	Payment Authorization	
10.	REST	AURANT OPERATING STANDARDS	18
	10.1	Importance of Uniformity	18
	10.2	Condition and Appearance of Restaurant	18
	10.3	Relocation of Franchise; Sale-Leaseback of Real Estate	19
	10.4	Restaurant Menu and Service Methods	
	10.5	Approved Products, Distributors and Suppliers	21
	10.6	Compliance with Laws and Good Business Practices	23
	10.7	Ownership, Management and Personnel of Restaurant	
	10.8	Insurance	24
	10.9	Websites and Social Platforms	25
	10.10	Guest Experience Program	25
	10.11	Online Learning Management System	
	10.12	Nonattendance of Franchisee Convention or Manager Convention	26
11.	ADVI	ERTISING AND MARKETING	26
	11.1	Advertising Fund	
	11.2	Advertising Cooperatives	
	11.3	Participation in Certain Programs and Promotions, including Customer	
	-	Loyalty/Rewards Program	28
12.	ACCO	DUNTING, REPORTS AND FINANCIAL STATEMENTS	29
13.	INSPI	ECTIONS AND AUDITS	29
	13.1	Franchisor's Right to Inspect Restaurant	29
	13.2	Franchisor's Right to Audit	29
14.	TRAN	NSFER REQUIREMENTS	
	14.1	Interests in Franchisee	30
	14.2	Transfer by Franchisor	
	14.3	No Transfer Without Approval	
	14.4	Conditions for Approval of Transfer	
	14.5	Death and Total Disability	
	14.6	Effect of Consent to Transfer	
	14.7	Franchisor's Right of First Refusal	
	14.8	Public Offering	34
15.	TERN	MINATION OF FRANCHISE	34

	15.1	By Franchisee	34
	15.2	By Franchisor	34
	15.3	Cross Default and Cross Termination	36
16.	RIGH	TS AND OBLIGATIONS OF FRANCHISOR AND FRANCHISEE	UPON
	TERN	MINATION OR EXPIRATION OF FRANCHISE	36
	16.1	Payment of Amounts Owed to Franchisor	36
	16.2	Marks	
	16.3	De-Identification	37
	16.4	Confidential Information	37
	16.5	Covenant Not to Compete	37
	16.6	Franchisor's Option to Purchase Restaurant	38
	16.7	Franchisor's Right to Operate Restaurant & Receive Assignment of Lease	
	16.8	Continuing Obligations	39
17.	ENFO	DRCEMENT	39
	17.1	Invalid Provisions; Substitution of Valid Provisions	
	17.2	Unilateral Waiver of Obligations	
	17.3	Written Consents from Franchisor	
	17.4	Reasonableness	
	17.5	Duty of Good Faith	
	17.6	No Guarantees.	
	17.7	No Waiver	
	17.8	Cumulative Remedies	
	17.9	Specific Performance; Injunctive Relief	
	17.10	Arbitration	
	17.11	Franchisee's Right to Associate	
	17.12	Limitations of Claims	
	17.13	Binding Effect	
	17.14	Entire Agreement	
	17.15	No Liability to Others; No Other Beneficiaries	
	17.16	Governing Law	
18.	MISC	ELLANEOUS	43
	18.1	Construction and Definitions	
	18.2		
	18.3	Multiple Originals	43
	18.4	Timing Is Important	
	18.5	Independent Provisions	
	18.6	Delegation of Duties	
	18.7	Force Majeure	
19.	NOTI	CES	44
EXHI	BITS		
I	ADDE	NDUM TO FRANCHISE AGREEMENT	
II		ORIZATION AGREEMENT FOR DIRECT PAYMENTS	
III		EMENT OF OWNERSHIP	
IV		ANTY	
V		TRADITIONAL FACILITY RIDER	
VI		E RIDERS	
V 1	יוידעני		

TACO JOHN'S INTERNATIONAL, INC. FRANCHISE AGREEMENT

This Franchise Agreement (this "Agreement") is made, entered into and effective as of the date (the "Effective Date") set forth in the Addendum to Franchise Agreement attached as Exhibit I (the "Addendum"). The parties to this Agreement are you, the undersigned franchisee, as the Franchisee, and us (Taco John's International, Inc., a Wyoming corporation). This Agreement is for a Taco John's Restaurant to be located at the address listed on the Addendum.

1. INTRODUCTION

This Agreement has been written in an informal style to make it more easily readable and to be sure that you become thoroughly familiar with all of the important rights and obligations this Agreement covers before you sign it. In this Agreement, we refer to Taco John's International, Inc. as "we" or the "Franchisor." We refer to you as "you," or the "Franchisee." If the Franchisee is an entity, you will notice certain provisions that are applicable to its owners, upon whose business skill, financial capability and personal character we are relying in entering into this Agreement. Those individuals will be referred to in this Agreement as "owners."

We have a system for the establishment and operation of Mexican food restaurants (each, a "Taco John's Restaurant" or "Restaurant") with distinctive recipes, interiors and exteriors, color schemes, fixtures and equipment, which we refer to in this Agreement as the "System" and which we may modify, develop and supplement from time to time. We identify the System by the use of certain trademarks, service marks and other commercial symbols, including the mark TACO JOHN'S®, and certain associated designs, artworks and logos, which we may change or add to from time to time (the "Marks").

This Agreement is being presented to you because you desire to obtain the right to develop, own and be franchised to operate a Taco John's Restaurant.

There are currently two Taco John's Restaurant concepts offered by us pursuant to this Agreement. Our feature concept is our traditional restaurant concept, which has three prototype building models; a freestanding building, an endcap location in a shopping mall, strip mall or alternative facility with our approval, and a "DT (drive-thru) Digital Olé" building. The freestanding building and endcap location models typically have a premises size of 1,800 to 2,250 square feet with the capacity to seat 20 to 40 patrons and with drive-thru window service. The DT Digital Olé models are designed for smaller markets and typically have a Premises size of 946 to 1,500 square feet and focus on drive-thru, pickup, and walk-up service with limited or no seating. Our models of DT (drive-thru) Digital Olé buildings include the "DT Digital Olé" (without seating) and the "DT Digital Olé+" (with seating). Unless otherwise noted, references in this Agreement to our traditional restaurant refers to the freestanding, endcap, and DT Digital Olé models. The second franchise concept is for non-traditional locations. A non-traditional location is a venue or facility where the primary purpose of the consumer's visit is typically for reasons other than the purchase of goods from a Taco John's Restaurant. Examples include airports and other transportation hubs, hospitals, college campuses and other educational facilities, convention centers, grocery stores, department stores, "big box" retail centers, sports arenas and stadiums, hotels and office buildings, military installations (Army and Air Force Exchange Services), and food courts. Typically, these are full-service restaurants with a premises size of 675 to 2,000 square feet. Non-traditional locations typically do not have a drive-thru window.

In signing this Agreement, you acknowledge your understanding of the importance of our standards of quality and service and the necessity of operating your Taco John's Restaurant in conformity with our

standards and specifications. You also acknowledge that you have conducted an independent investigation of the Taco John's Restaurant business, you have had ample opportunity to seek independent advice and recognize that, like any other business, the nature of it may change over time, that an investment in a Taco John's Restaurant involves business risks, and that the success of this business venture is primarily dependent on your business abilities and efforts.

We expressly disclaim making, and you acknowledge that you have not received or relied on, any information as to the revenues, profits, or likelihood of success of the Taco John's Restaurant venture contemplated by this Agreement, other than as set forth in our Franchise Disclosure Document. You acknowledge that there have been no representations by us or our officers, directors, shareholders, employees, or agents, that are inconsistent with the statements made in our Franchise Disclosure Document or the provisions of this Agreement. You further represent to us, as an inducement to our entering into this Agreement with you, that there have been no misrepresentations to us in your application for the rights granted by this Agreement or in the financial information provided by you and your owners.

2. GRANT OF FRANCHISE

2.1 Grant and Term.

Subject to the provisions of this Agreement, we grant to you a franchise (the "Franchise") to operate a Taco John's Restaurant of the type designated in the Addendum attached hereto at the location listed on the Addendum (the "Premises"), under the name TACO JOHN'S®, and to use the System and the Marks in the operation of the Restaurant for an "Initial Term" as set forth in the Addendum beginning on the Effective Date of this Agreement, unless the Franchise is terminated earlier pursuant to this Agreement. Termination or expiration of this Agreement will constitute a termination or expiration of your Franchise.

2.2 Full Term Performance.

You specifically agree to continuously operate the Restaurant, perform your obligations under this Agreement, and exert your best efforts to promote and enhance the business of the Restaurant for the full term of this Agreement in accordance with the System and all of our standards and specifications, and we agree to perform all our obligations under this Agreement continuously throughout the term of this Agreement.

2.3 Protected Territory; Right of First Refusal; Reservation of Rights.

(a) <u>Protected Territory</u>. Provided you are in substantial compliance with all the provisions of this Agreement and except as described in this Section 2.3, we will not establish, operate, or grant a new franchise for the operation of a Taco John's Restaurant within the "Protected Territory." Your Protected Territory is defined in the Addendum. You agree that a delivery of goods, whether by a third-party delivery service or otherwise, to a customer located within your Protected Territory from a Taco John's Restaurant located outside of your Protected Territory is not a violation of your Protected Territory rights. Similarly, you agree that your Protected Territory rights do not prevent a Taco John's Restaurant located outside of your Protected Territory from providing catering services within your Protected Territory.

(b) <u>Right of First Refusal</u>. In the event:

- (i) we or a prospective franchisee desires to locate, open and operate a Taco John's Restaurant, other than a Restaurant in a non-traditional location described below, within two miles outside the boundary of your Protected Territory ("Right of First Refusal Area"); or
- (ii) we or a prospective franchisee desires to locate, open and operate an Affiliated Business (as defined below), other than in a non-traditional location described below, within your Protected Territory, then
- 1. we shall first notify you and permit you the first option to locate, acquire and operate a Taco John's Restaurant in the Right of First Refusal Area under the terms and conditions of our then-current Franchise Agreement for a Taco John's Restaurant (where Section 2.3(b)(i) applies) or in your Protected Territory for an Affiliated Business (where Section 2.3(b)(ii) applies). Such offer shall expire 30 days after notice of such offer is given to you. You must sign our then-current Franchise Agreement and pay the then-current initial franchise fee in full (which fee is non-refundable) on or before the expiration of the 30 day period. Otherwise, we will be entitled to locate and operate or grant another person the right to locate and operate a Taco John's Restaurant or Affiliated Business within the Right of First Refusal Area or Protected Territory, as applicable, and you will have no right to prevent or prohibit us from operating or granting a franchise for a Taco John's Restaurant or an Affiliated Business in such area in the future.
- 2. If you do sign our then-current Franchise Agreement and pay the then-current franchise fee in full, you will have 120 days to bring an acceptable site to us within the Right of First Refusal Area or Protected Territory, as is applicable, in accordance with our then-current requirements. If you fail to meet this requirement, we will be entitled to operate or grant a franchise for the operation of a Taco John's Restaurant or an Affiliated Business in such area, as applicable, and you will have no right to prevent or prohibit us from operating or granting a franchise for a Taco John's Restaurant or an Affiliated Business in such area in the future.
- 3. Our obligation to make the offers described in this Section 2.3(b) is subject to the condition that you are in substantial compliance with all of the provisions of this Agreement. In the event that you and another Taco John's franchisee have overlapping Right of First Refusal Areas, then you and the other franchisee will each be given consecutive periods of 30 days in which to exercise your respective rights of first option. The franchisee whose Taco John's Restaurant began operating first shall have the first 30 day period. If such franchisee does not exercise its rights by signing a franchise agreement by the end of such 30 day period, then the second franchisee shall have the next 30 day period in which to exercise its right of first refusal. If neither you nor the other franchisee signs a franchise agreement in accordance with this Section 2.3(b) during the applicable time period, we will be entitled to proceed to operate or grant a franchise to a third party in accordance with this Section.
- (c) <u>Non-traditional Locations</u>. We reserve the right to operate or grant licenses to others to operate, without providing any rights or compensation to you, Taco John's Restaurants in non-traditional locations that may be within your Protected Territory.
- (d) <u>Alternative Channels</u>. We also reserve the right for us and our affiliates to market, offer, and sell, and to authorize third parties to market, offer, and sell, any and all products and services through alternative channels of distribution (other than Taco John's Restaurants), at any location (including within your Protected Territory), and on any terms and conditions as we determine including, but not limited to, sales through channels such as wholesale or retail distribution, Internet marketing and distribution, social media marketing, mail order, catalog, retail store display, or made-to-order or pre-packaged product sales

in grocery stores, at any location. The products and services available through these alternative channels may include those that are the same as or similar to those which you will offer and sell, such as any menu items offered at your Restaurant, or entirely different services and products. The prices advertised and charged by us, our affiliates, and our approved suppliers for the sale of the products and services to operators of alternative channels, and the prices advertised and charged by the operators of the alternative channels to third parties and the public for the products and services, may be higher or lower than the prices at which the same or similar products and services are made available by us, our affiliates, and our approved suppliers to you and the prices charged by you to customers of your Restaurant.

- (e) <u>Delivery Services</u>. You must offer delivery services from your Restaurant to customers located in your Protected Territory, if third party delivery services are available in your Protected Territory. You may also offer delivery services from your Restaurant to customers located outside of your Protected Territory, even if the location of the customer is in the protected territory of another franchisee. Similarly, another franchisee may offer delivery services from its Restaurant to customers located in your Protected Territory.
- (f) <u>Catering Services</u>. You must offer catering services through online catering platforms that we designate or approve. You must comply with any terms imposed by the catering platform vendor from time to time. You may conduct the catering services outside of your Protected Territory, even if the location is within the protected territory of another franchisee. Similarly, another franchisee may conduct catering services at locations within your Protected Territory.
- (g) <u>Reservation of Rights</u>. Except as provided in Sections 2.3(a), 2.3(b) and 2.3(c) above, we retain all rights with respect to Taco John's Restaurants, the Marks and the System.
- (h) <u>Affiliated Business</u>. An "Affiliated Business" as referred to above shall be defined as a Mexican food restaurant or other Mexican food distribution outlet (other than a Taco John's Restaurant as defined below) which is operated, franchised, or licensed by us or an Affiliate (as defined below) under the Marks or other marks.
- (i) <u>Taco John's Restaurant</u>. A "Taco John's Restaurant" offers our "Core Menu" as defined by current policy and has facilities and equipment necessary to prepare the Core Menu under the Marks. Excluding for non-traditional locations, it is not associated with a business, the purpose of which is other than the operation of a Taco John's Restaurant.
- (j) Other Names and Trademarks. Except as described above with respect to the right of first refusal for Affiliated Businesses, we have the right to use and license the use of alternative proprietary marks or methods in connection with the operation of other businesses under trademarks and names which are not the same as or confusingly similar to the Marks, which businesses may be the same as, or similar to, or different from Taco John's Restaurants.

2.4 Renewal of Franchise.

- (a) <u>Franchisee's Right to Renew</u>. Subject to Sections 2.4 (b) below and provided you are in substantial compliance with all provisions of this Agreement, then upon expiration of the Initial Term, if:
- (i) you maintain possession of and agree to remodel, renovate, refresh, reimage, refurbish, and modernize the Premises and the Restaurant, including the building design, parking lot, landscaping, equipment, signs, interior and exterior décor items, fixtures, furnishings, trade dress, and color scheme, to meet our then-current standards, specifications, and design criteria required of new franchisees, and including, without limitation, such structural changes, remodeling, and redecoration, and

such modifications to existing improvements as may be necessary to do so, all within the time frame we require; or

(ii) you are unable to maintain possession of the Premises and you secure substitute premises which comply with the specifications and standards then applicable under new or renewal franchises for Taco John's Restaurants and which we have accepted;

then you will have the right to renew the Franchise for up to three additional terms of 10 years each. You shall pay us a renewal fee in the amount set forth in Section 9.9 each time you renew the Franchise.

(b) Notices and Other Requirements.

- (i) At least 18 months prior to the expiration of the Franchise, we will notify you of the date that your Franchise is scheduled to expire. You must give us written notice of your intent to renew the franchise not less than 12 months prior to its expiration.
- (ii) If you have provided us notice of your intent to renew your Franchise, then at least six (6) months prior to the expiration of the Franchise, we will give you written notice of any deficiencies in your operation or the historical performance of the Restaurant which could cause us not to renew the Franchise.
- (iii) If we fail to provide the six months' notice described above within the required time period, we shall have the right, by written notice to you, to extend the term of this Agreement for such time period as may be necessary to provide the required notice. Notwithstanding an extension, any renewal term granted pursuant to this Section shall expire on the appropriate anniversary date as though the notice was given when required.
- (iv) If you are entitled to renewal under Section 2.4(a) above, our notice will state what actions, if any, you must take to correct the deficiencies in the Premises and the operations and the historical performance of your restaurant, and will specify the time period in which those deficiencies must be corrected or other requirements satisfied.
- (v) You will not be entitled to renewal of the Franchise unless you comply with all the terms and conditions of this Agreement from the date you give us notice of your desire to renew the Franchise up to the date of expiration. If we send a notice that the Franchise will not be renewed, it will state the reasons for our refusal to renew. We will not refuse to renew your Franchise except for good cause. "Good Cause" shall mean cause based upon a legitimate business reason including, but not limited to, your failure to comply with any material lawful requirement of this Agreement (provided that our refusal to renew your Franchise is not arbitrary or capricious when compared to our actions with respect to other franchisees in the same general market area under similar circumstances) or our complete withdrawal from directly or indirectly distributing our products or services in the geographic area served by your Franchise. As a condition of renewal, we may require you (or another owner) or your manager to attend our thencurrent training program and meet our standard certification requirements. We will not charge a fee for training required upon renewal as provided in this Section. However, you agree to pay all expenses incurred by any owners, manager(s) and/or employees of the Restaurant in connection with such training.
- (vi) To renew the Franchise, the Franchiser, the Franchisee and its owners shall execute the form of standard Franchise Agreement and any ancillary agreements we are then offering in the grant of franchises to prospective new franchisees for the operation of a Taco John's Restaurant (with appropriate modifications to reflect the fact that the agreement relates to the grant of a renewal franchise). Those agreements may contain royalty, advertising and other fees that differ from those contained in this Agreement, except that no Initial Franchise Fee will be payable upon renewal of the Franchise.

3. DEVELOPMENT AND OPENING OF RESTAURANT

3.1 Site Location.

You agree that you will locate, within 90 days of the date of this Agreement, a site suitable for the operation of a Taco John's Restaurant ("Proposed Site"). You agree to work with a real estate broker approved by us to locate a site. We will not receive any payments from approved real estate brokers, and we will not have any liability to you for their acts or omissions. Upon location of a site which you believe in good faith to be suitable for development as a Taco John's Restaurant, you agree to submit to us full and complete site documentation in our required form, including a copy of the proposed purchase agreement or lease, and any other information about the site and its development that we may reasonably request. We will use reasonable efforts to evaluate and accept or reject a Proposed Site and we will not unreasonably withhold acceptance of sites that meet our minimum standards for market type, trade area, general location, physical and demographic characteristics of the market or sub-market, traffic pattern, parking, restaurant competition, proximity of other businesses (including other Taco John's Restaurants), the nature of other businesses in proximity to the site, layout and other physical characteristics, development costs, rental, lease duration and other lease terms and conditions for the Taco John's Restaurant.

If you request our assistance with site evaluation and or site acquisition, you must reimburse us for our actual expenses incurred in this process at our then-current rates.

3.2 Site Acceptance.

You agree that you will not execute a purchase agreement for a Proposed Site until you have provided us with a copy of the purchase agreement and we have accepted the site and purchase agreement in writing. You may not execute a lease for a Proposed Site until you have provided us a copy of the lease and we have accepted the site and lease in writing. We have 30 days after receipt of your full and complete site documentation and purchase agreement or lease to review and accept it. We may provide you with suggestions and recommendations concerning the terms and conditions of the lease or purchase agreement. We have the right to require that certain provisions are included in the lease relating to duration, use, default, notices, assignment, remodeling, personal property rights, the right for our employees to enter the Premises, our right to protect our Marks, and the right of first refusal. Some provisions may be contained in a Lease Rider, in the form designated by us. You further acknowledge that our involvement in the review and approval of a purchase agreement or lease for the Proposed Site and our acceptance of a site will not constitute a recommendation, endorsement, representation, guarantee, or warranty, express or implied, as to the suitability or profitability of the Proposed Site or of the purchase agreement or lease for a Taco John's Restaurant or for any other purpose. An "Accepted Site" is a Proposed Site for which we have reviewed your full and complete site documentation, a copy of the proposed purchase agreement or lease, and any other required information, and which we have accepted in writing. If you lease the land or building for your Restaurant, the initial lease term plus any renewal options must equal or exceed the entire Initial Term of this Agreement.

3.3 Site Acquisition.

Upon our written acceptance of a Proposed Site for your Restaurant, you agree to attempt to obtain lawful possession of the Accepted Site through purchase or lease. While our review and assistance may include recommendations to protect you under the terms of the lease or purchase agreement, you acknowledge that our involvement in the negotiation or approval of the documents for the Proposed Site is primarily for our benefit only. We make no representations or warranties as to the legal validity of any recommended provisions. It ultimately is your responsibility to negotiate and accept the final terms of these

documents. You should take all steps necessary to ascertain whether such location and purchase agreement or lease are acceptable to you, which may include engaging your own legal counsel to review the documents.

3.4 Construction Documents and Specifications.

You agree to use only qualified building design professionals, including architects and engineers, that are approved by us. You further agree to complete the construction and build-out of your Restaurant using only a qualified general contractor approved by us. Upon reasonable request, we will consider the approval of other professionals, but we are under no obligation to approve any particular professional or any minimum number of them. All professionals are required to be licensed if required by the state in which the Restaurant is located. You must deal directly with any such architects, engineers, or general contractors. We will not receive any payments from approved architects, engineers, or general contractors, and we will not have any liability to you for their acts or omissions.

We will furnish one copy of conceptual site plans, construction documents and specifications reflecting our requirements for design, decoration, furnishings, furniture, layout, equipment, fixtures and signs for a Taco John's Restaurant at no cost. You are responsible for paying our then-current fees for any additional layouts or revisions you request from us. We require that the construction documents be reviewed, modified and sealed by local architects and engineers approved by us for use in the state where your Restaurant will be located. You must use only qualified general contractors approved by us. You must submit complete construction documents and specifications to us for our review and approval before you begin construction at the Premises. Such review is not designed to assess compliance with federal, state or local laws and regulations, including the Americans with Disabilities Act, as compliance with such laws is your sole responsibility. It will be your sole responsibility to have prepared all required architectural designs, mechanical drawings, site plans, blueprints and construction plans to suit the dimensions of the Premises and to ensure compliance with all applicable federal, state and local laws, regulations and codes including the Americans with Disabilities Act, building codes, permit requirements, and any lease or purchase agreement. You acknowledge that our assistance related to providing the conceptual site plans, construction documents, and specifications and our review and approval of any construction document related to the Premises, do not constitute a recommendation, endorsement or guarantee by us of the suitability or profitability of the location, layout, or design of your Premises. Upon our request, you must submit all revised documents and specifications during the course of such construction. If you do not begin construction within six months after our approval, you must resubmit the drawings for approval to ensure that they still meet our current standards and specifications. You must construct the Restaurant in accordance with those documents and specifications. We will hold regularly-scheduled Stage Gate New Restaurant Opening meetings. We will conduct at least one inspection of the Restaurant during construction and may conduct other inspections as we deem appropriate for purposes of confirming compliance with our standards and specifications.

If you build a DT Digital Olé model Restaurant, we require you to use only the modular building manufacturers approved by us. We will not receive any payments from approved modular building manufacturers and we will not have any liability to you for their acts or omissions.

3.5 Equipment, Fixtures, Furnishings and Signs.

You agree to use in the development and operation of the Restaurant only those brands, types, and/or models of equipment, computer hardware and software, furniture, fixtures, furnishings, digital menu boards, and signs which meet the standards and specifications promulgated by us as to their nature and quality. You agree to purchase exterior digital menu boards, and you have the option to purchase interior digital menu boards, from our approved supplier. You agree to purchase equipment, furniture, fixtures,

furnishings, and signs and license software which meet our standards and specifications only from distributors and other suppliers we have approved. If you would like to acquire any items from any unapproved supplier, you must submit to us a written request for approval of the proposed supplier. We will follow the same approval process set forth in Section 10.5(f) below.

3.6 Restaurant Opening.

Unless different deadlines are set forth in the Addendum, you agree, at your own expense, to do the following within the earlier of: (a) 210 days after receiving your building permits for the Accepted Site, or (b) 10 days after obtaining the Certificate of Occupancy for the Restaurant, and in any event within 12 months after the signing of the Franchise Agreement, or any earlier date as may be provided in your lease or sublease: (a) secure all financing required to fully develop the Restaurant; (b) construct, decorate, equip and furnish the Restaurant according to the construction documents and specifications we have accepted; (c) cause the training requirements of Section 4.1 to be completed; (d) purchase an opening inventory of products, supplies and materials, including all marketing materials; (e) obtain all necessary permits, licenses and insurance; (f) do any other acts necessary to open the Restaurant for business; (g) obtain our approval to open the Restaurant for business; and (h) open the Restaurant for business. Furthermore, if you do not already operate a Taco John's Restaurant, you will not open the Restaurant for business unless a representative of the Franchisor is present to provide pre-opening and post-opening assistance (for the period applicable under Article 4 below). We agree to provide such assistance, provided you notify us in writing no less than 30 days prior to the scheduled opening of the Restaurant and comply with all preopening requirements set forth above. If your scheduled opening date is delayed for any reason, you may be required to pay a delayed opening fee as set forth in Section 9.12.

4. TRAINING

4.1 Initial Training and Certification.

You acknowledge that it is very important to the operation of the Restaurant that you and your employees receive appropriate training. For that reason, you agree as follows:

Before the Restaurant opens for business, we may, at our discretion, require a franchise owner to complete up to one week in a franchise ownership, business management, and operations training program (the "Owner Training Program" or "OTP"). If we require, the OTP will occur prior to the Initial Training, described below. You must pay all expenses incurred by any owners of the Restaurant in connection with attending the OTP. However, we do not charge a fee for one owner to attend the OTP. This program will be conducted in a Restaurant location or in our Restaurant Support Center at the time and location we designate. In addition, one owner, or, if the owner is not an individual, the person designated by the owner to assume primary operational responsibility for the Restaurant ("Designated Principal"), and a minimum of three managers must attend our Management Certification Program to receive training in the operation of a Taco John's Restaurant at the time we designate (the "Initial Training"). If you operate multiple Taco John's Restaurants, we may require you to have members of your management team who oversee multiple Restaurant locations (each an "Above Store Leader"), attend additional training. If you are a new franchisee or you open a new restaurant in an underpenetrated market, we may require you and the managers we designate to attend additional training for up to three weeks. If the owner or Designated Principal will not have primary operating responsibilities, then a minimum of four managers must attend the Initial Training. We will not charge a fee for the Initial Training for up to four individuals. However, you agree to pay all expenses incurred by any owners, Above Store Leaders, and manager(s) of the Restaurant in connection with the Initial Training. The Initial Training will include classroom instruction and restaurant operation training and will be furnished at a Taco John's Restaurant(s) location or in our

Restaurant Support Center at the time we designate. Training programs may differ for owners or employees depending upon their responsibilities at the Restaurant.

- (b) The owner, or if applicable, the Designated Principal, and the manager(s) of the Restaurant must complete all training requirements listed in subsection (a) above and must meet our standard certification requirements under the Initial Training. (A person who has met our standard certification requirements is referred to in this Agreement as being "certified.") If the owner or the manager(s) fail to complete the training program or are not certified, we will allow another owner or, manager, as applicable, to attend training, provided that he or they must complete our training program and be certified. If the latter owner or manager fails to complete the training program or to be certified, we will have the option, upon 30 days' written notice to you, to terminate this Agreement. We reserve the right to suspend or terminate the training program for any training candidate who fails to adhere to the established standards and guidelines. This includes, but is not limited to, non-compliance with operational procedures, failure to meet performance benchmarks, or any conduct deemed detrimental to the franchise system's reputation and integrity. The decision to suspend or terminate training will be at our sole discretion and is intended to ensure the highest quality and consistency across all franchised locations.
- (c) We will provide, at our expense, qualified person(s) who will, for a minimum of three days assist you and/or your manager with the pre-opening of the Restaurant and, for up to 10 days (which may or may not be consecutive days), assist you and/or your manager with the post-opening of the Restaurant (or, if you already operate another Restaurant, for up to eight days assist you and/or your manager with the opening and initial operation of the Restaurant), provided you have notified us in writing no less than 30 days prior to the scheduled opening of the Restaurant.
- (d) All owners, Designated Principals, Above Store Leaders, managers, or other employees who attend the Initial Training or later seek to become Active Certified Managers, as defined below, must participate in a food safety program and obtain a food safety certification from a third party licensed to teach food safety that we approve. You are solely responsible for all fees charged by that third-party provider for the training, as well as all expenses incurred by any of your owners and employees while participating in the program.

4.2 Active Certified Managers.

You agree to maintain the number of certified managers we specify from time to time in the Operations Manual (defined in Section 5.2 below), who have successfully completed our standard certification requirements and who work regularly in the Restaurant (each an "Active Certified Manager"), one of which may be you, your Designated Principal, or an Above Store Leader, for the term of this Agreement. You agree to pay our then-current fees for the required certification courses for each manager after the initial training program. We reserve the right to require additional levels of managers certified by us depending on the size and performance of the Restaurant.

4.3 Ongoing and Additional Training Programs.

(a) We provide you with access to our online learning management system, currently called The Olé! Way, which provides certain ongoing training programs and materials that we elect to provide. You agree to have your management and/or other employees complete additional training programs as we may require during the term of this Agreement, including training we generally require of all franchisees and training we require only for you or a limited number of franchisees based on our determination that you have failed to meet any of our standards. You may be required to enter into an agreement with the provider of that system and abide by all terms of the agreement, including paying the provider all required fees.

- (b) We may also conduct ongoing on-site training for some or all franchisees at their Restaurant locations on a rotating basis or as we otherwise deem appropriate. We will not charge a fee for any mandatory additional training. Your managers will also be permitted to attend other training programs, workshops, or seminars which we may offer from time to time. We may charge a fee equal to actual expenses incurred for any non-mandatory additional training.
- (c) Each ongoing optional and mandatory program may be offered via The Olé! Way online learning management system, at an in-person session, or in any other manner we designate. We may require attendance by your manager(s) at a manager convention. You must pay all expenses and salaries incurred by any owners and/or manager(s) of the Restaurant in connection with their attendance at all training programs.

4.4 Purpose of Training.

All training programs are provided to protect our brand and the Marks and not to control the day-to-day operations of your Taco John's Restaurant. We continuously evolve our training programs to provide the most thorough training to our franchisees and reserve the right to modify the training program requirements from time to time.

5. GUIDANCE; OPERATIONS MANUAL

5.1 Guidance and Assistance.

During the term of this Agreement, we will from time to time furnish you guidance and assistance with respect to the operation and management of a Taco John's Restaurant. This guidance and assistance will, in our discretion, be furnished in the form of the Taco John's Operations Manual (defined below), bulletins, written reports, computer disks, other electronically stored data, telephone consultations, and/or personal consultations at our offices or at the Restaurant. If you request additional, special on-premises training of your personnel or other assistance in operating the Restaurant, you must pay for all expenses for that training or assistance, including any per diem charges assessed by us and the documented and reasonable travel and living expenses of our personnel. Any information we provide to you does not relieve you of your responsibility to consult with your own legal advisor. You are solely responsible for complying with any legal requirements affecting you or your Taco John's Restaurant.

5.2 Operations Manual.

- (a) Operations Manual. We will make available to you during the term of the Franchise one or more manuals, technical bulletins or other written materials containing mandatory and suggested specifications, standards and operating procedures and rules prescribed from time to time by us and information relative to your other obligations hereunder and the operation of the Restaurant (the "Operations Manual"). The Operations Manual may consist of multiple volumes of printed text, computer disks, or other electronically stored data. We may provide a portion or all of the Operations Manuals (including updates and amendments), and other instructional information and materials in, or via, electronic media, including without limitation, through the online learning management referenced in Section 4.3 above or otherwise through the Internet. The Operations Manual is designed to protect our reputation and the goodwill of the Marks; it is not designed to control the day-to-day operations of your Taco John's Restaurant.
- (b) <u>Confidential Information</u>. You shall, at all times, treat the Operations Manual, and any other materials created for or approved for use in the operation of the Restaurant, as confidential, and

shall use all reasonable efforts to maintain such information as secret and confidential. You may not at any time, copy, duplicate, record or otherwise reproduce the foregoing materials, in whole or in part, nor otherwise make the same available to any unauthorized person. The Operations Manual shall at all times remain our sole property.

- (c) <u>Modifications to Operations Manual</u>. We shall have the right to add to and otherwise modify the Operations Manual from time to time, as we deem appropriate in our discretion, to improve the standards of service or product quality or the efficient operation of Taco John's Restaurants, to protect or maintain the goodwill associated with the Marks, to meet competition, or to otherwise improve or benefit the System.
- (d) Operations Manual Status. The Operations Manual, as modified from time to time, and mandatory specifications, standards and operating procedures and rules we prescribe from time to time and communicate to you in writing, shall constitute provisions of this Agreement as if fully set forth herein. All references herein to this Agreement shall include the provisions of the Operations Manual and all such mandatory specifications, standards and operating procedures and rules. The master copy of the Operations Manual maintained at our principal office will be controlling in the event of a dispute relative to the content of any Operations Manual.

6. MARKS

6.1 Ownership and Goodwill of Marks.

You acknowledge that your right to use the Marks is derived solely from this Agreement and is limited to your operation of the Restaurant in compliance with this Agreement and all standards, specifications, and operating procedures we prescribe from time to time during the term of the Franchise. If you make any unauthorized use of the Marks, it will constitute a breach of this Agreement and an infringement of our and our Affiliate's rights in and to the Marks. You acknowledge and agree that all your usage of the Marks and any goodwill established by your use will inure to our and our Affiliate's benefit exclusively, and that this Agreement does not confer any goodwill or other interests in the Marks on you, other than the right to operate a Taco John's Restaurant in compliance with this Agreement. All provisions of this Agreement applicable to the Marks will apply to any additional trademarks, service marks, commercial symbols, designs, artworks, and logos we may authorize you to use during the term of this Agreement.

6.2 Limitations on Franchisee's Use of Marks.

You agree to use the Marks as the sole trade identification of the Restaurant, except that you will display at the Restaurant a notice, in the form we prescribe, which states that you are the independent owner of the Restaurant pursuant to a Franchise Agreement with us. You agree not to use any Mark as part of any corporate or trade name or with any prefix, suffix, or other modifying words, terms, designs, or symbols (other than logos and additional trade and service marks licensed to you under this Agreement), or in any modified form, nor may you use any Mark or any name or symbol similar to the Marks, in connection with the performance or sale of any unauthorized services or products or in any other manner we have not expressly authorized in writing. You agree to display the Marks in the manner we prescribe at the Restaurant, on your computer system, on menus and in connection with advertising and marketing materials, and to use, along with the Marks, notices of trade and service mark registrations as we specify.

We will maintain, and provide to you upon your written request, a list of approved advertising and promotional materials which may be used by Taco John's franchisees. You agree to submit to us for our approval samples of all local advertising and promotional materials, including computer advertising, which

are not on the list of approved materials prior to using them. We will provide approval or disapproval within five business days from the date we receive the materials. You agree not to use any advertising, promotional materials or web sites that we have disapproved. You further agree to obtain any fictitious or assumed name registrations as may be required under applicable law.

6.3 Notification of Infringements and Claims.

You agree to immediately notify us in writing of any apparent infringement of or challenge to your use of any Mark, or claim by any person of any rights in any Mark or similar trade name, trademark or service mark of which you become aware. You agree not to communicate with anyone except us, our Affiliate and our counsel in connection with any such infringement, challenge or claim. We and our Affiliate will have the sole right to exclusively control any litigation or other proceeding arising out of any infringement, challenge or claim relating to any Mark. You agree to sign any documents, render any assistance, and do any acts that our and/or our Affiliate's attorneys say is necessary or advisable in order to protect and maintain the Marks for the benefit of Taco John's Restaurants.

6.4 Discontinuance of Use of Marks.

If it becomes advisable at any time in our sole judgment for the Restaurant to modify or discontinue the use of any Mark or for the Restaurant to use one or more additional or substitute trade or service marks, you agree at your expense to comply with our directions to modify or otherwise discontinue the use of the Mark, or use one or more additional or substitute trade or service marks, within a reasonable time after our notice to you.

6.5 Indemnification of Franchisee.

We agree to indemnify you against and to reimburse you for all damages for which you are held liable in any proceeding arising out of your use of any Mark, and for all costs you reasonably incur in the defense of any such claim in which you are named as a party, provided (i) you have operated fully and completely pursuant to and in compliance with this Agreement, and (ii) you have timely notified us of the claim and afforded us the opportunity to participate in the action, including, in our sole discretion, representing you in such action.

7. RELATIONSHIP OF PARTIES; INDEMNIFICATION

7.1 Independent Contractor; No Fiduciary Relationship.

Both of us understand and agree that this Agreement does not create a fiduciary relationship between us, that you and we are independent contractors, and that nothing in this Agreement is intended to make either party a general or special agent, joint venturer, partner, or employee of the other for any purpose whatsoever. You agree to conspicuously identify yourself in all your dealings with customers, suppliers, public officials, Restaurant personnel, and others as the owner of the Restaurant pursuant to a Franchise Agreement with us and agree to place any other notices of independent ownership on your forms, business cards, stationery, e-mails, advertising, and other materials as we may require from time to time.

7.2 No Liability, No Warranties.

You agree not to employ any of the Marks in signing any contract, check, purchase agreement, negotiable instrument or legal obligation, application for any license or permit, or in a manner that may result in liability to us for any indebtedness or obligation of yours. Except as expressly authorized by this

Agreement, neither of us will make any express nor implied agreements, warranties, guarantees or representations, or incur any debt, in the name of or on behalf of the other or represent that their relationship is other than that of franchisor and franchisee.

7.3 Indemnification.

We will not assume any liability or be deemed liable for any agreements, representations, or warranties you make that are not expressly authorized under this Agreement, nor will we be obligated for any damages to any person or property directly or indirectly arising out of the operation of the business you conduct pursuant to this Agreement, whether or not caused by your negligent or willful action or failure to act. We will have no liability for any taxes levied against you or your assets or on us in connection with the business you conduct, or any payments you make to us pursuant to this Agreement or any Franchise Agreement (except for our own income taxes). You agree to indemnify us, defend and hold us, our Affiliates, shareholders, directors, officers, employees, agents and their respective assignees, harmless against and to reimburse us for all such obligations, damages, and taxes for which we are held liable and for all costs we reasonably incur in the defense of any such claim brought against us or in any such action in which we are named as a party, including, without limitation, actual and consequential damages, reasonable attorneys', accountants' and expert witness fees, cost of investigation and proof of facts, court costs, other litigation expenses and travel and living expenses. However, your obligation to indemnify, defend and hold us harmless will not apply to any damages, costs or expenses which result from any actions taken by you in strict compliance with the then-current Operations Manual or any other then-current specification, standard, procedure, guidance, assistance or instruction we have given you or which result from our or our agents' negligence or willful or reckless misconduct. Your indemnification obligations described above will continue in full force and effect after the expiration or termination of this Agreement. We have the right to defend any such claim against us.

8. CONFIDENTIAL INFORMATION; NON-COMPETITION

8.1 Types of Confidential Information.

We and our Affiliates possess certain unique confidential and proprietary information and trade secrets consisting of the following information developed by us and our franchisees:

- (a) Formulas, recipes, menu analyses, and methods of preparing, packaging, and serving products sold at Taco John's Restaurants;
- (b) Knowledge of sales and profit performance and standard performance evaluations of individual Taco John's Restaurants and Taco John's Restaurants as a whole;
- (c) Knowledge of test programs, concepts or results relating to new menu items or categories; brands; sources of products; advertising and promotional programs; restaurant image and decor; and methods for the selection and training of Restaurant personnel; forms and employee information; and
- (d) Methods, techniques, formats, specifications, procedures, systems and knowledge of and experience in the development, operation, and franchising of Taco John's Restaurants.

We will disclose much of the above-described information to you in training, in the Operations Manual and in providing guidance and assistance to you under this Agreement. In addition, in the course of the operation of the Restaurant, you or your employees may develop ideas, concepts, methods, techniques of improvement, relating to the Restaurant which you agree to disclose to us and which we may then authorize you and others in writing to use in the operation of Taco John's Restaurants. (Any such

information disclosed to or developed by you will be referred to in this Agreement as "Confidential Information.")

8.2 Non-disclosure Agreement.

You agree that your relationship with us does not vest in you any interest in the Confidential Information other than the right to use it in the development and operation of the Taco John's Restaurant under this Agreement with us, and that the use or duplication of the Confidential Information in any other business would constitute an unfair method of competition. You acknowledge and agree that the Confidential Information belongs to us and our Affiliates, may contain trade secrets belonging to us and our Affiliates, and is disclosed to you or authorized for your use solely on the condition that you agree, and you therefore do agree, that you:

- (a) Will not use the Confidential Information in any other business or capacity;
- (b) Will maintain the absolute confidentiality of the Confidential Information during and after the term of this Agreement;
- (c) Will not make unauthorized copies of any portion of the Confidential Information; and
- (d) Will use all reasonable procedures we may prescribe from time to time to prevent unauthorized use or disclosure of the Confidential Information which may include, without limitation, requiring each employee and owner who has access to the Confidential Information to sign a confidentiality agreement in a form approved by us.

You agree that you will cause each owner, officer, or director to sign a separate non-disclosure agreement upon our request, and failure to do so may constitute a default of this Agreement.

8.3 Non-competition Agreement.

You agree that we would be unable to protect the Confidential Information against unauthorized use or disclosure and would be unable to encourage a free exchange of ideas and information among owners of Taco John's Restaurants if franchisees of Taco John's Restaurants were permitted to hold interests in any Competitive Businesses, as defined below. Therefore, during the term of this Agreement, neither the Franchisee, nor any owner, nor the spouse or immediate family member of the Franchisee or any owner shall perform services for or have any direct or indirect interest as a disclosed or beneficial owner, investor, partner, director, officer, employee, manager, consultant, representative or agent in any Competitive Business. For the purposes of this Agreement, a "Competitive Business" is any restaurant (other than a Taco John's Restaurant operated under a Franchise Agreement with us) that offers or sells Mexican food as more than 15% of its menu items. The ownership of 1% or less of a publicly traded company will not be deemed to be prohibited by this Section.

You agree that you will cause each owner, officer, or director to sign a separate non-compete agreement upon our request, and failure to do so may constitute a default of this Agreement.

9. FEES

All fees paid by you to us pursuant to this Agreement are non-refundable.

9.1 Initial Franchise Fee.

You agree to pay us an Initial Franchise Fee (the "Initial Franchise Fee") in the amount set forth in the Addendum. The Initial Franchise Fee is payable in full upon execution of this Agreement. You acknowledge and agree that the Initial Franchise Fee represents payment for the initial grant of the rights to use the Marks and System, and that we have earned the Initial Franchise Fee upon receipt thereof.

9.2 Royalty.

In consideration for your right to use the Marks and the System as provided in this Agreement, you agree to pay us a royalty (the "Royalty") equal to 5% of the Net Sales (defined below) of the Restaurant. The Royalty shall be payable monthly not later than the 20th day of the month on the Net Sales of the Restaurant for the preceding month and shall be accompanied by a report of the Net Sales for the preceding month in a form approved by us. We may change the frequency of the payment and/or the due date(s) of the payment of the Royalty on 60 days' prior advance written notice to you. The term "Net Sales" shall, for purposes of this Agreement, mean the total receipts from all sales of the Restaurant, including all receipts from sales made at or away from the premises of the Restaurant and revenue from the redemption of Taco John's gift certificates, customer loyalty/rewards program, gift cards and other prepaid cards, sales through mobile ordering, delivery, and catering, and all other income of every kind and nature related to, derived from, or originating from the Restaurant, including proceeds of any business interruption insurance policies, exclusive of sales and equivalent taxes. We reserve the right to modify our policies and practices regarding inclusion or exclusion of certain revenue from "Net Sales" as circumstances, business practices, and technology change.

9.3 Advertising and Marketing Fee.

You agree to pay to us for deposit in an Advertising Fund an "Advertising and Marketing Fee" currently equal to 4% of the Net Sales of the Restaurant. We will not change the Advertising and Marketing Fee percentage before January 1, 2026. Thereafter, we reserve the right to modify the Advertising and Marketing Fee on 60 days' notice to an amount equal to the amount we charge to new franchisees, but in no event shall the Advertising and Marketing Fee exceed 4.25% of the Net Sales of the Restaurant during the Initial Term of this Agreement. The Advertising and Marketing Fee is payable in the same manner as the Royalty. The Advertising and Marketing Fee will be used in the manner described in Article 11.

You understand and acknowledge that Advertising and Marketing Fees expended by us or our designees will be used by us, as we determine in our sole discretion, to foster recognition and promotion of the Marks, the brand, and retail sales of Taco John's Restaurants generally, and to pay for the development, support, and dissemination of marketing programs. Although we will endeavor to use the Advertising and Marketing Fees to develop and place advertising in a manner that will benefit all Taco John's Restaurants, we cannot ensure that expenditures of Advertising and Marketing Fees in or affecting any geographic area are proportionate or equivalent to Advertising and Marketing Fees paid by Taco John's Restaurants operating in that geographic area or that any Taco John's franchisee will benefit from the placement of advertising directly or in proportion to the amounts it has paid as Advertising and Marketing Fees.

Taco John's Restaurants owned by us pay Advertising and Marketing Fees on the same basis as franchised Taco John's Restaurants.

9.4 Technology Fee.

We currently do not charge a technology fee ("Technology Fee"), but we may establish this fee to pay for technology related to implementation, subscriptions or license fees, and to assist with the

development and support of technology. We will give you not less than 60 days' notice of the requirement that you pay this fee. If we create a Technology Fee, it will replace some or all of the following as we designated from time to time: the license fee for your digital menu boards, The Olé! Way online learning management system license fee, financial reporting system fee, back of house information system fee, geofencing service fee, gift card fee, loyalty/rewards program fee, guest experience program fees, and other technology-related subscription or license costs, and can also be used for further development and support of technology by or for us, including changes or enhancements to ordering and AI platforms, the mobile app, data reporting platforms, custom development, or other technology innovations. The Technology License Fee will be in the amount charged by the suppliers for the underlying products and services, plus a charge of up to 10% of those amounts to be retained by us and expended in development and support of technology by us. Such development and support may be conducted in-house and/or with third parties, and we may compensate ourselves from the Technology Fee for any personnel time or expenditures that we incur in providing these services. When we collect this fee, we will submit a portion of it to the third-party vendors who provide the services covered by this fee on your behalf. The portion retained by us will be used for technology development and support. We may increase the Technology Fee at any time, provided that the increase will never be more than the increase in actual costs to provide the products and services plus 10% of the total cost to provide the products and services. The Technology Fee is payable in the same manner as the Royalty.

9.5 Grand Opening Marketing.

You agree upon execution of this Agreement to set aside the sum of money set forth in the Addendum to be spent for the grand opening of the Restaurant to be conducted pursuant to a grand opening marketing program approved by us. The grand opening marketing program is to be conducted within the first 6 to 12 weeks of operation of the Restaurant. We may require you to provide us with an accounting of your grand opening expenditures within 180 days after the date the Restaurant is opened.

The grand opening marketing program expenditure listed in the Addendum may be higher if you open a Restaurant in a market that is defined by us as underpenetrated. We will notify you that the Restaurant is in an underpenetrated market and help you develop a grand opening marketing program designed to introduce the market to the Taco John's concept. An underpenetrated market is one where there is limited awareness of the Taco John's concept and minimal Taco John's advertising penetration.

9.6 Interest on Late Payments; Nonsufficient Funds Fees.

All Royalty payments, Advertising and Marketing Fees, Learning System Fees, Technology Fees (if created), amounts due from you for purchases from us or our Affiliates, and other amounts which you owe us or our Affiliates (unless otherwise provided for in a separate agreement between us or our Affiliates) will begin to accrue interest after their respective due dates at a rate not to exceed 1.5% per month. If there are insufficient funds in your account to process payment of fees due to us, you agree to pay us the thencurrent nonsufficient funds fee (NSF fee), not to exceed \$100 per transaction, and shall reimburse us for all expenses we incur as a result of such nonsufficient funds. You acknowledge that the inclusion of this Section in this Agreement does not mean we agree to accept or condone late payments, nor does it indicate that we have any intention to extend credit to, or otherwise finance your operation of the Restaurant.

9.7 No Withholding.

You agree that under no circumstances will you withhold or suspend payment of, or reduce the amount of the Royalty, Advertising and Marketing Fees, or other fees payable under this Agreement. Notwithstanding the foregoing, in the event that you dispute in good faith the amount of an individual payment due under this Agreement, you may pay only the amount you believe is due, provided that you

give us prompt notice of the reasons for which you dispute the amount of the payment and proceed to make good faith efforts to resolve such dispute.

9.8 Transfer of Ownership Fee.

You agree to pay us a transfer fee of \$10,000 for each Transfer. Once the transferee is approved, the transfer fee is due whether or not a transfer actually occurs and is nonrefundable once paid.

9.9 Renewal Fee.

You agree to pay us a renewal fee equal to 25% of our then-current Initial Franchise Fee for Taco John's Restaurant franchises for each renewal.

9.10 Relocation Fee.

At our option, you will pay us a relocation fee of \$10,000, which is due at the time the relocation process commences with our acceptance of the new site. The relocation fee is due whether or not a relocation actually occurs and is nonrefundable once paid.

9.11 Non-compliance Fee.

- (a) If you are determined by us to be non-compliant with equipment requirements under Section 10.2(d), remodel, refresh, or other requirements under Section 10.2(e), financial reporting system or other technology requirements under Section 10.4(i), or fail to sign up for third-party delivery under Section 10.4(d) or catering under Section 10.4(l), upon our demand, you will pay us a non-compliance fee equal to 3% of Net Sales. This fee will be charged and calculated based on your Net Sales from the date that is fifteen (15) days after you are notified of the non-compliance issue, and must be paid weekly through the date when the non-compliance is cured, up to a maximum of six months for each act of non-compliance. During the 15 day notice period, you will have the opportunity to cure before we charge the non-compliance fee.
- (b) If you are determined by us to be non-compliant with our third-party delivery or catering requirements (other than the requirement to initially sign up for the services), or the required hours of operations in the Operations Manual, upon our demand, you will pay us a non-compliance fee equal to \$1,000 per day for any partial day of non-compliance, and \$3,000 per day for any full day of non-compliance. This fee will be payable immediately upon notification by us.

You acknowledge that the non-compliance fee is a reasonable estimate of our administrative and management costs and is not a penalty. Our right to seek the non-compliance fee and your payment of the non-compliance fee do not preclude us from seeking any other legal or equitable relief to which we are entitled under this Agreement.

9.12 Delayed Opening Fee.

If your scheduled opening date is delayed for any reason, then upon our demand, you will pay us a delayed opening fee equal to \$250 per team member for each of our team members who were scheduled to attend the opening, up to a maximum of \$2,000, to offset the costs we incur in preparing to attend your opening, including rescheduling our resources and modifying our travel arrangements. The fee is due within five days after notice to you.

9.13 Media Relations Support Fee.

If you are determined by us to need media relations support for a crisis as defined in Section 10.6(c) and fail to provide the required notice to us, you will pay us a "Media Relations Support Fee" of \$500. The fee is due within five days after notice to you.

9.14 Allocation of Payments.

Unless other written instructions accompany a specific payment, all payments made by you pursuant to this Agreement shall be applied in such order as we may designate from time to time.

9.15 Payment Authorization.

We require that any payments to be made by you to us pursuant to this Agreement be made by automatic bank draft in accordance with the Authorization Agreement for Direct Payments (ACH), attached as Exhibit II to this Agreement. We reserve the right to require payment of fees by other similar technology now or hereafter developed.

10. RESTAURANT OPERATING STANDARDS

10.1 Importance of Uniformity.

By signing this Agreement, you indicate that you understand and acknowledge that every detail of the appearance and operation of the Restaurant is important - not only to you, but to us and to other Taco John's Restaurant franchisees - in order to develop and maintain high and uniform operating standards, to increase the demand for the products and services sold by all franchisees, to establish and maintain a reputation for operating uniform, efficient, high quality restaurants, and to protect the goodwill of all Taco John's Restaurants. You also acknowledge that a fundamental requirement of the System, this Agreement, and other Taco John's Restaurant franchises is adherence by all franchisees to our uniform standards and policies, including but not limited to those set forth in the Operations Manual, except for certain regional or local variations we may from time to time approve or require. These standards and policies are designed to protect our reputation and the goodwill of the Marks and are not designed to control the day-to-day operations of your Taco John's Restaurant.

10.2 Condition and Appearance of Restaurant.

You agree that:

- (a) Neither the Restaurant nor the Premises will be used for any purpose other than the operation of a Taco John's Restaurant in compliance with this Agreement;
- (b) You will maintain the condition and appearance of the Restaurant, its equipment, furniture, fixtures, signs, and the Premises in accordance with our standards and consistent with the image of a Taco John's Restaurant as an efficiently operated business offering high quality products, beverages and services; observing the highest standards of cleanliness and sanitation; with efficient, courteous service and pleasant ambiance; and in that connection will take, without limitation, the following actions during the term of this Agreement: (i) thorough cleaning, repainting and redecorating of the interior and exterior of the Premises at reasonable intervals; (ii) interior and exterior repair of the Premises; (iii) repair or replacement of damaged, worn out or obsolete equipment, furniture, furnishings and signs; and (iv) engage a pest control service on an ongoing basis in accordance with our requirements;

- (c) You will not make any material alterations to the Premises, or to the appearance of the Restaurant, as originally developed, without our advance approval. If you do so, we have the right, at your expense, to rectify alterations we have not previously approved;
- (d) You will replace or add new equipment when we reasonably specify in order to meet changing standards or accommodate new menu items and new methods of preparation and service. For any non-compliance with this subsection 10.2(d), you may be subject to a non-compliance fee as set forth in Section 9.11;
- (e) On notice from us, you will remodel, renovate, refresh, reimage, refurbish, and modernize the Premises and the Restaurant, including the building design, parking lot, landscaping, equipment, signs, interior and exterior décor items, fixtures, furnishings, trade dress, and color scheme, to meet our then-current standards, specifications, and design criteria required of new franchisees, and including, without limitation, such structural changes, remodeling, and redecoration, and such modifications to existing improvements as may be necessary to do so, all within the time frame we require. Material alterations to remodel or renovate shall not be required more often than once every 10 years. Nonmaterial alterations to refresh, reimage, refurbish, or modernize shall not be required more often than once every five years. The frequency limitations shall not apply to your obligations to regularly maintain, repair, and replace the Restaurant and related property as described in Section 10.2(b) or your obligations to purchase equipment and other materials as described in Section 3.5. Material or non-material alterations may be required within 12 months after a Transfer in accordance with Section 14.4(d). For any non-compliance with this subsection 10.2(e), you may be subject to a non-compliance fee as set forth in Section 9.11; and
- (f) You will place or display at the Premises only those signs, emblems, designs, artworks, lettering, logos, and display and advertising materials that we from time to time approve.

10.3 Relocation of Franchise; Sale-Leaseback of Real Estate.

- (a) Provided that you are in substantial compliance with the provisions of this Agreement, if you lose the right to possession of the Premises through no fault on your part or if the Premises are destroyed, condemned or otherwise rendered unusable, or if you request the relocation of your Restaurant and, in our reasonable judgment, the existing site for the Restaurant does not meet our then-current standards and specifications for sites for Taco John's Restaurants, we will grant permission for the relocation of the Restaurant to an alternate location and site which meets our then-current site criteria for Taco John's Restaurants and is approved by us. If we agree on the relocation of the Restaurant, you agree to expeditiously develop the alternate site in compliance with our then-current specifications and requirements for Taco John's Restaurants, all at your sole cost and expense, and to open the relocated Restaurant within 150 days after receiving our permission to relocate, unless otherwise agreed to by us. You agree to de-identify, to our standards, the vacated location within 30 days of relocation. At our option, you shall pay us a relocation fee in the amount set forth in Section 9.10 upon the commencement of relocation of your Restaurant.
- (b) If you own and later sell the real estate where your Restaurant is located and subsequently lease it for operation of the Restaurant (a sale-leaseback), you must provide us with notice at least 90 days in advance of the sale and provide a copy of the proposed lease for our review and acceptance. You and the lessor may be required to sign a Lease Rider in a form and with terms acceptable to us.

10.4 Restaurant Menu and Service Methods.

You agree that:

- (a) Your Restaurant will be open during all required hours of operations and will offer for sale all products and services included in the Core Menu as specified in the Operations Manual, and you will not remove any product from the Core Menu unless we authorize such removal. For any non-compliance with the required hours of operations in the Operations Manual, you may be subject to a non-compliance fee as set forth in Section 9.11;
- (b) Your Restaurant will offer and sell approved products and services only in the manner we have prescribed, and you will follow all of our specifications as to content, weight and quality of products served to customers;
- (c) You will not offer for sale or sell at the Restaurant, the Premises or any other location any products or services we have not approved;
- (d) Your Restaurant will offer delivery services through all approved third-party delivery providers available in your community. For any non-compliance with this subsection 10.4(d), you may be subject to a non-compliance fee as set forth in Section 9.11;
- (e) You will not use the Premises for any purpose other than the operation of the Restaurant;
- (f) You will discontinue selling and offering for sale any products or services that we decide at any time to withdraw from Taco John's Restaurants; and
- (g) You will maintain an inventory of approved products, beverages, ingredients and other goods sufficient in quantity and variety to realize the full potential of the Restaurant;
- (h) You will accept gift cards and loyalty/rewards redemptions, and specific credit and debit cards using the service provider we designate, and other methods of payment from customers as we may direct from time to time. You shall not charge customers any additional fees or service charges if they elect to use a credit card or debit card for payment of any services or products provided in your Taco John's Restaurant. The Payment Card Industry ("PCI") requires all companies that process, store, or transmit credit or debit card information to protect the cardholders' information by complying with the PCI Data Security Standard ("PCI DSS"). Therefore, you will be PCI compliant by following and adhering to the then-current PCI DSS, currently found at www.pcisecuritystandards.org, or any similar or subsequent standard for the protection of cardholder data throughout the term of this Agreement. PCI mandates the PCI DSS compliance;
- (i) You agree to purchase and maintain, at your own cost and expense from an approved vendor, the specific electronic cash register, freestanding kiosks, kitchen video monitors, receipt printers, a drive-thru speed of service timing system, and headset system (the "Point of Sale System"), along with the computer hardware, software, Internet connections and service, and other computer related accessories, peripherals and equipment necessary to support the Point of Sale System, mobile ordering and delivery platforms that utilize geofencing services, firewall and security service, and the required back of house information system (which includes the specific computer and software required to manage inventory, purchasing and other back of house functions) we specify from time to time (collectively, with the Point of Sale system, the "Computer System") with at least the minimum specifications we require from time to time for use in operating the Restaurant. Your Computer System must be set up in accordance with

our specifications to record all sales information in connection with the Restaurant and to comply with your reporting requirements to us. Your Computer System must also be capable of processing credit card and debit card payment and complying with PCI DSS standards as described in Section 10.4(h) and be able to receive sales from mobile ordering, third-party delivery, and catering. We may require you to upgrade, at your expense, any part or all of the Computer System when deemed necessary by us. We may also change the required Point of Sale System or other elements of the Computer System from time to time in our discretion, and you will be required to acquire any updated system at your sole expense. You must enter into any contracts required by the company that oversees these systems and services and you must pay the fee for those systems and services to us or the supplier, as we designate. We require you to use a financial reporting system provided by us or our designated supplier that allows you to upload and deliver your Net Sales reports, Profit and Loss statements, and other data, information, and records we require. There is currently no fee due for this system, but we may impose a fee upon 180 days' notice, in which event you must pay the fee charged for that service to us or the supplier, as we designate. The fee for the financial reporting system, once established, is subject to change upon 30 days' notice from us or the supplier. You must purchase exterior digital menu boards, and you have the option to purchase interior digital menu boards, from our approved third-party supplier. You will be required to pay the license fee for the digital menu boards, which may be charged directly by the supplier or which we may collect on behalf of the supplier. For any non-compliance with this subsection 10.4(i), you may be subject to a non-compliance fee as set forth in Section 9.11.

- (j) We may, from time to time, conduct market research and testing to determine consumer trends and the salability of new products and services. If you are chosen by us, and if you agree, you will participate in our market research programs, in test marketing new food products and services in the Restaurant, and by providing us with timely reports and other relevant information regarding that market research. In connection with any such test marketing in which you agree to participate, you agree to purchase a reasonable quantity of the products being tested and to effectively promote and make a good faith effort to sell them.
- (k) We may also provide guidance and assistance to you in recommending the prices to be charged by you for the services and products provided by you. Unless prohibited by applicable law, we may periodically set a maximum or minimum price that you may advertise and charge for products and services offered by your Taco John's Restaurant for limited time offerings or offers and items we designate for inclusion in our value programs. If we establish a maximum price for any products or services, you shall not offer or sell those products or services at any greater price. If we establish a minimum price for any products or services, you shall not offer or sell those products or services at any lesser price. If we do not establish pricing limits, we may establish suggested prices. In that case, any prices that we recommend to you are merely recommendations and you may establish your own prices, which may be higher or lower than our recommended prices. You must abide by our advertising policies related to advertising prices.
- (l) We require you to offer catering service through online catering platforms that we designate or approve. For any non-compliance with this subsection 10.4(l), you may be subject to a non-compliance fee as set forth in Section 9.11.

10.5 Approved Products, Distributors and Suppliers.

(a) The reputation and goodwill of Taco John's Restaurants is based on, and can be maintained only by, the sale of distinctive, high quality food products and beverages and the presentation, packaging, service, and delivery of those products in an efficient and appealing manner. We have developed standards and specifications for food products, ingredients, beverages, materials and supplies incorporated in or used in the preparation, serving, packaging and delivery of prepared food and beverage products authorized for sale at Taco John's Restaurants. We have approved and will continue to periodically approve

suppliers and distributors of the above products that meet our standards and requirements, which standards we will provide to prospective suppliers as described below in this Section. You agree to purchase all food products and beverages, menus, paper and plastic products, packaging or other materials, and utensils only from distributors and other suppliers we have approved. If we have approved a single distributor for your area, you must obtain products only from that distributor.

- (b) We may approve a single supplier for any product or service and may approve a supplier only as to certain products and services. We may concentrate purchases with one or more suppliers to obtain lower prices or the best advertising support or services for any group of Taco John's Restaurants franchised or operated by us, but we will use our reasonable efforts to negotiate agreements with suppliers which, in our good faith belief, are in the best interest of all Taco John's Restaurants. Approval of a supplier may be conditioned on requirements relating to the frequency of delivery, concentration of purchases, standards of service or other criteria and may be temporary, pending our continued evaluation of the supplier from time to time. We reserve the right to negotiate a contract with a distributor that will include a program whereby the distributor shall pay us administrative allowances to be used in our discretion to address supply chain issues or other supply chain initiatives. Such administrative allowances shall not be considered Rebates as defined in 10.5(c) below.
- (c) You hereby authorize us to act as your sole agent to enter into contracts with parties offering promotion, discount or other programs whereby you would receive rebates or other marketing allowances ("Rebates") from handling items offered for sale by such parties. All Rebates will be paid directly to us, unless we designate otherwise. Rebates paid to us will be used at our discretion. By signing this Agreement you assign all of your right, title and interest in all Rebates to us, and authorize us to furnish any proof of purchase evidence as may be required in accordance with the contracts with the suppliers. At our discretion, we may agree with certain suppliers for certain of these Rebates to be divided between us and our franchisees, but we have no obligation to arrange for any particular Rebate amounts to be paid to you.
- (d) We reserve the right to require you to use the brand of fountain service beverages we designate from time to time, upon our reasonable prior notice.
- If you would like to purchase any items or services from any unapproved supplier, (e) you must submit to us a written request for approval of the proposed supplier. Upon receipt of your request to approve an alternate supplier, we will furnish the supplier with our relevant standards and specifications (except standards and specifications which are or include trade secrets) for suppliers of items or services of the type to be furnished by the proposed supplier provided the proposed supplier agrees to execute a confidentiality agreement in a form acceptable to us. We have the right to inspect the proposed supplier's facilities, and to require product samples from the proposed supplier to be delivered at our option either directly to us or to any independent, certified laboratory which we designate for testing in order to determine if the proposed supplier meets our requirements for approved suppliers. We are not required to make available to you the criteria used in approving suppliers. We will inform you of our decision to approve or disapprove a proposed supplier within 90 to 120 days from the date we receive all required information about the supplier. We reserve the right to charge the proposed supplier a fee (not to exceed the reasonable cost of the inspection and the actual cost of the test) to make the evaluation. We reserve the right to periodically re-inspect the facilities and products of any approved supplier and to revoke our approval if the supplier does not continue to meet any of our criteria.
- (f) We, for ourselves and on behalf of our Affiliates, agree not to charge a markup on proprietary products sold by us or the Affiliates to you.

10.6 Compliance with Laws and Good Business Practices.

- (a) You will secure and maintain in force in your name all required licenses, permits, and certificates relating to the construction and operation of the Restaurant. You also agree to familiarize yourself with all the federal, state and local laws, ordinances and regulations that apply to the construction and operation of the Restaurant, and you will operate the Restaurant in full compliance with all applicable laws, ordinances, and regulations. From time to time we may, but have no requirement to do so, advise you of any legislative or other legal developments that may affect you or your Restaurant. Any information we provide to you does not relieve you of your responsibility to consult with your own legal advisor. You are solely responsible for complying with any legal requirements affecting you or your Taco John's Restaurant.
- (b) All advertising you employ must be completely factual, in good taste (in our judgment), and must conform to the highest standards of ethical advertising. You and we agree that in all dealings with each other, with customers, with suppliers, and with public officials, you and we will both adhere to the highest standards of honesty, integrity, fair dealing, and ethical conduct. You and we further agree to refrain from any business or advertising practice which may be harmful to the reputation of Taco John's Restaurants and the goodwill associated with the Marks.
- (c) You will notify us in writing within five days of the commencement of any action, suit, or proceeding, and of the issuance of any order, writ, injunction, award, or decree of any court, agency, or other governmental unit, which may materially adversely affect your operation or financial condition or that of the Restaurant, or of any notice of violation of any law, ordinance, or regulation relating to health or safety. You will also promptly notify us in writing of any crisis that could generate media attention for the Taco John's brand. For purposes of this provision, a crisis means any event that occurs at your Restaurant or within your organization that could result in negative media attention for the Taco John's brand including, but not limited to, the events listed in our crisis management procedures described in the Operations Manual. If you do not report a crisis to us within 24 hours of your discovery of the crisis, we have the right to charge you a Media Relations Support Fee as set forth in Section 9.13.
- (d) You, your Active Certified Managers, and your other employees, will at all times present and conduct business in a professional manner with all customers, suppliers, and us. All communications with customers, suppliers, and us must be conducted in a dignified, respectful, courteous, and professional manner. No hostile, threatening, or offensive statements or behavior are permitted. If you or any of your agents fail to comply with these requirements in your communications with us, we may require you to communicate with us only through our legal staff or other person we designate.

10.7 Ownership, Management and Personnel of Restaurant.

(a) You agree that the Restaurant will, during substantially all operating hours, be under the direct, on-premises supervision of the Franchisee, an owner or a manager who has completed our Initial Training and has been certified by us, or an Active Certified Manager, who devotes full time and efforts to the management of the Restaurant. You are solely responsible for recruiting, appointing, hiring, firing, and supervising your Active Certified Managers, Above Store Leaders, employees, independent agents, and other representatives. You will have sole authority and control over the day-to-day operations of your Taco John's Restaurant and its employees and representatives. The employees of the Restaurant will be your employees, and they are not our employees or agents. You are solely responsible for implementing training and other programs for employees related to the legal, safe, and proper performance of their work, regardless of the fact that we may provide advice, suggestions, and certain training programs regarding the overall operation of a Restaurant as described in this Agreement. Such advice, suggestions, and training by us are provided to protect our brand and the Marks and not to control the day-to-day operations of your Taco John's Restaurant.

- (b) The Franchisor and the Franchisee agree that they will not recruit or hire, either directly or indirectly, any employee (or a former employee for six months after his or her employment has ended) of the other party without the advance written permission of the other party.
- (c) You represent that the Statement of Ownership attached to this Agreement as Exhibit III is true, complete, and accurate. You agree to promptly provide us with advance written notification of any changes to this information during the term of this Agreement and to comply with the provisions of Article 14, Transfer Requirements. If the franchisee is an entity, we may require the officers, directors, shareholders, partners, managers, and members, whichever are applicable, to sign the Guaranty which is attached to this Agreement as Exhibit IV.

10.8 Insurance.

- (a) Prior to commencement of construction of the Restaurant, you must obtain and have in effect the following insurance coverage under policies of insurance issued by carriers of your choice that we approve, unless we designate a specific required carrier for systemwide brand protection policies: (i) comprehensive public, product and automobile liability insurance against claims for personal injury, death, and property damage caused by or occurring in conjunction with the operation of the Restaurant or your conducting business pursuant to this Agreement under one or more policies of insurance containing minimum liability coverage we prescribe from time to time; (ii) general casualty insurance including fire and extended coverage, vandalism, and malicious mischief insurance for the replacement value of the Restaurant and its contents; (3) cyber liability insurance; (4) loss of business income and incident response for foodborne illness insurance; (5) employment practices liability insurance, (6) umbrella insurance, (7) workers' compensation insurance required by your state, and (8) business interruption insurance, each in amounts we may require from time to time. You must maintain these policies in force during the entire term of this Agreement. We may periodically increase or decrease the amounts of coverage required under these insurance policies and require different or additional kinds of insurance at any time, including excess liability insurance, to reflect inflation, additional risks, changes in law or standards of liability, higher damage awards or other relevant changes in circumstances. Each insurance policy must be endorsed to name us as Additional Insured - Grantor of Franchise, and must provide for 30 days' advance written notice to us of any material modification, cancellation, or expiration of the policy. We may require that you carry additional insurance policies in the future upon 60 days' notice to you. Notwithstanding the foregoing, we currently maintain the insurance coverage for loss of business income and incident response for foodborne illness, which covers our locations and each individual franchised location in the required amounts. We may designate a specific required carrier and collect your share of the premium for this insurance (as determined by us in our discretion) from you and pay it to the insurance carrier on your behalf. However, we may require you to obtain this insurance directly in the future.
- (b) You must before the expiration of the term of each insurance policy furnish us with a copy of each policy you are to maintain for the upcoming term, along with evidence of the payment of the premium for each. We reserve the right to require you to send us a complete copy of each insurance policy on request, evidence of the payment of the premium for each, and copies of the Additional Insured Endorsements. If you do not maintain the required insurance coverage, or do not furnish us with satisfactory evidence of your insurance coverage and the payment of the premiums for them, we may obtain, at our option and in addition to our other rights and remedies under this Agreement, any required insurance coverage on your behalf. If we do that, you agree to fully cooperate with us in our effort to obtain the insurance policies, promptly execute all forms or instruments required to obtain or maintain the insurance, allow any inspections of the Restaurant which are required to obtain or maintain the insurance and pay to us, on demand, any costs and premiums we incur.

(c) Your obligation to maintain insurance coverage as described in this Agreement will not be reduced in any manner by reason of any separate insurance we maintain on our own behalf, nor will our maintenance of such insurance relieve you of any obligations under Article 7 of this Agreement.

10.9 Websites and Social Platforms.

Presence on the Internet, as it may evolve over time, including Websites and Social Platforms, is considered to be advertising and is subject to our review and written approval before you may use it in connection with the Restaurant. As used in this Agreement, the term Internet includes, but is not limited to, what are today known as World Wide Web home pages and Social Platforms. However, these types of global electronic communications evolve and change over time and such changes are intended to be included within the reasonable intent of this clause. Collectively, and for ease of reference, the term "Internet presence" will encompass the current Internet, along with websites and social platforms, and future similar concepts. If we permit you to have an independent Internet presence, then the following requirements will apply:

- (a) You may not establish or use an Internet presence without our prior written approval;
- (b) Before establishing any Internet presence, you must submit to us for our written approval, a proposal in the form, manner, and containing the content that we may reasonably require, including: (i) the domain name or platform name, (b) format, and (c) visible and non-visible content (including meta-tags). If you propose any material revisions to the Internet presence or any of the information contained in the Internet presence, you must submit the revisions to us for our prior written approval;
- (c) If requested by us, you must establish hyperlinks to our website and others as we may require from time to time; and
- (d) You must properly use our Marks as described in Article 6 of this Agreement on your Internet presence.

We reserve the right to revoke our approval of any Internet presence at any time and require you to discontinue using it. We may, in the future, restrict any Internet presence to web pages on our Taco John's website or other similar Internet presence as the same may evolve. In such event, you will be required, at your own expense, to convert your Internet presence to the format we reasonably specify.

10.10 Guest Experience Program.

We have established a guest experience program to evaluate guest satisfaction of your Restaurant, which consists of mystery shops and guest surveys. You are required to participate in this program and pay the fees or charges associated with it. We may modify this program from time to time at our discretion. If we have approved vendors for this program, you must use our approved vendors and utilize the services at a frequency and in the manner prescribed by us. You will be required to pay all fees of the vendors for this program, which fees may change from time to time. We may require that you pay these fees to us, for us to transmit to the vendors. We have the right to review the results of the service evaluations and reserve the right to charge you for the cost of resolving guest complaints.

10.11 Online Learning Management System.

We currently maintain and provide you with access to the online learning management system referenced in Section 4.3. In order to obtain access to this system, you and any other users we approve may be required to sign an agreement in the form required, which may be modified from time to time, as well as pay the provider of this system the fees it charges. We have no obligation to maintain this system indefinitely. We can, after 60 days' advance written notice to you, replace this system entirely or designate a different provider of this or another system, and require you to subscribe, at your sole cost and expense, to the new service or provider we designate to obtain access to this or similar information, including the Operations Manual.

10.12 Non-attendance of Franchisee Convention or Manager Convention.

We require you (or at least one of your owners if the Franchisee is an entity) or your Above Store Leader to attend the convention ("Franchisee Convention") held by the Association of Taco John's Franchisees (the "Association") and we may require at least one manager per Restaurant to attend the manager convention. If you do not have the required attendees at the Franchisee Convention or the manager convention, then we may charge you a non-attendance fee of \$1,000 for each such event, at our sole discretion.

Not less than 60 days prior to the opening date of a Franchisee Convention or manager convention, you may submit a written request to us for a waiver of this requirement based on your specific circumstances. We will use our best business judgment to determine whether to approve the request.

11. ADVERTISING AND MARKETING.

11.1 Advertising Fund.

- (a) The Advertising and Marketing Fee described in Section 9.3 will be paid into a fund (the "Advertising Fund"). The monies in the Advertising Fund will be used by us, as we determine in our sole discretion, for expenses incurred in connection with the development, support, and dissemination of marketing programs and promotion of the Marks and the brand.
- (b) The funds in the Advertising Fund are spent to foster recognition and promotion of the Marks, the brand, and retail sales of Taco John's Restaurants, generally, and to pay for the development, support, and dissemination of marketing materials and programs. Materials produced using the Advertising Fund may include television and radio spots, newspaper and billboard displays, freestanding inserts ("FSI") and flyers, signs, menus, menu boards, email or other social media campaigns, digital displays, and visual displays. We may use a variety of sources for advertising materials including in-house and regional or national advertising agencies. The Advertising Fund is also used to create and produce advertising materials that are made available to you, to other Taco John's Franchisees, and to advertising cooperatives formed by Taco John's Franchisees ("Advertising Cooperatives"), as they may exist from time to time. We or our advertising agency may charge a fee for any advertising materials acquired by a Franchisee or Advertising Cooperative.
- (c) The Advertising Fund will be administered by us. Unless we have established a National Marketing Program as provided below, we will confer with an advertising production committee (the "Advertising Production Committee"), which will be made up of seven franchisees and one of our management level employees. The franchisee members of the Advertising Production Committee will be either elected by franchisees or appointed by the Board of Directors of the Association of Taco John's

Franchisees, Inc., and serve three year terms. The Advertising Production Committee will be organized and governed pursuant to its bylaws that we approve. We may disband the Advertising Production Committee in our sole discretion. The Advertising Production Committee serves in an advisory role only. The artistic and creative aspects of advertising/marketing materials will be determined by us after conferring with the Advertising Production Committee.

- (d) If you are required to join a local or regional Advertising Cooperative, the Advertising Fund will contribute to the Advertising Cooperative monthly in an amount equal to the lesser of the amount you have paid toward your Advertising and Marketing Fee or 2.85% of your total net sales to be used for media placement in your DMA, as defined below. At our discretion, we may also agree to contribute additional amounts of your Advertising and Marketing Fee to the Advertising Cooperative. If you are not currently assigned by us to an Advertising Cooperative, then you will be required, at your expense, to use one of our approved media buyers to assist you in the daily administration of your local advertising.
- (e) If you are operating a traditional Taco John's Restaurant and are not in an Advertising Cooperative, you will be required to conduct local advertising within your DMA and you will be required, at your expense, to use one of our approved media buyers to assist you in the daily administration of your local advertising. In that event, we will pay from the Advertising Fund 2.85% of your Net Sales to the media buyer you select to be applied against the media buyer's fees and your local advertising program developed by the media buyer. You must be current on paying us your Advertising and Marketing Fee or we may withhold amounts paid to your media buyer, in which case you will be responsible for all amounts due to the media buyer. We will not pay any amounts from the Advertising Fund to your media buyer if you are in an Advertising Cooperative or if you operate a non-traditional Restaurant. You are permitted to use advertising and promotional materials you develop and to conduct independent local promotions so long as you obtain our prior written consent to do so. You must submit to us for our approval any advertising and promotional materials you develop at least 10 days before you use those materials.
- (f) We will withhold the distribution of marketing materials to any franchisee who fails to timely remit the Advertising and Marketing Fee.
 - (g) We do not reimburse food cost for coupons redeemed at your restaurant.
- (h) We will prepare an annual unaudited statement showing all amounts collected and expended by the Advertising Fund, and make it available to you on written request.
- (i) We may in the future establish a program for national, regional, or local media placement, digital advertising, and national advertising (the "National Marketing Program"). If we create a National Marketing Program, we may also use some or all of the Advertising Fund to support it. At that time, the Advertising Production Committee will be disbanded, and the functions of that committee shall become the responsibility of a National Marketing Program advisory committee (the "National Marketing Program Advisory Committee"). We currently do not have any standards for a National Marketing Program or for a National Marketing Program Advisory Committee. We will provide you at least six months' advance notice of the establishment of a National Marketing Program.

11.2 Advertising Cooperatives.

We reserve the right to establish and require your participation in either a local or regional Advertising Cooperative at our sole discretion (each an Advertising Cooperative). If we require you to join a local Advertising Cooperative, it will consist of Franchisees in your Designated Market Area ("DMA")

or such other contiguous areas determined by us, in our sole discretion, to be appropriate in your market area. Currently, a DMA is determined by the A.C. Nielsen Company's Nielsen Station Index and is defined as "generally a group of counties in which the commercial TV stations in the Metro/Central area achieve the largest audience share." We may change how the DMA is established based on changes in the advertising industry.

If we require you to join a regional Advertising Cooperative, we will designate the market areas included within the regional Advertising Cooperative. We may change your assignment to an Advertising Cooperative in our sole discretion. We will provide you at least six months' advance notice of any change in your assignment to an Advertising Cooperative.

We require that every local or regional Advertising Cooperative be legally incorporated, and as a corporation, each Advertising Cooperative must operate under bylaws that we approve. We will provide model bylaws to all Advertising Cooperatives. We may dissolve an Advertising Cooperative or change the structure and/or function of an Advertising Cooperative.

Franchisees assigned to a local or regional Advertising Cooperative shall elect officers of their Advertising Cooperatives, and these officers are responsible for administration of the Advertising Cooperative and the expenditure of funds contributed to the Advertising Cooperative. We require that at a minimum, a President and Treasurer for each Advertising Cooperative be elected. We may require the Advertising Cooperative to delegate the daily administration of the Advertising Cooperative to an advertising agency, media buyer, or other professional marketing services provider.

If you are required to join a local or regional Advertising Cooperative, then the Fund will contribute to your Advertising Cooperative monthly an amount equal to the lesser of the amount you have paid toward your Advertising and Marketing Fee or 2.85% of your total Net Sales (or 1% for non-traditional facility franchisees) to be used for media placement in your DMA. An Advertising Cooperative may determine, by a vote of its members, to collect additional advertising fees for use by the Advertising Cooperative on approved advertising and marketing programs, although non-traditional locations are not required to contribute additional advertising fees to an Advertising Cooperative. Advertising Cooperatives are also required to prepare annual financial statements and to present these to its members (or prospective members) and to us upon request.

If we establish a National Marketing Program, it may replace any local and regional Advertising Cooperatives.

11.3 Participation in Certain Programs and Promotions, including Customer Loyalty/Rewards Program.

You must use your best efforts to promote and advertise your Restaurant and you must participate in all national or special advertising and promotional programs we establish in the manner we direct. You must, at your expense, participate in any customer loyalty/rewards program we may establish, including honoring loyalty/rewards redemptions presented in your Restaurant. We reserve the right to modify or terminate the program. You must, at your expense, also participate in any gift card program, as it may be modified from time to time, and honor all Taco John's gift cards presented in your Restaurant. In particular, you must pay any fees imposed by any gift card processor, which we may collect on behalf of the gift card processor. You must enter into any contracts required by the company that oversees our gift card program or customer loyalty/rewards program and pay any fees required by that company.

12. ACCOUNTING, REPORTS AND FINANCIAL STATEMENTS

You agree to establish and maintain bookkeeping, accounting, and recordkeeping systems conforming to our requirements as we prescribe from time to time. We have the right to electronically access, either directly from your Computer System or through a third-party vendor, at our discretion, your Restaurant sales information at any time. You agree to furnish us, by the 20th day of each month, a report of the Net Sales of the Restaurant and a Profit and Loss Statement for the preceding month using the financial reporting system described in Section 10.4(i), and any other data, information, and supporting records concerning the operation and financial condition of the Restaurant that we reasonably require. You agree to certify and sign each report and financial statement required by this Section if we so request. Reports are to be submitted electronically when we direct. By submitting reports or any other financial information to us, you authorize us to utilize this information to prepare a financial performance representation, to release this information as necessary to substantiate any financial performance representation we make, to share such information in summary form as we deem necessary or desirable to share with other franchisees at any annual convention or other franchise business meetings, and to utilize this information in any other manner and with any other parties that we deem appropriate without obtaining any further written consent from you.

13. INSPECTIONS AND AUDITS

13.1 Franchisor's Right to Inspect Restaurant.

To determine whether you and the Restaurant are complying with this Agreement, and with specifications, standards, and operating procedures we prescribe for the operation of Taco John's Restaurants, we or our agents have the right, at any reasonable time and without advance notice to you, to:

- (a) Inspect, photograph and videotape the Premises;
- (b) Observe the operations and sample products of the Restaurant for such consecutive or intermittent periods as we deem necessary;
 - (c) Interview personnel of the Restaurant;
 - (d) Interview customers of the Restaurant; and
- (e) Inspect your Computer System including, but not limited to, inspecting your Internet presence, if any has been previously approved.

You agree to fully cooperate with us in connection with any of those inspections, observations and interviews. You agree to present to your customers any evaluation forms we periodically prescribe and agree to participate and/or request your customers to participate in any surveys performed by or on our behalf.

13.2 Franchisor's Right to Audit.

We have the right at any time during business hours to inspect, copy and audit, or cause to be inspected, copied and audited, the business records, bookkeeping and accounting records, sales and income tax records and returns, computer records and other records of the Restaurant and the books and records of you and any owner. You agree to fully cooperate with our representatives and independent accountants we may hire to conduct any inspection or audit. If an audit determines there is a deficiency in previously reported Net Sales, you agree to pay us the additional royalties owed, plus interest as provided in Section

9.6, within five days after written demand. If the deficiency is more than 3% of actual Net Sales, you agree to pay us, within five days after demand, the costs actually incurred by us in the examination of your books and records.

14. TRANSFER REQUIREMENTS

14.1 Interests in Franchisee.

The Franchisee and each owner represent, warrant and agree that all "Interests" in the Franchisee are owned in the amount and manner you have disclosed to us. (An "Interest" means any shares, membership interests, or partnership interests of the Franchisee, and any other equitable or legal right in any of Franchisee's stock or other interests, revenues, profits, rights or assets. When referring to the Franchisee's rights or assets, an "Interest" also includes this Agreement and the Franchisee's rights under and interest in this Agreement, the Restaurant and the revenues, profits or assets of the Restaurant.) The Franchisee and each owner also represent, warrant and agree that no change will be made in the ownership of an Interest other than as permitted by this Agreement or as we may otherwise approve in writing. The Franchisee and each owner agree to furnish us with evidence as we may request from time to time to assure ourselves that the Interests of the Franchisee and each owner remain as permitted by this Agreement, including a list of all persons or entities owning any Interest.

14.2 Transfer by Franchisor.

This Agreement is fully transferable by us and will inure to the benefit of any person or entity to whom it is transferred, or to any other legal successor to our interests in this Agreement provided, however, that we will make reasonable provisions for the performance of our obligations under this Agreement by our successor.

14.3 No Transfer Without Approval.

You understand and acknowledge that the rights and duties created by this Agreement are personal to you and that we have entered into this Agreement in reliance on the individual or collective character, skill, aptitude, attitude, business ability, and financial capacity of the Franchisee and the owners. Accordingly, neither this Agreement nor any part of your interest in it, nor any other Interest of the Franchisee or an owner, may be transferred without our advance written approval. However, such approval will be given provided that the transferee satisfies our then-current reasonable qualifications for new Taco John's Restaurant franchisees, which qualifications shall be based on legitimate business reasons. You acknowledge and agree that the qualifications and conditions for such transfers and transferees shall include, without limitation, the conditions set forth in this Agreement. If the proposed transferee (the "Proposed New Owner") does not meet our then-current reasonable qualifications, we may refuse to permit the transfer, provided that our refusal is not arbitrary or capricious when compared to our actions in other similar circumstances. Any Transfer (as defined below) that is made without compliance with this Agreement will constitute a breach of this Agreement and convey no rights to or interests in this Agreement, in you, or in any Taco John's Restaurant. Your articles of incorporation, articles organization, partnership agreement, or other formation documents and all stock certificates or other evidence of ownership must describe or bear a legend reflecting the transfer restrictions of this Section 14.3.

Except as described below, the term "Transfer" as used in this Agreement means any voluntary, involuntary, direct or indirect assignment, sale, gift, exchange, or occurrence of any other event which would or might change the ownership of any Interest, and includes, without limitation, the transfer of ownership of capital stock, membership interest or partnership interests, merger or consolidation, or issuance of additional ownership interests in the Franchisee, sale of common stock of the Franchisee sold

pursuant to a private placement or registered public offering, transfer of an Interest in a divorce proceeding or otherwise by operation of law, or transfer of an Interest to a trust or upon the death of an owner of the Interest whether by will or otherwise. Notwithstanding the foregoing and any other provision of this Agreement, the following occurrences shall not require our consent and shall not make applicable the right of first refusal as set forth in Section 14.7:

- (a) The Transfer of ownership of the Franchise upon your death or total disability, or upon the death or total disability of an owner of the Franchise if the Franchisee is a corporation, limited liability company or partnership, to such person's spouse, child or children, or to a co-owner in the Franchisee is an entity, unless the transferee fails to meet our then-current reasonable qualifications for Taco John's Restaurant franchisees.
- (b) A Transfer of the Franchise to a corporation or other entity wholly owned by you. You agree that, in connection with such Transfer:
- (i) you will submit to us such documents relating to the entity as we may require;
- (ii) you and all other owners of such entity will sign our then-current Franchise Agreement and execute a guaranty of such entity's obligations under the Franchise Agreement; and
- (iii) the organizational documents of the entity, including all stock certificates or operating agreements, shall state that they are subject to the restrictions on transfer contained in this Agreement.
- (c) A Transfer of less than a Controlling Interest in the Franchise to an existing owner of the Franchise, provided that 50% or more of the Interests in the Franchise are held by persons who meet our then-current qualifications for Taco John's Restaurant franchisees. If less than 50% of the Franchise would be owned by persons who meet our then-current reasonable qualifications, we may refuse to authorize the Transfer.
- (d) A Transfer of less than a Controlling Interest in the Franchise to your spouse, child or children, provided that 50% or more of the Interests in the Franchise are held by persons who meet our then-current qualifications. If less than 50% of the Franchise would be owned by persons who meet our then-current reasonable qualifications for Taco John's Restaurant franchisees, we may refuse to authorize the Transfer.
- (e) A Transfer of less than a Controlling Interest in the Franchise to an employee stock ownership plan, or employee incentive plan, provided that 50% or more of the Interests in the Franchise are held by persons who meet our then-current qualifications. If less than 50% of the Franchise would be owned by persons who meet our then-current reasonable qualifications for Taco John's Restaurant franchisees, we may refuse to authorize the Transfer.
- (f) A grant or retention of a security interest in the Restaurant or its assets, provided the security agreement establishes an obligation on the part of the secured party enforceable by us to give us notice of the secured party's intent to foreclose on the collateral simultaneously with notice to the debtor, and a reasonable opportunity to redeem the interests of the secured party and recover the secured party's interest in the Restaurant by paying the secured obligation.

- (g) A Transfer of less than 50% of the Interests in the Franchise or the Restaurant, unless such Transfer is one of a series of Transfers which taken together would constitute the Transfer of 50% or more of the total of such Interests.
- (h) A Transfer to a trust provided the trustee and beneficiaries meet our current reasonable qualifications for new franchisees.

You must give us written notice 60 days in advance of any of the occurrences described in Section 14.3 (a) through (h) above.

14.4 Conditions for Approval of Transfer.

If the Franchisee and the owners are in full compliance with this Agreement, we will not unreasonably withhold our approval of a Transfer that meets all the applicable requirements of this Section. The person or entity to whom you wish to make the Transfer or its principal owners (defined above as the "Proposed New Owner"), must be individuals of good moral character and otherwise meet our then-current reasonable qualifications for Taco John's Restaurant franchisees including, without limitation, qualifications relating to business experience, aptitude and financial resources and all of the following conditions must be met before or at the time of the Transfer:

- (a) You must pay any amounts owed to us or our Affiliates which are unpaid;
- (b) The Proposed New Owner must have completed our training program and have been certified. The Proposed New Owner shall be responsible for the travel and living expenses (including all transportation costs, room, board and meals) incurred during the training program;
- (c) If your existing lease or sublease for the Premises requires it, the Lessor must have consented to the assignment or sublease of the Premises to the Proposed New Owner; or, if the Proposed New Owner is entering a new lease, sublease, or other financing arrangement including a sale-leaseback for the Premises, you must provide us with a copy of the documents for our review and acceptance;
- (d) Either you have remodeled, reimaged, expanded, equipped, modernized and redecorated the interior and exterior of the Premises in accordance with our then-existing image and specifications for a Taco John's Restaurant in connection with the Transfer, or the Proposed New Owner shall have agreed, in writing, to make such reasonable capital expenditures to remodel, reimage, expand, equip, modernize and redecorate the interior and exterior of the Premises in accordance with our then-existing image and specifications for a Taco John's Restaurant;
- (e) The Proposed New Owner shall agree to sign a Franchise Agreement in a form then currently offered by us, which may include different fees and duties, except that the term of which shall end on the expiration date of this Agreement, and which shall supersede this Agreement in all other respects;
- (f) You must give us written notice 60 days in advance of your intended Transfer date. Once the transferee is approved, you shall pay us a transfer fee in the amount set forth in Section 9.8 for each Transfer;
- (g) You must provide us a copy of the proposed purchase agreement or other agreement for the transaction between you and the Proposed New Owner, and that agreement must be approved by us;

- (h) You must have offered us the opportunity to exercise our right of first refusal pursuant to Section 14.7, and we must have declined to exercise it; and
- (i) You must sign a general release, in a form approved by us, of any and all claims against us, our Affiliates and the respective officers, directors, employees and agents.

14.5 Death and Total Disability.

If, upon the death or total disability of the Franchisee (or the death or total disability of an owner) the Restaurant is not under the direct on-premises supervision of an owner or manager who has completed our training programs and has been certified by us, the executor, administrator, conservator or other personal representative of such person shall appoint a competent manager within a reasonable time, not to exceed 45 days from the date of death or total disability. The appointment of such manager shall be subject to our prior written approval, such manager must complete our training program and be certified by us. If the Restaurant is not being managed by an approved and certified manager within 45 days after the death or total disability, we shall have the right, but not the obligation, to immediately appoint a manager to maintain the operation of the Restaurant until an approved assignee shall be able to assume the management and operation of the Restaurant. We shall have the right to charge a fee for such management service of 15% of the net sales for the days we managed the Restaurant.

14.6 Effect of Consent to Transfer.

Our consent to a proposed Transfer pursuant to this Section will not constitute a waiver of any claims we may have against you under this Agreement, and will not be deemed a waiver of our right to demand exact compliance with any of the terms or conditions of the new Franchise Agreement entered into by the Proposed New Owner.

14.7 Franchisor's Right of First Refusal.

If you or any owner of the Franchisee wishes to Transfer an Interest, we will have a right of first refusal to purchase that Interest. The party proposing the Transfer (the "Transferor") must submit to us a written offer to sell the Interest, which offer shall include the price and other material terms and conditions of the offer. The Transferor must also provide us with any other information we need to evaluate the offer, if we request it, within seven days after receipt of the offer. We have the right, exercisable by delivering written notice to the Transferor within 15 days after the date of last delivery to us of the offer and any other documents we have requested, to purchase the Interest for the price and on the terms and conditions contained in the offer. We will have at least 60 days to prepare for closing and we will be entitled to all customary representations and warranties. If the proposed Transfer includes assets not related to the operation of the Restaurant, we may purchase only the assets related to the operation of the Restaurant for an equitable price.

If we do not exercise our right of first refusal, the Transferor may offer that Interest and complete the sale to a Proposed New Owner pursuant to and on the terms of the offer to us, as long as we have approved the Transfer as provided in Section 14.3 of this Section. You must immediately notify us of any changes in the terms of an offer to a Proposed New Owner. Any material change in the terms of an offer before closing will make it a new offer, revoking any previous approval or previously made election to purchase and giving us a new right of first refusal effective as of the day we receive formal notice of a material change in the terms. For purposes of this Section, a material change in an offer shall include, but not be limited to, a deviation of greater than 10% in the price you originally offered to us. If the sale to the Proposed New Owner is not completed within 120 days after we have approved the Transfer, our approval of the proposed Transfer will expire.

14.8 Public Offering.

Notwithstanding anything in this Section to the contrary, we will not withhold consent to a public offering of securities (debt or equity) of the Franchisee provided:

- (a) The parties who are the owners of the Franchisee prior to such offering retain control of more than 50% of the voting power of the Franchisee after the offering;
- (b) The Franchisee submits to us prior to the public offering the offering documents and such other information relating to the public offering as we may reasonably request;
- (c) The Franchisee agrees to reimburse us for our actual expenses incurred in connection with the offering or proposed offering (including reasonable attorneys' fees); and
- (d) The Franchisee complies with all our requirements in connection with such offering including, without limitation, adding appropriate disclaimers to the offering documents and execution of appropriate indemnification agreements indemnifying us against any costs or liabilities arising out of such public offering.

15. TERMINATION OF FRANCHISE

15.1 By Franchisee.

- (a) <u>Termination Due to our Material Breach</u>. If we breach this Agreement and fail to cure such breach within 90 days after written notice thereof is delivered to us, you may terminate this Agreement and the Franchise effective 10 days after delivery to us of notice of termination. However, if the nature of the breach is such that it cannot be cured within 90 days, you will not have the right to terminate if we promptly commence and continue diligent, good faith efforts to cure such breach.
- (b) <u>No Other Rights of Termination by You</u>. Except for a termination of this Agreement <u>and</u> the Franchise in full compliance with the requirements set forth in subpart (a) of this Section 15.1, you have no rights to terminate this Agreement or the Franchise prior to the expiration of the Term.

15.2 By Franchisor.

We shall have the right to terminate this Agreement if you:

- (a) Fail to open the Restaurant as provided in this Agreement;
- (b) Fail to obtain and maintain the certifications required pursuant to Article 4;
- (c) Transfer control of, lose the right to occupy the Premises through no fault of your own and you fail to locate a substitute site as provided in Section 10.3, or the lease for the Premises of the Restaurant is terminated because of your default thereunder;
 - (d) Attempt to make a Transfer without compliance with the provisions of Article 14;

- (e) Fail to follow our directions and guidelines concerning use of the Marks or otherwise misuse the Marks and fail to correct the failure or misuse within 10 days after notification from us;
- (f) Fail to comply with any other provision of this Agreement or any mandatory specifications, standard or operating procedure;
- (g) Fail to obtain our permission to relocate a Restaurant or to de-identify the vacated location;
 - (h) Fail to offer our Core Menu, or offer unapproved menu items;
- (i) Are found to be in violation of any applicable health, safety or sanitation laws, regulations or codes, by any governmental official, and fail to cure any such violation within 72 hours after receiving notice therefor;
- (j) File or otherwise commence litigation, arbitration, or any other legal action against us that is not in compliance with the dispute resolution terms agreed upon in Article 17 of this Agreement as may be modified by any applicable rider in Exhibit VI, and fail to dismiss such action within seven days after notification from us; or
- (k) Fail to abide by any other provision of the Agreement, provided you will have 30 days after receipt of written notice from us to cure any of the aforementioned defaults (unless a shorter time period is noted above); provided however, that termination by us will not be arbitrary or capricious when compared to our actions with respect to other franchisees in the same market area in similar circumstances.

Furthermore, we shall have the right to terminate this Agreement effective upon delivery of notice of termination to you, if:

- (l) You are declared bankrupt or judicially determined to be insolvent;
- (m) You voluntarily abandon the Franchise by failing to operate the Restaurant for five consecutive days without our prior written approval, unless the failure to operate is due to circumstances beyond your control, other than financial inability;
- (n) You or your owners have made or knowingly make any false or material misrepresentations, or have knowingly omitted or knowingly omit to state any material facts relating to the acquisition, ownership or operation of the Franchise, including but not limited to sales reporting;
- (o) The Restaurant or the Premises are lawfully seized, taken over, or foreclosed by a government authority or official;
- (p) You are convicted, or enter a no contest plea, for criminal misconduct which materially and adversely affects the operation, maintenance, or goodwill of the Restaurant;
- (q) You operate the Restaurant in a manner that imminently endangers the public health and safety;
- (r) You fail on three or more separate occasions in any three year period to comply with the material provisions of this Agreement, , whether or not such failures to comply are corrected after notice thereof is given to you;

- (s) In the event of a force majeure event as described in Section 18.7 below that occurs and continues for a period of six consecutive months or longer, and which prevents either party from performing its obligations hereunder;
- (t) We receive credible evidence, which we verify to our satisfaction, that you, your Designated Principal, or any other management level employee, has sexually harassed or intimidated any individual or intentionally engaged in any racial, ethnic, religious, sexual, or other offensive discrimination against any individual or group; or
- (u) Fail to pay when due any monies owed to us (unless such monies are the subject of a bona fide dispute subject to Section 9.7 hereof); provided that, you will have 10 days after receipt of written notice from us to cure.

15.3 Cross-Default and Cross Termination.

- (a) A default by you under this Agreement will be deemed a default of all agreements between you and/or any Franchisee Affiliate and Franchisor (the "Other Agreements"). A default by you and/or any Franchisee Affiliate under any of the Other Agreements will be deemed a default under this Agreement. A default by any guarantor(s) of this Agreement or of any of the Other Agreements will be deemed a default of this Agreement.
- (b) If this Agreement is terminated as a result of a default by you, we may, at our option, elect to terminate any or all of the Other Agreements. If any of the Other Agreements is terminated as a result of a default by you and/or any Franchisee Affiliate, we may, at our option, elect to terminate this Agreement. It is agreed that an incurable or uncured default under this Agreement or any of the Other Agreements will be grounds for termination of this Agreement and/or any and all of the Other Agreements without additional notice or opportunity to cure.

16. RIGHTS AND OBLIGATIONS OF FRANCHISOR AND FRANCHISEE UPON TERMINATION OR EXPIRATION OF FRANCHISE

16.1 Payment of Amounts Owed to Franchisor.

You agree to pay to us within five days after the effective date of termination or expiration of the Franchise all fees and all other amounts due us pursuant to this Agreement or any other agreement, whether written or oral, between the parties. In the event of a termination due to a default by you, you shall also pay to us, as liquidated damages and not as a penalty, an amount equal to the Royalty that would have been due from you in accordance with Section 9.2 of this Agreement based on the monthly average of the Net Sales (i) over the past 60 full months of operation of the Restaurant prior to the date of default, (ii) from the date of commencement of operation of the Restaurant if the Restaurant had not been in operations for a full 60 months prior to the date of default or (iii) for the entire term of operation of the Restaurant by you if you did not own the franchise for a full 60 months prior to the date of default, multiplied by the number of months remaining on the term of this Agreement but not to exceed 24.

16.2 Marks.

You and your owners agree that after the termination or expiration of the Franchise, you will:

- (a) Not directly or indirectly at any time identify any business with which you are associated as a current or former Taco John's Restaurant or franchisee;
- (b) Not use any Mark or any imitation of any Mark in any manner or for any purpose, or use for any purpose any trademark or other commercial symbol that suggests or indicates an association with us;
- (c) Return to us or destroy (whichever we specify) all menus, forms and materials containing any Mark or otherwise relating to a Taco John's Restaurant;
- (d) Remove all Marks affixed to uniforms or, at our direction, cease to use those uniforms;
- (e) Remove all Marks from any of your websites or other Internet presence, discontinue all use of any domain name which contains the Marks and, upon our request, transfer it to us; remove your listing from all Internet directories; and notify all search engines that you are no longer part of the System; and
- (f) Take any action that may be required to cancel all fictitious or assumed name or equivalent registrations relating to your use of any Mark.

16.3 De-Identification.

If you retain possession of the Premises or if required by the terms of your lease, you agree to completely remove or modify, at your sole expense, any part of the interior and exterior decor that we deem necessary to disassociate the Premises with the image of a Taco John's Restaurant and to ensure that the location is not recognizable as a Taco John's Restaurant, including painting and removing any trade dress and signage bearing the Marks. If you do not take the actions we request within 30 days after notice from us, we have the right to enter the Premises and make the required changes at your expense, and you agree to reimburse us for those expenses on demand.

16.4 Confidential Information.

You agree that on termination or expiration of the Franchise, you will immediately cease to use any of the Confidential Information. You further agree to immediately return to us all copies of the Operations Manual and any other confidential materials which we have loaned to you.

16.5 Covenant Not to Compete.

On termination or expiration of this Agreement in accordance with its terms, the Franchisee and the owners agree that, for a period of two years after the effective date of termination or the date on which you start complying with this covenant, whichever is later, neither you nor your owners will have any direct or indirect interest (through a member of your immediate Family or that of an owner, or otherwise) as a disclosed or beneficial owner, investor, manager, or consultant, in any Competitive Business (as defined in Section 8.3) which is located or operating within five miles of the Protected Territory or within five miles of the Protected Territory of any other Taco John's Restaurant.

The prohibitions of this Section will not apply to the ownership of the 1% or less of a publicly traded company.

16.6 Franchisor's Option to Purchase Restaurant.

Upon the termination or expiration of the Franchise, we shall have the option, but not the obligation, exercisable for thirty (30) days after written notice to you, to purchase at fair market value all of the assets of the Restaurant including all approved equipment, fixtures, furniture and signs, and all utensils, supplies, inventory, materials and other items imprinted with any Mark.

Additionally, if you or a Franchisee Affiliate owns the Premises, we may elect, at our option (but with no obligation to do either), to (i) purchase the Premises (including all improvements thereon) at its fair market value and on commercially reasonable terms, or (ii) lease the Premises from you or the Franchisee Affiliate for an initial five year term with one (1) renewal term of five years, at fair market value and on commercially reasonable terms. We will notify you of our election to purchase or lease the Premises within the same 30 day period in which we will notify you of our election to purchase the assets of the Restaurant.

For purposes of determining the fair market value of any assets or property, the goodwill attributable to the Marks and System shall not be considered. If we cannot agree with you on the fair market value of the Premises or assets of the Restaurant or on commercially reasonable terms for the transaction within a reasonable time, such value and terms shall be determined by an independent appraiser who shall have been selected jointly by you and us. The expenses of the appraiser will be split equally between us. We shall not assume any liabilities, debts or obligations of the Restaurant in connection with any such transfer and you shall indemnify us from any and all claims made against us arising out of any such transfer of the assets of the Restaurant. The parties shall comply with all applicable laws in connection with any such transfer and you shall cooperate with us in complying with all such requirements. If we elect to exercise the option to purchase (and/or to lease the Premises), we may set off all amounts you owe us under this Agreement against any payments for the purchase (and/or lease). You agree to cause any Franchisee Affiliate to comply with these requirements.

The terms and conditions of this right and option to purchase may be recorded, if deemed appropriate by us, in the real property records and we or our designee and you (or the applicable Franchisee Affiliate) shall execute such additional documentation as may be necessary and appropriate to effectuate such recording.

We will have the unrestricted right to assign our option to purchase under this Section 16.6 and our right to assume the lease under Section 16.7 at any time prior to the closing of those transactions. At the closing of the purchase of the Restaurant, as above provided, both you and we will execute and deliver all documents necessary to vest title in us or our nominee free and clear of all liens and encumbrances. We shall have the right to set off against the purchase price of the assets of the Restaurant all amounts due to the Franchisor under this Agreement or any other agreement between the parties.

16.7 Franchisor's Right to Assume Lease.

If the Premises is leased, then upon termination or expiration of the Franchise, we will have the option (but not the obligation) to require you to assign to us, or another franchisee we designate, your interest in any lease for the Premises, and you will vacate the Restaurant promptly and completely, rendering all necessary assistance to us to enable us to take prompt possession. We may exercise this option by providing you with our notice that we intend to assume the lease within 30 days of the expiration or termination.

16.8 Franchisor's Right to Operate Restaurant.

If we terminate this Agreement pursuant to Section 15.2, we will have the right to immediately enter and take possession of the Restaurant to maintain continuous operation of the Restaurant, provide for orderly change of management and disposition of personal property, and otherwise protect our interests. If we exercise this right, you will vacate the Restaurant promptly and completely, rendering all necessary assistance to us to enable us to take prompt possession, and you will have no right to any revenue that we earn while operating the Restaurant. If you dispute the validity of our termination of this Agreement, we will nevertheless have the option, which you irrevocably grant, to operate the Restaurant pending the final, unappealed determination of the dispute under this Agreement. If an arbitrator or court of competent jurisdiction makes a final, unappealed determination that the termination was not valid, we will make a full and complete accounting for the period during which we operated the Restaurant, except that we will be entitled to retain amounts equal to the Royalty and all other fees that would have been payable to us under this Agreement had you still been operating the Restaurant and our out-of-pocket expenses incurred in operating the Restaurant. You acknowledge that we will have a duty to utilize only reasonable efforts and will not be liable to you or your owners for any debts, losses, or obligations the Restaurant incurs, while we manage the Restaurant.

16.9 Continuing Obligations.

All obligations of this Agreement (whether yours or ours) which expressly or by their nature survive the expiration or termination of this Agreement will continue in full force and effect after and notwithstanding its expiration or termination until they are satisfied in full or by their nature expire.

17. ENFORCEMENT

17.1 Invalid Provisions; Substitution of Valid Provisions.

In the event that any arbitrator or court of competent jurisdiction determines that any provision of this Agreement, including but not limited to any of the restrictive covenants contained either Section 8.3 or Section 16.5 hereof, are unenforceable as written for any reason, including for purposes of the restrictive covenants, reasons that the areas of restriction exceed the reasonable maximum time period, geographic area or scope, then the parties hereby request and authorize the arbitrator or court to "blue pencil" such provision so as to make it enforceable and to best carry out the intent of the parties, or to deem such provision severed from this Agreement if it cannot be so modified. The holding, declaration or pronouncement shall not adversely affect any other provisions of this Agreement, which shall otherwise remain in full force and effect.

If any lawful requirement or court order of any jurisdiction (a) requires a greater advance notice of the termination or non-renewal of this Agreement than is required under this Agreement, or the taking of some other action which is not required by this Agreement, or (b) makes any provision of this Agreement or any specification, standard or operating procedure we prescribed invalid or unenforceable, the advance notice and/or other action required or revised version of the specification, standard or operating procedure will be substituted for the comparable provisions of this Agreement in order to make the modified provision enforceable to the greatest extent possible. You agree to be bound by the modification to the greatest extent lawfully permitted.

17.2 Unilateral Waiver of Obligations.

Either of us may, by written notice, unilaterally waive or reduce any obligation or restriction of the other party under this Agreement. The waiver or reduction may be revoked at any time for any reason on 10 days' prior written notice.

17.3 Written Consents from Franchisor.

Whenever this Agreement requires our advance approval or consent, you agree to make a timely written request for it. Our approval or consent will not be valid unless it is in writing.

17.4 Reasonableness.

We both agree to act reasonably in all dealings with each other pursuant to this Agreement. Whenever the consent or approval of either party is required or contemplated hereunder, the party whose consent is required agrees not to unreasonably withhold the same, unless such consent is expressly subject to such party's sole discretion pursuant to the terms of this Agreement.

17.5 Duty of Good Faith.

This Agreement imposes upon both parties a duty of good faith in performance of this Agreement. "Good faith" as used herein means honesty in fact and the observance of reasonable standards of fair dealing in the restaurant industry.

17.6 No Guarantees.

If in connection with this Agreement, we provide to you any waiver, approval, consent, or suggestion, or if we neglect or delay our response or deny any request for any of those, we will not be deemed to have made any warranties or guarantees which you may rely on, and will not assume any liability or obligation to you.

17.7 No Waiver.

If at any time we do not exercise a right or power available to us under this Agreement or do not insist on your strict compliance with the terms of this Agreement, or if there develops a custom or practice which is at variance with the terms of this Agreement, we will not be deemed to have waived our right to demand exact compliance with any of the terms of this Agreement at a later time. Similarly, our waiver of any particular breach or series of breaches under this Agreement or of any similar term in any other agreement between the Franchisor and any franchisee will not affect our rights with respect to any later breach. It will also not be deemed to be a waiver of any breach of this Agreement for us to accept payments which are due to us under this Agreement. If, after expiration or termination of this Agreement, you continue to operate the Franchise or to perform your obligations under this Agreement, it will not be deemed to be a waiver by us or to constitute an implied contract or extension of this Agreement regardless of whether we accept payments, provide support to you, or otherwise perform our obligations.

17.8 Cumulative Remedies.

The rights and remedies specifically granted to either you or us by this Agreement will not be deemed to prohibit either of us from exercising any other right or remedy provided under this Agreement or permitted by law or equity.

17.9 Specific Performance; Injunctive Relief.

Notwithstanding Section 17.10 below, each party reserves the right to seek injunctions or orders of specific performance in a proper case from a court of competent jurisdiction without being required to post a bond.

17.10 Arbitration.

Except insofar as we elect to enforce this Agreement pursuant to Section 17.9, and except for claims relating to the Marks, or to enforce Section 16.5 of this Agreement, all controversies, disputes or claims arising between the Franchisor, its Affiliates, officers, directors, agents, employees and attorneys (in their representative capacity) and the Franchisee, the owners and any guarantors of the Franchisee's obligations, arising out of or related to:

- (a) This Agreement or any provision thereof or any related agreement (except for any sublease with any Affiliates of the Franchisor);
 - (b) The relationship of the parties hereto;
- (c) The validity of this Agreement or any related agreement, or any provision thereof; or
- (d) Any specification, standard or operating procedure relating to the establishment or operation of the Franchise shall be subject to the following dispute resolution process:
- Arbitration. All controversies, disputes or claims which are subject to this Section 17.10 shall be submitted for binding arbitration by either party. The arbitration will be administered by the American Arbitration Association. If the American Arbitration Association should cease doing business during the term of this Agreement, including any renewals, we will notify you of our selection of another arbiter group. Such arbitration proceedings shall be conducted before one arbitrator who is a member of the American Arbitration Association Panel of Franchise Arbitrators and shall be conducted in Minneapolis, Minnesota. Except as otherwise provided in this Agreement, the arbitration shall be conducted in accordance with the then-current commercial arbitration rules of the American Arbitration Association. The decision as to whether a claim is subject to mandatory arbitration shall be made by an arbitrator, not a court. The arbitrator shall have the right to award or include in the award any relief available and appropriate under the applicable law (as set forth in Section 17.16) and this Agreement. The award and decision of the arbitrators shall be conclusive and binding upon all parties to the arbitration and on no other parties, and judgment upon the award may be entered in any court of competent jurisdiction. The parties further agree that in connection with any such arbitration proceeding they shall file any compulsory counterclaim (as defined by Rule 13 of the Federal Rules of Civil Procedure) within 30 days after the date of the filing of the claim to which it relates.

The parties agree that arbitration shall be conducted on an individual basis. The parties agree further that arbitration shall not be conducted on a class-wide basis. In the event you control, are controlled by, or are in active concert with another Taco John's franchisee, or if there is a guarantor of some or all of your obligations to us, then we may join those parties to any arbitration allowed pursuant to this Section. The officers, directors, managers, partners or other agents or representatives of both you and us also may be joined in the arbitration.

This Section 17.10 shall continue in full force and effect subsequent to and notwithstanding the expiration or termination of this Agreement.

17.11 Franchisee's Right to Associate.

We will not restrict your ability to join and participate in any trade association of Taco John's franchisees nor will we retaliate against you for joining or participating in such an association. We will regularly consult with any such association which annually provides us with documentation demonstrating that it represents 50% or more of the then-current total of both franchised Taco John's Restaurants and Taco John's franchisees.

17.12 Limitations of Claims.

Any and all claims arising out of or relating to this Agreement or the relationship among the parties hereto shall be barred unless an action or legal or arbitration proceeding is commenced by the party asserting such claim within one year from the date that such party knew or should have known of the facts giving rise to the claim.

17.13 Binding Effect.

This Agreement is binding on and will inure to the benefit of our successors and assigns and will be binding on and inure to the benefit of your successors and assigns, and if you are an individual, on and to your heirs, executors and administrators.

17.14 Entire Agreement.

- (a) This Agreement, together with the introduction and exhibits to it, constitutes the entire agreement between us, and there are no other oral or written understandings or agreements between us concerning the subject matter of this Agreement. Nothing in this Agreement is intended to disclaim any representations made by us in the Franchise Disclosure Document delivered to you.
- (b) This Agreement may be modified only by written agreement signed on behalf of the Franchisee by all named individual Franchisees and/or by the authorized officer or agent if the named Franchisee is an entity, and on behalf of us by our President or CEO, or, at our option, upon notice of the approval of a Super-Majority as defined in Section 17.14(c) below.
- (c) This Agreement may be modified by us at our option whenever we and a Super-Majority, as hereinafter defined, of our franchisees agree to any such modification. A "Super-Majority" of our franchisees shall consist of the owners of at least 75% of all Taco John's Restaurant franchises, or, if only a portion of Taco John's Restaurant franchises are affected by the modification, at least 75% of those Taco John's Restaurant franchises affected by the modification. Whenever a modification is approved by a Super-Majority, we may elect to treat the modification as effective to all franchisees or the applicable group thereof, including you, to the same extent and in the same manner as if the modification was unanimously approved by them, and regardless of whether you may or may not desire to be bound by the modification. We shall provide you with notice of any modification to this Agreement based on a Super-Majority approval at least 30 days prior to the date such modification is to be effective. By signing this Agreement, you appoint our officers as your attorneys in fact with irrevocable power and authority to execute any such modification so approved.

17.15 No Liability to Others; No Other Beneficiaries.

We will not, because of this Agreement or by virtue of any approvals, advice or services provided to you, be liable to any person or legal entity who is not a party to this Agreement, and no other party shall have any rights because of this Agreement.

17.16 Governing Law.

All disputes to be arbitrated will be governed by the Federal Arbitration Act (the "FAA"). Except to the extent governed by the FAA and the United States Trademark Act of 1946 (Lanham Act, 15 U.S.C. Sections 1051 et seq.) or other federal law, this Agreement shall be interpreted under the laws of the state of Wyoming and any dispute between the parties shall be governed by and determined in accordance with the substantive laws of the state of Wyoming, which laws shall prevail in the event of any conflict of law.

18. MISCELLANEOUS

18.1 Construction and Definitions.

All headings of the various sections and paragraphs of this Agreement are for convenience only and do not affect the meaning or construction of any provision. All references in this Agreement to masculine, neuter or singular usage will be construed to include the masculine, feminine, neuter or plural, wherever applicable. The term "Franchisee Affiliate" is any of your direct or indirect owners and any entity in which you or any of your owners own an interest, whether directly or indirectly. The term "Affiliate" as used in this Agreement is applicable to any company directly or indirectly owned or controlled by us. The term "Control" or "Controlling Interest," as used in this Agreement shall mean 51% or more of the general partnership interest of a partnership or the voting power of a corporation or limited liability company. The term "Immediate Family" as used in this Agreement means a person's spouse, parents, siblings or natural or adopted children.

18.2 Joint and Several Liability.

If two or more persons are the Franchisee under this Agreement, their obligation and liability to us shall be joint and several.

18.3 Multiple Originals.

This Agreement will be executed using multiple copies, each of which will be deemed an original.

18.4 Timing Is Important.

Time is of the essence of this Agreement. This means that it will be a material breach of this Agreement to fail to perform any obligation within the time required or permitted by this Agreement.

18.5 Independent Provisions.

The provisions of this Agreement are deemed to be severable. In other words, the parties agree that each provision of this Agreement will be construed as independent of any other provision of this Agreement.

18.6 Delegation of Duties.

We have the right to delegate the performance of any portion or all of our obligations and duties hereunder to third parties from time to time.

18.7 Force Majeure.

Neither of us will be liable for loss or damage or deemed to be in breach of this Agreement if a failure or delay in performance results from: (a) transportation shortage, inadequate supply or unavailability from manufacturers or suppliers of equipment, food products, supplies, labor, material or energy, or the voluntary surrender of the right to acquire or use any of the foregoing in order to accommodate or comply with any order, request, regulation, recommendation or instruction of any federal, state or municipal government; (b) compliance with any law, ruling, order, regulation, requirement or instruction of any federal, state or municipal government; (c) Act of God; (d) viral or bacterial epidemic, declared pandemic, or other declared public health crisis; (e) fire, strike embargo, war, terrorism, riot, hurricane, tornado, earthquake; or (f) other similar event or cause beyond the control of either party whose performance was prevented or delayed. Any delay resulting from any of these causes shall extend the time for performance or excuse performance, as may be reasonable, except that the causes shall not excuse payments of amounts owed at the time of the occurrence or payment of any amount due thereafter. A party seeking relief under this Section shall as soon as practicable notify the other party of its inability to perform.

19. NOTICES

All notices required or permitted to be given under this Agreement will be given in writing and deemed delivered by hand delivery; by certified mail with return receipt requested; by express mail or overnight delivery service that provides evidence of delivery or attempted delivery; or by e-mail or other electronic delivery system. When made to the Franchisee, notice will be addressed to the address or e-mail address set forth on the Addendum or, if none, to the registered agent or Restaurant address, at our option. If to the Franchisor, notice will be addressed to its principal place of business or the following e-mail address: tjilegal@tacojohns.com. Either party may change its address or e-mail address for notice from time to time by written notice to the other party in accordance with this Article. Notice will be deemed given three days after depositing in the United States mail, postage prepaid; one day after depositing in express mail or overnight delivery; when received by hand delivery; or with proof of delivery by e-mail or other electronic delivery system.

The parties to this Agreement now execute and deliver this Agreement in multiple counterparts on the dates indicated below. This Agreement shall not become effective until the Effective Date set forth in the Addendum.

TACO JOHN'S INTERNATIONAL, INC.	FRANCHISEE:
By: Hasthan Lord Noony President & CEO	By:
Heather Leed Neary, President & CEO	Nama:
Date:	Name:(Print Name)
	Its:
	(Print Title)
	Date:

EXHIBIT I

ADDENDUM TO TACO JOHN'S INTERNATIONAL, INC. FRANCHISE AGREEMENT

This is an Addendum ("Addendum") to the Franchise Agreement (the "Agreement") by and between you, the undersigned franchisee, and Taco John's International, Inc. This Addendum modifies the terms of the Agreement and in the event of a conflict in terms between the Agreement and this Addendum, the terms of this Addendum shall be controlling.

	The parties agree as follows:
1. this Agreement	Concept and Type of Restaurant. The concept and type of Restaurant to be operated under pursuant to Section 2.1 of the Agreement, is as follows:
	Traditional:
	☐ Freestanding Building
	☐ Endcap
	☐ DT Digital Olé
	☐ DT Digital Olé+
	Non-Traditional: (if checked, Exhibit V will apply)
2. and Section 2.1	Restaurant Location. The location of the Restaurant, referenced in the header paragraph of the Agreement, is
3. Agreement, sha	<u>Initial Term</u> . The Initial Term of the Agreement, as referenced in Section 2.1 of the all be as follows:
	20 years from the Effective Date of this Agreement
	☐ 10 years from the Effective Date of this Agreement
4. the Agreement	<u>Protected Territory</u> . The Protected Territory of the Restaurant referenced in Section 2.3 of is (check one):
	☐ Not Applicable (Non-Traditional only).
5. Agreement, will Agreement.	<u>Initial Franchise Fee</u> . The Initial Franchise Fee, as referenced in Section 9.1 of the ll be the sum of \$, which is payable in full upon execution of this
6. opening marke	Grand Opening Marketing. The required amount Franchisee must spend on a grand ting plan, as referenced in Section 9.5 of the Agreement, will be the sum of \$

7.	Effective Date. The Effective Date	of this Agreement is
8.	Expiration Date: The expiration dat	e of this Agreement is
9. any Notice red	Notices. Pursuant to Section 19 of quired or permitted to be given under	the Agreement, Franchisee's contact information for the Agreement is:
	Franchisee Name:	
	Contact Person:	
	Address	
10.	Other terms (if appropriate).	
Executed by the	ne parties on the date indicated below	their respective signatures.
TACO JOHN	S'S INTERNATIONAL, INC.	FRANCHISEE:
By:	Neary, President & CEO	By:
	•	N
Date:		Name:(Print Name)
		Its:(Print Title)
		(Print Title)
		Date:

EXHIBIT II

AUTHORIZATION AGREEMENT FOR DIRECT PAYMENTS



AUTHORIZATION AGREEMENT FOR DIRECT PAYMENTS (ACH)

Partnership, Corporation, Lin	mited Liability Company)
	Zip Code:
rity is to remain in full force a in such time and manner as	or the original debit amount plus any and effect until TJI has received written to afford TJI a reasonable opportunity
Date:	
Date:	
	tional, Inc. to initiate variable tion identified. In the event thate a debit to the account for the initial force a in such time and manner as a Date:

Please

- Complete and sign form
- Attach a copy of a voided check
- Return by email to: accounting@tacojohns.com

EXHIBIT III

STATEMENT OF OWNERSHIP

CURRENT STRUCTURE, OWNERSHIP AND MANAGEMENT OF FRANCHISEE

Franchisee:	City/	ST:	
☐ Individual ☐ Partne	Form of Ov (Check ership Limited Liability	-	Other (Describe
partner, showing perce		ide the name and address of nat person is active in ma s formed.	
Date of formation:	State	of formation:	
NAME	ADDRESS	PERCENT OWNED	ACTIVE IN MANAGEMENT
(continue on additional she	eets as necessary)		
	e names, addresses and own active in management.	of business entity, give the nership interest percentage of of organization:	f each member, and
MEMBER NAME	ADDRESS	PERCENT OF OWNERSHIP	ACTIVE IN MANAGEMENT
(continue on additional she	eets as necessary)		

Date of incorporation:	State	of incorporation:	
STOCK OWNER NAME	ADDRESS	PERCENT OF SHARES OWNED	ACTIVE IN MANAGEMENT
(continue on additional shee	ets as necessary)	l.	
(continue on additional since	cts as necessary)		
	above information must be rep	ported to Company in writing.	
Any and all changes to the a Below are the names, Franchisee's owners, pri	above information must be reparted addresses, titles, and telencipal officers, or partners	elephone numbers (if applications, and Above Store Leaders of the franchised business)	r other individuals
Any and all changes to the a Below are the names, Franchisee's owners, pri	above information must be reparted addresses, titles, and telencipal officers, or partners	elephone numbers (if applices, and Above Store Leaders of	r other individuals
Any and all changes to the a Below are the names, Franchisee's owners, pri who will devote their ful	addresses, titles, and te incipal officers, or partners ill-time efforts to the opera	elephone numbers (if applices, and Above Store Leaders of the franchised business	r other individuals s.
Any and all changes to the a Below are the names, Franchisee's owners, pri who will devote their ful	addresses, titles, and te incipal officers, or partners ill-time efforts to the opera	elephone numbers (if applices, and Above Store Leaders of the franchised business	r other individuals s.
Any and all changes to the a Below are the names, Franchisee's owners, pri who will devote their ful	addresses, titles, and te incipal officers, or partners ill-time efforts to the opera	elephone numbers (if applices, and Above Store Leaders of the franchised business	r other individuals s.
Any and all changes to the a Below are the names, Franchisee's owners, pri who will devote their ful	addresses, titles, and te incipal officers, or partners ill-time efforts to the opera	elephone numbers (if applices, and Above Store Leaders of the franchised business	r other individuals s.

Signature

Print name (and title if applicable)

Date

EXHIBIT IV

GUARANTY

GUAL	KANI Y
or "us") of the Franchise Agreement (the "Agreement (the "Franchisee"), for a franchise to be located at/ir for purposes of this Guaranty only) hereby person successors and assigns that the Franchisee will pur agreement and covenant set forth in the Agreement as	the execution by Taco John's International, Inc. ("we" ent") with, each of the undersigned ("you," hally and unconditionally (1) guarantees to us and our nctually pay and perform each and every undertaking, and (2) agrees to be personally bound by, and personally the Agreement including, without limitation, monetary om taking certain actions.
obligations guaranteed by this Guaranty; (2) any ri against the Franchisee or any other person as a c reimbursement from, or subrogation against, the	ght you may have to require that an action be brought condition of your liability; (3) all right to payment or Franchisee which you may have arising out of your y and all other legal or equitable defenses to which you
and several; (2) you will make any payment or ren demand if the Franchisee fails or refuses to do so will conditioned on our pursuit of any remedies against will not be diminished, relieved or otherwise affect which we may from time to time grant to the Franchisthe acceptance of any partial payment or performant addition of any rider to the Agreement or any of	and immediate liability under this Guaranty shall be joint der any performance required under the Agreement on hen required; (3) your liability will not be contingent or to the Franchisee or any other person; (4) your liability and by any extension of time, credit or other indulgence isee or to any other person including, without limitation, are, or the compromise or release of any claim or by the other amendment to the Agreement accepted by the dispersion of the Agreement and igations under the Agreement.
	covenants and confidentiality provisions contained in e indemnification provisions contained in Section 7.3 of
such proceeding, we shall be entitled to reimbursem to, reasonable accountants', attorneys', attorneys' a investigation and proof of facts, court costs, othe whether incurred prior to, in preparation for or in co	in a judicial or arbitration proceeding, and prevail in the ent of our costs and expenses including, but not limited assistants', arbitrators' and expert witness fees, costs of er litigation expenses and travel and living expenses, ontemplation of the filing of any such proceeding. If we with any failure by you to comply with this Guaranty, osts and expenses incurred by us.
This Guaranty is now executed as of the Eff	fective Date of the Agreement.
GUARANTOR	GUARANTOR
GUARANTOR	GUARANTOR

EXHIBIT V

NON-TRADITIONAL FACILITY RIDER

This Non-Traditional Facility Rider ("Rider") is m	ade this day of,
by and between	("Franchisee") and Taco John's International,
Inc. ("Franchisor").	
In the event of any conflicts between the, (the "Franchise Agreement' Rider shall control. This Rider amends and supplements condition contained therein remain in full force and effect, set forth below.	') and the terms of this Rider, the terms of this the Franchise Agreement, and all terms and
1. The third paragraph of Article 1 of the Francinserted in its place:	nchise Agreement is deleted and the following

This Agreement is being presented to you because you desire to obtain the right to develop, own and be franchised to operate a Taco John's Restaurant at a non-traditional location, as described below. You acknowledge that the signs, equipment installation and configuration, size and appearance of this Restaurant may differ from that of a traditional free-standing Taco John's restaurant.

2. The following is added to Section 2.1 of the Franchise Agreement:

You agree that this grant is for a Restaurant to be located at a non-traditional location. For purposes of this Agreement, the Premises shall include the real estate, site, building, improvements, and the Common Areas. "Common Areas" means the areas of the Premises which may be shared by the Restaurant and the other businesses operated on the Premises, if any, including but not limited to seating areas, parking, restrooms, garbage and storage areas.

- Sections 2.3(a), 2.3(b), 2.3(c), and 2.3(d) are deleted in their entirety. It is understood that Franchisee shall not receive any Protected Territory or Right of First Refusal Area, or any other kind of exclusivity rights, for its Restaurant. Unless otherwise deleted or modified in this Rider, all references to a "Protected Territory" throughout the Franchise Agreement shall mean the Premises.
 - 4. Section 2.4(a)(ii) is deleted with the following substituted in its place:
 - (ii) you are unable to maintain possession of the Premises and you secure substitute premises at a non-traditional location which comply with the specifications and standards then applicable under new or renewal franchises for Taco John's Restaurants to be operated at a non-traditional location and which we have accepted:

5. Section 7.3 of the Franchise Agreement is deleted with the following inserted in its place:

7.3 Indemnification.

We will not assume any liability or be deemed liable for any agreements, representations, or warranties you make that are not expressly authorized under this Agreement, nor will we be obligated for any damages or contractual liabilities to any person or property directly or indirectly arising out of the possession, ownership or operation of the Restaurant, the Common Areas and the other businesses operated on the Premises, or for any claims or demands for damages to property, or for injury, illness or death of persons directly or indirectly resulting therefrom, whether or not caused by your negligent or willful action or failure to act. We will have no liability for any taxes levied against you or your assets or on us in connection with the business you conduct, or any payments you make to us pursuant to this Agreement or any Franchise Agreement (except for our own income taxes). You agree to indemnify us, defend and hold us, our Affiliates, shareholders, directors, officers, employees, agents and their respective assignees, harmless against and to reimburse us for all such obligations, damages, and taxes for which we are held liable and for all costs we reasonably incur in the defense of any such claim brought against us or in any such action in which we are named as a party, including, without limitation, actual and consequential damages, reasonable attorneys', accountants' and expert witness fees, cost of investigation and proof of facts, court costs, other litigation expenses and travel and living expenses. However, your obligation to indemnify, defend and hold us harmless will not apply to any damages, costs or expenses which result from any actions taken by you in strict compliance with the then-current Operations Manual or any other then-current specification, standard, procedure, guidance, assistance or instruction we have given you or which result from our or our agents' negligence or willful or reckless misconduct. Your indemnification obligations described above will continue in full force and effect after the expiration or termination of this Agreement. We have the right to defend any such claim against us.

6. The first paragraph of Section 9.3 of the Franchise Agreement is deleted with the following inserted in its place:

9.3 Advertising and Marketing Fee.

You agree to pay to us for deposit in an Advertising Fund an "Advertising and Marketing Fee" equal to 2% of the Net Sales of the Restaurant. The Advertising and Marketing Fee is payable in the same manner as the Royalty. The Advertising and Marketing Fee will be used in the manner described in Article 11.

7. The following paragraphs are added to Section 10.2 of the Franchise Agreement:

If the Restaurant shares a common storage area with any other business operated on the Premises, you will maintain the common storage area in accordance with our standards for cleanliness and repair. If there is only one restroom facility located on the Premises, it shall be maintained in accordance with our standards for cleanliness and repair.

If the Premises include another restaurant concept, the drive-thru window, if any, on the Taco John's side of the Premises shall only service Taco John's customers with Taco John's approved products. If the other concept has a drive-thru window, it may not be used to distribute Taco John's products.

8. Section 10.3 of the Franchise Agreement is deleted with the following inserted in its place:

Provided that you are in substantial compliance with the provisions of this Agreement, if you lose the right to possession of the Premises through no fault on your part or if the Premises are destroyed, condemned or otherwise rendered unusable, or if you request the relocation of your Restaurant and, in our reasonable judgment, the existing site for the Restaurant does not meet our then-current standards and specifications for sites for Taco John's Restaurants, we will grant permission for the relocation of the Restaurant to an alternate non-traditional location and site which meets our then-current site criteria for Taco John's Restaurants located at non-traditional locations and is approved by us. If we agree on the relocation of the Restaurant, you agree to expeditiously develop the alternate site in compliance with our then-current specifications and requirements for Taco John's Restaurants, all at your sole cost and expense, and to open the relocated Restaurant within 150 days after receiving our permission to relocate, unless otherwise agreed to by us. You agree to de-identify, to our standards, the vacated location within 30 days of relocation. You shall pay us a relocation fee of 25% of the then-current Initial Franchise Fee for Taco John's Restaurant franchisees.

9. Section 10.4(i) of the Franchise Agreement is deleted with the following inserted in its place:

(i) You agree to purchase and maintain, at your own cost and expense from an approved vendor, the specific electronic cash register, freestanding kiosks, kitchen video monitors, receipt printers, and a drive-thru speed of service timing system and headset system if your Restaurant has a drive-thru (the "Point of Sale System"), along with the computer hardware, software, Internet connections and service, required dedicated power lines, and other computer related accessories, peripherals and equipment necessary to support the Point of Sale System, mobile ordering and delivery platforms that utilize geofencing services (if applicable), firewall and security service, and the required back of house information system (which includes the specific computer and software required to manage inventory, purchasing and other back of house functions) we specify from time to time (collectively, with the Point of Sale system, the "Computer System") with at least the minimum specifications we require from time to time for use in operating the Restaurant. Your Computer System must be set up in accordance with our specifications to record all sales information in connection with the Restaurant and to comply with your reporting requirements to us. Your Computer System must also be capable of processing credit card and debit card payment and complying with PCI DSS standards as described in Section 10.4(h) and be able to receive sales from mobile ordering, third-party delivery, and catering (if applicable). We may require you to upgrade, at your expense, any part or all of the Computer System when deemed necessary by us. We may also change the required Point of Sale System or other elements of the Computer System from time to time in our discretion, and you will be required to acquire any updated system at your sole expense. You must enter into any contracts required by the company that oversees these systems and services and you must pay the fee for those systems and services to us or the supplier, as we designate. We require you to use a financial reporting system provided by us or our designated supplier that allows you to upload and deliver your Net Sales reports, Profit and Loss statements, and other data, information, and records we require. There is currently no fee due for this system, but we may impose a fee upon one hundred eighty (180) days' notice, in which event you must pay the fee charged for that service to us or the supplier, as we designate. The fee for the financial reporting system, once established, is subject to

change upon notice from us or the supplier. You have the option to purchase interior digital menu boards from our approved third-party supplier. You will be required to pay the license fee for the digital menu boards, which may be charged directly by the supplier or which we may collect on behalf of the supplier. For any non-compliance with this subsection 10.4(i), you may be subject to a non-compliance fee as set forth in Section 9.11.

10. The following paragraph is added at the end of Section 10.4 of the Franchise Agreement:

Employees (i) shall not wear Taco John's restaurant uniforms when working at any other business operated on the Premises or within the same facility or center, and (ii) shall not be assigned to work at the Restaurant and at any other business operated on the Premises or within the same facility or center contemporaneously during any shift of work hours.

- 11. Section 11.1(d) is deleted and the following inserted in its place:
- (d) If you are required to join a local or regional Advertising Cooperative, the Advertising Fund will contribute to the Advertising Cooperative in an amount at least equal to the lesser of the Advertising and Marketing Fee you have paid or 1% of your total net sales from time to time, to be used for media placement in your DMA, as defined below.
- 12. Section 16.5 is deleted with the following inserted in its place:

On termination or expiration of this Agreement in accordance with its terms, the Franchisee and the owners agree that, for a period of two years after the effective date of termination or the date on which you start complying with this covenant, whichever is later, neither you nor your owners will have any direct or indirect interest (through a member of your immediate Family or that of an owner, or otherwise) as a disclosed or beneficial owner, investor, manager, or consultant, in any Competitive Business (as defined in Section 8.3) which is located or operating within five miles of the Protected Territory or within five miles of the Premises of any other Taco John's Restaurant.

This Rider is effective as of the date first above set forth.

FRANCHISOR:	FRANCHISEE:
TACO JOHN'S INTERNATIONAL, INC.	
By: Heather Leed Neary, President & CEO	By:
Date:	Name:(Print Name)
	Its:(Print Title)
	Date:

EXHIBIT VI

RIDERS TO THE FRANCHISE AGREEMENT FOR SPECIFIC STATES

If any one or more of the following Riders to the Franchise Agreement for Specific States ("Riders") is checked as an "Applicable Rider" below, then that Rider shall be incorporated into the Franchise Agreement entered into by Taco John's International, Inc. and the undersigned Franchisee. To the extent any terms of an Applicable Rider conflict with the terms of the Franchise Agreement, the terms of the Rider shall supersede the terms of the Franchise Agreement.

APPLICABLE RIDER:	
☐ California ☐ North Dakota	
☐ Illinois ☐ South Dakota	
☐ Indiana ☐ Virginia	
☐ Minnesota ☐ Washington	
☐ New York ☐ Wisconsin	
TACO JOHN'S INTERNATIONAL, INC. FRANCHISEE:	
By: By: Heather Leed Neary, President & CEO Date: Name:	Name)
By: By: Heather Leed Neary, President & CEO Date: Name:	Name)

CALIFORNIA RIDER TO FRANCHISE AGREEMENT

1. No statement, questionnaire, or acknowledgment signed or agreed to by a franchisee in connection with the commencement of the franchise relationship shall have the effect of (i) waiving any claims under any applicable state franchise law, including fraud in the inducement, or (ii) disclaiming reliance on any statement made by any franchisor, franchise seller, or other person acting on behalf of the franchisor. This provision supersedes any other term of any document executed in connection with the franchise.

ILLINOIS RIDER TO FRANCHISE AGREEMENT

Illinois law governs the Franchise Agreement.

In conformance with Section 4 of the Illinois Franchise Disclosure Act, any provision in a franchise agreement that designates jurisdiction and venue in a forum outside of the State of Illinois is void. However, a franchise agreement may provide for arbitration to take place outside of Illinois.

Your rights upon Termination and Non-Renewal of an agreement are set forth in sections 19 and 20 of the Illinois Franchise Disclosure Act.

In conformance with section 41 of the Illinois Franchise Disclosure Act, any condition, stipulation or provision purporting to bind any person acquiring any franchise to waive compliance with the Illinois Franchise Disclosure Act or any other law of Illinois is void.

No statement, questionnaire, or acknowledgment signed or agreed to by a franchisee in connection with the commencement of the franchise relationship shall have the effect of (i) waiving any claims under any applicable state franchise law, including fraud in the inducement, or (ii) disclaiming reliance on any statement made by any franchisor, franchise seller, or other person acting on behalf of the franchisor. This provision supersedes any other term of any document executed in connection with the franchise.

INDIANA RIDER TO FRANCHISE AGREEMENT

- 1. <u>Conditions for Approval of Transfer.</u> The following statement is added at the end of Section 14.4(i):
 - , excluding only such claims as Franchisee may have that have arisen under the Indiana Franchise Disclosure Law and the Indiana Deceptive Franchise Practices Act.
- 2. <u>Termination of the Franchise, By Franchisor</u>. Section 15.2(k) is amended by adding the following language:
 - , except as provided in Indiana Code §23-2-2.7-1(7).

3. **Entire Agreement.** The first sentence of Section 17.14 is deleted in its entirety and the following is substituted in its place.

This Agreement, together with the introduction and exhibits to it and the Franchise Disclosure Document, constitute the entire agreement between us, and there are no other oral or written understandings or agreements between us concerning the subject matter of this Agreement.

4. **Governing Law.** Section 17.16 is deleted and the following is inserted in its place:

All disputes to be arbitrated will be governed by the Federal Arbitration Act (the "FAA"). Disputes related to a violation of the Indiana Franchises Act or the Indiana Deceptive Franchise Practices Act shall be governed thereby. Otherwise, except to the extent governed by the FAA and the United States Trademark Act of 1946 (Lanham Act, 15 U.S.C. Sections 1051 et seq.) or other federal law, this Agreement shall be interpreted under the laws of the state of Wyoming and any dispute between the parties shall be governed by and determined in accordance with the substantive laws of the state of Wyoming, which laws shall prevail in the event of any conflict of law.

MINNESOTA RIDER TO FRANCHISE AGREEMENT

1. <u>Limitation of Claims</u>. The following phrase is added at the beginning of Section 17.12:

Except for any claims arising under the Minnesota Statutes 1984, Chapter 80C, which will be governed thereby,

- 2. <u>Non-Solicitation</u>. Minnesota Statutes Section 181.991 prohibits a franchisor from restricting, restraining, or prohibiting in any way a franchisee from soliciting or hiring an employee of the franchisor or an employee of a franchisee of the same franchisor. Any such restrictions in Section 10.7(b) are hereby deemed deleted.
- 3. **Addition of Paragraphs.** The following is added to the Franchise Agreement:

Minnesota law provides franchisees with certain termination and non-renewal rights. Minn. Stat. §80C.14 Subd. 3, 4 and 5 require, except in certain specified case, that a franchisee be given 90 days' notice of termination (with 60 days to cure) and 180 days' notice for non-renewal of the Agreement. To the extent that any provisions of this Agreement are inconsistent with such Minnesota law, such law shall control.

Minn. Stat. §80C.21 and Minn. Rule 2860.4400J prohibit us from requiring litigation to be conducted outside Minnesota. In addition, nothing in the disclosure document or agreement can abrogate or reduce any of your rights as provided for in Minnesota Statutes, Chapter 80C, or your rights to any procedure, forum, or remedies provided for by the laws of the jurisdiction.

NEW YORK RIDER TO FRANCHISE AGREEMENT

1. <u>Initial Training and Certification</u>. The second to last sentence of Section 4.1(a) is deleted in its entirety and the following sentence is substituted in its place:

The Initial Training will include classroom instruction and restaurant operation training and will be furnished at a Taco John's Restaurant(s) location or in our Restaurant Support Center at a time we designate.

2. **Operations Manual.** Section 5.2(c) is amended by adding the following sentence:

Modifications to the Operations Manual will not unreasonably affect your obligations, including economic requirements, under the Agreement.

3. **Indemnification.** The following sentence is added at the end of Section 7.3:

However, you shall not be required to indemnify us for any liability which arises as a result of our breach of this Agreement or other civil wrongs committed by us.

4. <u>Transfer by Franchisor</u>. The following sentence is added at the end of Section 14.2:

However, no assignment will be made except to an assignee who, in our good faith judgment, is willing and able to assume our obligations under the Franchise Agreement.

5. **Transfer by Franchisee.** Section 14.4(i) is amended by adding the following language:

; provided however, that all rights enjoyed by you and any causes of action arising in your favor from the provisions of Article 33 of the General Business Law of the State of New York and the regulations issued thereunder shall remain in force; it being the intent of this proviso that the non-waiver provisions of GBL 687.4 and 687.5 be satisfied.

6. **Governing Law.** The following sentence is added at the end of Section 17.16:

The foregoing choice of law should not be considered a waiver of any right conferred upon Franchisee by the General Business Law of the State of New York, Article 33. This language has been included in this Agreement as a condition to registration. Franchisor and Franchisee do not agree with the above language and believe that each of the provisions of the Agreement, including all choice-of-law provisions, are fully enforceable. Franchisor and Franchisee intend to fully enforce all of the provisions of the Agreement and all other documents signed by them, including but not limited to, all venue, choice-of-law, arbitration provisions and other dispute avoidance and resolution provisions and to rely on federal pre-emption under the Federal Arbitration Act.

NORTH DAKOTA RIDER TO FRANCHISE AGREEMENT

1. **Transfer by Franchisee.** The following is added at the end of Section 14.4 (i):

(Any release executed in connection herewith will not apply to any claims that may arise under the North Dakota Franchise Investment Law.)

2. <u>Addition of Paragraphs.</u> The Franchise Agreement is amended by adding the following paragraphs:

The Securities Commission has held the following to be unfair, unjust or inequitable to North Dakota franchisees (Section 51-19-09, N.D.C.C.):

Restrictive Covenants: Franchise disclosure documents which disclose the existence of covenants restricting competition contrary to Section 9-08-06, N.D.C.C., without further disclosing that such covenants will be subject to this statute.

Sites of Arbitration Proceedings: Franchise agreements providing that the parties must agree to the arbitration of the disputes at a location that is remote from the site of the franchisees' business. (This language has been included in this Agreement as a condition to registration. Franchisor and Franchisee do not agree with the foregoing language and believe that each of the provisions of the Agreement, including all arbitration provisions, are fully enforceable. Franchisor and the Franchisee intend to fully enforce all of the provisions of the Agreement and all other documents signed by them, including but not limited to, all arbitration provisions and other dispute avoidance and resolution provisions and to rely on federal pre-emption under the Federal Arbitration Act. If there is a valid and legal restriction of the North Dakota Securities Commissioner to prohibit the parties from agreeing to a site of arbitration in Minneapolis, Minnesota, then the arbitration proceedings under Section 17.10(d) shall be conducted at a place mutually agreed upon by the parties.)

Restriction on Forum: Requiring North Dakota franchisees to consent to the jurisdiction of courts outside of North Dakota.

Liquidated Damages and Termination Penalties: Requiring North Dakota Franchisees to consent to liquidated damages or termination penalties.

Applicable Laws: Franchise agreements which specify that they are to be governed by the laws of a state other than North Dakota.

Waiver of Trial by Jury: Requiring North Dakota franchisees to consent to the waiver of a trial by jury.

Waiver of Exemplary & Punitive Damages: Requiring North Dakota franchisees to consent to a waiver of exemplary and punitive damages.

General Release: Franchise agreements that require the franchisee to sign a general release upon renewal of the franchise agreement.

Limitation of Claims: Franchise agreements that require the franchisee to consent to a limitation of claims. The statute of limitations under North Dakota law applies.

SOUTH DAKOTA RIDER TO FRANCHISE AGREEMENT

1. Acknowledgments. The following language is added to Section 1 of the Franchise Agreement:

Pursuant to SDCL 37-5B, any acknowledgement provision, disclaimer or integration clause or a provision having a similar effect in a franchise agreement does not negate or act to remove from judicial review any statement, misrepresentation or action that would violate this chapter or a rule or order under this chapter.

VIRGINIA RIDER TO THE FRANCHISE AGREEMENT

1. <u>Termination of the Franchise, By Franchisor</u>. The following language is added to the end of Section 15.2:

Pursuant to Section 13.1 564 of the Virginia Retail Franchising Act, it is unlawful for a franchisor to cancel a franchise without reasonable cause. If any ground for default or termination stated in this Agreement does not constitute "reasonable cause," as that term is defined in the Virginia Retail Franchising Act or the laws of Virginia, that provision may not be enforceable.

2. No statement, questionnaire, or acknowledgment signed or agreed to by a franchisee in connection with the commencement of the franchise relationship shall have the effect of (i) waiving any claims under any applicable state franchise law, including fraud in the inducement, or (ii) disclaiming reliance on any statement made by any franchisor, franchise seller, or other person acting on behalf of the franchisor. This provision supersedes any other term of any document executed in connection with the franchise.

WASHINGTON RIDER TO FRANCHISE AGREEMENT, AREA DEVELOPMENT AGREEMENT, AND RELATED AGREEMENTS

The provisions of this Addendum form an integral part of, are incorporated into, and modify the Franchise Disclosure Document, the franchise agreement, and all related agreements regardless of anything to the contrary contained therein. This Addendum applies if: (a) the offer to sell a franchise is accepted in Washington; (b) the purchaser of the franchise is a resident of Washington; and/or (c) the franchised business that is the subject of the sale is to be located or operated, wholly or partly, in Washington.

- 1. <u>Conflict of Laws</u>. In the event of a conflict of laws, the provisions of the Washington Franchise Investment Protection Act, chapter 19.100 RCW will prevail.
- 2. <u>Franchisee Bill of Rights.</u> RCW 19.100.180 may supersede provisions in the franchise agreement or related agreements concerning your relationship with the franchisor, including in the areas of termination and renewal of your franchise. There may also be court decisions that supersede the franchise agreement or related agreements concerning your relationship with the franchisor. Franchise agreement provisions, including those summarized in Item 17 of the Franchise Disclosure Document, are subject to state law.
- 3. <u>Site of Arbitration, Mediation, and/or Litigation</u>. In any arbitration or mediation involving a franchise purchased in Washington, the arbitration or mediation site will be either in the state of

Washington, or in a place mutually agreed upon at the time of the arbitration or mediation, or as determined by the arbitrator or mediator at the time of arbitration or mediation. In addition, if litigation is not precluded by the franchise agreement, a franchisee may bring an action or proceeding arising out of or in connection with the sale of franchises, or a violation of the Washington Franchise Investment Protection Act, in Washington.

- 4. <u>General Release</u>. A release or waiver of rights in the franchise agreement or related agreements purporting to bind the franchisee to waive compliance with any provision under the Washington Franchise Investment Protection Act or any rules or orders thereunder is void except when executed pursuant to a negotiated settlement after the agreement is in effect and where the parties are represented by independent counsel, in accordance with RCW 19.100.220(2). In addition, any such release or waiver executed in connection with a renewal or transfer of a franchise is likewise void except as provided for in RCW 19.100.220(2).
- 5. <u>Statute of Limitations and Waiver of Jury Trial</u>. Provisions contained in the franchise agreement or related agreements that unreasonably restrict or limit the statute of limitations period for claims under the Washington Franchise Investment Protection Act, or rights or remedies under the Act such as a right to a jury trial, may not be enforceable.
- 6. <u>Transfer Fees</u>. Transfer fees are collectable only to the extent that they reflect the franchisor's reasonable estimated or actual costs in effecting a transfer.
- 7. <u>Termination by Franchisee</u>. The franchisee may terminate the franchise agreement under any grounds permitted under state law.
- 8. <u>Certain Buy-Back Provisions</u>. Provisions in franchise agreements or related agreements that permit the franchisor to repurchase the franchisee's business for any reason during the term of the franchise agreement without the franchisee's consent are unlawful pursuant to RCW 19.100.180(2)(j), unless the franchise is terminated for good cause.
- 9. <u>Fair and Reasonable Pricing</u>. Any provision in the franchise agreement or related agreements that requires the franchisee to purchase or rent any product or service for more than a fair and reasonable price is unlawful under RCW 19.100.180(2)(d).
- 10. <u>Waiver of Exemplary & Punitive Damages</u>. RCW 19.100.190 permits franchisees to seek treble damages under certain circumstances. Accordingly, provisions contained in the franchise agreement or elsewhere requiring franchisees to waive exemplary, punitive, or similar damages are void, except when executed pursuant to a negotiated settlement after the agreement is in effect and where the parties are represented by independent counsel, in accordance with RCW 19.100.220(2).
- 11. <u>Franchisor's Business Judgement.</u> Provisions in the franchise agreement or related agreements stating that the franchisor may exercise its discretion on the basis of its reasonable business judgment may be limited or superseded by RCW 19.100.180(1), which requires the parties to deal with each other in good faith.
- 12. <u>Indemnification</u>. Any provision in the franchise agreement or related agreements requiring the franchisee to indemnify, reimburse, defend, or hold harmless the franchisor or other parties is hereby modified such that the franchisee has no obligation to indemnify, reimburse, defend, or hold harmless the franchisor or any other indemnified party for losses or liabilities to the extent that they are caused by the indemnified party's negligence, willful misconduct, strict liability, or fraud.

- 13. <u>Attorneys' Fees</u>. If the franchise agreement or related agreements require a franchisee to reimburse the franchisor for court costs or expenses, including attorneys' fees, such provision applies only if the franchisor is the prevailing party in any judicial or arbitration proceeding.
- 14. <u>Noncompetition Covenants</u>. Pursuant to RCW 49.62.020, a noncompetition covenant is void and unenforceable against an employee, including an employee of a franchisee, unless the employee's earnings from the party seeking enforcement, when annualized, exceed \$100,000 per year (an amount that will be adjusted annually for inflation). In addition, a noncompetition covenant is void and unenforceable against an independent contractor of a franchisee under RCW 49.62.030 unless the independent contractor's earnings from the party seeking enforcement, when annualized, exceed \$250,000 per year (an amount that will be adjusted annually for inflation). As a result, any provision contained in the franchise agreement or elsewhere that conflicts with these limitations is void and unenforceable in Washington.
- 15. <u>Nonsolicitation Agreements</u>. RCW 49.62.060 prohibits a franchisor from restricting, restraining, or prohibiting a franchisee from (i) soliciting or hiring any employee of a franchisee of the same franchisor or (ii) soliciting or hiring any employee of the franchisor. As a result, any such provisions contained in the franchise agreement or elsewhere are void and unenforceable in Washington.
- 16. Questionnaires and Acknowledgments. No statement, questionnaire, or acknowledgment signed or agreed to by a franchisee in connection with the commencement of the franchise relationship shall have the effect of (i) waiving any claims under any applicable state franchise law, including fraud in the inducement, or (ii) disclaiming reliance on any statement made by any franchisor, franchise seller, or other person acting on behalf of the franchisor. This provision supersedes any other term of any document executed in connection with the franchise.
- 17. <u>Prohibitions on Communicating with Regulators</u>. Any provision in the franchise agreement or related agreements that prohibits the franchisee from communicating with or complaining to regulators is inconsistent with the express instructions in the Franchise Disclosure Document and is unlawful under RCW 19.100.180(2)(h).
- 18. <u>Advisory Regarding Franchise Brokers</u>. Under the Washington Franchise Investment Protection Act, a "franchise broker" is defined as a person that engages in the business of the offer or sale of franchises. A franchise broker represents the franchisor and is paid a fee for referring prospects to the franchisor and/or selling the franchise. If a franchise is working with a franchise broker, franchisees are advised to carefully evaluate any information provided by the franchise broker about a franchise.

WISCONSIN RIDER TO FRANCHISE AGREEMENT

1. <u>Termination of the Franchise, By Franchisor</u>. The following is added to Section 15.2 of the Franchise Agreement.

The conditions under which this Agreement can be terminated or not renewed may be affected by the Wisconsin Fair Dealership Law, Wisconsin Statutes 1981-82, Title XIV-A, Chapter 135.

ATTACHMENT B AREA DEVELOPMENT AGREEMENT

Developer	

033125, as amended 071725

TABLE OF CONTENTS

<u>Secti</u>	<u>on</u>	<u>Page</u>
1.	INTRODUCTION	1
2.	DEVELOPMENT AREA	2
3.	DEVELOPMENT FEE	3
4.	DEVELOPMENT SCHEDULE AND MANNER OF EXERCISING DEVELOPMENT RIGHTS	
5.	DEVELOPMENT EXECUTIVE	5
6.	TERM	5
7.	DEFAULT AND TERMINATION	5
8.	TRANSFERABILITY	6
9.	COVENANTS	7
10.	NOTICES	7
11.	INDEPENDENT CONTRACTOR RELATIONSHIP AND INDEMNIFICATION	8
12.	MISCELLANEOUS	8
13.	APPLICABLE LAW	9

EXHIBITS

I ADDENDUM TO AREA DEVELOPMENT AGREEMENT

TACO JOHN'S INTERNATIONAL, INC.

AREA DEVELOPMENT AGREEMENT

This Area Development Agreement (this "Agreement") is made, entered into and effective as of the date determined in accordance with Section 12.G of this Agreement (the "Effective Date"). The parties to this Agreement are you, the undersigned, as the Developer, and us (Taco John's International, Inc., a Wyoming corporation). This Agreement is for a Taco John's Restaurant to be located at the address listed on the Addendum to Area Development Agreement attached as Exhibit I (the "Addendum").

1. <u>INTRODUCTION</u>

This Agreement has been written in an informal style to make it more easily readable and to be sure that you become thoroughly familiar with all of the important rights and obligations this Agreement covers before you sign it. In this Agreement, we refer to Taco John's International, Inc. as "we" or "TJI." We refer to you as "you," or the "Developer." If the Developer is an entity, you will notice certain provisions that are applicable to its shareholders, members, or partners, upon whose business skill, financial capability and personal character we are relying in entering into this Agreement. Those individuals will be referred to in this Agreement as "owners."

We have a system for the establishment and operation of Mexican food restaurants, which we will refer to in this Agreement collectively as "Taco John's Restaurants" or "Restaurants," or individually as a "Taco John's Restaurant" or a "Restaurant." Taco John's Restaurants operate utilizing distinctive recipes, interior and exterior designs, color schemes, and fixtures and equipment, which we refer to in this Agreement as the "System" and which we may modify, develop and supplement from time to time. We identify the System by the use of certain trademarks, service marks and other commercial symbols, including the mark TACO JOHN'S®, and certain associated designs, artworks and logos, which we may change or add to from time to time (the "Marks").

This Agreement is being presented to you because you desire to obtain the right to develop, own and be franchised to operate multiple Taco John's Restaurants. In signing this Agreement, you acknowledge your understanding of the importance of our standards and the necessity of complying with our standards and specifications in developing and opening multiple Taco John's Restaurants. You also acknowledge that you have conducted an independent investigation of the Taco John's Restaurant business, you have had ample opportunity to seek independent advice and recognize that, like any other business, the nature of it may change over time, that an investment in a Taco John's Restaurant involves business risks, and that the success of this business venture is primarily dependent on your business abilities and efforts.

We expressly disclaim making, and you acknowledge that you have not received or relied on, any information as to the revenues, profits, or likelihood of success of the Taco John's Restaurant venture contemplated by this Agreement, other than as set forth in our Franchise Disclosure Document. You acknowledge that there have been no representations by us or our officers, directors, shareholders, employees, or agents, that are inconsistent with the statements made in our Franchise Disclosure Document or the provisions of this Agreement. You further represent to us, as an inducement to our entering into this Agreement with you, that there have been no misrepresentations to us in your application for the rights granted by this Agreement or in the financial information provided by you and your owners.

2. <u>DEVELOPMENT AREA</u>

- A. Subject to the terms and conditions of this Agreement and Developer's continuing performance hereof, we grant to you the exclusive right to establish the number of Taco John's Restaurants specified in the Addendum to this Agreement attached hereto as Exhibit I (the "Addendum") and incorporated herein by reference, in the geographic area also specified in the Addendum (the "Development Area"). Developer accepts this grant and undertakes the obligation to develop and operate the Taco John's Restaurants in compliance with our standards.
- B. Each Taco John's Restaurant shall be established and operated pursuant to a separate Franchise Agreement to be entered into between the parties. The term "Franchise Agreement" means the then-current form of Franchise Agreement used by us for granting franchises to Taco John's Restaurant franchisees, including all ancillary documents we may require. Developer agrees to comply with the terms of all Franchise Agreements between TJI and Developer as a part of its obligations hereunder and acknowledges that failure to execute and comply with its Franchise Agreements is a breach of this Agreement. Upon the execution of each Franchise Agreement, the terms and conditions of each Franchise Agreement shall control the establishment and operation of the applicable Taco John's Restaurant. The parties acknowledge that the Franchise Agreement governing the operation of Franchisee's first Taco John's Restaurant to be opened hereunder (the "Initial Franchise Agreement") is being executed concurrently with this Agreement.
- C. We will not establish, nor will we grant a franchise to any other party to establish, additional Taco John's Restaurants using the Marks and System anywhere within the Development Area during the term, as defined in Article 6 below, of this Agreement, except as otherwise provided herein or with your written permission. Notwithstanding the foregoing, there may be existing Franchise Agreements in effect as of the date of this Agreement under which we have granted rights to third parties to operate Taco John's Restaurants in the Development Area (the "Existing Restaurants"). You agree and acknowledge that the Franchise Agreements for such Existing Restaurants may remain in effect, and nothing in this Agreement shall prevent, prohibit, or restrict the operations of such Existing Restaurants. We may renew the Franchise Agreements or enter into successor Franchise Agreements for any Existing Restaurants. In addition, subject to the exclusivity set forth in Section 2.A above, all rights reserved to us in each of the Franchise Agreements executed hereunder are reserved under this Agreement. Furthermore, you agree that we also reserve the right, either ourselves or through an affiliate, to exercise any and all rights that we have reserved in any Franchise Agreement with you within your Development Area, excluding only the right to establish or grant rights to a third party to establish a Taco John's Restaurant using the Marks and System within the Development Area during the term hereof. Specifically, this includes but is not limited to all of those rights reserved under Section 2.3 of the Initial Franchise Agreement, such as our right to operate or allow others to operate non-traditional units within the Development Area, sales through other channels in the Development Area, the operation of other businesses under different trademarks within the Development Area, and the conduct of delivery and catering services by other Taco John's Restaurants in your Development Area.
- D. Provided you are in full compliance with this Agreement, including the Development Schedule, defined in Section 4.A below, and with each Franchise Agreement executed pursuant to this Agreement, commencing on the date on which you execute a Franchise Agreement for the last of your Taco John's Restaurants to be developed hereunder and for a period of two years thereafter (the "Exercise Period"), we grant to you a right of first refusal to establish and operate additional Taco

John's Restaurants in the Development Area in excess of the number of Taco John's Restaurants specified in the Development Schedule, in accordance with this Section 2.D. If we identify a location for a third party who desires to establish and operate a Taco John's Restaurant in the Development Area, or if we identify a location in the Development Area that we believe to be suitable for development of a company or affiliate-owned Taco John's Restaurant (in each case, a "Subsequent Restaurant Location"), we will first offer you the right to acquire a franchise for the Subsequent Restaurant Location under the terms and conditions of our then-current Franchise Agreement for a Taco John's Restaurant, including the Initial Franchise Fee set forth therein. Such offer shall expire 30 days after notice of such offer is given to you (the "Offer Period"). If you do not sign our then-current Franchise Agreement for the Subsequent Restaurant Location prior to the expiration of the Offer Period, this right of first refusal shall expire and we will be entitled to operate or grant a franchise for the operation of a Taco John's Restaurant at such Subsequent Restaurant Location, as applicable. You will have this right of first refusal for all Subsequent Restaurant Locations during the Exercise Period for so long as you have exercised your rights to acquire all prior Subsequent Restaurant Locations offered to Developer under this Section 2.D. If you refuse or decline to acquire a franchise for a Subsequent Restaurant Location, this right of first refusal shall expire and be null and void as to all future Subsequent Restaurant Locations in the Development Area and you will have no right to prevent or prohibit us from operating or granting a franchise for a Taco John's Restaurant in the Development Area. Upon the expiration of the Exercise Period, we may resume adding franchised or our own Taco John's Restaurants in the Development Area to the extent permitted by the terms of the Franchise Agreements then in effect for Taco John's Restaurants located in the Development Area.

E. This Agreement is not a franchise agreement and we do not grant you any franchise rights or other similar rights to use the Marks or System under this Agreement. You have no rights to license or subfranchise others to use the Marks or System. Other than the right to enter into Franchise Agreements, nothing in this Agreement grants you the right to enter into any agreement with respect to the Marks or System.

3. **DEVELOPMENT FEE**

- A. Concurrently with the execution of this Agreement, you will pay us as consideration for the development rights granted herein, an amount set forth in the Addendum ("Development Fee") equal to \$5,000 for each Taco John's Restaurant to be opened hereunder.
- B. We will apply \$5,000 of the Development Fee toward the initial franchise fee ("Initial Franchise Fee") due under the Initial Franchise Agreement and each subsequent Taco John's Restaurant to be developed under this Agreement (each, a "Subsequent Franchise Agreement"), which shall be in the amount determined in accordance with Section 3.C below.
- C. The Initial Franchise Fee for each Franchise Agreement signed hereunder shall be in the amount of the then-current Initial Franchise Fee applicable for that type of Taco John's Restaurant.
- D. The balance of the Initial Franchise Fee for each Taco John's Restaurant to be developed hereunder will be due in accordance with the terms of the applicable Franchise Agreement for the Taco John's Restaurant to be developed, provided that if you have not signed a Franchise Agreement for the applicable Taco John's Restaurant by the date set forth in the Development Schedule, as defined below, for that particular Taco John's Restaurant, the balance of the Initial Franchise Fee will be due in full on the date set forth in the Development Schedule for the applicable Taco John's Restaurant.

E. Other than to have applied portions of the Development Fee toward the Initial Franchise Fee for each Franchise Agreement to be executed pursuant to this Agreement, the amounts paid pursuant to this Agreement are nonrefundable once paid. Under no circumstances will you be entitled to a refund, return or rebate of any portion of Development Fee paid hereunder.

4. <u>DEVELOPMENT SCHEDULE AND MANNER OF EXERCISING DEVELOPMENT RIGHTS</u>

- A. You will be bound by and strictly follow the schedule for developing the Taco John's Restaurants set forth in the Addendum (the "Development Schedule"). Time is of the essence. By the date set forth under the Development Schedule for each applicable Taco John's Restaurant, you must exercise your development rights by entering into a Franchise Agreement with us for that Taco John's Restaurant and opening the Restaurant for business. Further, you must have open and operating the cumulative number of Taco John's Restaurants set forth on the Development Schedule. If you do not meet your Development Schedule under this Agreement, then, at our option, we may treat such failure as a default pursuant to Section 7.B(i) below, or, in the alternative, we may terminate your exclusivity rights to the Development Area and your right of first refusal rights described in Section 2. At your discretion, you may request an extension to the Development Schedule to avoid these consequences. If we agree to grant that extension, you will be charged a non-refundable Extension Fee as set forth in Section 4.A(i) below.
- (i) The "Extension Fee" shall be 4.5% of the applicable Monthly System Sales Average, as defined hereafter. The "Monthly System Sales Average" means an amount equal to the average net sales of all Taco John's Restaurants (franchised and company-owned) reported to us over the previous month. The Extension Fee must be paid for each Restaurant to be developed hereunder for which a Franchise Agreement is not signed or which is not open as required by the Development Schedule until such time as you comply with the Development Schedule up to a maximum of 13 months. The Extension Fee is charged on the 20th day of each month. If you are not in compliance with the Development Schedule after 13 months, we may exercise our right to terminate this Agreement.
- B. You will exercise your right for development of each Taco John's Restaurant by giving us written notice of your intention to develop such Taco John's Restaurant at least 45 days in advance of the deadline set forth in the Development Schedule for executing each Franchise Agreement. Subject to our approval, you must execute the then-current form of Franchise Agreement for the particular Taco John's Restaurant and pay the balance of the then-current Initial Franchise Fee by the deadline set forth in the Development Schedule. We will enter a Franchise Agreement with you only if you are in compliance with all requirements and obligations of this Agreement and all other agreements between the parties, including all other individual Franchise Agreements. You acknowledge and agree that we will have the right to refuse to offer you the right to enter into a Subsequent Franchise Agreement if we believe, in our sole discretion, that you do not have sufficient financial resources or other ability to properly develop and operate the proposed subsequent Taco John's Restaurant.
- C. Each Subsequent Franchise Agreement to be executed by you for each Taco John's Restaurant to be developed hereunder shall be in the then-current form of Franchise Agreement being offered to franchisees by us, which may contain terms substantially different than the terms of the Initial Franchise Agreement and earlier Subsequent Franchise Agreements. You acknowledge that we have the right to charge the then-current Initial Franchise Fee and the then-current rates for all other

initial fees and deposits, the royalty fees, product purchases, advertising contributions, and other fees, products and services.

5. DEVELOPMENT EXECUTIVE

You agree to designate, in writing, an individual (hereinafter referred to as the "Development Executive") who will be personally responsible for your development activities during the term of this Agreement, and who will devote the individual's best efforts to the development of Taco John's Restaurants in the Development Area. You agree that the Development Executive shall be an owner, shareholder, member, or partner of your business.

EXAMPLE 6. TERM

Unless sooner terminated in accordance with Article 7 of this Agreement, subject only to the right of first refusal set forth in Section 2.D of this Agreement, which shall survive the expiration hereof, the term of this Agreement and all rights granted to Developer shall expire on the earlier of (i) the date on which you execute the Subsequent Franchise Agreement for the last of your Taco John's Restaurants to be developed hereunder; or (ii) the date of the deadline set forth in the Development Schedule for you to execute the Subsequent Franchise Agreement for the last of your Taco John's Restaurants to be developed hereunder.

7. <u>DEFAULT AND TERMINATION</u>

- A. The rights granted to you in this Agreement have been granted in reliance on your representations and warranties, and strictly on the conditions set forth in this Agreement, including, without limitation, the condition that you strictly comply with the Development Schedule and each of your Franchise Agreements.
- B. You shall be deemed in default and this Agreement may be terminated by us in the following circumstances:
- (i) You default on any term or condition of this Agreement and fail to cure such default after 30 days' written notice to you; including without limitation, the failure to execute the required Franchise Agreements or open and continuously operate the cumulative number of Taco John's Restaurants required by the Development Schedule set forth in the Addendum; or
- (ii) You are in default under any of the Franchise Agreements executed in furtherance of this Agreement or any other agreement between us and you or any of your affiliates and fail to cure such default within the time periods specified in such other agreements.
- C. Upon termination of this Agreement pursuant to Section 7.B above, all remaining development rights and the right of first refusal granted to you under this Agreement shall automatically be revoked and shall be null and void. You shall not be entitled to any refund of any portion of the Development Fee under any circumstances. You shall have no right to establish or operate any Taco John's Restaurant for which a Franchise Agreement has not been executed by the parties.
- D. If this Agreement is terminated solely pursuant to Section 7.B.(i) above, TJI and Developer agree that such termination shall not constitute a default or result in a termination of any

Franchise Agreements executed between the parties in effect as of the date of termination of this Agreement. In that case, those Franchise Agreements shall continue in full force and effect notwithstanding the termination of this Agreement. TJI and Developer agree that any statements to the contrary in the Franchise Agreements executed by them, including any cross-default and cross-termination provisions, will be inapplicable in the situation of a termination of this Agreement based solely on Developer's failure to meet the Development Schedule.

- E. No right or remedy herein conferred upon or reserved by TJI is exclusive of any other right or remedy provided or permitted by law or equity.
- F. In the event of expiration or termination of this Agreement for any reason, you will remain subject to the provision of Article 9 of this Agreement regarding restrictive covenants and all other provisions of this Agreement that by their terms are intended to survive expiration and/or termination of this Agreement, and all terms and conditions of any and all Franchise Agreements executed in furtherance of this Agreement, whether or not any of such Franchise Agreements have been terminated prior to the termination of this Agreement.

8. TRANSFERABILITY

- A. This Agreement is fully assignable by us and will inure to the benefit of any assignee or other legal successor in interest. If we assign this Agreement to a third party who expressly assumes our obligations under this Agreement, we will be relieved from any performance or other obligations under this Agreement.
- (i) This Agreement is entered into by us with specific reliance upon your personal experience, skills, and managerial and financial qualifications. Consequently, this Agreement, and your rights and obligations under it, are and shall remain personal to you and you will not consent to or undertake a Transfer, as defined below, without our prior written consent, which consent can be withheld or conditioned in our sole and absolute discretion. As used herein, the term "Transfer" shall mean any sale, assignment, gift, pledge, mortgage or any other encumbrance, transfer by bankruptcy, transfer by judicial order, merger, consolidation, share exchange, transfer by operation of law or otherwise, whether direct or indirect, voluntary or involuntary, of this Agreement or any ownership interest in it, or any rights or obligations arising under it, or of any material portion of Developer's assets, or of any interest in Developer. Our consent to the Transfer under this Agreement may be subject to the same conditions for a Transfer of the Franchise Agreement most recently executed between TJI and Developer, except that a different transfer fee will apply as stated below.
- B. A condition of our approval of a Transfer of this Agreement is that you will Transfer all of your interest under this Agreement. You will not be permitted to Transfer any portion of this Agreement, such as the development rights of a particular Restaurant, separate and apart from the remainder of this Agreement. We will not approve any Transfer of this Agreement without the Transfer of the rights to all Franchise Agreements entered into between TJI and Developer hereunder. You and the transferee shall comply with the requirements under each Franchise Agreement entered into between TJI and you for the Transfer of each such Franchise Agreement, including the payment of the applicable transfer fee and any other fees due for each Franchise Agreement.
- C. In the event of any proposed Transfer of this Agreement as described herein, you and/or the proposed transferee shall pay to us, in addition to any fees that apply related to the Transfer of any Franchise Agreements, a development rights transfer fee equal to one-half of the transfer fFee set forth

in our then-current form of Franchise Agreement as of the date of the Transfer for each Taco John's Restaurant remaining to be developed under this Agreement as of the date of the Transfer (the "Development Rights Transfer Fee"). The Development Rights Transfer Fee shall be payable in a lump sum to us at the start of the transfer process as one of the pre-conditions to obtaining our written consent to a proposed Transfer. The Development Rights Transfer Fee paid to us will not be applied to the initial fees due for future Franchise Agreements to be executed by the transferee pursuant to this Agreement, but the transferee will receive credit for your earlier payment of the Development Fee as set forth herein.

- D. As one of the pre-conditions of obtaining our written consent to a proposed Transfer, you must provide us a copy of, and obtain our approval of, the proposed purchase agreement or other agreement for the transaction between you and the proposed transferee.
- E. You may Transfer one or more of the individual Franchise Agreements or any interest in those Franchise Agreements, or all or a substantial portion of the assets of your Taco John's Restaurants associated with such Franchise Agreements without a Transfer of your interest under this Agreement in accordance with the terms of each such Franchise Agreement, provided that the Taco John's Restaurant(s) associated with the Franchise Agreement(s) being Transferred are open and operating as of the date of the Transfer (a "Franchise Transfer"). In that event, (i) the Taco John's Restaurant(s) associated with the Franchise Transfer shall continue to be counted as Taco John's Restaurant(s) to be developed under this Agreement, (ii) the Franchise Agreement(s) associated with the Franchise Transfer shall continue to be included in the cumulative number of Taco John's Restaurants required to be opened and continuously operated in the Development Area, regardless of whether the applicable Franchise Agreement is transferred to the transferee or the transferee executes a new Franchise Agreement, (iii) the exclusive rights granted to you in this Agreement shall be modified to permit the transferee of the Franchise Transfer to operate in the Development Area, and (iv) a default by the transferee of the Franchise Agreement related to a Franchise Transfer shall constitute grounds for a default to be declared under this Agreement pursuant to Section 7.B.(ii) hereof.

9. <u>COVENANTS</u>

During the term of this Agreement, Developer and its officers, directors, shareholders, partners, members, owners, managers, representatives, and agents and their immediate family members shall be subject to all restrictive covenants as set forth in each Franchise Agreement executed pursuant to this Agreement and in any Nondisclosure and Noncompetition Agreements executed in conjunction with each Franchise Agreement, which covenants by this reference are incorporated herein. Upon termination of this Agreement, Developer and its officers, directors, shareholders, partners, members, owners, managers, representatives, and agents and their respective immediate families, shall be subject to all restrictive covenants as set forth in the last Subsequent Franchise Agreement executed hereunder, and in any Nondisclosure Agreements executed in conjunction with such Franchise Agreement. Until such time as the Initial Franchise Agreement has been signed, all restrictive covenants as set forth in the franchise agreement and nondisclosure and noncompetition agreement attached to the latest version of the Franchise Disclosure Document delivered to Developer shall apply.

10. NOTICES

All written notices required or permitted to be given under this Agreement will be given in writing by hand delivery; by certified mail with return receipt requested; by express mail or overnight delivery service that provides evidence of delivery or attempted delivery; or by e-mail or other

electronic delivery system. When made to the Developer, notice will be addressed to the address or e-mail address set forth on the Addendum and if to TJI, notice will be addressed to its principal place of business or the e-mail address: tjilegal@tacojohns.com. Either party may change its address or e-mail address from time to time by written notice to the other party in accordance with this Article. Notice will be deemed given three days after depositing in the United States mail, postage prepaid; one day after depositing in express mail or overnight delivery; when received by hand delivery; or with proof of delivery by e-mail or other electronic delivery system.

11. INDEPENDENT CONTRACTOR RELATIONSHIP AND INDEMNIFICATION

- A. The parties agree that each of them is an independent businessperson or entity, their only relationship is by virtue of this Agreement and the Franchise Agreements between them, and that no fiduciary relationship is created hereby. Neither party is liable or responsible for the other's debts or obligations, nor will either party be obligated for any damages to any person or property directly or indirectly arising out of the operation of the other party's business authorized by or conducted pursuant to this Agreement. TJI and Developer agree that neither of them will hold itself out to be the agent, employer or partner of the other and that neither of them has the authority to bind or incur liability on behalf of the other.
- B. Developer agrees to indemnify, release, defend and hold TJI, its subsidiaries and affiliates, and its and their respective shareholders, directors, officers, employees, agents, successors and assignees (the "Indemnified Parties") harmless against, and to reimburse them for all Claims, defined below, any and all third party obligations of Developer, and any and all claims, obligations and liabilities directly or indirectly arising out of this Agreement or the operation of any Taco John's Restaurants. For purposes of this indemnification, "Claims" means all obligations, actual and consequential damages and costs reasonably incurred in the defense of any claim against the Indemnified Parties, including, without limitation, reasonable accountants', attorneys' and expert witness fees, costs of investigation and proof of facts, court costs, other litigation expenses and travel and living expenses. TJI will have the right to defend any such Claim against it. This indemnity will continue in full force and effect subsequent to and notwithstanding the expiration or termination of this Agreement.

12. <u>MISCELLANEOUS</u>

- A. No failure by us to exercise any power reserved to us in this Agreement or to insist upon compliance by you with any obligation or condition in this Agreement, and no custom or practice of the parties at variance with the terms hereof, shall constitute a waiver of our rights to demand exact compliance with the terms of this Agreement. Our waiver of any particular default shall not affect or impair our right in respect to any subsequent default of the same or of a different nature, nor shall any delay, forbearance, or omission of our right to exercise any power or right arising out of any breach or default by you of any of the terms, provisions, or covenants of this Agreement, affect or impair our rights, nor shall such constitute a waiver by us of any rights hereunder or rights to declare any subsequent breach or default.
- B. In the event that any arbitrator or court of competent jurisdiction determines that any provision of this Agreement is unenforceable as written for any reason, then the parties hereby request and authorize the arbitrator or court to "blue pencil" such provision so as to make it enforceable and to best carry out the intent of the parties, or to deem such provision severed from this Agreement if it

cannot be so modified. The holding, declaration or pronouncement shall not adversely affect any other provisions of this Agreement, which shall otherwise remain in full force and effect.

- C. All captions in this Agreement are intended solely for the convenience of the parties, and none shall be deemed to affect the meaning or construction of any provision hereof. All acknowledgements, promises, covenants, agreements and obligations herein made or undertaken by or on behalf of Developer and by any of Developer's owners, shareholders, members, directors, officers or employees shall be deemed jointly and severally undertaken by all such parties on Developer's behalf.
- D. This Agreement (including the riders, attachments and exhibits to this Agreement) contains the entire agreement and understanding between the parties and supersedes any and all prior agreements concerning its subject matter. We do not authorize and will not be bound by any representation of any nature other than those expressed in this Agreement. You acknowledge and agree that no representations have been made to you by us or our representatives regarding projected sales volumes, market potential, revenues, or profits of a Taco John's Restaurant. Additionally, you acknowledge and agree that, in entering into this Agreement, you are not relying on the existence or non-existence of any particular fact or matter not set forth in this Agreement, a Franchise Agreement between the parties, or in the Franchise Disclosure Document provided to you. You agree and understand that we will not be liable or obligated for any oral representations or commitments made prior to the execution hereof, for claims of negligent or fraudulent misrepresentation based on any such oral representations or commitments, or for claims of negligent or fraudulent omissions or nondisclosure of facts or information. Nothing in this Agreement or in any related agreement is intended to disclaim any representations made by us in the Franchise Disclosure Document provided to you. This Agreement may be modified only upon execution of a written agreement signed on behalf of the Developer by all named Developers and/or by the authorized officer or agent if the named Developer is an entity and on behalf of us by our President and CEO.
- E. In the event of any conflict between the terms of this Agreement and the terms of any Franchise Agreement executed pursuant hereto, the terms of this Agreement shall control.
- F. The prevailing party in any action arising out of, or related to, this Agreement (including an action to compel arbitration) is entitled to recover from the other party all costs and expenses related to the action, including reasonable attorneys' fees, and all costs of collecting monies owed. If both parties are awarded a judgment in any dollar amount, the court or arbitrator, as applicable, shall determine the prevailing party taking into consideration the merits of the claims asserted by each party, the amount of the judgment received by each party and the relative equities between the parties.
- G. This Agreement shall not be effective until accepted by us as evidenced by dating and signing of this Agreement by our President and CEO.

13. <u>APPLICABLE LAW</u>

A. THIS AGREEMENT TAKES EFFECT UPON ITS ACCEPTANCE AND EXECUTION BY TJI AND, EXCEPT TO THE EXTENT GOVERNED BY UNITED STATES FEDERAL LAW, SHALL BE INTERPRETED AND CONSTRUED UNDER THE LAWS OF THE STATE OF WYOMING.

- TJI AND DEVELOPER AGREE THAT ANY DISPUTE BETWEEN THE PARTIES ARISING OUT OF THE TERMS OF THIS AGREEMENT SHALL BE GOVERNED BY THE APPLICABLE PROVISIONS OF THE LAST FRANCHISE AGREEMENT ENTERED INTO BETWEEN TJI AND DEVELOPER AT THE TIME OF THE DISPUTE. WHICH TERMS AND CONDITIONS ARE BY THIS REFERENCE INCORPORATED HEREIN, AND SHALL APPLY REGARDLESS OF WHETHER THE APPLICABLE FRANCHISE AGREEMENT IS IN FULL FORCE AND EFFECT AT THE TIME OF THE DISPUTE. IN THE EVENT NO FRANCHISE AGREEMENT HAS BEEN SIGNED AT THE TIME OF THE DISPUTE, ANY DISPUTE ARISING OUT OF THIS AGREEMENT SHALL BE SUBMITTED FOR BINDING ARBITRATION BEFORE ONE ARBITRATOR WHO IS A MEMBER AMERICAN ARBITRATION ASSOCIATION PANEL OF FRANCHISE ARBITRATORS AND SHALL BE CONDUCTED IN MINNEAPOLIS, MINNESOTA. THE ARBITRATION SHALL BE CONDUCTED IN ACCORDANCE WITH THE THEN-CURRENT COMMERCIAL ARBITRATION RULES OF THE **AMERICAN** ARBITRATION ASSOCIATION.
- C. NO RIGHT OR REMEDY CONFERRED UPON OR RESERVED BY TJI OR FRANCHISEE BY THIS AGREEMENT IS INTENDED TO BE, NOR SHALL BE DEEMED, EXCLUSIVE OF ANY OTHER RIGHT OR REMEDY HEREIN OR BY LAW OR EQUITY PROVIDED OR PERMITTED, BUT EACH SHALL BE CUMULATIVE OF EVERY OTHER RIGHT OR REMEDY.
- D. NOTHING HEREIN CONTAINED SHALL BAR TJI'S RIGHT TO OBTAIN INJUNCTIVE RELIEF AGAINST THREATENED CONDUCT THAT WILL CAUSE IT LOSS OR DAMAGES, UNDER THE USUAL EQUITY RULES, INCLUDING THE APPLICABLE RULES FOR OBTAINING RESTRAINING ORDERS AND PRELIMINARY INJUNCTIONS.

of, 20		
тл:	DEVELOPER:	
Taco John's International, Inc.	Company Name	
By:	By:	
Its:	Its:	
Date:	Date:	

EXHIBIT I ADDENDUM TO AREA DEVELOPMENT AGREEMENT

This is an Addendum ("Addendum") to the certain Area Development Agreement dated of even date (the "Agreement") by and between Taco John's International, Inc., a Wyoming corporation ("TJI"), and the undersigned developer ("Developer"). This Addendum modifies the terms of the Agreement, and in the event of a conflict in terms between the Agreement and this Addendum, the terms of this Addendum are controlling.

The parties agree as follows:

	1. <u>Description or</u>	Development Are	ea. The Deve	lopment A	ea, referenced	in Section 2.A
of the A	of the Agreement will be the geographical area described as follows:					
develop	2. <u>Number of Ta</u> ned in the Developmen Restaurant under the In		in Section 2.	A of the A	greement (inclu	iding the Taco
	3. Development	Fee. The Develop	ment Fee to	be paid by	Developer to T	IJI pursuant to
Section	3.A of the Agreement			1 3	1	1
respons	4. <u>Development</u> sible for your development	Executive. The nent activities is:	designated	Developmo	ent Executive	who will be
		Print Name and Ti	tle			
	Address	City	State	Zip		
	Phone Number					
E-mail Address						
		[Space Inten	tionally Left Blo	ank]		

of the Agreement is that it must execute Franchise Agreements for those Taco John's Restaurants set

Development Schedule. Developer acknowledges and agrees that a material provision

5.

forth in Paragraph 2 of this Addendum and continuously operate those Taco John's Restaurants in accordance with the following Development Schedule:

Number of Restaurants to be Developed	Date by which Franchise Agreement must be Signed	Date by which Restaurant must be Open and Continuously Operating	Cumulative number of Restaurants Required to be Open and Continuously Operating as of the Date in the Preceding Column
1	Date of this Agreement		
2			
3			
4			

6. Other terms (if appropriate).

тл:	DEVELOPER:
Taco John's International, Inc.	
	Company Name
By:	By:
Its:	Its:
Date:	Date:

ATTACHMENT C

NONDISCLOSURE AND NONCOMPETITION AGREEMENT

033125, as amended 071725

NONDISCLOSURE AND NONCOMPETITION AGREEMENT

This Nondisclosure and Noncompetition A	greement (this "Agreement") is made and entered into
effective the on the day of	, 20 by and among Taco John's International,
Inc., a Wyoming corporation ("TJI"), 1650 West Er	nd Blvd., Suite 200, St. Louis Park, Minnesota 55416, the
franchisee named on the signature page of this Agree	ement (the "Franchisee"), and the associate of Franchisee
named on the signature page of this Agreement (the '	'Associate'').

RECITALS

- A. TJI is engaged in the business of selling franchises for the operation of restaurants which serve Mexican food operating under the name and service mark "TACO JOHN'S" and related design ("**Franchised Businesses**").
- B. TJI and its affiliates have developed proprietary methods for establishing, operating and promoting Franchised Businesses utilizing certain confidential information as more fully described herein ("Licensed Methods"), and have established substantial goodwill and an excellent reputation with respect to the quality of the services available in a Franchised Business, which goodwill and reputation have been and will continue to be of major benefit to TJI.
- C. Franchisee is a franchisee under an effective franchise agreement ("**Franchise Agreement**") with T.J.
- D. Associate is or will become involved with Franchisee in the capacity of an owner, director, officer, manager, agent, employee, or independent contractor (such capacities collectively referred to as "Affiliation") or is related to a person who has an Affiliation with Franchisee, and will become privileged as to certain confidential information related to TJI, its operations and the Franchised Business.
- E. Associate, Franchisee and TJI have reached an understanding and agreement with regard to nondisclosure by Associate of confidential information and with respect to noncompetition by Associate with TJI and Franchisee.
- **NOW THEREFORE,** in consideration of the foregoing, the mutual promises contained herein and other good and valuable consideration, the receipt and sufficiency of which are acknowledged, Associate, Franchisee and TJI, intending legally to be bound, agree as follows:
- 1. <u>Confidential Information</u>. Associate recognizes and agrees that certain proprietary information relating to TJI and its operations and the operations of Franchised Businesses ("Confidential Information") is owned by TJI and is treated as confidential by TJI and Franchisee, including without limitation, all proprietary information concerning Franchised Businesses; the Licensed Methods; all financial information of TJI or Franchisee other than financial information filed with any government regulatory agency; marketing methods; sales and promotional methods; all nonpublic statistical information; the strategic plan, budgets and projections for TJI; all information concerning negotiations of any kind conducted by TJI whether pending or completed; all marketing research data and marketing plans; all information contained in the TJI operations manuals, and any other manual or other nonpublic written information; internal lists of franchisees and customers of Franchised Businesses; and all other information which may be considered a trade secret or proprietary and such Confidential Information as may be further developed from time to time by TJI.
- 2. Operations Manuals as Trade Secret. It is understood that Confidential Information, constituting "trade secrets," as used in this Agreement is deemed to include, without limitation, all information contained in the Operations and Training Manuals, which may be provided as one or more separate manuals, written instructional guides, computer disks, or other electronically stored data, as the same are changed or supplemented from time to time, and any information of whatever nature which gives TJI an opportunity to

obtain an advantage over its competitors who do not have access to, know of, or use such written materials or information.

- 3. Use and Disclosure of Confidential Information. Associate acknowledges that, in connection with Associate's Affiliation with TJI or Franchisee, TJI or Franchisee will disclose in strict confidence certain Confidential Information necessary for the operation of a Franchised Business. Associate specifically acknowledges that the Confidential Information is valuable, unique and comprises a substantial portion of the assets of TJI; and Associate agrees that he or she will not utilize all or any portion of the same for Associate's personal benefit during the term of Associate's Affiliation with Franchisee, nor in any manner use the same subsequent to the termination of Associate's Affiliation with TJI or Franchisee or the termination or expiration of the Franchise Agreement, nor disclose any of the same to any person, firm, corporation or other entity whatsoever at any time for any reason or purpose, without the prior written consent of TJI. Associate shall not copy, publish or otherwise duplicate the Confidential Information or permit others to do so and shall return all Confidential Information to Franchisee upon termination of Associate's Affiliation with Franchisee. Associate may disclose to other employees, agents, or representatives of TJI or Franchisee the Confidential Information only to the extent necessary for such employees, agents or representatives to carry out their intended function.
- 4. Noncompetition Covenant. Associate agrees that TJI and Franchisee would be unable to protect the Confidential Information against unauthorized use or disclosure and would be unable to encourage a free exchange of ideas and information among TJI, Franchisee and Associate if Associate was permitted to hold interests in any Competitive Businesses, as defined below. Therefore, during the term of this Agreement, neither Associate, nor the spouse or immediate family member of Associate shall perform services for or have any direct or indirect interest as a disclosed or beneficial owner, investor, partner, director, officer, employee, manager, consultant, representative or agent in any Competitive Business. For the purposes of this Agreement, a "Competitive Business" is any restaurant (other than a Taco John's Restaurant operated under a Franchise Agreement with TJI) that offers or sells Mexican food as more than 15% of its menu items. The ownership of 1% or less of a publicly traded company will not be deemed to be prohibited by this Section.
- 5. Audit of Business Records. TJI or its authorized agent may request, receive, inspect, and audit any business records, financial or otherwise, of Associate, Associate's immediate family members, or any party affiliated with Associate or its immediate family members, including any companies or entities associated with Associate or its immediate family members, that TJI in its sole discretion determines may be relevant in determining Associate's compliance with the terms of this Agreement or Franchisee's business results in its Franchised Business. Any such inspection or audit shall be conducted in accordance with the audit provisions set forth in the Franchise Agreement, which are deemed incorporated herein. Inspections and audits conducted at Associate's business location or other location where the records are held may take place without prior notice, during normal business hours. TJI may audit and inspect documents covering a period beginning with the date on which Associate's Affiliation commenced and ending on the date such audit is concluded. All documents provided for TJI's inspection or audit must be certified by Associate and the appropriate affiliated party, if applicable, as true, complete and correct. Inspections and audits may be conducted following the expiration or termination of Associate's Affiliation for any reason.
- **6.** <u>Injunction.</u> Associate acknowledges and agrees that in the event of any breach or threatened breach of this Agreement, TJI and Franchisee, or either one separately, shall be authorized and entitled to seek, from any court of competent jurisdiction, preliminary and permanent injunctive relief in addition to any other rights or remedies to which TJI and/or Franchisee may be entitled.
- 7. <u>Assignment</u>. Both Franchisee and TJI may assign all or part of this Agreement and the rights which inure to either of them hereunder without the consent of Associate, provided that any assignment by Franchisee shall require the written consent of TJI. This Agreement shall not be assignable by Associate.

- **8.** Effect of Waiver. The waiver by Associate, Franchisee or TJI of a breach of any provision of this Agreement shall not operate or be construed as a waiver of any subsequent breach thereof, and in no event shall such a waiver be binding upon TJI unless it is in writing and signed by an authorized representative of TJI.
- 9. <u>Binding Effect</u>. This Agreement shall be binding upon and inure to the benefit of Associate, Franchisee and TJI and their respective heirs, executors, representatives, successors and assigns.
- 10. Entire Agreement. This instrument contains the entire agreement of Associate, Franchisee and TJI relating to the matters set forth herein. It may not be changed orally, but only by an agreement in writing, signed by the party against whom enforcement of any waiver, change, modification, extension or discharge is sought. Further, both Associate and Franchisee agree that no change to this Agreement shall be made without the written consent of TJI having first been obtained.
- 11. Governing Law. This instrument shall be governed by and construed under the laws of the State of Wyoming.

12. <u>Arbitration</u>.

- a. Any and all controversies, disputes or claims between TJI, its subsidiaries and affiliated companies or their shareholders, officers, directors, agents, employees and attorneys (in their representative capacity); Franchisee, its shareholders, officers, directors, agents and employees; and/or Associate arising out of or related to this Agreement or the validity hereof shall be submitted for binding arbitration; except for actions for injunctive relief pursuant to Section 6 above, which actions TJI and/or Franchisee at their option may bring either in a court of competent jurisdiction or in arbitration.
- If TJI is a party to any controversy, dispute or claim, the arbitration will be administered before one (1) arbitrator who is a member of the American Arbitration Association Panel of Franchise Arbitrators, on demand of either party. The arbitration shall be conducted in Minneapolis, Minnesota and in accordance with the then-current Commercial Arbitration Rules of the American Arbitration Association. If the American Arbitration Association should cease doing business during the term of this Agreement, including any renewals, TJI will notify Franchisee and Associate of its selection of another arbiter group. The decision as to whether a claim is subject to mandatory arbitration shall be made by an arbitrator, not a court. The arbitrator shall have the right to award or include in the award any relief which is deemed proper in the circumstances including, without limitation, money damages (with interest on unpaid amounts from date due), specific performance, injunctive relief, attorneys' fees and costs. The award and decision of the arbitrators shall be conclusive and binding upon all parties to the arbitration and on no other parties, and judgment upon the award may be entered in any court of competent jurisdiction, and both parties waive any right to contest the validity or enforceability of such award. The parties further agree to be bound by the provisions of any applicable limitation on the period of time in which claims must be brought. The parties further agree that in connection with any such arbitration proceeding they shall file any compulsory counterclaim (as defined by Rule 13 of the Federal Rules of Civil Procedure) within 30 days after the date of the filing of the claim to which it relates.
- c. If TJI is not a party to such controversy, dispute or claim, such arbitration proceedings shall be conducted within the Protected Territory (as applicable, as defined in the Franchise Agreement between the Franchisee and TJI) of the Franchisee and will be heard by one arbitrator in accordance with the then current commercial arbitration rules of any arbitration group mutually acceptable to Franchisee and Associate, and if Franchisee and Associate cannot agree on an arbitration group within 30 days after demand for arbitration, then the American Arbitration Association shall conduct such arbitration in accordance with its then current commercial arbitration rules. The decision as to whether a claim is subject to mandatory arbitration shall be made by an arbitrator, not a court.
- 13. <u>Severability</u>. If any provision of this Agreement is held, declared or pronounced void, voidable, invalid, unenforceable or inoperative for any reason, by any court of competent jurisdiction, government authority, arbitrator or otherwise, the parties authorize and request such court, governmental

authority, or arbitrator to modify the provision held to be void, voidable, invalid, unenforceable or inoperative to contain such lesser covenants that impose the maximum duty permitted by law so that the provision is upheld as valid, and the parties agree to be bound by the modified provision. The holding, declaration or pronouncement shall not affect adversely any other provisions of this Agreement, which shall otherwise remain in full force and effect.

- Agreement in any action at law or in equity and if the TJI and/or Franchisee is successful in such litigation or arbitration as determined by the court or arbitrator in a final judgment or decree, then the Associate shall pay TJI or Franchisee, as applicable, all costs, expenses and reasonable attorneys' fees incurred by TJI and/or Franchisee (including without limitation such costs, expenses and fees on any appeals), and if TJI and/or Franchisee receives a judgment in any such action or proceeding, such costs, expenses and attorneys' fees shall be included as part of such judgment.
- **15. <u>Definitions.</u>** All capitalized terms not defined in this Agreement have the respective meanings set forth in the effective Franchise Agreement between Franchisee and TJI.
- **16.** <u>Cross Default.</u> A default by Associate under this Agreement will be deemed a default of all agreements between Franchisee and TJI, unless waived by TJI in writing.

The parties have executed this Agreement as of the date first above written.

ТЛ:	FRANCHISEE:
TACO JOHN'S INTERNATIONAL, INC., a Wyoming corporation	a
By: Its:	By: Its:
	ASSOCIATE:
	Print Name:

ATTACHMENT D

LIST OF FRANCHISEES AND AREA DEVELOPERS

AREA DEVELOPERS AS OF DECEMBER 31, 2024

RELIANCE HOSPITALITY GROUP, LLC 148 Cottonwood Road Newington, Connecticut 06111 (860) 778-6179

B&B HOLDING, LLC 1808 Pascoli CV Lexington, Kentucky 40509 (502) 681-6144

OLE 2021, LLC (2 Area Development Agreements) 210 Broadway N, Ste. 200 Fargo, North Dakota 58107 (701) 235-2014

OLE ENTERPRISES LLC 210 Broadway N, Suite 200 Fargo, North Dakota 58102 (701) 235-2014

PRGA OPERATIONS – MONTANA, LLC P.O. Box 937 Minot, North Dakota 58702 (701) 340-2312

LOVE'S TRAVEL STOPS AND COUNTRY STORES, INC. 10601 N. Pennsylvania Avenue Oklahoma City, Oklahoma 73120 (605) 214-1065

ACTIVE FRANCHISEES AS OF DECEMBER 31, 2024

ARKANSAS

J&P Tacos LLC 1819 E Main Street Russellville, Arkansas 72801 (479) 280-1932

COLORADO

Love's Travel Stops & Country Stores Inc. 3800 E State Highway 56 Berthoud, Colorado 80513 (970) 532-2507

4 Cities, Inc. 200 E Horsetooth Road Ft Collins, Colorado 80525 (970) 204-4335

4 Cities, Inc. 1609 Specht Point Road Ft Collins, Colorado 80525 (970) 472-0086

TJH Ventures LLC 1430 North Avenue Grand Junction, Colorado 81501 (970) 314-9620

4 Cities, Inc. 2508 11th Avenue Greeley, Colorado 80631 (970) 352-2857

4 Cities, Inc. 1635 N Lincoln Avenue Loveland, Colorado 80538 (970) 667-2066

PJM Inc. 9675 Washington Street Thornton, Colorado 80229 (303) 484-9828 4 Cities, Inc. 7910 S 6th Street Wellington, Colorado 80549 (970) 568-3967

PJM Inc. 280 W 136th Avenue Westminster, Colorado 80234 (720) 728-6886

IDAHO

D Squared Enterprises LLC 1643 Washington Street N Twin Falls, Idaho 83301 (208) 933-2190

ILLINOIS

L&R Industries Inc. 1525 S Main Street Bloomington, Illinois 61701 (309) 828-4336

Will-Ring Enterprises Inc.; Alexander G Habeeb 562 William R Latham Sr Drive Bourbonnais, Illinois 60914 (815) 929-1333

Will-Ring Enterprises Inc. 484 S Kennedy Drive Bradley, Illinois 60915 (815) 939-3064

L&R Industries Inc. 725 S Main Street Creve Coeur, Illinois 61610 (309) 698-0027

L&R Industries Inc. 2422 E Washington Street East Peoria, Illinois 61611 (309) 699-7412 Galesburg Mexican Inc. 1922 N Henderson Street Galesburg, Illinois 61401 (309) 297-4072

Hall & Walsh Properties, Inc. 8304 N 2nd Street Machesney Park, Illinois 61115 (815) 877-3141

OLE 2021, LLC 903 N Court Street Marion, Illinois 62959 (618) 993-8901

Wendy & Tom, Inc. 120 E 10th Avenue Milan, Illinois 61264 (309) 948-2018

Roadways Plaza, LLC 2855 Overpass Road Riverton, IL 62561 (217) 606-7666

Love's Travel Stops & Country Inc. 201 Loves Drive South Jacksonville, Illinois 62650 (217) 243-0400

INDIANA

Evansville West Hospitality TN LLC 604 N St Joseph Avenue Evansville, Indiana 47712 (812) 467-0804

Evansville East Hospitality TN LLC 2509 Washington Avenue Evansville, Indiana 47714 (812) 473-7841

Love's Travel Stops & Country Stores Inc. 1379 Flaxmill Road Huntington, Indiana 46750 (260) 609-5286 Love's Travel Stops & Country Stores Inc. 10157 N. Love's Avenue Michigan City, Indiana 46361 (260) 609-5286

David Van Baalen, Patricia Van Baalen, Ben Van Baalen 850 N Broadway Peru, Indiana 46970 (765) 919-4554

IOWA

Taco John's of Iowa Inc. 2501 Adventureland Drive Altoona, Iowa 50009 (515) 967-8777

Taco John's of Iowa Inc. 500 Lincoln Way Ames, Iowa 50010 (515) 232-3176

Taco John's of Iowa Inc. 319 S Ankeny Blvd Ankeny, Iowa 50021 (515) 964-9800

Eagle Landing Avoca Foods LLC 7005 N Chestnut Street Avoca, Iowa 51521 (712) 343-4007

Taco John's of Iowa Inc. 415 Story Street Boone, Iowa 50036 (515) 432-7600

Burlington Mexican LLC 119 S Roosevelt Avenue Burlington, Iowa 52601 (319) 754-5114

Carroll Hospitality TN LLC 513 E 6th Street Carroll, Iowa 51401 (712) 792-2646 C&M Enterprises, Inc.6210 University Avenue Cedar Falls, Iowa 50613 (574) 934-4103

PenCore Holdings Inc. 223 Collins Road NE Cedar Rapids, Iowa 52402 (319) 393-6101

Double L Investments, Inc. 226 Edgewood Road NW Cedar Rapids, Iowa 52405 (319) 396-2025

Cherokee Hospitality TN LLC 902 N 2nd Street Cherokee, Iowa 51012 (712) 225-0374

Ole Enterprises LLC 2601 James Street Coralville, Iowa 52241 (319) 519-5041

Bremer Restaurant Management Inc. 2300 W Broadway Council Bluffs, Iowa 51501 (712) 323-7088

Bremer Restaurant Management Inc. 217 E Broadway Council Bluffs, Iowa 51501 (712) 323-5121

Taco John's of Iowa Inc. 708 S Walnut Street Creston, Iowa 50801 (641) 782-4731

Davenport Hospitality TN, LLC 1915 E Kimberly Road Davenport, Iowa 52807 (563) 355-2044 Manakamana, Inc. 320 W Kimberly Road #3 Davenport, Iowa 52806 (563) 388-7322

Taco John's of Iowa Inc. 2727 SW 9th Street Des Moines, Iowa 50315 (515) 282-8800

Taco John's of Iowa Inc. 1250 E Army Post Road Des Moines, Iowa 50315 (515) 285-1331

Taco John's of Iowa Inc. 5824 Douglas Avenue Des Moines, Iowa 50310 (515) 270-4624

Taco John's of Iowa Inc. 921 E Euclid Avenue Des Moines, Iowa 50317 (515) 262-0110

Taco John's of Iowa Inc. 2121 Hubbell Avenue Des Moines, Iowa 50317 (515) 262-5211

Taco John's of Iowa Inc. 2600 Ingersoll Avenue Des Moines, Iowa 50312 (515) 243-6900

Eastern Iowa TJ's LLC 17 E 20th Street Dubuque, Iowa 52001 (563) 582-3443

Eastern Iowa TJ's LLC 2258 Flint Hill Drive Dubuque, Iowa 52003 (563) 583-3642 Fairfield Mexican Inc. 703 W Burlington Avenue Fairfield, Iowa 52556 (641) 472-7833

Taco John's of Iowa Inc. 250 SE Destination Drive Grimes, Iowa 50111 (515) 986-3360

Taco John's of Iowa Inc. 1020 West Street Grinnell, Iowa 50112 (641) 236-4863

Ole Enterprises LLC 1940 Blairs Ferry Road Suite 100 Hiawatha, Iowa 52233 (319) 200-4800

Taco John's of Iowa Inc. 1103 N Jefferson Way Indianola, Iowa 50125 (515) 961-3335

OLE 2021, LLC 320 5th Avenue SW Le Mars, Iowa 51031 (712) 546-4008

Ole Enterprises LLC 907 S Center Street Marshalltown, Iowa 50158 (641) 753-6211

Ole Enterprises LLC 603 S Federal Avenue Mason City, Iowa 50401 (641) 423-8503

Ole Enterprises, LLC 2516 Park Avenue Muscatine, Iowa 52761 (563) 264-8911

Newton Hospitality TN LLC 2106 1st Avenue E Newton, Iowa 50208 (641) 792-1798

Taco John's of Iowa Inc. 1611 Sunset Drive Norwalk, Iowa 50211 (515) 981-0598

KGT Inc. 224 8th Street SW Orange City, Iowa 51041 (712) 737-4555

Taco John's of Iowa Inc. 711 A Avenue W Oskaloosa, Iowa 52577 (641) 673-3264

Taco John's of Iowa Inc. 423 Church Street Ottumwa, Iowa 52501 (641) 682-0965

Taco John's of Iowa Inc. 911 W 16th Street Pella, Iowa 50219 (641) 628-8818

Sheldon Hospitality TN LLC 100 34th Avenue Sheldon, Iowa 51201 (712) 324-5287

Murray 5G Inc. 2111 27th Street Sioux City, Iowa 51104 (712) 255-8709

Murray 5G Inc. 3601 Gordon Drive Sioux City, Iowa 51106 (712) 255-9996 Murray 5G Inc. 1721 Hamilton Blvd Sioux City, Iowa 51103 (712) 258-8023

Murray 5G Inc. 3218 Singing Hills Blvd Sioux City, Iowa 51106 (712) 222-1223

Nissen Inc. 2329 Circle Drive W Spirit Lake, Iowa 51360 (712) 336-4849

Taco John's of Iowa Inc. 3924 NW Urbandale Drive Urbandale, Iowa 50322 (515) 252-9977

C&M Enterprises Inc. 1620 La Porte Road Waterloo, Iowa 50702 (319) 233-3651

Taco John's of Iowa Inc. 205 Hickman Road Waukee, Iowa 50263 (515) 216-2204

Kuhl Corporation of Waverly; Laura A Kuhlmann 617 W Bremer Avenue Waverly, Iowa 50677 (319) 352-5055

Taco John's of Iowa Inc. 350 Grand Avenue West Des Moines, Iowa 50265 (515) 255-4485

Taco John's of Iowa Inc. 7253 Vista Drive West Des Moines, Iowa 50266 (515) 457-9100

KANSAS

Sickler Investment Company; Stone Investments Inc. 216 S 10th Street Atchison, Kansas 66002 (913) 367-0276

Taco Springs LLC 13032 Kansas Avenue Bonner Springs, Kansas 66012 (913) 422-7304

Chanute Hospitality TN LLC 1503 S Santa Fe Avenue Chanute, Kansas 66720 (620) 433-4056

Western Kansas Mexican Foods Inc. 625 W Willow Colby, Kansas 67701 (785) 460-8226

Condordia Hospitality TN LLC 2122 Lincoln Street Concordia, Kansas 66901 (785) 243-4335

FAST N FRIENDLY, LLC Kansas Turnpike Milepost 132 Emporia, Kansas 66801 (620) 203-2160

LOSDIAS LLC 420 E 30th Avenue Hutchinson, Kansas 67502 (620) 259-7270

Taco Springs, LLC 480 N Main Street Lansing, Kansas 66043 (913) 364-5036

Sodak Tacos Inc.; Kenneth R Creasey 1101 W 6th Street Lawrence, Kansas 66044 (785) 843-0936 TJ's East Inc.; Kenneth R Creasey 2309 Haskell Avenue Lawrence, Kansas 66044 (785) 842-5533

Love's Travel Stops & Country Stores Inc. 1001 E Pancake Blvd Liberal, Kansas 67901 (620) 626-5409

Flint Hills Holdings LLC 214 Leavenworth Street Manhattan, Kansas 66505 (785) 320-6133

KCTJ's Inc.; Stacey Frederes; Jamie Frederes; Fred 18617 W 151st Street Olathe, Kansas 66062 (913) 829-8226

LOSDIAS LLC 418 E Iron Avenue Salina, Kansas 67401 (785) 825-6382

LOSDIAS LLC 3049 Riffel Drive Salina, Kansas 67401 (785) 404-1600

TMEX-Rest LLC 3706 S. Burlingame Rd. Topeka, KS 66609

KENTUCKY

Benton Mex LLC; HMC Inc.; Kimberly Inman; Tammi S Halvorson 360 Main Street Benton, Kentucky 42025 (270) 527-1670

Henderson Hospitality TN LLC 1949 US Highway 41 N Henderson, Kentucky 42420 (270) 830-6585 Madisonville Mex LLC; Kimberly Inman 1076 N Main Street Madisonville, Kentucky 42431 (270) 821-6568

HMC Inc.; Mayfield TJ's LLC; Kimberly Inman Tammi S Halvorson 1016 Paris Road Mayfield, Kentucky 42066 (270) 247-0545

HMC Inc.; TJ's of Murray LLC; Kimberly Inman; Tammi S Halvorson 1100 Chestnut Street Murray, Kentucky 42071 (270) 753-9697

B&B Holding, LLC 525 E Brannon Road Nicholasville, Kentucky 40356 (859) 724-7492

Bridge Mex LLC; HMC Inc.; Kimberly Inman; Tammi S Halvorson 2201 Bridge Street Paducah, Kentucky 42001 (270) 443-6716

Paducah Hospitality TN LLC 5266 US Highway 60 Paducah, Kentucky 42001 (270) 558-5206

HMC Inc.; TJ's of Jackson LLC; Kimberly Inman; Tammi S Halvorson 2956 Jackson Street Paducah, Kentucky 42001 (270) 442-9399

Princeton Hospitality TN LLC 530 US Highway 62 W Princeton, Kentucky 42445 (270) 365-7112

MASSACHUSETTS

Reliance Hospitality Group LLC 50 S Broadway Lawrence, Massachusetts 01846 (978)291-5907

Reliance Hospitality Group LLC 536 N Main Street Leominster, Massachusetts 01453 (978) 952-0775

MINNESOTA

Ole Enterprises LLC 2225 E Main Street Albert Lea, Minnesota 56007 (507) 373-9818

JAT Restaurants LLC 5952 Large Avenue Albertville, Minnesota 55301 (763) 276-7187

ECHOS Management Group Inc. 1702 Broadway Street Alexandria, Minnesota 56308 (320) 763-7885

Ole Enterprises LLC 308 W Oakland Avenue Austin, Minnesota 55912 (507) 433-3870

E&G, Inc. 1285 Paul Bunyan Drive NW Bemidji, Minnesota 56601 (218) 444-2266

Twin City TJ's Inc.; Tamra Kennedy 11724 Ulysses Street NE Blaine, Minnesota 55449 (763) 767-3344

PenCore Holdings Inc. 301 Washington Street NE Brainerd, Minnesota 56401 (218) 829-7375 JAT Restaurants LLC 1201 Highway 25 N Buffalo, Minnesota 55313 (763) 682-9100

Cambridge Tacos Inc. 225 2nd Avenue NE Cambridge, Minnesota 55008 (763) 689-4553

Walt Cressman; Gwen F Cressman; Judd Selland; Kay Selland; Brian Miller 1200 Highway 33 S Cloquet, Minnesota 55720 (218) 879-0902

TJC Dakota Inc. 605 N Main Street Crookston, Minnesota 56716 (218) 281-7357

ECHOS Management Group 233 E Frazee Street Detroit Lakes, Minnesota 56501 (218) 844-4660

SAM Enterprises Inc. 312 E Central Entrance Duluth, Minnesota 55811 (218) 727-5667

SAM Enterprises Inc. 4601 Grand Avenue Duluth, Minnesota 55807 (218) 624-1044

SAM Enterprises Inc. 1810 London Road Duluth, Minnesota 55812 (218) 728-4797

E-dot Restaurant Management Inc. 18033 Zane Street NW Elk River, Minnesota 55330 (763) 441-6380 L & T Peterson Enterprises Inc.; Lance Peterson; Tanya Peterson 901 E Blue Earth Avenue Fairmont, Minnesota 56031 (507) 238-2322

ECHOS Management Group 1212 N Union Avenue Fergus Falls, Minnesota 56537 (218) 739-3563

Twin City TJ's Inc.; Tamra Kennedy 1966 W Broadway Avenue Forest Lake, Minnesota 55025 (651) 464-7590

A&E Erickson Inc. 910 S Pokegama Avenue Grand Rapids, Minnesota 55744 (218) 999-7474

Walt Cressman; Gwen F Cressman; Judd Selland; Kay Selland; Brian Miller 2502 13th Avenue E Hibbing, Minnesota 55746 (218) 262-5052

Twin City TJ's Inc.; Tamra Kennedy 134 4th Avenue NW Hutchinson, Minnesota 55350 (320) 587-3424

J-Ron Enterprises Inc. 719 E Depot Litchfield, Minnesota 55355 (320) 693-3554

Bieg Deal LLC 109 LeMieur Street Little Falls, Minnesota 56345 (320) 632-2754

Cassie Baum 8-22 Aviation Road Little Falls, Minnesota 56345 320-639-2277 J-Ron Enterprises Inc. 900 S Kniss Avenue Luverne, Minnesota 56156 (507) 283-4562

LTP Restaurants Inc. 2112 N Broad Court Mankato, Minnesota 56001 (507) 388-2836

TLP Ventures Inc. 1040 Madison Avenue Mankato, Minnesota 56001 (507) 386-8226

JAT Restaurants LLC 13320 Grove Drive Maple Grove, Minnesota 55369 (763) 416-1460

JAT Restaurants LLC 9495 Deegan Avenue NE Monticello, Minnesota 55362 (763) 295-6913

E-dot Restaurant Management Inc. 108 Atlantic Avenue Morris, Minnesota 56267 (320) 589-2133

Twin City TJ's Inc.; Tamra Kennedy 4217 Winnetka Avenue N New Hope, Minnesota 55428 (763) 432-7294

Lilleodden Enterprises LLC 1301 S Broadway Street New Ulm, Minnesota 56073 (507) 354-8226

Stillwater Tacos Inc. 5910 Neal Avenue N Oak Park Heights, Minnesota 55082 (651) 439-7771 Oakdale Tacos Inc. 1010 Gershwin Avenue N Oakdale, Minnesota 55128 (651) 738-6831

Adahway LLC 38668 US Highway 169 Onamia, Minnesota 56359 (320) 738-1399

Steele County Taco Inc. 1839 S Cedar Avenue Owatonna, Minnesota 55060 (507) 451-2023

E-dot Restaurant Management Inc. 301 S Rum River Drive Princeton, Minnesota 55371 (763) 389-5478

Cassie Baum 1258 Fromelt Loop NW Rice, MN 56367 (320) 393-2914

Dual Brands Inc.; Walt Cressman 1045 7th Street NW Rochester, Minnesota 55901 (507) 281-9409

Panama Enterprises, Inc. 15085 Canada Avenue Rosemount, Minnesota 55068 (651) 423-6005

JAT Restaurants LLC 155 County Road 120 Sartell, Minnesota 56377 (320) 281-5588

TJ's of Savage Inc.; Cindy J Lund-Geer 4111 Egan Drive Savage, Minnesota 55378 (952) 890-6100

Ed Bounds 1301 Heather Street Shakopee, Minnesota 55379 (952) 496-2798

E-dot Restaurant Management Inc. 22945 Lake George Blvd NW St Francis, Minnesota 55070 (763) 753-7885

JAT Restaurants LLC 211 County Road 75 W St Joseph, Minnesota 56374 (320) 557-0557

Twin City TJ's Inc.; Tamra Kennedy 3340 Rice Street St Paul, Minnesota 55126 (651) 483-5776

TRF Hospitality TN LLC 122 3rd Street W Thief River Falls, Minnesota 56701 (218) 681-5987

JAT Restaurants LLC 29 Division Street Waite Park, Minnesota 56387 (320) 251-4220

Twin City TJ's Inc.; Tamra Kennedy 4430 S Lake Street White Bear Lake, Minnesota 55110 (651) 653-5997

Beef or Beans Inc.; 1303 S 1st Street Willmar, Minnesota 56201 (320) 231-2716

DDI of Winona Inc. 1213 Gilmore Avenue Winona, Minnesota 55987 (507) 452-6057

MISSOURI

CG Hospitality TN LLC 111 N Kings Highway Cape Girardeau, Missouri 63701 (573) 651-3555

V&Y LLC 2300 Bernadette Drive Space 716 Columbia, Missouri 65203 (573) 499-9992

Independence Hospitality TN LLC 4242 S Noland Road Independence, Missouri 64055 (816) 381-7827

ESS Foods LLC 1215 Millcreek Road Lebanon, Missouri 65536 (417) 533-7000

OLE 2021, LLC 1015 S Main Street Maryville, Missouri 64468 (660) 582-8056

Brenda White; Donna White; Larry A White; Larry G White 309 N 2nd Street Odessa, Missouri 64076 (816) 633-8662

D&S Bluff LLC 2780 Tucker Road Poplar Bluff, Missouri 63901 (573) 727-8145

KTA Enterprises Inc.; Thomas (Sonny) Pratt 1409 E Malone Avenue Sikeston, Missouri 63801 (573) 471-3344

OLE 2021, LLC 6110 Lake Avenue St Joseph, Missouri 64504 (816) 364-2131 OLE 2021, LLC 2205 N Belt Highway St Joseph, Missouri 64506 (816) 238-5685

OLE 2021, LLC 1315 S Belt Highway St Joseph, Missouri 64507 (816) 364-3355

MONTANA

PenCore Holdings Inc. 1105 Grand Avenue Billings, Montana 59101 (406) 252-3318

PenCore Holdings Inc. 402 Main Street Billings, Montana 59101 (406) 248-8002

PenCore Holdings Inc. 2655 Central Avenue Billings, Montana 59101 (406) 656-3688

PenCore Holdings Inc. 4780 King Street E Billings, Montana 59102 (406) 655-9764

PenCore Holdings Inc. 1104 Shiloh Crossing Billings, Montana 59102 (406) 534-6313

Best Taco Investments LLC 303 N 7th Avenue Bozeman, Montana 59715 (406) 587-0373

Les Torgerson; Terri LaFontaine Torgerson 232 Central Avenue W Browning, Montana 59417 (406) 338-4831 Best Taco GF 10th, LLC 1301 10th Avenue S Great Falls, Montana 59401 (406) 727-2222

Best Taco GF Northside, LLC. 1001 14th Street N Great Falls, Montana 59401 (406) 727-2758

Ole Enterprises LLC 825 Northwest Bypass Great Falls, Montana 59404 (406) 453-4141

TORCO LLC 215 W 14th Street Hardin, Montana 59034 (406) 665-1370

OLE 2021, LLC 505 1st Street W Havre, Montana 59501 (406) 265-6100

OLE 2021, LLC 2301 N Montana Avenue Helena, Montana 59601 (406) 443-3710

OLE 2021, LLC 150 N Main Street Kalispell, Montana 59901 (406) 257-1721

Best Taco Investments LLC 1115 W Lewis Street Livingston, Montana 59047 (406) 222-1911

Best Taco Investments LLC 911 S Haynes Street Miles City, Montana 59301 (406) 234-2627 Best Taco Missoula, LLC 2600 S Reserve Street Missoula, Montana 59801 (406) 543-7275

TRT Inc. 724 S Central Avenue Sidney, Montana 59270 (406) 433-2295

NEBRASKA

AL Hospitality TN LLC 824 E 3rd Street Alliance, Nebraska 69301 (308) 762-6590

Bremer Restaurant Management, Inc. 1215 N 6th Street Beatrice, Nebraska 68310 (402) 228-1334

Bremer Restaurant Management Inc. 1314 Harlan Drive Bellevue, Nebraska 68005 (402) 934-9655

Taco John's of Nebraska Inc. 102 W 5th Street Bridgeport, Nebraska 69336 (308) 641-7756

Taco Queen Inc. 930 G Street Central City, Nebraska 68826 (308) 946-2801

CH Hospitality TN LLC 1310 W Highway 20 Chadron, Nebraska 69337 (308) 432-4509

Bremer Restaurant Management Inc. 3805 Howard Blvd Columbus, Nebraska 68601 (402) 564-1022

Bremer Restaurant Management Inc. 3310 Elk Lane Fremont, Nebraska 68025 (402) 620-4385

GRG Hospitality TN LLC 3650 N 10th Street Gering, Nebraska 69341 (308) 632-3100

Wiltgen Inc. II; Jim Wiltgen 1408 S Locust Street Grand Island, Nebraska 68801 (308) 382-0380

Grand Island TJ's Inc.; Jim Wiltgen 2210 N Webb Road Grand Island, Nebraska 68803 (308) 381-8267

LRH Inc.
729 S Burlington Street
Hastings Nebraska 689

Hastings, Nebraska 68901 (402) 462-4211

Wiltgen Inc. 419 E 25th Street Kearney, Nebraska 68847 (308) 234-9515

TACOED Inc. 105 W 56th Street Kearney, Nebraska 68847 (308) 224-2400

DeRock Inc. 501 N Madison Street Lexington, Nebraska 68850 (308) 324-6610

Echo Valley Enterprises Inc. 3301 N 27th Street Lincoln, Nebraska 68521 (402) 466-8226 Echo Valley Enterprises Inc. 5641 S 56th Street Lincoln, Nebraska 68516 (402) 423-8230

Triple H Companies LLC 2301 N 84th Street Lincoln, Nebraska 68507 (402) 325-0105

Echo Valley Enterprises Inc. 1126 South Street Lincoln, Nebraska 68502 (402) 477-2943

MSB Enterprises LLC; Jerry Calvin; Robin Calvin 710 West B Street McCook, Nebraska 69001 (308) 345-1980

Top Tier LLC 2410 S 11th Street Nebraska City, Nebraska 68410 (402) 873-7956

Murray 5G Inc. 801 S 13th Street Norfolk, Nebraska 68701 (402) 316-7832

T-J Inc. 1216 E 4th Street North Platte, Nebraska 69101 (308) 534-2258

T-J Inc.
3211 Detco Circle
North Platte, Nebraska 69101
(308) 534-0200

Epic Eats LLC 302 Oregon Trail Suite 100 Ogallala, Nebraska 69153 (308) 284-8226 Bremer Restaurant Management Inc. 2901 N 108th Street Omaha, Nebraska 68164 (402) 964-2111

Bremer Restaurant Management Inc. 350 N 168th Street Omaha, Nebraska 68118 (402) 330-5253

Bremer Restaurant Management Inc. 5445 L Street Omaha, Nebraska 68117 (402) 731-8250

Bremer Restaurant Management Inc. 1115 S Saddle Creek Road Omaha, Nebraska 68106 (402) 553-7400

Bremer Restaurant Management Inc. 15375 Weir Street Omaha, Nebraska 68137 (402) 614-4250

Richard L Shelhamer; Margaret A Shelhamer; Carolyn Hahlbeck 1002 E Douglas Street O'Neill, Nebraska 68763 (402) 336-3421

SB Hospitality TN LLC 3206 Avenue I Scottsbluff, Nebraska 69361 (308) 633-3100

Love's Travel Stops & Country Inc. 645 Chase Blvd Sidney, Nebraska 69162 (308) 254-4832

Murray 5G Inc. 2701 Dakota Avenue S Sioux City, Nebraska 68776 (402) 494-4002 Michalek York LLC 3609 S Lincoln Avenue York, Nebraska 68467 (402) 362-4736

NEVADA

TJ's Mexican Food Inc.; Julie Ann Raum 770 S Wells Avenue Reno, Nevada 89502 (775) 786-8422

NORTH DAKOTA

DAC Foods Inc. 1911 N 13th Street Bismarck, North Dakota 58501 (701) 223-7547

DAC Foods Inc. 320 S 3rd Street Bismarck, North Dakota 58501 (701) 222-1878

DAC Foods Inc. 1518 Burnt Boat Drive Bismarck, North Dakota 58503 (701) 258-1688

OLE 2021, LLC 354 Highway 2 W Devils Lake, North Dakota 58301 (701) 662-2227

Preferred Restaurant Group Inc. 180 14th Street W Dickinson, North Dakota 58601 (701) 483-0975

Love's Travel Stops & Country Stores Inc 7988 160th Avenue NE Drayton, North Dakota 58225 (701) 454-0370

OLE 2021, LLC 2522 Gateway Drive Grand Forks, North Dakota 58201 (701) 772-5712 PenCore Holdings Inc. 605 25th St. SW Jamestown, North Dakota 58401 (701) 252-2508

DAC Foods Inc. 814 E Main Street Mandan, North Dakota 58554 (701) 663-2828

Preferred Restaurant Group Inc. 1725 S Broadway Minot, North Dakota 58702 (701) 838-3289

Parshall/Lucky Mound Community Development Board & Corporation 305 2nd Avenue NE Parshall, North Dakota 58763 (701) 627-4444

Robert Bergan; Joni Bergan 148 2nd Street NE Valley City, North Dakota 58072 (701) 845-5495

ECHOS Management Group 801 Dakota Avenue Wahpeton, North Dakota 58075 (701) 642-4680

Preferred Restaurant Group Inc. 1215 2nd Avenue W Williston, North Dakota 58801 (701) 572-7980

OHIO

Taco John's of Ohio Inc. 392 Richland Avenue Athens, Ohio 45701 (740) 594-4766

Love's Travel Stops & Country Stores Inc. 972 State Road 97 Bellville, Ohio 44813 (419) 886-0209 Love's Travel Stops & Country Stores Inc. 25727 Durdoc Street Circleville, Ohio 43113 (740) 474-4126

SOUTH DAKOTA

PenCore Holdings Inc. 2300 SE 6th Avenue Aberdeen, South Dakota 57401 (605) 225-2533

Cowboytacos, LLC 1401 Mill Street Belle Fourche, South Dakota 57717 (605) 892-6436

OLE 2021, LLC 101 Split Rock Blvd Brandon, South Dakota 57005 (605) 582-7287

Ole Enterprises, LLC 609 Medary Avenue Brookings, South Dakota 57006 (605) 692-7524

KFP Enterprises Inc. 704 E 5th Street Canton, South Dakota 57013 (605) 987-2410

Chamberlain Hospitality TN LLC 126 S Front Street Chamberlain, South Dakota 57325 (605) 734-4115

IBEats, LLC 86 Charles Street Deadwood, South Dakota 57732 (605) 578-3975

Dakotamart Inc. 24327 US Highway 212 Eagle Butte, South Dakota 57625 (605) 964-8146 Traden Inc. 1145 Jensen Highway Hot Springs, South Dakota 57747 (605) 745-6880

PenCore Holdings Inc. 410 Dakota Avenue N Huron, South Dakota 57350 (605) 352-4474

MB Hospitality TN LLC 219 W 4th Avenue Milbank, South Dakota 57252 (605) 432-4906

GF Buche Company Hwy 18 Mission, South Dakota (605)384-4333

Midwest Mexican Foods Inc. 701 N Sanborn Blvd Mitchell, South Dakota 57301 (605) 996-4445

Murray 5G Inc. 1240 River Drive N Sioux City, South Dakota 57049 (605) 232-5809

PenCore Holdings Inc. 415 E Sioux Avenue Pierre, South Dakota 57501 (605) 224-2083

Angela Charlson Main Street Pine Ridge, South Dakota 57770 (605) 867-5333

Ole Enterprises LLC 1710 Cambell Street Rapid City, South Dakota 57701 (605) 343-6778 Ole Enterprises LLC 3812 Eglin Street Rapid City, South Dakota 57703 (605) 791-2062

Ole Enterprises LLC 1828 Haines Avenue Rapid City, South Dakota 57701 (605) 343-3119

Ole Enterprises LLC 2810 W Main Street Rapid City, South Dakota 57701 (605) 343-7844

Midwest Mexican Foods Inc. 1700 E 10th Street Sioux Falls, South Dakota 57103 (605) 339-9558

Midwest Mexican Foods Inc. 1920 W 12th Street Sioux Falls, South Dakota 57103 (605) 338-8518

Midwest Mexican Foods Inc. 1420 W 41st Street Sioux Falls, South Dakota 57105 (605) 361-7970

Goten S&B Inc. 1224 E 57th Street Sioux Falls, South Dakota 57108 (605) 332-0300

Goten S&B Inc. 4209 N Cliff Avenue Sioux Falls, South Dakota 57104 (605) 338-7121

Goten S&B Inc. 5100 S Louise Avenue Sioux Falls, South Dakota 57108 (605) 361-9920 Goten S&B Inc. 6729 W 41st Street Sioux Falls, South Dakota 57106 (605) 362-1020

Midwest Mexican Foods Inc. 1804 S Sycamore Avenue Sioux Falls, South Dakota 57103 (605) 271-1191

Pond Enterprises Inc. 406 E Hickory Street Sisseton, South Dakota 57262 (605) 698-7310

Gar-Shar Incorporated 504 W Jackson Blvd Spearfish, South Dakota 57783 (605) 642-4620

IBEats, LLC 2314 S Junction Avenue Sturgis, South Dakota 57785 (605) 347-3556

Hennessey Enterprises Inc. 825 E Cherry Street Vermillion, South Dakota 57069 (605) 624-9431

PenCore Holdings Inc. 24 9th Avenue SE Watertown, South Dakota 57201 (605) 886-3149

G.F. Buche Company 401 SD-46 Wagner, SD 57380 (605) 384-4010

G.F. Buche Company 1124 E. 5th St. Winner, SD 57580 (605) 384-4333 Murray 5G Inc. 1014 Broadway Street Yankton, South Dakota 57078 (605) 665-4114

TENNESSEE

Elizabethton Hospitality TN LLC 601 Broad Street Elizabethton, Tennessee 37643 (423) 543-4999

Morristown Hospitality TN LLC 2770 W Andrew Johnson Highway Morristown, Tennessee 37814 (423) 587-2640

Love's Travel Stops & Country Stores Inc. 1003 Highway 76 White House, Tennessee 37188 (615) 672-6902

UTAH

Love's Travel Stops and Country Stores, Inc 915 S 1100 Avenue W Fillmore, UT 84631 (435) 334-7001

WASHINGTON

Love's Travel Stops and Country Stores, Inc. 4315 Prichard Rd. Moses Lake, WA 98837 (509) 761-5421

Lee's of Washington Inc. E 1002 Wellesley Spokane, Washington 99207 (509) 487-6093

WISCONSIN

Herman Basuki; Julie L Basuki 2405 Highway 45 N Antigo, Wisconsin 54409 (715) 627-0500 James M Tysdal; Mary Tysdal 1416 N Richmond Street Appleton, Wisconsin 54911 (920) 733-0544

Z's Taco Casa LLC 2672 Eaton Road Bellevue, Wisconsin 54311 (920) 406-7630

BB Taco LLC 147 Liberty Avenue Beloit, Wisconsin 53511 (608) 365-7055

Sparks Tacos BRF LLC 400 Oasis Road Black River Falls, Wisconsin 54615 (715) 284-3010

PZA Inc. 257 W Prairie View Road Chippewa Falls, Wisconsin 54729 (715) 723-5011

PZA Inc. 2982 Commercial Blvd Chippewa Falls, Wisconsin 54729 (715) 726-2335

PZA Inc. 2001 Brackett Avenue Eau Claire, Wisconsin 54701 (715) 834-0031

BB Taco LLC 11180 Goede Road Edgerton, Wisconsin 53534 (608) 884-0196

PenCore Holdings Inc. 634 W. Johnson Street Fond Du Lac, Wisconsin 54935 (920) 922-4400 Wisconsin Tacos Inc. 710 11th Street S Hudson, Wisconsin 54016 (715) 386-5522

Stoneridge Kaukauna LLC 300 E Ann Street Kaukauna, Wisconsin 54130 (920) 766-6080

Ole 2021, LLC 6211 75th Street Kenosha, WI 53142 (262) 228-6205

Jeffrey D Fennie; Karla M Fennie 1211 Jackson Street La Crosse, Wisconsin 54601 (608) 782-1890

Jeffrey D Fennie; Karla M Fennie 602 Monitor Street La Crosse, Wisconsin 54603 (608) 782-1750

Wisconsin Tacos Inc. 813 N Central Avenue Marshfield, Wisconsin 54449 (715) 387-4884

Wisconsin Tacos Inc. 1526 Broadway Street N Menomonie, Wisconsin 54751 (715) 235-4477

Ole 2021, LLC 2170 Miller Parkway Milwaukee, WI 53219 (262) 330-0450

Ole 2021, LLC 4650 S. 5th Street (formerly Layton Ave.) Milwaukee, WI 53221 (414) 285-5069 Wisconsin Tacos Inc. 6305 Monona Drive Monona, Wisconsin 53716 (608) 221-4622

Ole 2021, LLC 6915 Washington Avenue Mount Pleasant, WI 53406 (262)-977-7650

Midwest Taco Maker LLC 105 N Sawyer Street Oshkosh, Wisconsin 54902 (920) 230-8226

Sparks Taco's, LLC 12910 Cox Lane Osseo, Wisconsin 54758 (715) 597-6819

Victual Inc. 1465 E Business Highway 151 Platteville, Wisconsin 53818 (608) 348-5400

Victual Inc. 910 Viking Drive Reedsburg, WI 53959 (608) 668-2864

Wisconsin Tacos Inc. 2941 S Decker Drive Rice Lake, Wisconsin 54868 (715) 234-3011

Daryl Kleine; Kelly Guralski-Kleine 5512 Business Highway 51 S Schofield, Wisconsin 54476 (715) 298-3413

Ole Enterprises LLC 1616 Academy Avenue Stevens Point, Wisconsin 54481 (715) 544-4402 SAM Enterprises Inc. 2437 Tower Avenue Superior, Wisconsin 54880 (715) 395-1725

PenCore Holdings Inc. 317 N Grand Avenue Waukesha, Wisconsin 53186 (262) 524-9898

Wautoma Shell LLC W7684 State Road 21 & 73 Wautoma, Wisconsin 54982 (920) 787-1050

Ole 2021, LLC 6749 W. Greenfield Avenue West Allis, WI 53214 (414)285-6204

Ole 202, LLC 1750 S. Main Street West Bend, WI 53095 (262) 276-0163

PenCore Holdings Inc. 610 E Grand Avenue Wisconsin Rapids, Wisconsin 54494 (715) 421-0602

Herman Basuki; Julie L Basuki 412 Genesee Street Wittenberg, Wisconsin 54499 (715) 253-2340

WYOMING

Lodar Enterprises Inc.; Lori Stilwell; Daryl Stilwell 4031 E 2nd Street Casper, Wyoming 82609 (307) 235-4016

Lodar Enterprises Inc.; Lori Stilwell; Daryl Stilwell 4035 Plaza Drive Casper, Wyoming 82604 (307) 262-0591 4 Cities, Inc. 2207 17th Street Cody, Wyoming 82414 (307) 527-6424

Lodar Enterprises Inc.; Lori Stilwell; Daryl Stilwell 400 Teton Way Douglas, Wyoming 82633 (307) 358-3744

Northwest Distributing Company 808 E 2nd Street Gillette, Wyoming 82716 (307) 682-2224

Northwest Distributing Company 610 E Boxelder Road Gillette, Wyoming 82718 (307) 685-6950

4 Cities, Inc. 870 Main Street Lander, Wyoming 82520 (307) 332-0291

T-J Inc. 2114 Grand Avenue Laramie, Wyoming 82070 (307) 745-5085

Lodar Enterprises Inc.; Lori Stilwell; Daryl Stilwell 3533 W Yellowstone Highway Mills, Wyoming 82609 (307) 235-8102 Gateway Travel Center LLC 5007 Highway 16 Newcastle, Wyoming 82701 (307) 746-2002

4 Cities, Inc. 525 E Coulter Avenue Powell, Wyoming 82435 (307) 754-9527

T-J Inc. 623 W Main Street Riverton, Wyoming 82501 (307) 856-9649

4 Cities, Inc. 450 N Center Street Rock Springs, Wyoming 82901 (307) 362-2252

Northwest Distributing Company; Mandy Phillips 2175 Sugarland Drive Sheridan, Wyoming 82801 (307) 673-4709

The LLL Company 407 Park Street Thermopolis, Wyoming 82443 (307) 864-2262

FRANCHISE AGREEMENTS SIGNED BUT RESTAURANTS NOT OPEN AS OF DECEMBER 31, 2024

<u>TBD – Boston, MA</u>Tacos of Commonwealth, LLC (978) 291-5907

TBD - FARGO, ND Ole 2021, LLC (605) 228-8468

ATTACHMENT E

LIST OF FORMER FRANCHISEES

LIST OF FRANCHISEES WHO HAVE CEASED DOING BUSINESS IN THE ONE YEAR PERIOD ENDED DECEMBER 31, 2024

Listed below are the names and last known city, state and telephone numbers of every franchisee who has had a Taco John's Restaurant terminated, cancelled, not renewed, or otherwise voluntarily or involuntarily ceased to do business under the Franchise Agreement during the most recently completed fiscal year or who has not communicated with us within 10 weeks of the date of this Disclosure Document.

If you buy this franchise, your contact information may be disclosed to other buyers when you leave the franchise system.

FRANCHISEE NAME	CITY	STATE	LAST KNOWN PHONE	TRANSFER	NO. OF RESTAURANTS
CRDC 58th Ave LLC	Denver	CO	970-978-0351		1
Ohare QSR LLC	Chicago	IL	563-676-0993		1
We Love Tacos, LLC	Cadillac	MI	616-776-2604		1
We Love Tacos, LLC	Caledonia	MI	616-776-2602		1
We Love Tacos, LLC	Hudsonville	MI	616-776-2601		1
We Love Tacos, LLC	Monroe	MI	616-776-2603		1
We Love Tacos, LLC	Wyoming	MI	616-776-2600		1
Bierschenk, Inc.	Marshall	MN	507-530-2511		1
Bierschenk-					-
Worthington, Inc.	Worthington	MN	507-530-2511		1
Big Sky Taco Vendors, Inc.	Great Falls	MT	406-868-8255	X	2
Doering, Inc.	Missoula	MT	406-550-3680	X	1
We Love Tacos, LLC	Kettering	ОН	616-776-2606	11	1
We Love Tacos, LLC	West Chester	ОН	616-776-2605		1
Supreme Tacos, LLC	Clarksville	TN	207-705-2384		2
Bridge City Taco					
LLC	Mobridge	SD	605-850-4118		1
Twin Lakes Food					
Services, LLC	Dover	TN	931-249-3920		1

There was one franchisee who transferred one Restaurant to a third party, but the franchisee or its owners also own other Restaurants, and as a result, did not leave the system and are not listed in this chart.

There were two corporate entity franchisees who transferred a total of two Restaurants to a related entities under common ownership for their own business purposes and, as a result, the transferors did not leave the system and are not listed in this chart.

There were 10 franchisees who closed a total of 15 Restaurants, but the franchisee or its owners also continued to operate other Restaurants they owned, and as a result did not leave the system.

In addition, there were three franchisees owning three restaurants who transferred an equity interest in a franchisee entity in 2024. However, the transferor franchisees remain in the system and are not listed in this chart.

Because none of these franchisees left the system, they are not included in the list above, although the changes in these outlets are shown in Item 20 of the Disclosure Document. As a result, there are 10 franchisees, owning 18 Restaurants who left the system and are shown in this chart.

ATTACHMENT F

FINANCIAL STATEMENTS

FINANCIAL REPORT

DECEMBER 31, 2024, 2023 AND 2022

TABLE OF CONTENTS

INDEPENDENT AUDITOR'S REPORT	1-2
FINANCIAL STATEMENTS	
Balance Sheets	3-4
Statements of Income	5
Statements of Retained Earnings	6
Statements of Cash Flows	7-8
Notes to Financial Statements	9-21



INDEPENDENT AUDITOR'S REPORT

To the Board of Directors Taco John's International, Inc. Cheyenne, Wyoming

Opinion

We have audited the financial statements of Taco John's International, Inc. (the Company), which comprise the Balance Sheets as of December 31, 2024, 2023 and 2022; the related Statements of Income, Retained Earnings, and Cash Flows for the years then ended; and the related notes to the financial statements.

In our opinion, the accompanying financial statements present fairly, in all material respects, the financial position of the Company as of December 31, 2024, 2023 and 2022, and the results of its operations and its cash flows for the years then ended in accordance with accounting principles generally accepted in the United States of America.

Basis for Opinion

We conducted our audits in accordance with auditing standards generally accepted in the United States of America (GAAS). Our responsibilities under those standards are further described in the Auditor's Responsibilities for the Audit of the Financial Statements section of our report. We are required to be independent of the Company and to meet our other ethical responsibilities, in accordance with the relevant ethical requirements relating to our audits. We believe that the audit evidence we have obtained is sufficient and appropriate to provide a basis for our audit opinion.

Responsibilities of Management for the Financial Statements

Management is responsible for the preparation and fair presentation of the financial statements in accordance with accounting principles generally accepted in the United States of America, and for the design, implementation, and maintenance of internal control relevant to the preparation and fair presentation of financial statements that are free from material misstatement, whether due to fraud or error.

In preparing the financial statements, management is required to evaluate whether there are conditions or events, considered in the aggregate, that raise substantial doubt about the Company's ability to continue as a going concern within one year after the date that the financial statements are issued or available to be issued.

Auditor's Responsibilities for the Audit of the Financial Statements

Our objectives are to obtain reasonable assurance about whether the financial statements as a whole are free from material misstatement, whether due to fraud or error, and to issue an auditor's report that includes our opinion. Reasonable assurance is a high level of assurance but is not absolute assurance and, therefore, is not a guarantee that an audit conducted in accordance with GAAS will always detect a material misstatement when it exists. The risk of not detecting a material misstatement resulting from fraud is higher than for one resulting from error, as fraud may involve collusion, forgery, intentional omissions, misrepresentations, or the override of internal control. Misstatements are considered material if there is a substantial likelihood that, individually or in the aggregate, they would influence the judgment made by a reasonable user based on the financial statements.

In performing an audit in accordance with GAAS, we:

- Exercise professional judgment and maintain professional skepticism throughout the audit.
- Identify and assess the risks of material misstatement of the financial statements, whether due to fraud or error, and design and perform audit procedures responsive to those risks. Such procedures include examining, on a test basis, evidence regarding the amounts and disclosures in the financial statements.
- Obtain an understanding of internal control relevant to the audit in order to design audit procedures that are appropriate in the circumstances, but not for the purpose of expressing an opinion on the effectiveness of the Company's internal control. Accordingly, no such opinion is expressed.
- Evaluate the appropriateness of accounting policies used and the reasonableness of significant accounting estimates made by management, as well as evaluate the overall presentation of the financial
- Conclude whether, in our judgment, there are conditions or events, considered in the aggregate, that raise substantial doubt about the Company's ability to continue as a going concern for a reasonable period of time.

We are required to communicate with those charged with governance regarding, among other matters, the planned scope and timing of the audit, significant audit findings, and certain internal-control-related matters that we identified during the audit.

Other Information Included in the Franchise Disclosure Document

Management is responsible for the other information included in the Franchise Disclosure Document. The other information does not include the financial statements and our auditor's report thereon. Our opinion on the financial statements does not cover the other information, and we do not express an opinion or any form of assurance thereon.

In connection with our audit of the financial statements, our responsibility is to read the other information and consider whether a material inconsistency exists between the other information and the financial statements or the other information otherwise appears to be materially misstated. If, based on the work performed, we conclude that an uncorrected material misstatement of the other information exists, we are required to describe it in our report.

Cheyenne, Wyoming

MHP Assurance Services, LLP

March 27, 2025

BALANCE SHEETS December 31, 2024, 2023 and 2022

		2024	(1	2023	(1	2022
ASSETS		2024	((Consolidated)	- (0	Consolidated)
Current Assets						
Cash	\$	4,498,391	\$	6,532,231	\$	6,810,546
Restricted cash, advertising funds	J	616,024	Ф	873,864	Ф	1,687,082
Receivables:		010,024		673,004		1,007,002
Trade, less allowance for credit losses 2024 \$40,000; 2023						
\$28,000; 2022 \$22,000		1,764,681		1,880,724		1,876,176
Notes, current portion, less allowance for credit losses 2024		1,704,001		1,000,721		1,070,170
\$10,000; 2023 \$20,000; 2022 \$19,000 (Note 2)		126,604		117,909		26,675
Inventories		79,365		80,401		86,607
Prepaid expenses and other current assets		204,367		223,840		160,440
Advertising fund restricted assets, less allowance for credit		201,00		,		100,
losses 2024 \$10,000; 2023 \$11,000; 2022 \$6,000		1,735,921		1,795,596		2,053,209
Kitchen deposits held for franchisees, current portion		762,000		170,000		1,140,000
Assets held for sale (Note 4)		321,058		321,058		224,012
Total current assets		10,108,411		11,995,623		14,064,747
Kitchen Deposits Held for Franchisees, less current portion		767,685		1,529,922		1,699,397
Notes Receivable, less current portion and allowance for credit						
losses 2024 \$0; 2023 \$4,000; 2022 \$5,000 (Note 2)		•		196,251		72,618
Property and Equipment						
Company-owned restaurants (Note 6):						
Land		2,187,603		2,187,603		2,187,603
Buildings and leasehold improvements		7,419,287		7,419,287		7,252,739
Furniture and equipment		2,990,673		2,948,710		2,898,126
Administrative offices:						
Building and improvements		1,578,848		1,563,864		1,532,503
Furniture and equipment		1,527,889		1,496,035		1,253,976
Construction in progress		-		16,560		78,305
		15,704,300		15,632,059		15,203,252
Less accumulated depreciation		5,286,315		4,265,032		3,276,627
		10,417,985		11,367,027		11,926,625
Operating Lease Right-of-Use Assets (Note 5)		5,970,344		6,387,265		6,780,592
Finance Lease Right-of-Use Assets (Note 5)		26,988		51,453		108,394
Property Held for Investment				-		321,058
Total assets	_\$_	27,291,413	\$	31,527,541	\$	34,973,431

Continued

BALANCE SHEETS, *Continued* December 31, 2024, 2023 and 2022

		2024	2023 (Consolidated)		2022 (Consolidate	
LIABILITIES AND STOCKHOLDERS' EQUITY			(o one one or
Current Liabilities						
Accounts payable and accrued expenses (Note 7)	\$	2,581,803	\$	1,761,291	\$	2,883,209
Due to affiliate (Note 3)		668,206		703,961		619,411
Deferred revenue		201,824		270,685		594,496
Franchise deposits		290,000		615,000		825,000
Unearned gift card revenue		1,026,895		1,049,937		1,064,000
Advertising fund restricted liabilities		2,376,604		2,699,838		3,679,903
Current maturities of note payable (Note 6)		155,958		-		-
Current maturities of operating lease liabilities (Note 5)		453,366		421,743		8
Current maturities of finance lease liabilities (Note 5)		17,547		24,611		47,582
Total current liabilities		7,772,203		7,547,066		9,713,601
Note payable, less current maturities, net of unamortized debt issuance costs 2024 \$37,741; 2023 \$45,549; 2022						
\$53,358 (Note 6)		3,093,717		2,009,110		1,301
Operating Lease Liabilities, less current maturities (Note 5)		6,803,398		7,256,764		7,061,207
Finance Lease Liabilities, less current maturities (Note 5)		10,537		28,084		52,695
Total liabilities	_	17,679,855		16,841,024		16,828,804
Commitments and Contingencies (Notes 7, 8, 9 and 10)						
Stockholders' Equity						
Capital stock:						
Common, Class A, voting - \$0.01 par value; 1,000,000 shares						
authorized; 2,000 shares issued and outstanding		20		20		20
Additional paid-in capital		550,670		550,670		550,670
Retained earnings		9,060,868		14,135,827		17,593,937
Total stockholders' equity		9,611,558		14,686,517		18,144,627
Total liabilities and stockholders' equity	\$	27,291,413	\$	31,527,541	\$	34,973,431

STATEMENTS OF INCOME Years Ended December 31, 2024, 2023 and 2022

		2023	2022
	2024	(Consolidated)	(Consolidated)
Revenues			
Franchise revenues:			
Franchise royalty fees, net of incentives 2024			
\$734,748; 2023 \$655,857; 2022 \$434,246 (Note 3)	\$ 19,489,415	\$ 20,304,653	\$ 20,346,731
Advertising and marketing fees (Note 3)	4,853,648	5,126,186	5,619,540
Forfeitures and liquidating damages	305,703	147,314	327,896
Sale of franchise rights, development agreements, and	442 550	420,000	205.000
supplies	443,750	420,000	305,000
Foodservice distribution fees	486,305	872,678	165,694
Total franchise revenues	25,578,821	26,870,831	26,764,861
Sales by Company-owned restaurants	9,200,999	9,659,004	7,059,600
Marketing allowances	1,184,397	1,215,447	1,063,064
Gift card breakage	130,658	132,025	127,173
Other (Note 3)	275,403	274,992	274,153
Total revenues	36,370,278	38,152,299	35,288,851
Costs and Expenses			
Franchise expenses:			
Affiliate royalty fees (Note 3)	4,142,446	4,286,390	4,228,600
Sale of franchise rights, development agreements, and	4,142,440	4,200,390	4,228,000
supplies (Note 3)	20.000	56,000	41 000
	20,000	56,000 4,342,390	41,000
Total franchise expenses	4,162,446	4,342,390	4,269,600
Company-owned restaurants cost of sales	2,876,172	3,157,189	2,451,649
Company-owned restaurants operating expenses	5,472,915	5,891,922	4,451,144
Advertising funds expense (Note 9)	6,646,911	6,947,091	7,147,797
General and administrative expenses (Notes 5, 7 and 8)	22,104,605	18,959,419	15,876,156
Total costs and expenses	41,263,049	39,298,011	34,196,346
Gain on Disposal of Property and Equipment (Note 4)	40	54,391	1,012,735
Operating (loss) income	(4,892,771)	(1,091,321)	2,105,240
Pii-I (P) I			
Financial (Expense) Income	0= 4==	47.040	41.000
Interest income (Note 2)	97,477	47,940	41,288
Interest (expense) (Notes 5 and 6)	(98,999)	(51,566)	(10,024)
Net (loss) income	\$ (4,894,293)	\$ (1,094,947)	\$ 2,136,504

STATEMENTS OF RETAINED EARNINGS Years Ended December 31, 2024, 2023 and 2022

		Retained Earnings
Balance, December 31, 2021	\$	18,083,022
Net income		2,136,504
Dividends		(2,625,589)
Balance, December 31, 2022		17,593,937
Net (loss)		(1,094,947)
Dividends		(2,363,163)
Balance, December 31, 2023	9	14,135,827
Net (loss)		(4,894,293)
Dividends	1	(180,666)
Balance, December 31, 2024		9,060,868

STATEMENTS OF CASH FLOWS Years Ended December 31, 2024, 2023 and 2022

	2024	2023 (Consolidated)	2022 (Consolidated)
Cash Flows From Operating Activities			
Net (loss) income	\$ (4,894,293)	\$ (1,094,947)	\$ 2,136,504
Adjustments to reconcile net (loss) income to net cash			
(used in) operating activities:			
Depreciation and amortization	1,045,748	1,038,529	627,606
Noncash operating lease expense	416,921	393,327	308,796
Amortization of debt issuance costs	7,808	7,809	1,301
(Recovery of) provision for credit losses	(3,000)	11,000	(98,000)
(Gain) on disposal of property and equipment	-	(27,094)	(876,365)
(Gain) on disposal of finance lease assets		(27,297)	(136,370)
Gift card breakage	(130,658)	(132,025)	(127,173)
Changes in assets and liabilities:			
Decrease (increase) in:			
Trade receivables	62,340	(228,564)	(69,449)
Inventories	1,036	6,206	(22,555)
Prepaid expenses and other current assets	19,473	(63,400)	222,970
Advertising fund restricted assets	60,675	252,613	(225,951)
Kitchen deposits held for franchisees	170,237	1,139,475	(2,839,397)
Increase (decrease) in:			
Accounts payable and accrued expenses	820,512	(1,152,988)	(567,491)
Due to affiliate	(35,755)	84,550	(53,716)
Deferred revenue	(68,861)	(323,811)	319,951
Franchise deposits	(325,000)	(210,000)	(135,000)
Unearned gift card revenue	107,616	117,962	155,173
Advertising fund restricted liabilities	(323,234)	(980,065)	(951,872)
Operating lease liabilities	(421,743)	617,300	(28,181)
Net cash (used in) operating activities	(3,490,178)	(571,420)	(2,359,219)
Cash Flows From Investing Activities			
Payments on notes receivable	243,259	110,649	3,153,214
Purchase of property and equipment	(72,241)	(402,648)	(5,172,868)
Proceeds from sale of property and equipment	-	129,052	2,162,497
Proceeds from sale of finance lease assets	-	27,579	142,121
Net cash provided by (used in) investing			N W S 10
activities	171,018	(135,368)	284,964
Subtotal (forward)	(3,319,160)	(706,788)	(2,074,255)

Continued

STATEMENTS OF CASH FLOWS, *Continued* Years Ended December 31, 2024, 2023 and 2022

	2023 2022				2022	
		2024	((Consolidated)	((Consolidated)
Subtotal (forwarded)	_\$_	(3,319,160)	\$	(706,788)	\$	(2,074,255)
Cash Flows From Financing Activities						
Proceeds from note payable		1,321,341		2,000,000		<u>u</u>
Principal payments on note payable		(88,584)		2,000,000		_
Principal payments on finance lease liabilities		(24,611)		(21,582)		(83,068)
Cash dividends paid		(180,666)		(2,363,163)		(2,625,589)
Net cash provided by (used in) financing	_	(======================================		(, ,)		
activities		1,027,480		(384,745)		(2,708,657)
(Decrease) in cash and restricted cash		(2,291,680)		(1,091,533)		(4,782,912)
		,				
Cash and Restricted Cash						
Beginning		7,406,095		8,497,628		13,280,540
			10000			
Ending		5,114,415	\$	7,406,095	\$	8,497,628
Complemental Diselection of Control Disease in Control						
Supplemental Disclosures of Cash Flows Information	e.	00 000	ø	£1 £((¢	10.024
Cash payments for interest	\$	98,999	\$	51,566	\$	10,024
Cash payments for state taxes		2,464		2,370		7,295
Supplemental Schedule of Noncash Operating Activities						
Initial recognition of operating lease asset and liability	\$	£P	\$	Star	\$	7,000,032
initial recognition of operating lease asset and flability	Þ	-	Φ	-	Φ	7,000,032
Supplemental Schedule of Noncash Investing and Financing Activities						
Trade receivable reclassified as note receivable	\$	41,703	\$	218,016	\$	-
Sale of capital asset financed by note receivable		-		107,500		-
Capital asset additions financed by accounts payable		-		31,070		360,099
Initial recognition of finance lease asset and liability		=				86,664
Residual value of finance lease assets upon disposal				(26,000)		(81,055)
Property held for investment reclassified as property				, ,		
held for sale		-		321,058		99,013

NOTES TO FINANCIAL STATEMENTS

Note 1. Nature of Business and Summary of Significant Accounting Policies

Nature of business: Taco John's International, Inc. (the Company) principally franchises and operates Taco John's Mexican food restaurants throughout the United States.

Principles of consolidation: The Company considers for consolidation entities in which a controlling financial interest may be achieved through arrangements that do not involve voting interests. Such an entity, known as a variable interest entity (VIE), is required to be consolidated by its primary beneficiary. The primary beneficiary is the entity that possesses the power to direct the activities of the VIE that most significantly impact its economic performance and has the obligation to absorb losses or the right to receive benefits from the VIE that are significant to it.

The Company possessed a variable interest in Taco John's Advertising Production Fund Corporation (the Fund). The Fund was established as a requirement of the Company's franchise agreements. The purpose of the Fund was to develop advertising materials and disburse advertising funds for the benefit of the Company's franchisees. The Fund ceased operations during the year ended December 31, 2023 and was legally dissolved.

A summary of the Company's significant accounting policies follows:

Cash: The Company maintains its cash in bank deposit accounts that, at times, may exceed Federally insured limits. The Company has not experienced any losses on such accounts and believes that it is not exposed to any significant credit risk on cash accounts.

Restricted cash: Restricted cash includes cash collected by the advertising funds, usage of which is restricted for advertising activities.

Trade accounts and notes receivable: Trade accounts and notes receivable are primarily generated from ongoing business relationships with the Company's franchisees as a result of the franchise agreements. Trade receivables bear interest of up to 1.5% per month. Notes receivable bear interest as stated in Note 2.

Receivables are stated net of an allowance for credit losses. Credit loss estimates include the consideration of the likelihood of default expected over the estimated life of the receivable. Management periodically assesses the need for an allowance for credit losses based upon several key credit quality indicators, such as outstanding past-due balances, the financial strength of the obligor, the estimated fair value of any underlying collateral, and agreement characteristics.

The Company believes that its vulnerability to risk concentrations in its receivables is mitigated by its favorable historical collectability on past-due balances and expectations for fluctuations in general market conditions. Receivables are written off when there is a mutually agreed-upon settlement of debt, a discharge due to bankruptcy, or a terminated franchise agreement and it is not cost-beneficial to pursue further collection. Recoveries of receivables previously written off are recorded when received.

The trade accounts receivable and allowance for credit losses balances were \$1,828,727 and (\$37,000), respectively, as of January 1, 2022.

Inventories: Inventories are stated at the lower of cost (first-in, first-out method) or net realizable value and consist primarily of restaurant food items and paper supplies.

NOTES TO FINANCIAL STATEMENTS

Note 1. Nature of Business and Summary of Significant Accounting Policies, Continued

Advertising fund restricted assets and liabilities: Under the Company's franchise agreements, franchisees contribute 4% of revenues to the advertising fund. On behalf of franchisees, the Company collects, holds, and distributes the advertising funds. Contributions from franchisees are recorded as advertising and marketing fees (1.15%) in the Statements of Income if the Company is acting as principal to the transactions. The remaining 2.85% of funds collected are not considered revenues or expenses of the Company, as the Company is acting as an agent. Advertising funds are used to build the Company's brand equity and awareness primarily through a national marketing strategy, including national television advertising, digital media, social media programs, email, loyalty, and public relations initiatives.

Restricted assets and liabilities of the advertising fund as of December 31 are as follows:

	 2024	2023	2022	
Trade receivables, net Prepaid expenses	\$ 1,462,373 273,548	\$ 1,572,612 222,984	\$	1,551,074 502,135
Advertising fund restricted assets	 1,735,921	\$ 1,795,596	\$	2,053,209
Accounts payable and accrued expenses Deferred revenue Assets held for others	\$ 226,895 840,255 1,309,454	\$ 304,266 999,887 1,395,685	\$	288,330 1,255,643 2,135,930
Advertising funds restricted liabilities	 2,376,604	\$ 2,699,838	\$	3,679,903

Kitchen deposits held for franchisees: To address supply chain backlog concerns, the Company paid a vendor for kitchen equipment on behalf of new franchisees. These deposits act as collateral until the new franchisees pay the vendor for, and take ownership of, the equipment. At that time, the deposits are eligible to be refunded to the Company. The Company believes that all deposits are fully refundable.

Assets held for sale: Noncurrent assets are classified as held for sale if it is highly probable that they will be recovered primarily through a sale rather than through continuing use. Such assets, or disposal groups, are generally measured at the lower of their carrying amount or fair value less costs to sell. Any impairment loss on a disposal group is allocated to assets and liabilities on a pro rata basis, except that no loss is allocated to inventories, which continue to be measured in accordance with the Company's other accounting policies. Impairment losses on the initial classification as held for sale and subsequent gains and losses on remeasurement are recognized in operating income. Once classified as held for sale, property and equipment are no longer depreciated.

Debt issuance costs: Debt issuance costs are presented as a direct deduction from the carrying amount of the associated debt liability. Debt issuance costs are amortized over the term of the loan using the straight-line method. Accounting principles generally accepted in the United States of America (U.S. GAAP) require that the effective yield method be used to amortize debt issuance costs; however, the effect of using the straight-line method is not materially different from the results that would have been obtained under the effective yield method. Amortization expense was \$7,808, \$7,809 and \$1,301 for the years ended December 31, 2024, 2023 and 2022, respectively, and is included in interest (expense) in the Statements of Income.

NOTES TO FINANCIAL STATEMENTS

Note 1. Nature of Business and Summary of Significant Accounting Policies, Continued

Property and equipment: Property and equipment are stated at cost. Assets with a cost of \$5,000 or more and a useful life of more than one year are capitalized and depreciated over their useful lives using the straight-line method. Leasehold improvements are amortized over the shorter of the lease term or the useful life of the assets. The estimated useful lives are as follows:

Buildings and leasehold improvements	5-30 years
Furniture and equipment	3-10 years
Vehicles	2-3 years

Maintenance and repairs of property and equipment are charged to operations, and major improvements are capitalized. When assets are retired or otherwise disposed of, the cost and related accumulated depreciation are removed from the accounts, and any resulting gain or loss is reflected in the results from operations in the period of disposal.

Leases: The Company determines if a contract contains a lease at inception. Generally, the lease term is the minimum of the noncancelable period of the lease, inclusive of renewal periods the Company is reasonably certain to exercise. The Company recognizes a right-of-use (ROU) asset and a lease liability for each lease with a contractual term greater than 12 months at the time of the lease inception. The Company does not record leases with an initial term of 12 months or less on the Balance Sheet but continues to record rent expense on a straight-line basis over the lease term.

Lease assets and liabilities are recognized at the lease commencement date, which is the date the Company controls the use of the property. Lease liabilities represent the present value of lease payments not yet paid. The Company did not make the policy election to combine lease and non-lease components. As such, common area maintenance payments are not part of future lease payments. The Company uses its incremental borrowing rate as disclosed in Note 6 as the discount rate for its leases.

ROU assets represent the Company's right to use an underlying asset and are based upon the lease liabilities adjusted for prepayments, initial direct costs, lease incentives, and the impairment of ROU assets, as applicable. Additionally, tenant incentives used to fund leasehold improvements are generally recognized when earned and reduce the Company's ROU asset related to the lease. These are amortized through the operating lease asset as reductions of expense over the lease term.

The Company's real estate operating leases typically include fixed rent escalations over the term of the lease. Operating lease expense is recognized on a straight-line basis over the lease term and is included in general and administrative expenses.

The Company (a) does not have lease agreements that contain any material residual value guarantees or material restrictive covenants, (b) has not entered into any leases with related parties, (c) does not have any leases in which it acts as the lessor, and (d) is not party to any subleasing arrangements.

NOTES TO FINANCIAL STATEMENTS

Note 1. Nature of Business and Summary of Significant Accounting Policies, Continued

Impairment of long-lived assets: Management reviews long-lived assets used in operations for impairment whenever events and circumstances indicate that the carrying amount of an asset may not be recoverable. Any impairment loss will be measured by the difference between the fair value of an asset and its carrying amount and will be recognized in the period the recognition criteria are first applied and met. Management determined that no impairment of long-lived assets existed as of December 31, 2024, 2023 or 2022.

Deferred revenue: Deferred revenue represents funds collected from franchisees that have yet to be expensed on dead inventory resolution and other supply chain issues. As the Company acts as a principal to these transactions, amounts received are deferred until the related expenses have been incurred. There deferred revenue balance was \$274,545 as of January 1, 2022.

Franchise deposits: The Company enters into area development agreements (ADA) with certain franchisees. The ADA generally provides the franchisee with the right to develop a specified number of new restaurants within a stated, non-exclusive territory for a specified period, subject to the franchisee meeting interim new restaurant development requirements. The ADA fee is \$5,000 per restaurant, which is recorded as a deposit. The ADA is applied to the initial franchise fee at the time the restaurant is opened or is recognized as forfeitures and liquidating damages if the ADA is subsequently terminated.

Unearned gift card revenue: The Company records a liability in the period a gift card is sold. As gift cards are redeemed or incur breakage, as discussed below, the Company reduces the liability. Unearned gift card revenue was \$1,036,000 as of January 1, 2022.

Revenue recognition: The Company recognizes revenue in accordance with Financial Accounting Standards Board (FASB) Accounting Standards Codification (ASC) Topic 606, *Revenue From Contracts With Customers*, which provides a five-step model for recognizing revenue from contracts with customers, as follows:

- 1. Identify the contract with a customer
- 2. Identify the performance obligations in the contract
- 3. Determine the transaction price
- 4. Allocate the transaction price to the performance obligations in the contract
- 5. Recognize revenue when or as performance obligations are satisfied

The Company's revenues consist primarily of franchise revenues, sales by Company-owned restaurants, marketing allowances, and gift card breakage. The Company's products and services are marketed and sold primarily to customers in the United States. Results of operations are substantially affected by economic conditions, which can vary significantly by market and are impacted by consumers' disposable income levels and spending habits.

NOTES TO FINANCIAL STATEMENTS

Note 1. Nature of Business and Summary of Significant Accounting Policies, Continued

Franchise revenues: The Company's franchise agreement provides a franchise the right to construct, own and operate a restaurant upon a site accepted by the Company and to use the Company's system in connection with the operation of the restaurant at that site. The franchise agreement generally provides for a term of 10 to 20 years and up to three 10-year renewals subject to certain conditions. The Company's franchise agreement includes (a) pre-opening services, such as training; (b) the right to use the Company's symbolic intellectual property over the term of each franchise agreement; and (c) ongoing services, such as the management of the advertising and marketing fund contributions, the development of training materials and menu items, and restaurant monitoring.

The franchise agreement provides for various fees, including, but not limited to, the following:

- (a) Initial franchise fees ranging from \$10,000 to \$25,000, including pre-opening services and the sale of franchise equipment, supplies, and rights, are recognized when the requirements of the franchise agreement have been substantially fulfilled, usually upon opening. These promises are accounted for as a single performance obligation. Additionally, the Company has elected the practical expedient to account for the pre-opening services as distinct from the franchise license.
- (b) Continuing royalties are calculated as a percentage of franchise restaurant sales that are related entirely to the Company's performance obligation under the franchise agreement. These royalties are considered variable consideration, but because they relate to a license of intellectual property, they are not included in the transaction price. Instead, royalty revenue is recognized as franchise restaurant sales occur. The royalty fee is 5% of sales. The performance obligation under the franchise agreement is the promise to provide daily access to the symbolic intellectual property over the term of each franchise agreement, which is a series of distinct services that represents a single performance obligation. Although the franchisor's underlying activities associated with the symbolic intellectual property will vary both within a day and day to day, the symbolic intellectual property is accessed over time, and the customer (the franchisee) simultaneously receives and consumes the benefit from the franchisor's performance of providing access to the symbolic intellectual property (including other related activities). At times, the Company offers reductions in the required royalty fee to certain franchisees. These incentives are accounted for as a reduction to the transaction price.
- (c) Advertising and marketing fees are calculated as 1.15% of franchise restaurant sales and are considered variable consideration. The performance obligation under the franchise agreement is the management of the advertising and marketing fees, which are used to promote the intellectual property of the Company and its franchisees.
- (d) Foodservice distribution fees are calculated at either \$0.08 (through June 2024) or \$0.10 (effective July 2024) per case for all products sold under the Company's foodservice distribution agreement and are considered variable consideration. The performance obligation under the purchase agreement is to provide dead inventory resolution and other supply chain support. The foodservice distribution fees are collected by the foodservice distribution vendor and remitted to the Company on a quarterly basis.

NOTES TO FINANCIAL STATEMENTS

Note 1. Nature of Business and Summary of Significant Accounting Policies, Continued

Sales by Company-owned restaurants: The Company records food and beverage revenue from Company-owned stores upon the delivery of the food or beverage to the customer (the consumer), which is when the performance obligation is satisfied. The Company collects and remits sales taxes on transactions with customers and reports such amounts under the net method in the Statements of Income. Accordingly, these taxes are not included in gross revenue. The Company owned seven restaurants at December 31, 2024, 2023 and 2022.

Marketing allowances: Marketing allowance revenue is recorded based on the transaction price, which includes variable consideration based on purchase requirements set forth by the Company's vendors. The performance obligation for the marketing allowances is the sale of certain soft drinks and other food items. Payment terms are set forth in the vendor agreements and are either (a) advanced to the Company at the beginning of the year based on the prior year's purchases or (b) remitted to the Company on a quarterly basis.

Gift card breakage: The Company sells gift cards that do not have an expiration date, and it does not deduct dormancy fees from outstanding gift card balances. The Company uses historical gift card redemption patterns to determine the probability of a gift card's redemption. When a gift card is redeemed by a customer, the Company remits the cash value to the applicable franchise. Gift card breakage is recognized when the likelihood of a gift card being redeemed by the customer is remote and the Company determines that there is not a legal obligation to remit the unredeemed gift card balance to the relevant jurisdiction. The determination of the gift card breakage rate is based upon the Company's historical redemption patterns. The Company recognizes gift card breakage by applying its estimate of the rate of gift card breakage on a pro rata basis over the period of estimated redemption.

Advertising: Advertising production costs are expensed in the period the advertising first takes place. Other advertising costs are expensed as incurred.

Self-insurance programs: The Company utilizes a self-insurance plan for health insurance. Predetermined loss limits have been arranged with an insurance company to limit the Company's per-occurrence cash outlay. Accounts payable and accrued expenses include the estimated cost to settle reported claims and incurred but not reported claims.

Income taxes: The Company, with the consent of its stockholders, elected to be taxed under sections of Federal and state income tax laws which provide that, in lieu of corporation income taxes, the stockholders separately account for their pro rata shares of the Company's items of income, deduction, losses, and credits. As a result of this election, no income taxes have been recognized in the accompanying financial statements.

Management evaluated the Company's tax positions and concluded that the Company had taken no uncertain tax positions that require adjustments to the financial statements. With few exceptions, the Company is no longer subject to income tax examinations by U.S. Federal, state, or local tax authorities except for the last three years filed.

NOTES TO FINANCIAL STATEMENTS

Note 1. Nature of Business and Summary of Significant Accounting Policies, Continued

Estimates: The preparation of financial statements in conformity with U.S. GAAP requires management to make estimates and assumptions that affect the reported amounts of assets and liabilities and the disclosure of contingent assets and liabilities at the date of the financial statements, and the reported amounts of revenue and expenses during the reporting period. Actual results could differ from those estimates.

Reclassifications: Certain assets and liabilities on the Balance Sheets as of December 31, 2023 and 2022, and certain revenues and expenses on the Statements of Income for the years then ended, have been reclassified to be consistent with the classifications adopted for the year ended December 31, 2024, with no effect on retained earnings or net (loss) income.

Subsequent events: Events occurring subsequent to the Balance Sheet date have been evaluated for financial statement impact or disclosure through March 27, 2025, the date the financial statements were available to be issued.

Note 2. Notes Receivable

At December 31, 2024, the Company has one note receivable outstanding with a balance of \$136,604. The note receivable is noninterest-bearing, requires monthly payments through December 2025, and is unsecured.

The Company had four notes receivable from a franchisee for the sale of four Company-owned restaurants. In March 2022, the Company received \$3,092,660, comprising the principal and accrued interest, as payment in full for the outstanding notes.

Note 3. Related-Party Transactions and Subsequent Event

Spicy Seasonings, LLC: The Company is related to Spicy Seasonings, LLC through common ownership. Spicy Seasonings, LLC pays the Company a monthly administrative fee for accounting services and office space provided to Spicy Seasonings, LLC. In addition, under the terms of a royalty agreement with Spicy Seasonings, LLC, the Company pays Spicy Seasonings, LLC 1% of systemwide restaurant sales and 20% of fees earned for each franchise sold. All Company trademarks, including the responsibility for defending them, rest with Spicy Seasonings, LLC. During the years ended December 31, the Company had the following transactions with Spicy Seasonings, LLC:

	2024		2023		2022		
Administrative fees (other revenue) Affiliate royalty fee expense, of which	\$	8,400	\$	8,400	\$	8,400	
\$668,206, \$703,961 and \$619,411, respectively, is included in due to affiliates		4,142,446		4,286,390		4,228,600	
Initial sale of franchise expense		20,000		56,000		41,000	

NOTES TO FINANCIAL STATEMENTS

Note 3. Related-Party Transactions and Subsequent Event, Continued

Related-party franchisees: Certain franchisees are owned by the Company's shareholders, and the Company recognizes revenue from these franchisees under the terms of the franchise agreements. In addition, the franchisees pay the Company a monthly administrative fee for accounting services. During the years ended December 31, the Company had the following transactions with related-party franchisees:

	-	2024		2023	2022	
Franchise royalty fees	\$	236,584	\$	238,195	\$	225,972
Advertising and marketing fees		189,647		190,934		174,817
Administrative fees (other revenue)		82,340		70,725		67,200

Effective January 1, 2025, the royalty for systemwide restaurant sales was reduced to 0.75%. Effective January 1, 2026, 2027, and 2028, the royalty for systemwide restaurant sales will be 1.1%, 1.15%, and 1%, respectively.

Note 4. Assets Held for Sale

Assets held for sale comprise the following as of December 31:

	2024	2023	2022
Company-owned restaurants: Property and equipment, net Land	\$ 321,058	\$ 321,058	\$ 224,012
Total assets held for sale	\$ 321,058	\$ 321,058	\$ 224,012

In December 2023, management committed to a plan to sell a parcel of land previously held for investment, and the asset was presented as held for sale at December 31, 2023. The carrying amount of the asset was \$321,058 at December 31, 2024 and 2023.

In December 2021, management committed to a plan to sell two Company-owned restaurants and land previously held for investment, and all assets associated with those restaurants and the land were presented as a disposal group held for sale at December 31, 2021. An impairment loss of \$441,131 was recognized during the year ended December 31, 2021 for the write-down of the disposal group to its fair value less costs to sell. The carrying amount of the disposal group was \$125,000 and \$458,025 at December 31, 2022 and 2021, respectively. In June 2022, the Company closed the sale of one of the Company-owned restaurants for a sales price of \$322,482, resulting in the recognition of a loss on the disposal of \$2,518. In April 2023, the Company closed on the sale of the second Company-owned restaurant and land for a net sales price of \$112,333, resulting in the recognition of a loss on the disposal of \$12,667.

In June 2022, management committed to a plan to sell the food truck, and all assets associated with the food truck were presented as a disposal group held for sale at December 31, 2022. The carrying amount of the disposal group was \$99,012 at December 31, 2022. In July 2023, the Company closed on the sale of the food truck for a net sales price of \$110,000, resulting in the recognition of a gain on the disposal of \$10,988.

NOTES TO FINANCIAL STATEMENTS

Note 4. Assets Held for Sale, Continued

In December 2021, management committed to a plan to sell the Wyoming headquarters building and land, and all assets associated with the building and land were presented as a disposal group held for sale at December 31, 2021. The carrying amount of the disposal group was \$932,412 at December 31, 2021. In December 2022, the Company closed on the sale of the building and land for a net sales price of \$1,777,249, resulting in the recognition of a gain on the disposal of \$844,837.

Note 5. Leases

The Company leases land for Company-owned restaurants, office space, vehicles, and office equipment under leases that have initial terms ranging from three to 20 years. Real estate leases typically include options to extend the leases for additional five-year periods, although such renewals have been excluded from the calculation of the lease liability as it is not reasonably certain that the Company will exercise them.

Operating lease cost is recognized on a straight-line basis over the lease term. Finance lease cost is recognized as a combination of the amortization expense for the ROU assets and interest expense for the outstanding lease liabilities, and results in a front-loaded expense pattern over the lease term. The components of lease cost for the years ended December 31 are as follows:

	-	2024	2023	2022
Operating lease cost	\$\$	645,101	\$ 651,210	\$ 330,898
Finance lease cost: Amortization of ROU lease assets		24,465	30,659	76,413
Interest on finance lease liabilities		1,184	2,255	6,223
		25,649	32,914	82,636
Short-term lease cost		31,363	44,731	220,570
Total lease cost	_\$	702,113	\$ 728,855	\$ 634,104

NOTES TO FINANCIAL STATEMENTS

Note 5. Leases, Continued

Supplemental cash flow and noncash information related to leases for the years ended December 31 is as follows:

	2024	2023	2022
Cash paid for amounts included in the measurement of lease liabilities: Operating cash flows from operating leases Operating cash flows from finance leases Financing cash flows from finance leases	\$ 648,470 1,184 24,611	\$ 633,330 2,255 21,582	\$ 50,284 6,223 83,068
ROU assets obtained in exchange for lease obligations: Operating lease liabilities Finance lease liabilities	\$ į	\$ E	\$ 7,000,032 86,664

Supplemental information related to leases for the years ended December 31 is as follows:

	 2024	2023		2022
Weighted-average remaining lease term (years):				
Operating leases	13.71	14.49		16.33
Finance leases	1.76	2.45		2.52
Weighted-average discount rate:				
Operating leases	3.03%	3.05%	E	3.07%
Finance leases	3.00%	3.00%	E.	3.79%
Supplemental Balance Sheet information:				
Operating lease ROU assets	 5,970,344	\$ 6,387,265	\$	6,780,592
Finance lease ROU assets, gross	\$ 86,664	\$ 86,664	\$	176,319
Accumulated amortization	 (59,676)	(35,211)		(67,925)
Finance lease ROU assets	\$ 26,988	\$ 51,453	\$	108,394

One of the Company's office leases included a \$1,000,000 lease incentive payable to the Company to fund a portion of leasehold improvements to the leased space, which was included in the calculation of the operating lease asset and liability. The incentive was received by the Company in February 2023, resulting in a corresponding increase to the operating lease liability.

NOTES TO FINANCIAL STATEMENTS

Note 5. Leases, Continued

The Company's lease term expirations range from 2025 through 2042, and the future minimum rental payments for noncancelable leases as of December 31, 2024 are as follows:

	Operating		Finance
		Leases	Leases
Years ending December 31:			
2025	\$	664,518	\$ 18,037
2026		585,529	7,175
2027		603,561	3,587
2028		651,174	=
2029		671,435	-
Thereafter		5,876,230	<u> </u>
Total minimum payments		9,052,447	28,799
Less interest	-	(1,795,683)	(715)
Present value of minimum lease payments	_\$_	7,256,764	\$ 28,084

Note 6. Long-Term Debt

In October 2022, the Company entered into a loan agreement with a bank that provided advances in accordance with a line of credit provision. The Company took advances on the loan through May 2024, at which time the line of credit provision converted to a loan payable in the amount of \$3,376,000. The loan agreement required monthly payments of interest through May 2024. Effective in June 2024, the loan agreement requires monthly principal and interest payments of \$21,153 through November 2029, at which time all unpaid principal and accrued interest is due and payable in full. The note accrues interest at a rate of 3%. The Company is subject to a prepayment penalty equal to 1% of the amount prepaid through November 2025. The loan agreement is collateralized by certain assets of Company-owned restaurants.

The loan agreement requires the submission of certain financial documents during the term of the agreement. There are additional covenants and restrictions, including additional reporting requirements, the maintenance of a certain financial ratio, and limitations on transferring real property and incurring or assuming future indebtedness.

NOTES TO FINANCIAL STATEMENTS

Note 6. Long-Term Debt, Continued

The outstanding balance of the note was \$3,287,416 at December 31, 2024. Aggregate maturities required at December 31, 2024 are as follows:

Years ending December 31:		
2025	\$	155,958
2026		160,757
2027		165,715
2028		170,590
2029	3	2,634,396
	\$	3,287,416

Note 7. Bonus Plans and Unvested Commitments

The Company has bonus plans under which certain officers, directors, and members of management meeting stated eligibility requirements are paid bonuses equal to varying percentages of their annual salary or stated incentive amounts, dependent on the achievement of established targets. These bonuses and the related taxes and benefits totaled \$1,337,300, \$292,000 and \$1,049,000 for the years ended December 31, 2024, 2023 and 2022, respectively, and are included in accounts payable and accrued expenses at each respective year-end.

Effective January 1, 2023, the Company adopted a new long-term incentive compensation plan that grants rights to incentive compensation on a deferred basis to eligible participants through 2025. The plan participants are primarily those individuals who have a substantial responsibility for the management and growth of the Company. The plan is designed to establish an annual incentive compensation pool based on the achievement of established targets, which vests ratably over a three-year period and is paid out annually in March following year-end. Vesting is accelerated upon certain events, such as death, disability, retirement, and other events, as defined by the plan document. Individuals must be a participant as of December 31 to receive vesting rights for that year's payout amount.

During the year ended December 31, 2024, the plan targets were not achieved and no incentive compensation pool was established. During the year ended December 31, 2023, an incentive compensation pool was established in the amount of \$835,000, including related taxes and benefits. As of December 31, 2024 and 2023, the vested balances of the 2023 pool totaled \$152,000 and \$458,900, respectively, which is included in accounts payable and accrued expenses for the years then ended. In addition, approximately \$48,000 of the original incentive compensation pool was forfeited during the year ended December 31, 2024.

The Company had a previous long-term incentive plan that terminated on December 31, 2022. The bonus and related taxes and benefits totaled \$538,450 at December 31, 2022, which is included in accounts payable and accrued expenses for the year then ended.

NOTES TO FINANCIAL STATEMENTS

Note 8. Employee Retirement Plan

The Company has established a salary deferral plan under Section 401(k) of the Internal Revenue Code. Employees are eligible to participate in the plan once they have completed six months of service and are at least 21 years of age. Eligible employees receive a safe harbor matching contribution equal to 100% of their salary deferrals that do not exceed 5% of their compensation. The Company may also make discretionary profit sharing contributions.

The Company's total contributions to the plan were \$317,574, \$366,117 and \$281,330 for the years ended December 31, 2024, 2023 and 2022, respectively. There were no discretionary profit sharing contributions for any year.

Note 9. Marketing Commitment

The Company had a master services agreement with an advertising agency to provide marketing services and projects to promote the Company's products through February 14, 2025. The agreement required an annual agency fee of \$1,600,000 to be paid in monthly installments. The monthly installments of \$66,667 and \$133,333 for February 2022 and March 2022, respectively, were waived. All hard costs and expenses are billed in addition to the agency fee.

The Company incurred agency fees of \$1,600,000, \$1,600,000 and \$1,400,000 for the years ended December 31, 2024, 2023 and 2022, respectively. The Company's future agency fees at December 31, 2024 total \$200,000 for the year ending December 31, 2025.

Note 10. Contingencies

Paycheck Protection Program (PPP) loan: Due to economic uncertainty and to retain its workforce, on April 15, 2020, the Company obtained a PPP loan in the amount of \$2,290,500, which was accounted for under FASB ASC Topic 470, *Debt*. The Company applied for full loan forgiveness, which was approved in full by the U.S. Small Business Administration (SBA) on August 8, 2021. As the Company received a PPP loan in excess of \$2,000,000, the SBA requires a document retention period of six years after the date of forgiveness. During this time period, the SBA has reserved the right to audit the loan.

Claims and litigation: The Company is subject to claims and litigation in the ordinary course of its operations. In the opinion of management and the Company's legal counsel, the liability, if any, arising from such claims will not have a material impact on the Company's financial position or cash flows as of and for the year ended December 31, 2024.

ATTACHMENT G

LIST OF STATE ADMINISTRATORS AND AGENTS FOR SERVICE OF PROCESS

LIST OF STATE ADMINISTRATORS AND AGENTS FOR SERVICE OF PROCESS

Listed below are the names, addresses and telephone numbers of the state agencies that are responsible for franchising disclosure and or registration laws. Please note that we may not yet be registered to sell franchises in any or all of the listed states.

If a state is not included in the list, then we have not appointed an agent for service of process in that state pursuant to the requirements of that state's franchise laws. Some states listed below may have other agents appointed in addition to those listed below. There may be states in addition to those listed below in which we have appointed an agent for service of process.

STATE	STATE ADMINISTRATOR	AGENT FOR SERVICE OF PROCESS
CALIFORNIA	Department of Financial Protection and Innovation 320 West 4th Street, Suite 750 Los Angeles, California 90013-2344 (213) 576-7500 One Sansome Street, Suite 600 San Francisco, California 94104-4448 (415) 972-8565	Commissioner Department of Financial Protection and Innovation 320 West 4th Street, Suite 750 Los Angeles, California 90013-2344 (213) 576-7500 (866) 275-2677 (toll free)
	2101 Arena Boulevard Sacramento, California 95834 (916) 445-7205 (866) 275-2677 (toll free) www.dfpi,ca.gov ask.dfpi@dfpi.ca.gov	
FLORIDA	Florida Department of Agriculture and Consumer Services Division of Consumer Services Attn: Finance & Accounting 407 South Calhoun Street Tallahassee, Florida 32399-0800 (850) 410-3800	None
HAWAII	Commissioner of Securities Department of Commerce & Consumer Affairs Business Registration Division Securities Compliance Branch 335 Merchant Street, Room 203 Honolulu, Hawaii 96813 (808) 586-2722	Hawaii Commissioner of Securities Same Address
ILLINOIS	Franchise Bureau Office of the Attorney General 500 South Second Street Springfield, Illinois 62706 (217) 782-4465	Illinois Attorney General Same Address

STATE	STATE ADMINISTRATOR	AGENT FOR SERVICE OF PROCESS
INDIANA	Indiana Secretary of State Division of Securities 302 West Washington Room E-111 Indianapolis, Indiana 46204 (317) 232-6681	Indiana Secretary of State 201 State House 200 West Washington Street Indianapolis, Indiana 46204 (317) 232-6531
IOWA	Iowa Secretary of State 321 E. 12 th Street Des Moines, Iowa 50319 (515) 281-5204	Same
MARYLAND	Office of Attorney General Maryland Division of Securities 200 St. Paul Place Baltimore, Maryland 21202-2020 (410) 576-6360	Maryland Securities Commissioner Same Address
MICHIGAN	Michigan Attorney General Franchise Section - Consumer Protection Division G. Mennen Williams Building, 1st Floor 525 W. Ottawa Street Lansing, Michigan 48933 P.O. Box 30213 Lansing, Michigan 48909 (517) 373-7117	Michigan Department of Commerce Same Address
MINNESOTA	Minnesota Department of Commerce 85 7 th Place East, Suite 280 St. Paul, Minnesota 55101 (651) 539-1600	Minnesota Commissioner of Commerce Same Address
NEBRASKA	Department of Banking and Finance 1526 K Street, Suite 300 Lincoln, Nebraska 68508-2732 P.O. Box 95006 Lincoln, Nebraska 68509-5006 (402) 471-3445	None
NEW YORK	NYS Department of Law Investor Protection Bureau 28 Liberty Street, 21st Floor New York, New York 10005 (212) 416-8222	Secretary of State 99 Washington Avenue Albany, New York 12231
NORTH DAKOTA	North Dakota Securities Department 600 East Boulevard Avenue, Fourteenth Floor, Dept 414 Bismarck, North Dakota 58505-0510 (701) 328-4712	North Dakota Securities Commissioner Same

STATE	STATE ADMINISTRATOR	AGENT FOR SERVICE OF PROCESS
OREGON	Department of Consumer and Business Services Division of Finance and Corporate Securities Labor and Industries Building 350 Winter Street NE, Room 410 Salem, Oregon 97301-3881 (503) 378-4140	Director of Oregon Department of Consumer and Business Services Division of Finance and Corporate Securities Labor and Industries Building 350 Winter Street NE, Room 410 Salem, Oregon 97301-3881 (503) 378-4140
RHODE ISLAND	State of Rhode Island and Providence Plantations Department of Business Regulation Securities Division 1511 Pontiac Avenue John O. Pastore Complex - Building 69-1 Cranston, Rhode Island 02920 (401) 462-9500	Director of Rhode Island Department of Business Regulation Same Address
SOUTH DAKOTA	South Dakota Division of Insurance Securities Regulation 124 S Euclid, Suite 104 Pierre, South Dakota 57501 (605) 773-3563	Director of South Dakota Division of Insurance Same Address
TEXAS	Secretary of State Statutory Documents Section James E. Rudder Building 1019 Brazos Street Austin, Texas 78701 P.O. Box 13550 Austin, Texas 78711 (512) 463-5705	None
UTAH	Utah Department of Commerce Consumer Protection Division 160 East 300 South, 2 nd Floor Salt Lake City, UT 84114 801-530-6601	Same
VIRGINIA	State Corporation Commission Division of Securities and Retail Franchising 1300 E. Main Street, 9th Floor Richmond, Virginia 23219 (804) 371-9051	Clerk of the State Corporation Commission State Corporation Commission 1300 E. Main Street, 1st Floor Richmond, Virginia 23219 (804) 371-9733
WASHINGTON	Department of Financial Institutions Securities Division P.O. Box 41200 Olympia, Washington 98504-1200 (360) 902-8760	Director of Financial Institutions 150 Israel Road SW Tumwater, Washington 98501 (360) 902-8760

STATE	STATE ADMINISTRATOR	AGENT FOR
		SERVICE OF PROCESS
WISCONSIN	Department of Financial Institutions	Administrator, Division of Securities
	Division of Securities	Same Address
	4822 Madison Yards Way, North Tower	
	Madison, Wisconsin 53705	
	P.O. Box 1768	
	Madison, Wisconsin 53701-1768	
	(608) 266-8557	

ATTACHMENT H

OPERATIONS MANUAL TABLE OF CONTENTS



Franchise Operations Manual



Table of Sections

Introduction

Guest Services

Food Safety and Security

Cleanliness

Operating Systems

Quality and Recipes

Appendix Quality and Recipes

Miscellaneous

Menu Assembly

Human Resources and Training

Equipment and Maintenance

introduction

Contents

Manual Organization	4
Manual Purpose and Limitations	5
Importance of Confidentiality	5
Electronic Updates	5
Submitting Suggestions	5
Manual Disclaimer	6
Statement of Gender Neutrality	6
The History of Taco John's®	6
What Sets Taco John's® Apart?	6
Services Provided to Franchisees	6
Initial Management Training Program	7
Initial On-Site Assistance	7
Ongoing Training and Support	7
Franchisee Committees	8
Ongoing Research and Development	8
Visits from the Support Team Members	8



guest service



©2023 Taco John's International, Inc. CP:04/2023

Contents

Team Member Appearance Standards	13
Cleanliness and Sanitation Standards	14
Guest-obsessed Service	14
Guest-obsessed in Action	14
Good Service	14
Guest-obsessed Service at Taco John's®	14
See you next time	15
Table Service	15
Alternative Delivery Service	16
Quick Standards at Taco John's®	16
Guest Service Steps of Service	17
Marketing and Merchandising	17
Suggestive Selling	18
When and How to Suggestive Sell	18
Helpful Selling	18
<u>Upselling</u>	19
Rounding Out.	19
Sampling	19
A Summary of Suggestive Selling - Ten Secrets to Success	19
New Guests	20
<u>Children</u>	20
Rowdy People	21
Elderly People	21
Disabled People	21
People under the Influence of Drugs or Alcohol	21
Preventing Complaints	22
Hopefully, This Never Happens, But	22
Coupons	23
Handling Complaints with a Written Response	23
General Cash Handling Policies	
Credit Cards	24

Voids and Overrings	25
Quick-Change Artists	25
Counterfeit Money	26
Bank Deposits	26
Safe Procedures	27
Gift Card Program	27
Gift Card Swipe Procedures	28



food safety and security



Food Safety	5
Types of Food Hazards	5
Biological Hazards	5
Physical Hazards	5
Chemical Hazards	5
Handwashing	6
When to Wash Your Hands	7
Hand Maintenance	7
Handwashing Stations	7
Glove Use	8
Avoiding Cross-Contamination	9
The FQT Log	10
Breaking the Cycle	10
Temperature Control During Thawing, Cooking, Cooling, and Reheating	10
Proper Storage - Refrigerators and Freezers	12
Dry Storage	13
The Time Factor: Freshness and Stock Rotation	13
The People Factor.	14
Pest Control	14
Working with the Health Department	15
Alleged Foreign Object or Alleged Illness	16
Team Member Safety Recommendations.	16
Protective Equipment	17
Avoiding Slips and Falls	17
Avoiding Burns	18
Avoiding Cuts	19
Avoiding Chemical Burns/Toxic Fumes from Cleaning Products	20
Preventing Back Injuries	20
Recommendations for Working on Equipment Safely	20
Avoiding Bio-Hazardous Pathogens	21
Reporting of Injuries	21
Responding to a Team Member Injury	21
First Aid Training	22
©2023 Taco John's International, Inc.	CP:04/2023

Guest Safety Recommendations	22
Recommended Preventive Measures	22
Restaurant Security Recommendations	24
Facility Considerations	24
Robbery	26
Preventing Internal/Team Member Theft	27
Follow the Regulations of Your Local Regulatory Authority	28
Implement the Four-Hour Sanitation Program	28



cleanliness



Taco John's® Cleaning System	31
Cleaning Equipment Needed	32
Required Cleaning Agents	
Safe Use of Cleaning Agents	33
Scheduling Cleaning	34



operating systems



©2023 Taco John's International, Inc. CP 04/2023

Operating Systems	38
Opening the Restaurant	38
Planning the Day	39
Team Member Deployment	41
Managing Meal Periods	43
Managing Prep	45
Team Member Breaks Recovery Activities	45
Recovery Activities	45
Managing Slow Periods	
Managing Shift Changes	47
Closing the Restaurant	48
Maintaining Quality	
Productivity and Speed Tips	
Brand Standards Assessment Manual (BSM)	53
Policies and Guidelines	55
Taco John's® Core Menu	55
Gift Card Program	57
Management Tools	58



quality recipes



Introduction to Quality Standards	91
The Distribution Process	91
Purpose of Supply Chain	91
Approved Distributors	91
Working with the Distributor	91
The Order Guide	92
Adjusting When Ordering	92
Placing the Order	92
Pricing and Payment	92
Returns and Credits	92
Ongoing Supplier Problems	93
Receiving	93
Accepting a Delivery	93
General Guidelines for Receiving an Order	93
The Delivery Vehicle	94
Verifying the Order	94
Quality Assurance Protocol	96
Storage	97
General Guidelines for Storing an Order	97
Recipe Preparation	100
Using Quality Raw Ingredients	100
Food Safety Points	100
Team Member Safety	101
Preparation or Prep Planning	101
Establishing Normal Build-To Amounts	102
Adjusting Build-To Amounts	102
Completing the Daily Prep Guide	103
Prioritizing Prep	103
Managing Prep Needs during the Day	103
Preparing Recipes	

Using the Right Smallwares and Equipment	104
Holding	
Holding Equipment	105
Hot Holding	106
Frying, Cooking, and Reheating Guidelines	108
Frying Guidelines	108
Production Line and Cooking and Heating Equipment	110
<u>ConvoTherm</u>	110
FryMaster Rethermalizer	
Steam jacketed kettles	110
MerryChef Oven	111
General Reheating Guidelines	111
Fountain Beverage Quality	113
Maintenance	
Hot Beverages Quality	115
Standard Equipment Schedule	116

Charts and Forms - Definitions	118
Microwave Cook Chart	118
Rotation Log	118
Shelf-Life Chart	118
Fry Quality Chart	118
Recipes	118



miscellaneous



©2023 Taco John's International, Inc.



menu assembly



©2023 Taco John's International, Inc. CP 04/2023

TABLE OF CONTENTS

Introduction to menu assembly	5
<u>Temperature</u>	5
Ingredient Consistency	6
<u>Freshness</u>	
Correct Ingredients	6
Ingredient Placement	6
Portion Control	
Finger Portion Charts	
Portion Build Cards	7
Side Orders of Ingredients	10
Ingredient General Information	11
Wrapping, Packaging, and Bagging	12
Six-Pack And A Pound [®]	
Breakfast Combos	16
Breakfast items	17
Forms and Charts	
Portion Chart for Sides and Extra Portions	117



human resources and training



<u>Human Resources Overview</u>	3
Recruiting Employees	3
Determining Hiring Needs	3
Getting the Word Out	4
The Application	4
Training & Development at Taco John's®	5
The Elite Training Specialist – Management Level 101	6
Training Methodology	6
Management Training	6
Training & development for the new franchisee	7
Nationally Certified Training Restaurants	7
The New Franchisee Management Training Program	7
Management Certification Process for all Franchisees	8
Managing Your Team	8
Coaching Best Practices.	8
Developing a Team Atmosphere	8
Communicating with Employees	9
Hosting Employee Meetings	9
Motivating Employees	9



equipment maintenance



Maintaining Equipment	12
Equipment Programming	12
Pepsi-Cola Preventative Maintenance	16
Pepsi Equipment Service/Repair Hotline	21
Preventative Maintenance	22
Beverage Unit	22
<u>Blender</u>	22
<u>Dishwasher</u>	23
Drive-thru Window	23
Food Processor/Hand Dicer	23
Fryer – (Frymaster)	23
HVAC	25
Ice Machine	25
Bunn Coffee/Tea	26
Curtis Coffee/Tea	27
Lobby Tables and Chairs	28
Microwave	28
Refrigerator and Freezer, Walk-in – Roof-top Condensing Unit	29
Rethermalizer	29
Roof	30
Steam-Jacketed Kettle	31
Water Filter	32
ConvoTherm	32
MercoMax and MercoEco	32
MerryChef Oven	33
Franke	33

ATTACHMENT I <u>DISCLOSURE ACKNOWLEDGMENT</u>

DISCLOSURE ACKNOWLEDGMENT

law mi	To make sure that no misunderstanding exists between chisee") and Taco John's International, Inc. ("Franchisor"), and to make sure that no violations of ght have occurred, and understanding that Franchisor is relying on the statements made in this ent, the undersigned acknowledge and represent as follows:
1.	I received a copy of the Franchise Disclosure Document ("Disclosure Document") on, 20 I had an opportunity to review the TACO JOHN'S Disclosure Document and Franchise Agreement and understand the terms, conditions, and obligations of the Franchise Agreement. Check one: () Yes () No. If no, please explain:
2.	I received a copy of the Franchise Agreement or Area Development Agreement at least seven calendar days prior to the date on which the Franchise Agreement or Area Development Agreement was executed. Check one: () Yes () No. If no, please explain:
3.	I had an opportunity to seek professional advice regarding the Disclosure Document and the Franchise Agreement. Check one: () Yes () No. If no, please explain:
4.	I agree that no oral, written, visual or other promise, agreement, commitment, representation, understanding, side agreement, option or other representation was made to me with respect to any matter including, but not limited to, any representation or promise regarding marketing, site location, advertising, or operational assistance, which contradicted the disclosures in the Disclosure Document. Check one: () Yes () No. If no, please state in detail the claim or representation, who made it, and approximately when:
5.	Except as stated in Item 19 of the Disclosure Document, I agree that no oral, written, visual, or other claim or representation which stated or suggested a specific level or range of actual or potential sales, costs, income, expenses, profits, cash flow, tax effects or otherwise, was made to me by any representative of the Franchisor. Check one: () Yes () No. If no, please state in detail the claim or representation, who made it, and approximately when:
6.	I understand that it is my responsibility to find and secure a location for the TACO JOHN'S restaurant, and to retain an architect, contractor, and other construction specialists. Check one: () Yes () No. If no, please explain:

of loss or failure; that the pur that there exist no guarantees	
	titute all of the shareholders, partners, investors and principals wed the Franchise Disclosure Document and all exhibits
ANYONE HAS SUGGESTED THAT THE STATEMENTS BEING ANSWERED TO INFORM THE PRESIDENT OF TACO JOE	S INCONSISTENT WITH THIS STATEMENT OR IF IS DOCUMENT BE SIGNED WITHOUT ALL OF THE RUTHFULLY AND ACCURATELY, IMMEDIATELY IN'S INTERNATIONAL, INC. AT 307-635-0101. Ing and represent and warrant that all of the above statements
Print Name	Print Name
Signature	Signature
Date	Date
Print Name	Print Name
Signature	Signature
Date	Date

ATTACHMENT J

GIFT CARD ONBOARDING AND CHANGE FORM



GIFT CARD ONBOARDING AND CHANGE FORM

FRANCHISEE	INFORMATION	IRS TAX I	REPORTING REQUIREMENTS:
Franchise Owner Name:		INCLUDE A CO	MPLETED W-9 WITH THIS FORM
Franchise Phone Number:		1099k Name: (Same as Line 1 on W-9)	
Franchise Owner Email:		1099k Tax ID #:	
Company DBA Name:		1099k Address 1:	
Store Location #:		1099k Address 2:	
Store Street Address:		1099k City:	
City:		1099k State:	
State:		1099k Zip Code:	
Zip Code:			TJI USE ONLY
Store Phone Number:		MID #:	
Request Type: check one New Store, expected opening date Ownership Change, expected date of transfer Update Bank Info, requested effective date Other Change, (describe)		Point of Sale Provider: check one	Qu Beyond Other: YES List Store #'s:
		Do you currently own other stores that are operating gift cards?	
			NO [
BANKING I	NFORMATION:		AUTHORIZATION:
Bank Name:		Signature:	
Bank Account Name:		Printed Name:	
Bank Account Number:		Date:	
Bank Routing Number:		A signature on this form confirms that all information submitted on this form is accurate.	
Bank Address:		INCLUDE A BA	ANK LETTER WITH THIS FORM

Please return this form to your contact at Taco John's International, Inc. for gift card processing.

ATTACHMENT K

LEASE RIDER

LEASE RIDER

THIS LEASE RIDER (the "RI	der") is made effective this	s day of	, 202 (the "Effective
Date") by and among	("Landlord"),	("Tenant"), and	Taco John's International,
Inc. ("TJI").			
WHEREAS, Landlord leases o	or will lease certain premise	s to Tenant located a	t ("Premises")
under that certain lease agreem	ent dated	between Landl	ord and Tenant ("Lease");
and			

WHEREAS, Tenant will operate a Taco John's Restaurant at such Premises under a Franchise Agreement ("Franchise Agreement") between Tenant and TJI; and

WHEREAS, the parties hereto desire to provide TJI with certain rights in the event of termination of the Franchise Agreement or default under the Lease.

NOW, THEREFORE, for good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged, the parties hereto agree as follows:

- 1. During the term of the Lease (the "Lease Term"), including any renewals, if any, the Premises shall not be used for any purpose other than the operation of a Taco John's restaurant business, and Landlord agrees that Tenant shall be authorized to make such modifications to the Premises as may be necessary to comply with TJI's requirements for remodeling, trade dress, and signage at regular intervals.
- 2. During the Lease Term and within 60 days after default or termination of the Lease, Landlord and Tenant grant to TJI or its agents the right to enter the Premises upon reasonable notice during regular business hours to:
 - (a) inspect and audit Tenant's business;
 - (b) make any modifications necessary to protect TJI's trademarks, provided such modifications do not violate the terms of the Lease; or
 - (c) remove Taco John's signage, trademarked items, trade dress, and other related materials.
- 3. No act, failure to act, event, condition, non-payment or other occurrence ("Event") shall constitute a breach or default under the Lease so as to allow to Landlord any right of acceleration of obligations thereunder, termination, cancellation, or rescission:
 - (a) if the Event is the non-payment of rent, unless such Event is not cured within ten (10) days after Notice of Default (as hereinafter defined) has been received by TJI;
 - (b) if the Event is anything other than the non-payment of rent, unless such Event is not cured within 25 days after Notice of Default (as hereinafter defined) has been received by TJI. Provided, however, if the Event is of such nature that it cannot reasonably be cured within such 25 day period, then, such 25 day period shall be extended to a period of such length as is reasonably necessary to cure such Event, provided, however, such period shall be extended only so long as Tenant and/or TJI diligently pursues the cure of such Event.
- 4. As used herein, "Notice of Default" means written notice mailed by registered or certified mail or delivered by a nationally recognized overnight carrier specifying the Event claimed and describing with specificity, the particular Event(s), the cure required by Landlord, and the cure period. Such Notice of Default shall be mailed to TJI at:

Taco John's International, Inc.
Office of the General Counsel
1650 West End Blvd., Suite 200
St. Louis Park, MN 55416 tjilegal@tacojohns.com
307-635-0101

- 5. Upon receipt of a Notice of Default, TJI shall have the right, but not the obligation, to cure any default or take any other action authorized herein. Nothing herein shall be construed to require TJI to make any payments, perform any obligations, or cure any default under the Lease. Any such decision to offer to make any payments, perform any obligation, or cure any default under the Lease shall be at the sole discretion of TJI. Upon written notice from TJI, Landlord agrees to accept from TJI any payment or performance required under the Lease, or to cure any default under the Lease.
- 6. In the event Landlord claims that an Event has occurred, or in the event TJI notifies Landlord in writing that TJI is exercising any right to take over possession of the Premises under the Franchise Agreement or any other agreement between TJI and Tenant, then, at TJI's option, Landlord shall accept TJI as substitute tenant under the Lease and will cooperate with TJI in turning actual, immediate possession of the Premises over to TJI. In such case, the Lease shall remain in full force and effect, but with TJI as the tenant thereunder. TJI's option, hereinabove granted, may be exercised only if TJI agrees to assume the obligations of the Tenant to Landlord under the Lease as of the date TJI or its affiliate or successor is given actual possession of the Premises. No Event shall give Landlord the right to operate the franchised business unless Landlord becomes properly qualified and approved as a franchisee under TJI's then-current requirements.
- 7. Landlord agrees that TJI may sublet or assign the Premises to a new or existing TJI franchisee on the same terms and conditions as are contained in the Lease provided that TJI provides written notice to Landlord, and TJI and the new franchisee execute an assignment and assumption of lease agreement in a form acceptable to Landlord in its reasonable discretion. In such event, TJI shall be released from any further obligation and liability under the Lease.
- 8. Tenant agrees that if Landlord claims that an Event has occurred, then TJI shall have the right to:
 - (a) immediate and actual possession of the Premises, and all equipment and inventory therein, which possession Tenant agrees to give peaceably, and which may be otherwise obtained by TJI by warrant, injunction, temporary restraining order, summary process or such other immediate legal, summary or equitable proceeding or action as TJI may choose. Tenant hereby waives any right to a jury in any such proceeding or action.
 - (b) become the tenant under the Lease to the exclusion of the Tenant.
- 9. If any provision of the Lease provides that alterations, improvements, or equipment in the Premises becomes the property of Landlord, the parties agree that such provision shall not apply to any proprietary equipment, trade dress, signage, or other items bearing the trademarks of TJI.
- 10. Landlord and Tenant agree not to amend the Lease in any material respect, except with prior written consent of TJI.
- 11. In the event of a conflict between the terms of this Rider and the terms of the Lease, the terms of this Rider shall control.
- 12. Tenant and Landlord understand that TJI is entering into or has entered into a Franchise Agreement with Tenant for a Taco John's Restaurant at the Premises in reliance on the agreements of Tenant and Landlord as herein contained and that TJI, in this instance, would not have otherwise entered into such Franchise Agreement.

033125, as amended 071725

IN WITNESS WHEREOF, the parties hereto have Date.	e duly executed and delivered this Rider as of the Effective
LANDLORD	
By:	
Name:	
Title:	
TENANT	
By:	
Name:	
Title:	
ТЛІ	
By:	
Name:	
Title:	

Any approvals or consents required by Landlord under this Rider shall not be unreasonably

13. A withheld.

ATTACHMENT L

OPERATIONAL ASSESSMENT PROGRAM CONFIDENTIALITY/RELEASE AGREEMENT

TACO JOHN'S INTERNATIONAL, INC.

Operational Assessment Program Confidentiality / Release Agreement

Taco John's International, Inc. ("TJI") and the undersigned applicant (the "Applicant") hereby agree that the Applicant will participate in TJI's Operational Assessment program ("OA") to determine Applicant's suitability for a potential Taco John's restaurant franchise. The OA will be conducted at the following location and on the following dates:

OA Location:	
OA will commence	and will finish
Applicant will be furnished with certain proprioperations of franchised Taco John's restaurar and is treated as confidential by TJI, including Taco John's franchised businesses; the franchised businesses; the franchised businesses; the franchised businesses; the franchised businesses; all information negotiations of any kind conducted by TJI who marketing plans; all information contained in nonpublic written information; internal lists businesses; and all other information which is Confidential Information as may be further de	licant's participation in the OA (or preparation for the OA) ietary information relating to TJI and its operations and the nt businesses ("Confidential Information") owned by TJg without limitation, all proprietary information concerning chise system; marketing methods; sales and promotionan; the strategic plan for TJI; all information concerning ether pending or completed; all marketing research data and the TJI operations manual, and any other manual or other of franchisees and customers of Taco John's franchisees may be considered a trade secret or proprietary and such eveloped from time to time by TJI. Applicant agrees not to tranchised Taco John's restaurant business, any Confidential of TJI.
waives, discharges and holds TJI and all of in agents harmless from any and all claims, den attorneys' fees and costs, on account of, arisin	al representatives, successors and assigns, forever releases its affiliated companies, directors, officers, employees and hands, causes of actions, loss, damage or injury, including g out of or attributable to the attendance or participation of other seminar, convention, program or meeting, or other well to or from such programs.
	olely responsible for all expenses incurred by Applicans participation in the OA, including travel, lodging, meals
APPLICANT:	TACO JOHN'S INTERNATIONAL, INC.
	By:
Print Name:	Print Name:
Date	Print Title:
Date	Date
Phone (Office)	Date
Phone (Home)	<u> </u>
Fax	_

ATTACHMENT M

STATE ADDENDA TO DISCLOSURE DOCUMENT

STATE LAW ADDENDA TO THE TACO JOHN'S INTERNATIONAL, INC. FRANCHISE DISCLOSURE DOCUMENT

The following modifications are made to the Taco John's, Inc. Disclosure Document for the states noted below.

CALIFORNIA

THE CALIFORNIA FRANCHISE INVESTMENT LAW REQUIRES THAT A COPY OF ALL PROPOSED AGREEMENTS RELATING TO THE SALE OF THE FRANCHISE BE DELIVERED TOGETHER WITH THE DISCLOSURE DOCUMENT.

OUR WEBSITES (www.tacojohns.com and www.tacojohnsfranchise.com) HAVE NOT BEEN REVIEWED OR APPROVED BY THE CALIFORNIA DEPARTMENT OF FINANCIAL PROTECTION AND INNOVATION. ANY COMPLAINTS CONCERNING THE CONTENT OF THESE WEBSITES MAY BE DIRECTED TO THE CALIFORNIA DEPARTMENT OF FINANCIAL PROTECTION AND INNOVATION AT www.dfpi.ca.gov.

1. The following paragraph is added to the end of Item 3:

Neither we nor any person listed in Item 2 of this Disclosure Document is subject to any currently effective order of any national securities association or national securities exchange, as defined in the Securities Exchange Act of 1934, 15 U.S.C. § 78a et seq., suspending or expelling these persons from membership in this association or exchange.

2. The following paragraph is added to the end of Item 5:

Initial Franchise Fees paid by franchisees are typically applied towards our general operating expenses, which include but are not limited to those expenses we incur in fulfilling our initial obligations to franchisees.

3. The following paragraphs are added to the end of Item 17:

The California Business and Professions Code Sections 20000 through 20043 provide rights to the franchisee concerning termination or nonrenewal of a franchise. If the Franchise Agreement contains a provision that is inconsistent with the law, the law will control

The Franchise Agreement provides for termination upon bankruptcy. This provision may not be enforceable under federal bankruptcy law. (11 U.S.C. § 101, et seq.)

The Franchise Agreement contains a covenant not to compete, which extends beyond the termination or expiration of the franchise. This provision may not be enforceable under California law.

The Franchise Agreement contains a liquidated damages clause that may apply if the Franchise Agreement is terminated due to a breach by you. Under California Civil Code Section 1671, certain liquidated damages clauses are unenforceable.

The Franchise Agreement requires binding arbitration. The arbitration will occur in Minneapolis, Minnesota, with the costs being awarded to the prevailing party. Prospective Franchisees are encouraged to consult private legal counsel to determine the applicability of California and Federal laws (such as Business and Professions Code 20040.5, Code of Civil Procedure Section 1281, and the Federal Arbitration Act) to any provision of the Franchise Agreement restricting venue to a forum outside the State of California.

The Franchise Agreement requires application of the laws of the State of Wyoming. This provision may not be enforceable under California law.

You may be required to sign a general release of claims if you transfer your franchise. California Corporations Code § 31512 voids a waiver of your rights under the Franchise Investment Law (California Corporations Code §§ 31000 through 31516). Business and Professions Code § 20010 voids a waiver of your rights under the Franchise Relations Act (Business and Professions Code §§ 20000 through 20043).

No statement, questionnaire, or acknowledgment signed or agreed to by a franchisee in connection with the commencement of the franchise relationship shall have the effect of (i) waiving any claims under any applicable state franchise law, including fraud in the inducement, or (ii) disclaiming reliance on any statement made by any franchisor, franchise seller, or other person acting on behalf of the franchisor. This provision supersedes any other term of any document executed in connection with the franchise.

ILLINOIS

Illinois law governs the Franchise Agreement.

In conformance with Section 4 of the Illinois Franchise Disclosure Act, any provision in a franchise agreement that designates jurisdiction and venue in a forum outside of the State of Illinois is void. However, a franchise agreement may provide for arbitration to take place outside of Illinois.

Your rights upon Termination and Non-Renewal of an agreement are set forth in sections 19 and 20 of the Illinois Franchise Disclosure Act.

In conformance with section 41 of the Illinois Franchise Disclosure Act, any condition, stipulation or provision purporting to bind any person acquiring any franchise to waive compliance with the Illinois Franchise Disclosure Act or any other law of Illinois is void.

No statement, questionnaire, or acknowledgment signed or agreed to by a franchisee in connection with the commencement of the franchise relationship shall have the effect of (i) waiving any claims under any applicable state franchise law, including fraud in the inducement, or (ii) disclaiming reliance on any statement made by any franchisor, franchise seller, or other person acting on behalf of the franchisor. This provision supersedes any other term of any document executed in connection with the franchise.

INDIANA

1. The Summary column of Items 17.v and we are subject to the Illinois Franchise Disclosure Act and the Indiana Deceptive Franchise Practices Act (specifically Indiana Code Section 23-2-2.7-1(10)).

MINNESOTA

1. The following legend is added to the "Special Risks to Consider about This Franchise" on the State Cover Page:

MINNESOTA STATUTES §80C.21 AND MINNESOTA RULES 2860.4400(J) PROHIBIT US FROM REQUIRING LITIGATION TO BE CONDUCTED OUTSIDE MINNESOTA, REQUIRING WAIVER OF A JURY TRIAL, OR REQUIRING THE FRANCHISEE TO CONSENT TO LIQUIDATED DAMAGES, TERMINATION PENALTIES OR JUDGMENT NOTES. IN ADDITION, NOTHING IN THE DISCLOSURE DOCUMENT OR AGREEMENT CAN ABROGATE OR REDUCE (1) ANY OF YOUR RIGHTS AS PROVIDED FOR IN MINNESOTA STATUTES, CHAPTER 80C, OR (2) YOUR RIGHTS TO ANY PROCEDURE, FORUM, OR REMEDIES PROVIDED FOR BY THE LAWS OF THE JURISDICTION.

2. The following statements are added at the end of Item 17:

Minnesota Rules 2860.4400(D) prohibits us from requiring you to assent to a general release from liability imposed by Minnesota Statutes, Sections 80C.01 to 80C.22, provided, it does not bar the voluntary settlement of disputes.

We will comply with Minnesota Statutes, Section 80C.14, Subd. 3-5, which require (except in certain specified cases) (1) that you be given 90 days' notice of termination (with 60 days to cure) and 180 days' notice for non-renewal of the Franchise Agreement and (2) that consent to the transfer of the franchise will not be unreasonably withheld.

Minnesota Statute §80C.17.5 provides that any claims and actions based on a violation of Chapter 80C of the Minnesota statutes or any rule or order thereunder shall be commenced within three years from the occurrence of the facts giving rise to such claim or action.

Minnesota Statute §80C.21 and Minnesota Rule 2860.4400(J) prohibit us from requiring litigation to be conducted outside Minnesota, requiring waiver of a jury trial, or requiring you to consent to liquidated damages, termination penalties, or judgment notes. In addition, nothing in the Franchise Disclosure Document or agreement can abrogate or reduce (1) any of your rights as provided for in Minnesota Statutes, Chapter 80C, or (2) your rights to any procedure, forum, or remedies provided for by the laws of the jurisdiction. (The above language has been included in this Disclosure Document as a condition to registration. We and you do not agree with the above language and believe that each of the provisions of the Franchise Agreement, including all choice of law provisions, are fully enforceable. We and you intend to fully enforce all of the provisions of the Franchise Agreement and all other documents signed by us, including but not limited to, all venue, choice-of-law, arbitration provisions and other dispute avoidance and resolution provisions and to rely on federal pre-emption under the Federal Arbitration Act.)

You cannot consent to us obtaining injunctive relief. We may seek injunctive relief. See Minnesota Rule 2860.4400(J). Also, a court will determine if a bond is required.

Minnesota Statutes Section 181.991 prohibits a franchisor from restricting, restraining, or prohibiting in any way a franchisee from soliciting or hiring an employee of the franchisor or a franchisee of the same franchisor.

Minnesota Statutes, Section 181.991 prohibits a franchisor from restricting, restraining, or prohibiting in any way a franchisee from soliciting or hiring an employee of the franchisor or an employee of a franchisee of the same franchisor. Any such restrictions in the Franchise Agreement are deemed deleted.

3. No statement, questionnaire, or acknowledgment signed or agreed to by a franchisee in connection with the commencement of the franchise relationship shall have the effect of (i) waiving any claims under any applicable state franchise law, including fraud in the inducement, or (ii) disclaiming reliance on any statement made by any franchisor, franchise seller, or other person acting on behalf of the franchisor. This provision supersedes any other term of any document executed in connection with the franchise.

NEW YORK

1. The following information is added to the Cover Page of the Disclosure Document:

INFORMATION COMPARING FRANCHISORS IS AVAILABLE. CALL THE STATE ADMINISTRATORS LISTED IN ATTACHMENT G OR YOUR PUBLIC LIBRARY FOR SERVICES OR INFORMATION. REGISTRATION OF THIS FRANCHISE BY NEW YORK STATE DOES NOT MEAN THAT NEW YORK STATE RECOMMENDS IT OR HAS VERIFIED THE INFORMATION IN THIS FRANCHISE DISCLOSURE DOCUMENT. IF YOU LEARN THAT ANYTHING IN THIS FRANCHISE DISCLOSURE DOCUMENT IS UNTRUE, CONTACT THE FEDERAL TRADE COMMISSION AND THE APPROPRIATE STATE PROVINCIAL AUTHORITY. THE FRANCHISOR MAY, IF IT CHOOSES, NEGOTIATE WITH YOU ABOUT ITEMS COVERED IN THE FRANCHISE DISCLOSURE DOCUMENT. HOWEVER. FRANCHISOR CANNOT USE THE NEGOTIATING PROCESS TO PREVAIL UPON A PROSPECTIVE FRANCHISEE TO ACCEPT TERMS THAT ARE LESS FAVORABLE THAN THOSE SET FORTH IN THIS FRANCHISE DISCLOSURE DOCUMENT.

2. The following paragraphs are added at the end of Item 3:

With the exception of what is stated above, the following applies to the franchisor, its predecessor, a person identified in Item 2, or an affiliate offering franchises under the franchisor's principal trademark:

- A. No such party has an administrative, criminal or civil action pending against that person alleging: a felony, a violation of a franchise, antitrust, or securities law, fraud, embezzlement, fraudulent conversion, misappropriation of property, unfair or deceptive practices, or comparable civil or misdemeanor allegations.
- B. No such party has pending actions, other than routine litigation incidental to the business, which are significant in the context of the number of franchisees and the size, nature or financial condition of the franchise system or its business operations.

- C. No such party has been convicted of a felony or pleaded nolo contendere to a felony charge or, within the 10-year period immediately preceding the application for registration, has been convicted of or pleaded nolo contendere to a misdemeanor charge or has been the subject of a civil action alleging: violation of a franchise, antifraud, or securities law; fraud; embezzlement; fraudulent conversion or misappropriation of property; or unfair or deceptive practices or comparable allegations.
- D. No such party is subject to a currently effective injunctive or restrictive order or decree relating to the franchise, or under a Federal, State, or Canadian franchise, securities, antitrust, trade regulation or trade practice law, resulting from a concluded or pending action or proceeding brought by a public agency; or is subject to any currently effective order of any national securities association or national securities exchange, as defined in the Securities and Exchange Act of 1934, suspending or expelling such person from membership in such association or exchange; or is subject to a currently effective injunctive or restrictive order relating to any other business activity as a result of an action brought by a public agency or department, including, without limitation, actions affecting a license as a real estate broker or sales agent.
- 3. The following statement is added to the end of the "Summary" sections of Item 17.c, titled "Requirements for you to renew or extend," and Item 17.m, titled "Conditions for our approval of transfer by you":

However, to the extent required by applicable law, all rights you enjoy and any causes of action arising in your favor from the provisions of Article 33 of the General Business Law of the State of New York and the regulations issued thereunder shall remain in force; it being the intent of this proviso that the non-waiver provisions of General Business Law Sections 687.4 and 687.5 be satisfied.

4. The following language replaces the "Summary" section of Item 17.d, titled "**Termination** by you":

You may terminate the Franchise Agreement on any grounds available by law.

5. The following statement is added to the end of the "Summary" sections of Item 17.v, titled "Choice of forum," and Item 17.w, titled "Choice of law":

The foregoing choice of law should not be considered a waiver of any right conferred upon the franchisor or upon the franchisee by Article 33 of the General Business Law of the State of New York.

- 6. Franchise Questionnaires and Acknowledgements -- No statement, questionnaire, or acknowledgment signed or agreed to by a franchisee in connection with the commencement of the franchise relationship shall have the effect of (i) waiving any claims under any applicable state franchise law, including fraud in the inducement, or (ii) disclaiming reliance on any statement made by any franchisor, franchise seller, or other person acting on behalf of the franchisor. This provision supersedes any other term of any document executed in connection with the franchise.
- 7. Receipts -- Any sale made must be in compliance with § 683(8) of the Franchise Sale Act (N.Y. Gen. Bus. L. § 680 et seq.), which describes the time period a Franchise Disclosure Document (offering prospectus) must be provided to a prospective franchise before a sale may be made. New York law requires a franchisor to provide the Franchise Disclosure Document at the earlier of the first personal meeting, 10

business days before the execution of the franchise or other agreement, or the payment of any consideration that relates to the franchise relationship.

WE REPRESENT THAT THIS DISCLOSURE DOCUMENT DOES NOT KNOWINGLY OMIT ANY MATERIAL FACT OR CONTAIN ANY UNTRUE STATEMENT OF A MATERIAL FACT.

NORTH DAKOTA

- 1. The North Dakota Securities Commissioner has held the following to be unfair, unjust or inequitable to North Dakota franchisees:
 - A. Restrictive Covenants: Franchise Disclosure Documents which disclose the existence of covenants restricting competition contrary to Section 9-08-06, N.D.C.C., without further disclosing that such covenants will be subject to this statute.
 - B. Situs of Arbitration Proceedings: Franchise agreements providing that the parties must agree to the arbitration of disputes at a location that is remote from the site of the franchisee's business.
 - C. Restrictions on Forum: Requiring North Dakota franchisees to consent to the jurisdiction of courts outside of North Dakota.
 - D. Liquidated Damages and Termination Penalties: Requiring North Dakota franchisees to consent to liquidated damages or termination penalties.
 - E. Applicable Laws: Franchise agreements which specify that they are to be governed by the laws of a state other than North Dakota.
 - F. Waiver of Trial by Jury: Requiring North Dakota franchisees to consent to the waiver of a trial by jury.
 - G. Waiver of Exemplary & Punitive Damages: Requiring North Dakota franchisees to consent to a waiver of exemplary and punitive damages.
 - H. General Release: Franchise agreements that require the franchisee to sign a general release upon renewal of the franchise agreement.
 - I. Limitation of Claims: Franchise agreements that require the franchisee to consent to a limitation of claims. The statute of limitations under North Dakota law applies.

VIRGINIA

1. The following legend is added to the "Special Risks to Consider about This Franchise" on the State Cover Page:

<u>Unregistered Trademarks</u>. Certain design trademarks that you may use in your business are not federally registered. If the franchisor's right to use these trademarks in your area is challenged, you may have to identify your business and its products or services with other trademarks that may differ from that used by other franchisees or the franchisor. This

change can be expensive and may reduce brand recognition of the products or services you offer.

2. No statement, questionnaire, or acknowledgment signed or agreed to by a franchisee in connection with the commencement of the franchise relationship shall have the effect of (i) waiving any claims under any applicable state franchise law, including fraud in the inducement, or (ii) disclaiming reliance on any statement made by any franchisor, franchise seller, or other person acting on behalf of the franchisor. This provision supersedes any other term of any document executed in connection with the franchise.

WASHINGTON

The provisions of this Addendum form an integral part of, are incorporated into, and modify the Franchise Disclosure Document, the franchise agreement, and all related agreements regardless of anything to the contrary contained therein. This Addendum applies if: (a) the offer to sell a franchise is accepted in Washington; (b) the purchaser of the franchise is a resident of Washington; and/or (c) the franchised business that is the subject of the sale is to be located or operated, wholly or partly, in Washington.

- 1. <u>Conflict of Laws.</u> In the event of a conflict of laws, the provisions of the Washington Franchise Investment Protection Act, chapter 19.100 RCW will prevail.
- 2. **Franchisee Bill of Rights.** RCW 19.100.180 may supersede provisions in the franchise agreement or related agreements concerning your relationship with the franchisor, including in the areas of termination and renewal of your franchise. There may also be court decisions that supersede the franchise agreement or related agreements concerning your relationship with the franchisor. Franchise agreement provisions, including those summarized in Item 17 of the Franchise Disclosure Document, are subject to state law.
- 3. <u>Site of Arbitration, Mediation, and/or Litigation</u>. In any arbitration or mediation involving a franchise purchased in Washington, the arbitration or mediation site will be either in the state of Washington, or in a place mutually agreed upon at the time of the arbitration or mediation, or as determined by the arbitrator or mediator at the time of arbitration or mediation. In addition, if litigation is not precluded by the franchise agreement, a franchisee may bring an action or proceeding arising out of or in connection with the sale of franchises, or a violation of the Washington Franchise Investment Protection Act, in Washington.
- 4. **General Release.** A release or waiver of rights in the franchise agreement or related agreements purporting to bind the franchisee to waive compliance with any provision under the Washington Franchise Investment Protection Act or any rules or orders thereunder is void except when executed pursuant to a negotiated settlement after the agreement is in effect and where the parties are represented by independent counsel, in accordance with RCW 19.100.220(2). In addition, any such release or waiver executed in connection with a renewal or transfer of a franchise is likewise void except as provided for in RCW 19.100.220(2).
- 5. <u>Statute of Limitations and Waiver of Jury Trial</u>. Provisions contained in the franchise agreement or related agreements that unreasonably restrict or limit the statute of limitations period for claims under the Washington Franchise Investment Protection Act, or rights or remedies under the Act such as a right to a jury trial, may not be enforceable.
- 6. <u>Transfer Fees</u>. Transfer fees are collectable only to the extent that they reflect the franchisor's reasonable estimated or actual costs in effecting a transfer.

- 7. <u>Termination by Franchisee</u>. The franchisee may terminate the franchise agreement under any grounds permitted under state law.
- 8. <u>Certain Buy-Back Provisions</u>. Provisions in franchise agreements or related agreements that permit the franchisor to repurchase the franchisee's business for any reason during the term of the franchise agreement without the franchisee's consent are unlawful pursuant to RCW 19.100.180(2)(j), unless the franchise is terminated for good cause.
- 9. <u>Fair and Reasonable Pricing</u>. Any provision in the franchise agreement or related agreements that requires the franchisee to purchase or rent any product or service for more than a fair and reasonable price is unlawful under RCW 19.100.180(2)(d).
- 10. Waiver of Exemplary & Punitive Damages. RCW 19.100.190 permits franchisees to seek treble damages under certain circumstances. Accordingly, provisions contained in the franchise agreement or elsewhere requiring franchisees to waive exemplary, punitive, or similar damages are void, except when executed pursuant to a negotiated settlement after the agreement is in effect and where the parties are represented by independent counsel, in accordance with RCW 19.100.220(2).
- 11. **Franchisor's Business Judgement.** Provisions in the franchise agreement or related agreements stating that the franchisor may exercise its discretion on the basis of its reasonable business judgment may be limited or superseded by RCW 19.100.180(1), which requires the parties to deal with each other in good faith.
- 12. <u>Indemnification</u>. Any provision in the franchise agreement or related agreements requiring the franchisee to indemnify, reimburse, defend, or hold harmless the franchisor or other parties is hereby modified such that the franchisee has no obligation to indemnify, reimburse, defend, or hold harmless the franchisor or any other indemnified party for losses or liabilities to the extent that they are caused by the indemnified party's negligence, willful misconduct, strict liability, or fraud.
- 13. Attorneys' Fees. If the franchise agreement or related agreements require a franchisee to reimburse the franchisor for court costs or expenses, including attorneys' fees, such provision applies only if the franchisor is the prevailing party in any judicial or arbitration proceeding.
- 14. Noncompetition Covenants. Pursuant to RCW 49.62.020, a noncompetition covenant is void and unenforceable against an employee, including an employee of a franchisee, unless the employee's earnings from the party seeking enforcement, when annualized, exceed \$100,000 per year (an amount that will be adjusted annually for inflation). In addition, a noncompetition covenant is void and unenforceable against an independent contractor of a franchisee under RCW 49.62.030 unless the independent contractor's earnings from the party seeking enforcement, when annualized, exceed \$250,000 per year (an amount that will be adjusted annually for inflation). As a result, any provision contained in the franchise agreement or elsewhere that conflicts with these limitations is void and unenforceable in Washington.
- 15. <u>Nonsolicitation Agreements</u>. RCW 49.62.060 prohibits a franchisor from restricting, restraining, or prohibiting a franchisee from (i) soliciting or hiring any employee of a franchisee of the same franchisor or (ii) soliciting or hiring any employee of the franchisor. As a result, any such provisions contained in the franchise agreement or elsewhere are void and unenforceable in Washington.
- 16. Questionnaires and Acknowledgments. No statement, questionnaire, or acknowledgment signed or agreed to by a franchisee in connection with the commencement of the franchise relationship shall have the effect of (i) waiving any claims under any applicable state franchise law, including fraud in the inducement, or (ii) disclaiming reliance on any statement made by any franchisor,

franchise seller, or other person acting on behalf of the franchisor. This provision supersedes any other term of any document executed in connection with the franchise.

- 17. **Prohibitions on Communicating with Regulators.** Any provision in the franchise agreement or related agreements that prohibits the franchisee from communicating with or complaining to regulators is inconsistent with the express instructions in the Franchise Disclosure Document and is unlawful under RCW 19.100.180(2)(h).
- 18. Advisory Regarding Franchise Brokers. Under the Washington Franchise Investment Protection Act, a "franchise broker" is defined as a person that engages in the business of the offer or sale of franchises. A franchise broker represents the franchisor and is paid a fee for referring prospects to the franchisor and/or selling the franchise. If a franchisee is working with a franchise broker, franchisees are advised to carefully evaluate any information provided by the franchise broker about a franchise.

WISCONSIN

- 1. REGISTRATION OF THIS FRANCHISE IN THE STATE OF WISCONSIN DOES NOT MEAN THAT THE STATE RECOMMENDS IT OR HAS VERIFIED THE INFORMATION IN THIS DISCLOSURE DOCUMENT.
- 2. The conditions under which the Franchise Agreement can be terminated or not renewed may be affected by the Wisconsin Fair Dealership Law, Wisconsin Statutes 1981-82, Title XIV-A, Chapter 135.

State Effective Dates

The following states have franchise laws that require that the Franchise Disclosure Document be registered or filed with the state, or be exempt from registration: California, Hawaii, Illinois, Indiana, Maryland, Michigan, Minnesota, New York, North Dakota, Rhode Island, South Dakota, Virginia, Washington, and Wisconsin.

This document is effective and may be used in the following states, where the document is filed, registered or exempt from registration, as of the Effective Date stated below:

State	Effective Date
California	March 31, 2025, as amended July 17, 2025
Hawaii	NOT EFFECTIVE
Illinois	March 31, 2025, as amended July 17, 2025
Indiana	March 31, 2025, as amended July 17, 2025
Maryland	NOT EFFECTIVE
Michigan	March 31, 2025, as amended July 17, 2025
Minnesota	PENDING
New York	April 25, 2025, as amended July 17, 2025
North Dakota	PENDING
Rhode Island	PENDING
South Dakota	March 31, 2025, as amended July 17, 2025
Virginia	PENDING
Washington	April 3, 2025, as amended July 17, 2025
Wisconsin	PENDING

Other states may require registration, filing, or exemption of a franchise under other laws, such as those that regulate the offer and sale of business opportunities or seller-assisted marketing plans.

ATTACHMENT N

RECEIPTS

RECEIPT

This Disclosure Document summarizes certain provisions of the Franchise Agreement and other information in plain language. Read this Disclosure Document and all agreements carefully. If Taco John's International, Inc. ("TJI," we" or "us") offer you a franchise, we must provide this Disclosure Document to you 14 calendar days before you sign a binding agreement with, or make a payment to, us or an affiliate in connection with the proposed franchise sale.

New York requires that we give you this Disclosure Document at the earlier of the first personal meeting or 10 business days before the execution of the franchise or other agreement or the payment of any consideration that relates to the franchise relationship. Michigan requires that we give you this Disclosure Document at least 10 business days before the execution of any binding franchise or other agreement or the payment of any consideration, whichever occurs first. Iowa requires that we give you this Disclosure Document at the earlier of the first personal meeting or 14 calendar-days before the execution of the franchise or other agreement or the payment of any consideration, whichever occurs first.

If we do not deliver this Disclosure Document on time or if it contains a false or misleading statement, or a material omission, a violation of federal law and state law may have occurred and should be reported to the Federal Trade Commission, Washington, D.C. 20580 and the appropriate state agency identified on Attachment G. We authorize the parties identified on Attachment G to receive service of process for us in the particular state.

The following Franchise Sellers were involved in the offering of this franchise:

The following employees of ours, having a principal business address of 1650 W End Boulevard, Suite 200, St. Louis Park, Minnesota 55416, and telephone number of 800-854-0819: Heather Leed Neary, Richard Bundy, Jackie Secor, Kevin Flaherty, Thomas A. Perella, Jr., Shannon Iverson, Jenna Leichty, Paul Guidera, Bill Crawford, and			
Issuance Date: March 31, 2025, as amended July 17	7, 2025		
I received a Disclosure Document dated March 31, 2	2025, as amended July 17, 2025, that included the following Attachments:		
Attachment A - Franchise Agreement (with Exhil Attachment B - Area Development Agreement (v Attachment C - Nondisclosure and Noncompetition Attachment D - List of Franchisees and Develope Attachment E - List of Former Franchisees Attachment F - Financial Statements Attachment G - State Administrators/Agents for State Administrators/Agents for State Attachment I - Disclosure Acknowledgment Attachment J - Gift Card Onboarding and Chang Attachment K - Lease Rider Attachment L - Operational Assessment Program Attachment M - State Addenda Attachment N - Receipts	with Exhibits) on Agreement ers Service of Process tents ge Form		
Date	Date		
Signature	Signature		
Please Print Name	Please Print Name		

RECEIPT

This Disclosure Document summarizes certain provisions of the Franchise Agreement and other information in plain language. Read this Disclosure Document and all agreements carefully. If Taco John's International, Inc. ("TJI," we" or "us") offer you a franchise, we must provide this Disclosure Document to you 14 calendar days before you sign a binding agreement with, or make a payment to, us or an affiliate in connection with the proposed franchise sale.

New York requires that we give you this Disclosure Document at the earlier of the first personal meeting or 10 business days before the execution of the franchise or other agreement or the payment of any consideration that relates to the franchise relationship. **Michigan** requires that we give you this Disclosure Document at least 10 business days before the execution of any binding franchise or other agreement or the payment of any consideration, whichever occurs first. **Iowa** requires that we give you this Disclosure Document at the earlier of the first personal meeting or 14 calendar-days before the execution of the franchise or other agreement or the payment of any consideration, whichever occurs first.

If we do not deliver this Disclosure Document on time or if it contains a false or misleading statement, or a material omission, a violation of federal law and state law may have occurred and should be reported to the Federal Trade Commission, Washington, D.C. 20580 and the appropriate state agency identified on <u>Attachment G</u>. We authorize the parties identified on <u>Attachment G</u> to receive service of process for us in the particular state.

The following Franchise Sellers were involved in the offering of this franchise:

The following employees of ours, having a principal business address of 1650 W End Boulevard, Suite 200, St. Louis Park, Minnesota 55416, and telephone number of 800-854-0819: Heather Leed Neary, Richard Bundy, Jackie Secor, Kevin Flaherty, Thomas A. Perella, Jr., Shannon Iverson, Jenna Leichty, Paul Guidera, Bill Crawford, and		
Issuance Date: March 31, 2025, as amended July 1	7, 2025	
I received a Disclosure Document dated March 31,	2025, as amended July 17, 2025, that included the following Attachments:	
Attachment A - Franchise Agreement (with Exhil Attachment B - Area Development Agreement (v Attachment C - Nondisclosure and Noncompetiti Attachment D - List of Franchisees and Develope Attachment E - List of Former Franchisees Attachment F - Financial Statements Attachment G - State Administrators/Agents for State Administrators/Agents for State Administrators Agents of Con Attachment I - Disclosure Acknowledgment Attachment J - Gift Card Onboarding and Chang Attachment K - Lease Rider Attachment L - Operational Assessment Program Attachment M - State Addenda Attachment N - Receipts	with Exhibits) on Agreement ers Service of Process tents ge Form	
Date	Date	
Signature	Signature	
Please Print Name	Please Print Name	

IMPORTANT: IF THE FDD WAS DELIVERED TO YOU ELECTRONICALLY, PLEASE IMMEDIATELY SIGN THIS PAGE ELECTRONICALLY.

ALTERNATIVELY, PLEASE IMMEDIATELY PRINT, SIGN, DATE, AND E-MAIL IT TO TJILEGAL@TACOJOHNS.COM, THEN PROMPTLY RETURN THE ORIGINAL BY MAIL OR COURIER TO TACO JOHN'S INTERNATIONAL, INC., 1650 WEST END BLVD., SUITE 200, ST. LOUIS PARK, MINNESOTA 55416.

UNIFORM FRANCHISE REGISTRATION APPLICATION

		File No. <u>638136</u>
		(Insert file number of immediately preceding filing of Applicant
State:	Wisconsin	Fee: \$200.00
APPL	LICATION FOR (Check only one):	
-	INITIAL REGISTRATION OF AN OF	FER AND SALE OF FRANCHISES
-	RENEWAL APPLICATION OR ANNU	JAL REPORT
	PRE-EFFECTIVE AMENDMENT	
_	POST-EFFECTIVE MATERIAL AME	NDMENT
1.	Full legal name of Franchisor: Taco John's l	International, Inc.
2.	Name of the franchise offering: Taco John's	;
3.	Franchisor's principal business address: 1650 West End Boulevard, St. Louis Park, Minnesota 5	
4.	Name and address of Franchisor's agent in to process: Department of Financial Inst Division of Securities 4822 Madison Yards Way, Madison, Wisconsin 53705	stitutions
5.	The states in which this application is or will	ll be shortly on file:
	California (Exemption), Illinois (Exemption) York (Exemption), North Dakota, Rhode Is (Exemption) & Wisconsin.	

6. Name, address, telephone and facsimile numbers, and e-mail address of person to whom communications regarding this application should be directed:

Douglas R. Ferguson, Esq. Robinson Waters & O'Dorisio, PC 1099 18th Street, Suite 2600 Denver, Colorado 80202 Phone: (303) 297-2600 Fax: (303) 297-2750

Certification

dferguson@rwolaw.com

I certify and swear under penalty of law that I have read and know the contents of this application, including the Franchise Disclosure Document with an issuance date of exhibit, March 31, 2025 as amended July 17, 2025 attached as an and documents are accurate and those material facts stated in all those do not contain any material omissions. I further certify that I am documents this certification on behalf of the Franchisor and that I do duly authorized to make so upon my personal knowledge. Signed at St. Louis Park, MN, guly 16, 2025.

Franchisor: TACO JOHN'S INTERNATIONAL, INC.

By: Heather Leed Neary Name: Heather Leed Neary Title: President + CEO STATE OF Minne Sota) ss

Personally appeared before me this day of day of 2025, the above-named person who executed the foregoing application as president + CEO of the above-named applicant and, being first duly sworn, stated upon oath that said application, and all exhibits submitted herewith, are true and correct.

Surcus Thombers Notary Public

Witness my hand and official seal.

My commission expires: 31 2027

SEAL



FRANCHISE SELLER DISCLOSURE FORM

1. List who w		who will solicit, offer or sell franchises for the Franchisor in this state:
	A.	Name: Jackie Secor
	B.	Business address and telephone number: 1650 W End Boulevard, Suite 200 St. Louis Park, Minnesota 55416 (800) 854-0819
	C.	Present Employer: Taco John's International, Inc.
	D.	Present Title: Chief Operating Officer
	E.	Employment during the past five years. For each employment, state the name of the employer, position held, and beginning and ending dates:
		Ms. Secor joined Taco John's International, Inc. in June 2025 as Chief Operating Officer. Prior to joining Taco John's, she served in various roles with GoTo Foods (formerly known as Focus Brands) located in Atlanta, Georgia, including as Senior Vice President-GTF Operations & Training for seven brands from December 2024 to May 2025, and as Senior Vice President-GTF Operations & Training, Specialty Category for four brands from October 2020 to December 2024. Ms. Secor is based out of Taco John's St. Louis Park, Minnesota office.
2.	State	whether the person identified in l above:
	A. viola	Has an administrative, criminal or material civil action pending against that person alleging a tion of franchise, antitrust or securities law, or alleging fraud, unfair or deceptive practices, or any comparable allegations?
		YES NO
		If you answered "yes", please provide:
		 Names of the parties:
	alleg	Has during the 10-year period immediately before the disclosure document's issuance date been icted of or pleaded nolo contendere to a felony charge; or been held liable in a civil action involving an ed violation of a franchise, antitrust or securities law, or allegations of fraud, unfair or deceptive ices, or comparable allegations?
		YES NO V
		If you answered "yes", please provide:
		1. Names of the parties:
		Case or proceeding identification number:

or concluded a	ect to a currently effective injunction or restrictive order or decree resulting from a pending ction brought by a public agency and relating to the franchise, or to a Federal, State or
Canadian franc	hise, securities, antitrust, trade regulation or trade practice law?
YES [□ NO ✓
If you	answered "yes", please provide:
1.	Names of the parties:
2.	Public agency or court:
3.	Case or proceeding identification number:

FRANCHISE SELLER DISCLOSURE FORM

1.	List wh	no will solicit, offer or sell franchises for the Franchisor in this state:
	A.	Name: Bill Crawford
	В.	Business address and telephone number: 1650 W End Boulevard, Suite 200 St. Louis Park, Minnesota 55416 517-898-5535
	C.	Present Employer: Taco John's International, Inc.
	D.	Present Title: Senior Construction Manager
	E.	Employment during the past five years. For each employment, state the name of the employer, position held, and beginning and ending dates:
		Mr. Crawford has served as Senior Construction Manager for Taco John's International, Inc. since June 2025. From September 2021 to April 2025, he served as Senior Construction Manager for Carrol's. From September 2018 to September 2021, he served as Director of Construction for Aeon Construction and Consultants.
2.	State w	hether the person identified in l above:
		Has an administrative, criminal or material civil action pending against that person alleging a on of franchise, antitrust or securities law, or alleging fraud, unfair or deceptive practices, or any omparable allegations?
		YES NO NO
		If you answered "yes", please provide:
		 Names of the parties: Forum, nature and current status of the pending action: Case or proceeding identification number:
	alleged	Has during the 10-year period immediately before the disclosure document's issuance date been ted of or pleaded nolo contendere to a felony charge; or been held liable in a civil action involving an l violation of a franchise, antitrust or securities law, or allegations of fraud, unfair or deceptive es, or comparable allegations?
		YES NO V
		If you answered "yes", please provide:
		 Names of the parties: The forum:
		Case or proceeding identification number:

or concluded a	ect to a currently effective injunction or restrictive order or decree resulting from a pending ction brought by a public agency and relating to the franchise, or to a Federal, State or hise, securities, antitrust, trade regulation or trade practice law?
Canadian Iranc	mise, securities, antitrust, trade regulation of trade practice taw:
YES [□ NO 🗸
If you	answered "yes", please provide:
1.	Names of the parties:
2.	Public agency or court:
3.	Case or proceeding identification number:



DOUGLAS R. FERGUSON

TELEPHONE: (303) 297-2600, EXT. 169 WRITER'S E-MAIL: DFERGUSON@RWOLAW.COM

July 17, 2025

VIA ELECTRONIC SUBMISSION

Securities Examiner
Wisconsin Department of Financial Institutions
345 West Washington Avenue, Suite 300
Madison, Wisconsin 53703

Re: Taco John's International, Inc.

Post-Effective Amendment of Franchise Registration

File No.: 638136

To Whom It May Concern:

On behalf of Taco John's International, Inc. (the "Franchisor"), we submit with this letter the following application documents for a post-effective amendment to the Franchisor's 2025 franchise registration. In particular, with this amendment, we are updating the Franchisor's address and updating Item 2 of the Franchisor's Franchise Disclosure Document ("FDD"). The following documents are included:

- 1. A full, clean copy of the FDD, as amended.
- 2. An executed Franchise Registration Post-Effective Amendment Application with a Certification.
 - 3. Franchise Seller Disclosure Forms for new franchise sellers.

We are also submitting with these materials payment for the amendment fee in the amount of \$200.00.

If you have any questions or comments regarding the Franchisor or this filing, please contact me at the telephone number or address indicated above.

Sincerely,

Douglas R. Ferguson

Douglas R. Ferguson

Enclosures

cc: Taco John's International, Inc. (w/o enclosures)